Report of the Board of Enquiry into Scientology

by Kevin Victor Anderson, Q.C.

Published 1965 by the State of Victoria, Australia



Contents

•	Prefatory	Note
---	-----------	------

- 1. Preliminary
- 2. Introduction
- 3. Scientology and claims made for it
- 4. The Organization of Scientology
- 5. Financial Aspects of Scientology in Victoria
- 6. Hubbard the founder of Scientology
 - Hubbard's Morbid Preoccupation with Perversion
- 7. Hubbard's scientific deficiencies
- Hubbard and Medicine
- Hubbard and Psycho-Analysis
- Hubbard as a Scientist
- Hubbard as a Nuclear Physicist
- 8. Hubbard's research
- 9. Dianetics
- 10. Association between Dianetics and Scientology
- 11. Scientology theories
- 12. The teaching of scientology
- 13. Scientology auditing processes
- 14. The E-Meter
- 15. Procurement and Dissemination
- 16. The preclear's progress
- 17. Dedication to Scientology
- 18. Scientology and hypnosis
- 19. The healing claims of Scientology
- 20. Alleged benefits of Scientology processing
- 21. General effects of processing
 - Short Term Effects
- Long Term Effects
- Individual Cases
- 22. Hostility to medical profession
- 23. Dangers to mental health
- 24. Dangers of coercion
- 25. Moral laxity
- 26. Family discord
- 27. Scientology and religion
- 28. Scientology and politics
- 29. On the fringe
- 30. Conclusions
- 31. Recommendations

Appendices

(In progress)

- 1. Alphabetical list of witnesses.
- 2. Qualifications of individual witnesses possessed of scientific or technical qualifications who gave evidence
 in favour of scientology.
- 3. Names, qualifications and appointments of expert witnesses who gave evidence on various aspects of scientology theory and practice and other scientific matters.
- 4. Lists of exhibits.
- 5. Ruling and reasons of the Board in relation to the operation of Parliamentary privileges.
 - 6. Glossary of scientology terms and abbreviations.
- 7. Consolidated statement of combined income and expenditure and profit or loss of the HASI and HCO from 1955 to 30th June, 1963.
- 8. Comparitive statement of the income and expenditure of the HASI for the financial years ended 30th June, 1961, 1962 and 1963.
- 9. Balance sheets of the HASI for the financial years ended 30th June, 1961, 1962 and 1963.
- 10. Statement of HASI receipts and payments for the six months ended 31st December, 1963.
- 11. HCO Pol. Lr. of 12th October 1961 "Refund of fee policy revised",
- 12. Comparative statement of the income and expenditure of the HCO for the financial years ended 30th June, 1961, 1962 and 1963.
- 13. Balance sheets of the HCO for the financial years ended 30th June, 1961, 1962 and 1963.
- 14. The Auditor's Code.
- 15. The Code of a Scientologist.
- 16. Extracts from *Brain-washing* Manual, with substitutions.
- 17. Scientology press policies Extracts from HCO Pol. Lr. of 14th August AD 13 (1963).
- 18. Text of "Loyalty Oath" appearing in Certainty Magazine, Vol. 5, No. 5.
- 19. (i) Extract from transcript.
 - (ii) Examples of matters dealt with during scientology processing.

The Anderson Report PREFATORY NOTE

There are some features of scientology which are so ludicrous that there may be a tendency to regard scientology as silly and its practitioners as harmless cranks. To do so would be gravely to misunderstand the tenor of the Board's conclusions. This Report should be read, it is submitted, with these prefatory observations constantly in mind. Scientology is evil; its techniques evil; its practice a serious threat to the community, medically, morally and socially; and its adherents sadly deluded and often mentally ill.

Its founder is Lafayette Ronald Hubbard, an American now resident at Saint Hill Manor, East Grinstead, Sussex, England, who falsely claims academic and other distinctions, and whose sanity is to be gravely doubted. He is the governing director of, and has absolute power in, a world-wide organization named Hubbard Association of Scientologists International (the HASI), which promotes the practice of scientology and canvasses for adherents in most of the countries of the Western World. In eight years to 1964, scientology became strongly entrenched in Victoria.

For many years Hubbard was a prolific and successful writer of science fiction; the theories of scientology which he has developed have altogether extraordinary and novel qualities, rivalling in fantasy the most advanced science fiction.

Many scientology techniques, beyond the elementary stages, are essentially those of command or authoritative hypnosis, and are potentially dangerous to mental health. Scientology processing or "auditing" is administered by scientology trained "auditors" who have no knowledge or appreciation of, or skill in, orthodox psychiatry or psychology; they are generally unaware of the dangers of the techniques which they practise and are unable to detect in their patients a variety of symptoms which would indicate to a medical practitioner or a trained psychologist mental and physical conditions which may require professional treatment.

A pseudo-science called "dianetics", also founded by Hubbard and claimed by him to be "the modern science of mental health", is an important part of scientology and categorically but falsely claims to cure 70 per cent. of man's ailments. Though scientology formally professes not to treat or cure physical or mental ailments, in a covert way it creates the impression that it does, and it frequently processes individuals for the purpose of curing or alleviating their ailments.

Scientology procedures have done very considerable mental harm to individuals who have been persuaded to undergo processing and training. In many cases, mental derangement and a loss of critical faculties have resulted from scientology processing and have continued long after the individuals concerned have ceased active association with scientology. In a number of instances the direct result of scientology processing has been to produce mental derangement which has required hospital treatment.

In the initial stages of scientology, its practitioners, using recognized psychological methods commonly found in schools, clubs and welfare and social organizations, have on occasions produced some apparent benefit for some people; but the enthusiasm resulting from such transient gains is deliberately exploited by the HASI to produce a subservience amounting almost to mental enslavement from which, because of fear, delusion, debilitation and other conditions induced by scientology processes, the individual finds it extremely difficult, and often impossible, to escape. The dubious and temporary benefit which some people claim to have received from scientology processing does not justify the continuance by unskilled and ignorant operatives of

practices which are positively harmful to mental health.

Though there was no evidence of blackmail in the popular sense, yet because of the domination which scientology asserts over its adherents who have been persuaded to reveal their innermost secrets in the course of processing, the potentiality for the misuse of confidences reposed in the HASI is great; the existence of files containing the most intimate secrets and confessions of thousands of individuals is a constant threat to them and is a matter of grave concern. This aspect is the more serious because copies of these reports are also held at Saint Hill Manor in England.

The appeal of scientology is at times deliberately directed towards the weak, the anxious, the disappointed, the inadequate and the lonely - those unduly concerned about themselves, who in some cases are mentally unwell. At other times it is directed at university and senior school students, ambitious business men and women, public servants, school teachers, housewives and others.

Scientology offers to "make the able more able", to remove "suppression", to improve IQ and personality, to proof people against mental and physical illness and to bestow a variety of other benefits, offering sure success by allegedly infallible means. All these claims are entirely unjustified. Public advertisements generally make no mention of scientology theory or advanced techniques; they merely offer lectures and benefits "without obligation" and make extravagant promises.

The principles and practices of scientology are contrary to accepted principles and practices of medicine and science, and constitute a grave danger to the health, particularly the mental health, of the community. Expert opinion to this effect was fully confirmed by the considerable number of specific cases of damage to mental health of which the Board heard evidence.

Many individuals have paid very large sums of money to the HASI for processing and other services, amounts of over £1,000 not being uncommon. The hourly rates for processing range from four and a half guineas downwards. Frequently, courses of processing extend over hundreds of hours; for these there are reduced rates. Financial hardship to the customer is the usual concomitant of processing. For the six years ended the 30th June, 1964, the gross income of the HASI was over a quarter of a million pounds, on which a flat 10 per cent. levy was payable to Hubbard's headquarters in England. In addition, considerable sums have been received by the Australian College of Personnel Efficiency at Geelong and the Melbourne College of Personal Efficiency at Hawthorn, two offshoots of the HASI.

Scientology has highly undesirable processes, many of which are hypnotic, wherein normal inhibitions and restraints are in abeyance. Sexual matters, normal and abnormal, are frequently dwelt upon extensively and erotically.

Scientology is not, and does not claim to be, a religion. The general attitude of its founder is hostile to and disparaging of religion.

Scientology is a grave threat to family and home life. As well as causing financial hardship, it engenders dissension, suspicion and mistrust amongst members of the family. Scientology has caused many family estrangements.

The Board has been unable to find any worth-while redeeming feature in scientology. It constitutes a serious medical, moral and social threat to individuals and to the community generally.

The Anderson Report CHAPTER 1 PRELIMINARY

Appointment of the Board.

This Board of Inquiry into Scientology was appointed by Order in Council made the 27th November, 1963, and published in the Victoria Government Gazette of the 28th November, 1963, No. 931, at p. 3547. The Order in Council was in the following terms:

BOARD OF INQUIRY INTO SCIENTOLOGY

At the Law Courts, Melbourne, the twenty-seventh day of November, 1963.

PRESENT:

The Lieutenant-Governor as Deputy for His Excellency the Governor of Victoria

Mr. Mack

Mr. Hamer

WHEREAS it is deemed expedient that a Board of Inquiry be appointed to inquire into and report upon Scientology:

Now therefore, the Lieutenant-Governor as Deputy for His Excellency the Governor of the State of Victoria, in the Commonwealth of Australia, by and with the advice of the Executive Council of the said State, doth by this Order constitute and appoint -

KEVIN VICTOR ANDERSON, Q.C.,

to be a Board to inquire into, report upon, and make recommendations concerning Scientology as known, carried on, practised and applied in Victoria and in particular and without derogating from the generality of the foregoing -

- 1. The aims, objects and purposes thereof and whether it is capable of achieving to any and what extent such aims, objects and purposes.
- 2. The persons and organizations by whom and the place or places at which the same is so carried on, practised, or applied.
- 3. The methods, processes, practices, techniques, principles and treatment involved or used or applied in the carrying on, practice, or application thereof.
- 4. The circumstances in which and the conditions under or in accordance with which the same is carried on, practised or applied.
- 5. The circumstances surrounding, the principles involved in, the conditions relating to, and the methods applied in the teaching thereof.

- 6. The nature and amount of fees charged or remuneration received by persons or organizations engaged in the carrying on, practice, application and teaching thereof.
- 7. The instruments, machines, devices or other apparatus used in the carrying on, practice, or application thereof, and the manner of use of and the effectiveness of the same.
- 8. The illnesses or ailments, mental or physical, treated in the course of the carrying on, practice or application thereof, the effectiveness of such treatment, and whether such treatment is in any and what way harmful or beneficial or likely to be harmful or beneficial to the persons treated.
- 9. The advertisements used in the carrying on, practice, or application thereof, and whether the same have been in any and what manner false or misleading, and whether the same should be prohibited, regulated, or restricted in any and what way.
- 10. Whether persons or organizations engaged in the carrying on, practice, or application thereof have indulged in any and what unlawful, improper, harmful or prejudicial practices or activities.
- 11. Whether children under the age of sixteen years have been treated by any and what persons or organizations in the carrying on, practice, or application thereof, the circumstances and conditions under which such children have been treated and whether such treatment has been or was likely to be in any and what manner harmful or beneficial.
- 12. The terms and conditions under which and the circumstances in which persons are engaged for employment and employed by persons or organizations carrying on, practising or applying the same.

And it is hereby directed that the said Kevin Victor Anderson shall, with as little delay as possible, report under his hand on this inquiry.

Whereof the said Kevin Victor Anderson and all other persons whom it may concern are to take notice and govern themselves accordingly.

And the Honorable Rupert James Hamer, for and on behalf of Her Majesty's Chief Secretary for the State of Victoria, shall give the necessary directions herein accordingly.

J. COLQUHOUN
Clerk of the Executive Council.

Development of Official Interest in Scientology.

For a few years before the appointment of this Board there had been an uneasy stirring in various places in respect of scientology. The harm which scientology was likely to cause has been of continuing and increasing concern to the Mental Health Authority. This Authority, which is the successor in law of the Mental Hygiene Authority, is now constituted under the Mental Health Act 1959, which provides by section 12 that the "The functions of the Authority shall, subject to this Act and the powers of the Governor in Council and the Minister under this Act, be to formulate control and direct general policy and administration in respect of the treatment and prevention of mental illness and intellectual defectiveness, and, in particular, and without affecting the generality of the foregoing

(a) to make provision for the improvement of the treatment and take measures for the prevention of

(b) to provide for the carrying out of research and investigations in relation to the causation and treatment of mental illness and intellectual defectiveness

The Authority's increasing concern developed because of information which it received of people being persuaded to take scientology courses and in some instances pay large sums of money, and particularly because such people had done so consequent upon misrepresentations concerning mental health and the mentally ill. The interest of the Authority in scientology did not develop because of any desire to interfere with people's beliefs, religious or otherwise. From as early as 1957, when the Chief Commissioner of Police raised the matter with the Authority, scientology had been to an increasing degree under the surveillance of the Authority because of the fear that it would influence people in such a way as to be injurious to their mental health.

The University of Melbourne was also alert to the potential harm which scientology could do to University students and, in 1961 and 1962, the University Student Counsellor became concerned at the attempts being made by scientology to exploit the anxieties of students. Students in their teens and early twenties frequently experience considerable anxiety concerning their courses and the manner in which they are coping with their studies, and this anxiety has been accentuated in recent years by the introduction of quotas, when failure in a year may mean the end of a student's academic career. In most cases such anxieties are normal, and the University of Melbourne has a staff of trained psychologists who act as student counsellors to assist students to cope with their difficulties. There are, however, no short-cut methods, and the University of Melbourne viewed the dangers inherent in scientology so seriously that the Vice-Chancellor on two separate occasions warned students against "city practitioners" who were offering services purporting to be psychological to assist them in their studies. The case of a University student who was attracted by scientology advertisements is dealt with in Chapter 15.

The Australian Medical Association was also alive to the potential harm arising from the activities of the scientologists, and an occasional statement in the press by the Medical Secretary of the Victorian Branch of that Association provoked scurrilous attacks on him by the scientologists.

Melbourne newspapers were also aware of undesirable features of scientology and the daily newspapers declined in 1961 to accept advertisements from the HASI. The Melbourne *Truth* newspaper made pointed attacks on scientology in a series of feature articles.

There was thus a growing public interest in Victoria in the activities of scientology, which culminated in statements in the Legislative Council by the Honorable J. W. Galbally, M.L.C., and the Honorable J. M. Walton, M.L.C.

Debate on aspects of scientology took place in November, 1963, in the Legislative Council, and a Private Member's Bill "to prohibit the teaching and practice of scientology for fee or reward and the use in relation to such teaching or practice of any apparatus or device for recording or measuring personal reactions, impulses or characteristics" was introduced into the Council by the Hon. J. W. Galbally, M.L.C., on the 26th November, 1963.

On the 27th November, 1963, this Board was appointed in the terms of the foregoing Order in Council.

Sittings of the Board.

The terms of reference required the Board to inquire into, report upon, and make recommendations

concerning scientology as known, carried on, practised and applied in Victoria they were in very wide terms which explicitly mentioned many matters and implicitly required the Board to investigate all phases of scientology in Victoria.

The Board sat to receive evidence and hear submissions and addresses on 160 days. The first formal sitting was on the 6th December, 1963, at the Flemington Court House, when counsel appeared for various interested parties, certain formalities took place, and some preliminary evidence was heard. On that day the Board adjourned to the premises of the Hubbard Association of Scientologists International (the HASI) at 15759 Spring-street, Melbourne, where counsel assisting the Board tendered in evidence the contents of 35 steel or wooden filing cabinets and a large quantity of other records of the HASI. These records were left in the custody of Peter Rogers Williams, then the Continental Director of the HASI, on his undertaking to produce them as and when required.

The Board resumed sitting on the 17th February, 1964, in the National Herbarium, Birdwood Avenue, Royal Botanic Gardens, Melbourne, and thereafter sat at the Herbarium, except on four days when it sat at Department of Health premises at 19 Park Street, South Yarra, and four days when it sat at Owen Dixon Chambers, 205 William Street, Melbourne. On some occasions, the Board adjourned to the HAST premises at 157-59 Spring Street, Melbourne, and at 2 Coates Lane, Melbourne, to hear evidence, witness demonstration sessions of scientology processing and listen to tape-recorded lectures of the founder of scientology, Lafayette Ronald Hubbard.

The final sitting of the Board was on the 21st April, 1965.

Appearances

Mr. Gordon Just of counsel (instructed by the Victorian Crown Solicitor) appeared to assist the Board.

Mr. J. R. O'Shea of counsel (instructed by Peter E. and Ewart F. Norris) appeared, until 26th May, 1964, for the Hubbard Association of Scientologists International, the Hubbard Association of Scientologists International Limited, Peter Rogers Williams, and Ernest Denzil Stephen Gogerly. On the 25th May, 1964, Mr. O'Shea announced that after the 26th May, 1964, he would not be further appearing before the Board and that Mr. Ian G. Abraham of counsel would be appearing instead. Mr. Abraham appeared with Mr. O'Shea on the 25th and 26th May, 1964, and thereafter he appeared alone for the clients formerly represented by Mr. O'Shea. Mr. Abraham continued to appear until the 30th November, 1964, when, on instructions from his clients, he withdrew from the hearing.

Mr. Warren C. Fagan of counsel (instructed by J. Robertson McMillan and Company) appeared for the Committee for Mental Health and National Security (an unincorporated association opposed to scientology) from the first sitting of the Board until the 6th May, 1964. (See below as to appearance permitted in lieu of Mr. Fagan.)

Mr. M. Ashkanasy, Q.C., with Mr. O'Shea (instructed by Petcr E. and Ewart F. Norris), appeared on one occasion on the 28th April, 1964, for Mr. Peter Rogers Williams.

Mr. Richard Searby of counsel (instructed by Keith Ness and Son) appeared on the 18th May, 1964, for Southdown Press Pty. Ltd., the publisher of the Truth newspaper.

Mr. A. R. Bieske appeared as solicitor, on the 1st September, 1964, for a person whose evidence at that stage was heard in camera.

Mr. Mervyn Kimm of counsel (instructed by Cedric Ralph) appeared on the I 0th September, 1964, for a person concerning whose possible interest in the Inquiry proceedings were held in camera.

Mr. J. F. Hill appeared as solicitor, on the 10th September, 1964, for three persons concerning whose possible interest in the Inquiry proceedings were held in camera.

In neither of these instances did the Board consider the matters raised in camera relevant to the terms of reference, and they were not further pursued.

Mr. B. L. Agapis, solicitor, of Sydney, was given leave on the 7th September, 1964, to appear for David Maxwell Tooley (commonly known as Marcus Tooley) the proprietor of the Australian Centre of Applied Psychology and The American College, both of 21 Bourke Street, Melbourne.

Mr. Phillip Bennett Wearne, the executive officer of the Committee for Mental Health and National Security, was given leave on the 27th May, 1964, to appear on behalf of that committee and himself, after Mr. Fagan of counsel ceased to appear for the Committee.

Mr. Douglas Moon was given leave to appear on the 18th August, 1964, in relation to an aspect of the Inquiry which related to him.

Mr. Michael Andrade Birch, the Melbourne representative of Marcus Tooley, was permitted to appear on the 17th August, 1964, on his own behalf.

Evidence and Witnesses

Though the Board was not bound by the rules of evidence, it endeavoured to conduct the Inquiry as nearly as the subject-matter and circumstances would allow, along lines conventionally followed by judicial and quasi-judicial tribunals.

The Board heard oral evidence on oath from 151 witnesses. The names of witnesses who gave evidence in the public sittings of the Board without restriction as to their identity, or whose names may be published without undue embarrassment, are listed in Appendix 1. The names of a small number of persons who gave evidence in public but in respect of whom the Board ruled that their identity be not disclosed are not included in Appendix I, nor are the names of witnesses who gave evidence entirely in camera (except where no undue embarrassment is attendant upon the inclusion of such names).

The evidence of all witnesses was taken in shorthand by the Government Shorthand Writer's staff and transcribed. It comprised nearly four. million words, covering 8,920 pages of transcript. The transcript of other than the strictly in camera evidence accompanies and forms part of this Report.

At all times, including during the in camera sessions, the scientology interests were permitted to be present, to participate in proceedings, to lead evidence and test evidence by cross-examination. This right was fully exercised until the 30th November, 1964, when counsel appearing for them was instructed by his clients to withdraw. Thereafter, they did not seek to exercise such right. The Board was thus denied the assistance of a final address by counsel for the scientology interests, and the burden imposed on counsel assisting the Board was greatly increased.

Unless there were compelling reasons to the contrary, the Board heard evidence in public. There were, however, a number of occasions when the Board considered it proper to sit in camera to hear evidence or submissions of and concerning people and matters, the general publication of

which was considered undesirable because of the intimate, personal and embarrassing nature of the subject-matter. At times, the Board sat strictly in camera; on other occasions, though not in camera, the Board directed that the name or any information which might identify a particular person or witness be not published. The Press co-operated, and observed directions given as to non-disclosure of identity in such circumstances.

Though many witnesses gave evidence in support of scientology, very few of them possessed any qualification which entitled them to speak authoritatively on scientific or technical matters. The Board repeatedly pointed out to the scientology interests the notable absence of any credible evidence to justify scientology theories and techniques; however, this deficiency was not remedied.

Some persons with scientific or technical qualifications gave evidence before the Board on behalf of the scientology interests; two of such witnesses were medical practitioners. However, none of these witnesses gave any reliable or acceptable evidence in attempted justification of scientology; and, in particular, neither of the two medical practitioners was able to explain or justify scientology.

Scientology witnesses with scientific or technical qualifications found it impossible to reconcile, and did not attempt to reconcile, their scientific and technical learning with the theories and practices of scientology, but they sought to justify such theories by saying that they were in a different frame of reference from conventional scientific knowledge.

Out of consideration for such witnesses the Board has decided to adopt the course of not separately listing the names of the scientology witnesses mentioned in the last two preceding paragraphs. It is aware that persons who are known to be scientologists have sometimes suffered ridicule and even opprobrium, and it has no desire unduly to expose these witnesses, some of them highly qualified and competent in their chosen professions, to the embarrassment which might be visited upon them should their association with scientology be publicized. They were not scientology practitioners in any real sense but were preclears who had "cognited" to some extent on scientology. Accordingly, Appendix 2 lists the particular qualifications of a number of unidentified persons.

Contrasting with the virtual absence of any reliable scientific evidence in support of scientology was the testimony of an imposing array of highly qualified experts in many fields, whose evidence was to the effect that scientology was entirely unscientific, at variance with well established principles and practices of medicine and other sciences and possessed many dangerous features. Much of this Report will be concerned with the detail of their evidence and the conclusions of the Board drawn from that evidence. These expert witnesses are listed by name, qualifications, and appointments in Appendix 3.

Non-attendance of Hubbard

As the founder of scientology, Lafayette Ronald Hubbard was an important figure in the Inquiry. He did not attend, though the Board indicated from time to time that it would have found his evidence informative and important. It appeared early in the Inquiry that Hubbard had no intention of visiting Victoria, having expressed himself as quite satisfied with the manner in which the Continental Director for Australia, New Zealand and Oceania, Peter Rogers Williams, was dealing with the Inquiry.

Subsequently, in October, 1964, the Melbourne solicitors of Hubbard requested the Government to pay the cost of bringing Hubbard from England to give evidence before the Board. This request and a similar request to the Board were refused. The Board is satisfied, however, that Hubbard had no

intention of appearing before it. The applications for his expenses to be paid were made, knowing they would be rejected; they were made merely for the purpose of founding a criticism of the conduct of the Board and of this Report. Hubbard's evidence would have been of great interest to the Board; to what extent it would have been of assistance to the Board it is, of course, impossible to say. The Board had before it a very large quantity of Hubbard's writings, and this Report is based to a very substantial extent on these writings. It is not to be expected that Hubbard would have repudiated all that he has been writing since 1950. As this Report will show, he stands sufficiently condemned on his writings. In any event, it was for Hubbard to decide whether or not he attended the Inquiry. His financial resources and those of his organization were more than adequate to defray the expenses of his attendance at the Inquiry. It was his decision alone which resulted in his non-attendance.

Exhibits

The Board received in evidence a vast quantity of documentary material. Though this material is listed as 621 numbered exhibits (see Appendix 4), in fact it comprised many thousands of documents, including books, magazines, brochures, pamphlets, bulletins, policy and other letters, correspondence files, processing files, transcripts of tape-recorded lectures, and a variety of other documents.

Parliamentary Privilege

As already mentioned, one of the events which eventually led to the appointment of this Board was the debate in Parliament concerning scientology. The Hon. J. W. Galbally, M.L.C., in the course of his second reading speech in the Legislative Council on the 27th November, 1963, said that he was prepared to give evidence before a Board of Inquiry into scientology. In answer to a request by the Board, Mr. Galbally attended before the Board on the 4th May, 1964. Counsel assisting the Board announced that, though Mr. Galbally had stated that he was prepared to give evidence before the Board of his knowledge of scientology, the guestion had arisen as to whether he was eligible to give evidence and whether the Board had jurisdiction to receive evidence from him, because of what appeared to be the very wide effect of Parliamentary privilege attaching to anything said by a Member of Parliament in the course of proceedings in Parliament. After considering argument by counsel, the Board was of the view that privilege of Parliament attached to the statements made by Mr. Galbally in the Legislative Council, and that such privilege was not that of the member but of the House in which the statements were made. Notwithstanding Mr. Galbally's stated desire to give evidence before the Board and his desire, if he were able, to waive the privilege, the Board considered that it should not allow him to enter the witness box and be sworn and thereafter be questioned concerning the subject matter of his statement in Parliament. Accordingly, Mr. Galbally was not called as a witness.

Because of the importance of the privileges of Parliament, the Board's ruling and the reasons therefor are set out in Appendix 5.

Secondary Sources of Information

As already mentioned, a number of articles in the Melbourne Truth newspaper over a period of about three years had directed attention to the practice of scientology in Victoria. One of the articles indicated that the newspaper had information which would be of assistance to the Board. Accordingly, the Editor of Truth was requested to arrange for the attendance before the Board of such of his staff as might be able to assist the Board. On the 18th May, 1964, the Editorial Director of Southdown Press Pty. Ltd. (the publisher of Truth newspaper) and another journalist employed

by that company attended and gave evidence before the Board as to the source of the information contained in certain of the articles above referred to.

The evidence which Mr. Galbally was prepared, but not permitted, to give and the evidence which the representatives of Southdown Press Pty. Ltd. gave would have been and was of a secondary nature. It did not relate to anything which the individuals concerned knew of their own knowledge, but consisted of information which they had received from other sources.

As events turned out, the Inquiry was not handicapped by the non-availability of Mr. Galbally's evidence, as other evidence given before the Board indicated the probable source of most of his information.

Course of the Inquiry

At the beginning of the Inquiry, consultation between the Board and counsel resulted in the adoption of a course of procedure whereby the Board would be informed at an early stage of the nature and purpose of scientology. This plan involved calling initially a number of executive officers of the HASI to explain the principles and practices of scientology. It was further proposed that a number of witnesses who claimed to have benefited from scientology would be called at an early stage to give evidence of such benefits.

There were many other persons who were available to give evidence. They may broadly be divided into two groups:

- (a) persons who had formed unfavourable opinions concerning scientology, either because they had formerly been associated with it and had broken away, or because they had observed its effects on other persons. Included in this group was a body of persons who had formed themselves into an unincorporated association which they named "The Committee for Mental Health and National Security.". Its executive officer was Phillip Bennett Wearne, who had formerly been an active scientologist.
- (b) a number of professional persons professors, medical practitioners (particularly psychiatrists), psychologists, physicists, scientists and the like whose evidence counsel assisting the Board considered should be heard because scientology purported to deal with a number of subjects which were their specialities.

Shortly after the Board commenced its regular sittings in February, 1964, the exceptionally wide scope of the terms of reference began to be appreciated and it became evident that the Inquiry would be of longer duration than originally expected. The subject-matter of scientology is dealt with in many thousands of pages of books, magazines, and pamphlets and in perhaps 30 million spoken words recorded on tapes. While, of course, the Board did not read and listen to all that material, much of which was repetitive, the subject matter of the Inquiry involved an investigation of a great variety of theories and practices and the effects, good and bad, of those practices.

After some 40 witnesses called at the request of the scientology interests had given evidence in favour of scientology (including Williams, whose evidence to that stage had occupied nineteen days and was not concluded) the Board heard evidence from several witnesses from each of the two categories mentioned above, namely, those opposed to scientology and expert witnesses.

Thereafter, the Board resumed hearing pro-scientology witnesses. At the beginning of September, 1964, the Board, after consultation with counsel, saw the prospect of finishing the hearing of

evidence by the end of September or, at the latest, early in October. It was with surprise, therefore, that on the 9th September, 1964, which was the one hundred and first day on which the Board had sat, counsel for the scientology interests complained that "at this stage we have not heard the case of my clients at all, and that one hundred days have been taken up hearing hostile evidence". (This manifestly incorrect statement deserves no further comment than that, at that stage, 56 proscientology witnesses had given evidence out of a total of 93 witnesses then heard.) Counsel then announced that he had somewhere in the vicinity of 50 more witnesses whom his clients desired to be called. (These had not even been hinted at when consultations had taken place as to the prospect of finishing the evidence by the end of September or early October.) The additional evidence which it was proposed to call, so counsel said, "ranges from some witnesses who wish to give evidence of a general nature against that already given to witnesses who are medical practitioners and who hold various University qualifications who will be giving expert evidence. They will cover the whole range of matters already dealt with in a way hostile to scientology".

Though the announcement that the scientology interests wanted many more witnesses called was entirely unexpected, the necessity of hearing them was, of course, evident, and thereafter the Board heard evidence from an additional 44 pro-scientology witnesses called at the request of the HASI. The promise that there would be witnesses giving expert evidence in support of scientology had not been fulfilled by the end of November, 1964, when there were no more witnesses whom the scientology interests desired to be called. (The significant failure by the HASI to call any evidence to refute the evidence of the expert witnesses is dealt with in Chapter 7.)

Thereafter, on the 30th November, 1964, counsel for the HASI announced that, upon instructions from his clients, he was withdrawing from further participation in the Inquiry. This was at a stage when all the evidence which the HASI had wanted to place before the Board had been received, and when the only evidence then remaining to be heard was that of one expert witness attending that day and of a few witnesses to be recalled to complete minor details of their earlier evidence.

Co-operation by the HASI

At the beginning of the Inquiry, the HASI hailed the appointment of the Board with great jubilation, proclaiming that the Inquiry would completely vindicate scientology and dianetics. In fact, Hubbard went so far as to declare that the Board had been appointed in response to scientology's demands. The HASI appeared to co-operate in the Inquiry by being represented before the Board and by complying with requests by the Board for literature and the like. It readily provided facilities for counsel assisting the Board, the Board secretary and an audit officer to examine its records. It presented a number of demonstration sessions of scientology auditing for the instruction of the Board and also provided facilities for the playing of a selection of Hubbard's taped lectures.

It was, however, noticeable that, as the inquiry proceeded, the general attitude of the scientology interests, while appearing to be helpful and informative, tended sometimes to cloud and confuse issues by too great attention to non-essentials and extremely lengthy expositions of certain aspects of scientology. Later reflection suggests the probability that this approach was deliberate and was really designed, notwithstanding the apparent readiness to co-operate, to so exhaust the patience of the Board or so overwhelm it with minute detail that it would be persuaded against prolonging the Inquiry until the conclusion of all relevant evidence. Such a devious plan may well have been inspired by the Brain-washing Manual (a remarkable document often referred to by Hubbard and dealt with at length in Chapters 12 and 28) which scientology appears to have followed in other respects, for that manual gives directions as to "conduct under fire", and directs that one "must have to hand innumerable documents which assert enormously encouraging figures Not one of these cases cited need be real, but they should be well documented and printed in such a fashion

as to form excellent court evidence". The manual further directs that "In order to defend . . . a great complexity should be made of psychiatric, psychoanalytical, and psychological technology. Any hearing should be burdened by terminology too difficult to be transcribed easily". The manual further directs that:

"Various types of in sanity should be characterized by difficult terms. The actual state should be made obscure, but by this verbiage it can be built into the court or investigating mind that a scientific approach exists and that it is too complex for him to understand. It is not to be imagined that a judge or a committee of investigation should inquire too deeply into the subject of insanity, since they, themselves, part of the indoctrinated masses, are already intimidated "

Mental Health

The task of the Board in giving effect to the very extensive terms of reference inevitably involved the investigation of certain aspects of mental health, but only so far as they related to scientology and its practice. The terms of reference did not involve investigation into the techniques or practices of psychiatry or psychology as such. From time to time attempts were made by the scientology interests to widen the scope of the Inquiry into a general investigation of the medical profession, or more particularly the practices of psychiatry and psychology, in an endeavour, it would seem, to deflect attention from scientology by attacking orthodox practices of medicine. The Board consistently declined to pursue such irrelevant exercises.

Glossary and Abbreviations

A glossary of scientology terms appears in Appendix 6.

Simplicity may be achieved by the abbreviation of common words, names, and phrases which will constantly recur in this Report. Unless some reason exists for amplification, the following abbreviations will generally be used for the words shown opposite them:

Com. Mag. Communication Magazine, official magazine of the Melbourne HASI.

Dianetics: MSMH "Dianetics: The Modern Science of Mental Health", the first major work by Lafayette Ronald Hubbard on dianetics.

Gogerly Ernest Denzil Stephen Gogerly

Gogerly, Mrs. Mrs. Catherine Mary Gogerly, wife of Gogerly.

HASI Hubbard Association of Scientologists International, the name given to the scientology organization, which has branches in several countries. This abbreviation, unless the context otherwise requires, will be used generally to refer to the Victorian headquarters of the organization and its administration at 157-59 Spring Street, Melbourne.

HCO Hubbard Communications Office, the organization whereby the founder of scientology keeps in communication with scientologists throughout the world and through which he promulgates directives, &c.

HCO Bull. Hubbard Communications Office Bulletin, a bulletin issuing frequently from HCO, containing directives, instructions, exhortations, reports on discoveries, &e.

HCO Infm. Lr. Hubbard Communications Office Information Letter.

HCO Pol. Lr. Hubbard Communications Office Policy Letter.

HGC Hubbard Guidance Centre, a section of the Melbourne HASI.

Hubbard Lafayette Ronald Hubbard, founder of scientology.

PAB Professional Auditors Bulletin.

Tampion Ian Kenneth Tampion.

Tampion, Mrs. Annie Tampion, wife of Tampion.

Williams Peter Rogers Williams.

Williams, Mrs. Penelope Elizabeth Berncastel Williams, wife of Williams.

The Anderson Report CHAPTER 2: INTRODUCTION

The Pattern of the Report

In order to facilitate the reading of this Report, it is considered advisable to set out in this introductory chapter a very brief review of what follows and a conspectus of the contents of the Report, so that initial references to scientology theory and practice, which occur before these aspects are fully dealt with, will be more readily understood.

In Appendix 6 is a list of abbreviations used in this Report and a glossary of scientology terms.

The subject matter of the Report has been divided into chapters. Though there is inevitably a degree of overlapping in the topics dealt with, an attempt has been made to deal reasonably completely in each chapter with a specific aspect of scientology. Where necessary, however, there are cross references to other chapters. The latter part of this Chapter is devoted to a reference to salient aspects of the Report.

Though the whole of the Report deals with scientology generally, Chapter 3 makes some reference to Hubbard's descriptions of scientology and the claims which he makes for it.

Chapter 4 deals with the organization of scientology, both on a world-wide basis and in Victoria.

Chapter 5 deals with the financial aspects of scientology in Victoria. This is given in some detail and serves to show the ramifications of scientology. One significant feature is the control that Hubbard exercises over scientology, extending even to retaining to himself and his wife exclusive control of at least two bank accounts in Victoria in which money is regularly deposited. At the conclusion of Chapter 5 reference is made in summary form to certain important financial aspects of scientology in Victoria.

Chapter 6 relates to Hubbard, the founder of scientology, and gives biographical and other details about him. His importance to scientology cannot be over-emphasized. Hubbard makes highly extravagant claims to learning and he parades as a scientist of colossal capacity. He claims also that the theories and practices of scientology are based upon scientific research. All these claims are entirely unjustified; his own deficiencies and those of his "research" are dealt with in Chapters 7 and 8.

Scientology developed out of a pseudo-science called "dianetics", which is still a significant and dangerous part of scientology. Chapters 9 and 10 deal with dianetics and its association with scientology. The link is an intimate one and, though scientologists claim that dianetics is not now practised, such a claim is false.

Chapter 11 deals with sufficient of scientology theory to give a fair picture of its entirely ridiculous qualities. There is the danger that when one seeks to classify and codify nonsense, the subject matter may acquire a deceptive quality of reasonableness. It would be unfortunate if any such quality were bestowed upon scientology theories by reason of the attempt in Chapter 11 to codify them.

Chapter 12 deals with the teaching of scientology.

Chapter 13 deals in some detail with scientology processing or auditing. A number of processes are dealt with at length.

Scientologists frequently use an electric instrument, called an E-meter, in processing. Very extravagant but unjustified claims are made for this meter, which is dealt with in Chapter 14.

Interest in and adherence to scientology is procured by cunning methods of deceptive advertising and carefully calculated and developed techniques for mental coercion and restraint. These are dealt with in Chapter 15.

Chapters 16 and 17 deal with a person's progress in and dedication to scientology.

One of the most alarming features of scientology, which is denied by Hubbard and his followers but abundantly proved by reliable and convincing evidence, is that beyond its elementary stages scientology techniques are mainly, or almost identical with, those of command or authoritative hypnosis and are potentially harmful to mental health. Chapter 18 illustrates the parallel between scientology and hypnotic techniques.

Scientology, in an indirect way, claims to heal. Chapter 19 deals with those claims.

Other benefits claimed by scientology are dealt with in Chapter 20.

In contrast with Chapters 19 and 20 are Chapters 21 and 23, dealing respectively with the actual general effects of processing and its danger to mental health. Of the many evil aspects of scientology, its danger to mental health is the most serious. Included in the phrase "danger to mental health" are many other evils of scientology, both moral and social, for these are often interwoven with mental ill health.

Chapter 22 deals with the viciousness with which Hubbard vilifies the medical and psychological professions, and engenders in his followers fear and hatred of those professions.

An alarming feature of scientology is its potentiality for coercion. This is dealt with in Chapter 24.

The titles of the next four chapters are self-explanatory: -

Chapter 25 - Moral laxity

Chapter 26 - Family discord

Chapter 27 - Scientology and religion;

Chapter 28 - Scientology and politics.

Scientology is a perverted form of psychology. Due to the absence of statutory control of those who profess to practise psychology, opportunities are afforded to charlatans and other persons without adequate qualifications to engage in exploitation of the gullible and impressionable. This aspect is dealt with in Chapter 29.

The conclusions of the Board, sufficiently evident from the whole tenor of the Report, are summarized in Chapter 30.

As directed by the terms of reference, the Board makes, in Chapter 31, the recommendations which it considers proper concerning scientology as known, carried on, practised and applied in

Victoria.

Conspectus of the Report.

This Report covers a very wide canvas and deals with many phases of a body of learning and a collection of techniques, compendiously described as "scientology", which appear strange and unreal to the average individual. Scientology claims to be "that branch of psychology which treats of (embraces) human ability." Its theories are, however, generally impossible, peculiar, and novel to itself. In that it deals with a variety of real and imagined activities and conditions of the mind, scientology may be classed as a kind of psychology, though often irrational and perverted. Its techniques are a conglomeration of procedures based on misconceptions of psychiatry, psychology, psycho-analysis and other sciences, as well as a very heavy leavening of procedures that are its founder's own brain-children.

The word "scientology" was first used probably in the middle thirties of this century by Lafayette Ronald Hubbard, the founder of the "science" of scientology, which is the subject matter of this Report. In about 1936, Hubbard, then a young American author, is said to have formulated certain "axioms" which later developed into the axioms of scientology and the axioms of "dianetics". The word "scientology" was not used, except perhaps sporadically, until 1951. In that year, Hubbard claims to have succeeded, at Phoenix, Arizona, U.S.A., in proving by "scientific rather than religious or humanitarian" means the existence of a spirit, akin to the soul, which he termed the "thetan", and thereupon to have founded or discovered scientology.

Scientology grew out of dianetics, which Hubbard had founded a year or so earlier. Dianetics, which he described as "the modern science of mental health", started life fully-fledged in 1950, when he published his first major book on the subject named, *Dianetics: The Modern Science of Mental Health*. The following year he published *Science of Survival*. In these two books he set out and developed his main theories concerning dianetics. He claims that the first book was based on a vast amount of experiment and data and that it was written in three weeks. The claim is made for both dianetics and scientology that each is a science, validated by its "workability". No scientific evidence was called to support such claim. On the other hand there was a large and impressive body of expert evidence which refuted such claim in no uncertain terms. The Board is satisfied on the evidence that there is no justification for this claim in respect of either dianetics or scientology. (See Chapters 7, 8, 9.)

In dianetics, Hubbard claimed that the hidden source of all psychosomatic ills was the "engram", and that dianetic skills had been developed for the invariable cure of all psychosomatic illnesses, "which constitute 70 per cent. of man's illnesses." The "engram" was defined as a "moment of unconsciousness containing pain or painful emotion and all perceptions and is not available to the analytical or conscious mind as experience". The engram was said to be received in one's past, the "past" in dianetics being either in this lifetime, or more likely and more frequently in the prenatal period, i.e., during the mother's pregnancy. (See Chapters 6, 9.)

In dianetics, Hubbard dealt with "the clear" who was the optimum individual and was the goal of dianetic therapy. A clear was an individual who was entirely without ills or aberrations, with an intelligence high above the current norm. In later years the concept of a clear developed and changed somewhat, but the state of clear became seductively more desirable of attainment, yet remained for all an elusive will o' the wisp. Adherents of dianetics and scientology vaguely thought and still think Of the state of clear as being a worthy goal and something highly desirable and, as "preclears", they strive to attain it. For many years Hubbard, both in dianetics and scientology, has been promising his followers that they would very soon be clear; but few have been recognized by

scientology as having attained that state or, if they attained it, to have remained long in it, in spite of the award of a silver bracelet by Hubbard to those whom he certified as clear. Beyond the "clear" lies the ultimate goal of scientology, namely, "operating thetan" or "OT", said sometimes to be a purely hypothetical state but, according to Hubbard, readily attainable if only scientology practitioners could do their jobs properly. The accolade on attaining such an ecstatic state is a gold bracelet bestowed by Hubbard. Hubbard has claimed that the state of OT has been attained, but that for the time being those so exalted are remaining incognito. (See Chapter 11.)

Many of the processes of dianetics and scientology are essentially hypnotic and induce hallucinations. (See Chapter 18.) As the practice of dianetics developed, processing purported to bring to light engrams which related to periods long before the prenatal period. Hubbard assumed that these hallucinations were real and he developed the theory that the incidents had occurred in a "past life", that is, in a previous lifetime of the person. On this basis he claimed to have discovered the thetan, which was really the person, the "I", which he declared had survived the deaths of countless bodies to which it had been successively assigned over the untold trillions of years of the thetan's existence.

In 1950 Hubbard had founded the Hubbard Dianetic Research Foundation in California. In 1951 differences between him and the board of control of that organization caused him to withdraw from it, and he then founded scientology. He proceeded to build up his scientology organization to promote the theories and practices which his prolific mind and pen produced. He is a man of great imagination and energy, and Hubbard Association of Scientologists International (commonly referred to as "the HASI"), which he incorporated in 1954 in Phoenix, Arizona, now has branches in many countries of the Western World. (See Chapters 4 and 6.)

Hubbard is the governing director of and is all powerful in his organization. He directs and controls everything and everyone to an astonishing degree. Though this is a report of an inquiry into scientology, it must necessarily and equally be a report on its founder and master, for his personality pervaded the whole Inquiry and an understanding of Hubbard is necessary for an understanding of scientology. Hubbard will be referred to constantly in this Report. Though Hubbard did not attend the Inquiry and presently resides at East Grinstead, Sussex, England, and was last in Victoria in 1959, scientology as known, carried on, practised and applied in Victoria is as much under Hubbard's direction and control as though he had been, for the past eight and more years, sitting in the chair behind the desk in the room at the Melbourne HASI which is always kept ready for him. Hubbard's writings, which are very voluminous, appear in several books, many magazines and pamphlets, a multitude of bulletins and many other communications. The Board considered that an important part of its inquiry related to Hubbard, the power which he exercised over the organization in Victoria, the influence which he wielded over members of the community who became interested in scientology and the literature and other communications and instructions emanating from him which were the foundation of scientology practice in Victoria. (See Chapters 4 and 8.)

Hubbard's theories developed along astonishing lines; he had been a successful science fiction writer before 1950, and it is fair to say that in his scientology writings thereafter he continued to demonstrate his remarkable flair for fiction and fantasy. He wrote thousands of pages and spoke millions of tape-recorded words about thetans, operating thetans, past lives, time tracks, engrams, implants, galaxies, events of countless trillions of years ago and a multitude of other similar topics. He propounded axioms, factors, logics and prelogics; and he wrote about dynamics and emotional tone scales. Parallel with and incorporated in his theoretical writings are the details of ever changing techniques which he claims to have developed by research at his home at Saint Hill Manor. (See Chapter 8.)

So far at least as Victoria is concerned - but the same seems to be generally true for other countries, too - scientology does not claim to be a religion. It claims to be an exact science. Hubbard claims, amongst other things, to have discovered by exact scientific method rather than by humanitarian or other means the existence of what he called the "thetan", which is imprecisely defined in scientology but which, for convenience of forming a concept of it, is sometimes described as being like the soul in Christian belief or a spirit or something to that effect. Hubbard claims that scientology changes no man's religion and changes no man's politics and that a person could consistently belong to any religious denomination and at the same time be a scientologist. In America, there are what are called "founding churches of scientology", founded by Hubbard. In Victoria there is no such establishment as the founding church of scientology, though in the early days of scientology in Victoria some of its adherents referred to themselves as "the reverend" and for a time there was even "a bishop", the illusion being heightened by their clerical garb. The title of "Doctor of Divinity", designated by the letters " D.D.", is used by some scientologists and on occasions even by Hubbard himself. This "doctorate" was bestowed by some Hubbardian institution in the United States. But these and a few other trappings do not make scientology a religion or its practices or beliefs religious. Nor do belated claims that scientology is a religion, made towards the close of the Inquiry in an attempt to obscure the real issues involved in the Inquiry, transform scientology into anything even remotely religious. (See Chapter 27.)

Dianetics specifically claimed to cure mental and physical ills. Though it was said that dianetics was not now practised, that is not so. Dianetics is said to be a part of scientology and in Victoria scientology has been actively practised in the fields of mental and physical health. The HASI claims to be "the world's largest mental health organization" and has frequently claimed that it "proofs people against mental and physical ills." The Board heard a large body of expert professional evidence which proved conclusively that from a psychiatric and psychological point of view there are grave dangers to mental health attendant upon treatment by scientology, and that the extravagant claims that dianetic and scientology processing could effect cures of mental and physical ailments or would proof against illness were entirely unjustified. The scientologists have devised means whereby, though not in terms claiming to heal, they nevertheless manage in a very skilful and deliberate way to spread the belief that scientology cures both mental and physical ills. Dianetic theory and practice are so interwoven with scientology theory and practice as to be often indistinguishable. All dianetic literature is recommended reading for scientologists and the two names are constantly linked. (Chapters 3, 9, 10, 19.)

From an early stage in dianetics, and thereafter consistently in dianetics and scientology, many of the techniques used were and still are potentially harmful hypnotic processes. They are administered by the wrong people, on the wrong people, for the wrong reasons and in the wrong way. Hubbard has said that his techniques are not hypnotic and some witnesses have stated they were not, but they plainly are, as the evidence of expert psychiatrist witnesses made clear; and it is obvious that in Victoria for the last ten years and more an increasing number of scientologists, quite unfitted for the task, have been meddling in an unskilled and dangerous way with the mental health of thousands of persons, some of them in urgent need of proper psychiatric treatment and care. (Chapters 18, 21, 23.)

Scientology practitioners are called "auditors". These auditors are often young persons, some still in their teens or just beyond, sometimes ill educated, and with only a few months' training in scientology techniques to equip them for their task of taking control of the minds of their patients. They are persons who earlier have found their way into scientology, lured by the promises of scientology which they accepted against reason and without question. In some instances, auditors are persons who themselves are or have been mentally ill and this circumstance has predisposed them to espouse scientology. (See Chapters 12, 13, 15, 16, 17.)

Scientology "infiltrates" in an insidious way. In its early stages, its practitioners pretended to be ministers of religion and, so garbed, they effected entry to various places. This practice is now discontinued. Instead, a system of apparently innocuous advertising has taken its place. Scientology organizations, under various names, advertise free lectures and courses, where "the able are taught to be more able", and promises are made that techniques are available to increase IQ and improve personality. None of the theories or advanced practices of scientology is mentioned in the advertisements. The appeal is made in the advertisements to the anxious, the worried, the inadequate, the lonely, the gullible, and (though he may not know of his condition) the mentally ill. The pleasant surprise of the free lectures and the impression made by the apparent efficiency and the glibness of the HASI staff hold promise of greater things to come, the unsuspecting individual is persuaded to undertake processing or a course of training and eventually, even if he realizes the deception, he is generally so indoctrinated as to be beyond caring. (See Chapters 15, 16, 17.)

Scientologists frequently have a strange and obsessive dedication to scientology, which enables them cheerfully to make considerable personal and economic sacrifices in the interests of scientology. (See Chapter 17.) This dedication arises because of the bemused state produced by its processes and the insidious appeal which it makes to those persons who seek escape from the realities of everyday existence and its problems. Adherence to scientology is sustained by a mixture of mental conditioning and fear. The mental conditioning is effected by hypnotic techniques and procedures which have a "brainwashing" effect. The fear develops because of the dependency in the preclear on the HASI which the various processes engender and because of the realization that the organization possesses detailed written records of intimate and shameful disclosures made by the preclear during processing. The Board did not hear any evidence of blackmail in the sense that money was extorted under threat of exposure; but it is satisfied that the knowledge that the organization possesses such records acts as a very potent force in compelling obedience to the organization, which generally involves continuing communication with it and the undergoing of repeated sessions of auditing. (Chapter 24.)

Scientology possesses certain features which are morally highly undesirable. Many of the processes are of a hypnotic nature and normal inhibitions and restraints are relaxed, with the result that when matters of sex and perversion are introduced into processing, as is frequently the case, they are discussed and probed and dwelt upon sometimes for hours on end. The quality of the filth and depravity recorded in the HASI files as being discussed between preclears and auditors almost defies description. One highly disturbing feature of the dangerous hypnotic techniques used is that the perverted and erotic disclosures of a preclear are very frequently merely hallucinatory, yet they persist as realities to the preclear who is thereby often morally disturbed. (Chapter 25.)

Though scientology affords for some people an escape from the realities of life, it is not in any way a healthy diversion or recreation. It is quite the reverse. It robs people of their initiative, their sense of responsibility, their critical faculties and sometimes their reason. It induces them mentally to debase and enslave themselves. It has done, and is capable of doing, grave harm to the mental and physical health of its victims by the practice of dangerous procedures and by persuading them that orthodox medical care and treatment, which some of them may urgently require, is evil and to be avoided. It consistently relieves them of large sums of money in payment of fees for processing and training. (See Chapters 16, 17, 21, 22, 23.)

The income of the Melbourne HASI for the six years ended the 30th June, 1963 was well over a quarter of a million pounds. During that period many preclears paid large sums of money for auditing and other services; in several individual cases the sum paid was substantially in excess of £1,000. (See Chapters 5, 17.)

Many of the theories and teachings of scientology are so fanciful that the reaction of the normal individual on hearing them is generally one of amusement and incredulity. On this account, the impression may exist in the community that scientology is just harmless nonsense and its followers merely queer people? that its theories are foolish but funny and that not much harm is being done by allowing silly people to have their silly beliefs and carry on their silly practices. Such an attitude is welcomed by the scientologists, for it serves to obscure the real nature of scientology. A tolerant "live and let live" attitude is what scientologists fervently desire, for it is on the inertia of the community, generated by tolerance and polite disinterest, that scientology thrives.

Hubbard is well aware of the value to him and his organization of good-natured tolerance and incredulity, and he trades on it. In HCO Bull. of the 29th July, 1963, he writes,

"Incredulity of our data and validity. This is our finest asset and gives us more protection than any other single asset. If certain parties thought we were real we would have infinitely more trouble. There's actual terror in the breast of a guilty person at the thought of OT, and without a public incredulity we never would have gotten as far as we have. And now its too late to be stopped. This protection was accidental but it serves us very well indeed. Remember that next time the ignorant scoff."

Tolerance of the beliefs of another, however silly they may seem to be, is one thing. Inaction, when the practice of those beliefs is positively harmful to others and may be permanently damaging, is quite another thing. The damage done and likely to be done by the practice of scientology is alarming and if this Report fails to make clear the great danger to the mental health of the community which scientology has been and is, the appointment of the Board will have been in vain.

The Anderson Report CHAPTER 3 SCIENTOLOGY AND CLAIMS MADE FOR IT

Although no adequate definition of scientology is found in Hubbard's millions of written and spoken words, many extravagant claims for it have been made, some of which may appropriately be considered at this stage.

"Scientology," writes Hubbard in *Scientology: 8-80*, "is a new word which names a new science. It is formed from a Latin word, 'scio', which means know, or distinguish, being related to the word 'scindo' which means cleave. (Thus, the idea of differentiation is strongly implied). It is formed from the Greek word 'logos', which means the word, or outward form by which the inward thought is expressed and made known: also, the inward thought or reason itself. Thus, scientology means knowing about knowing, or science of knowledge."

Such is the arrogance of Hubbard that he assumes that, having explained with some pomp that "scientology" is a hybrid term formed of the misalliance of two classical words, scientology is actually that which he says it means.

Scientology, says Hubbard, thus means "the study of knowledge"; it is the "science of knowing how to know; it is a route to knowledge, it is not knowledge itself"; it is a "gnostic faith in that it knows it knows"; "the science of scientology is an engineering science"; it is "a precision science"; it is "the common people's science of life and betterment". In 1965 it claims to be the science of "learning how to learn".

From Hubbard's writings and lectures comes an almost inexhaustible flow of pretentious claims. Scientology is "the first major breakthrough by the exact sciences in the field of the humanities - dedicated in the finest tradition of human dignity and freedom espoused by brilliant men in all periods of man's ascendancy over his relationship to the animal"; it is "the only successfully validated psycho-therapy in the world"; it is "the first precision science in the field of the humanities". It lays claim to a great number of other "firsts", which need not be enumerated.

Scientology is claimed to be "that branch of psychology that treats of human ability". It is said to do "things for people where nothing has been done before. It makes people well from illnesses which were once considered hopeless. It increases their intelligence. It changes their competence and betters their behaviour. In addition to those it brings them a better understanding of life".

"Scientology, the science of knowing, came out of the same crucible as the atom bomb. It is developed for good, not evil. For that reason Scientology has been called that branch of atomic science which deals with human ability. As its founder has been trained as one of the first nuclear physicists it can been seen why."

"Scientology is a system of organized axioms resolving problems of the spirit, life and thought, developed through the application of the methodology of the exact sciences to the humanities by L. Ron Hubbard, an American engineer and philosopher."

"The science of scientology is a broad well articulated science in the fullest sense of the word, which covers the mind, spirit, livingness and beingness of man. Scientology is the first and only

Anglo-American development in the field of the human mind."

"Scientology covers livingness as a complete subject. Dianetics covers only that part of livingness which applies immediately and directly to the human mind."

"Scientology brings order and knowingness out of confusion and chaos. Scientology is an exact science of the human mind and spirit. It used the methodology of mathematics and nuclear physics to create precise knowingness in a hitherto unknown field. Scientology and Dianetics are the only Anglo-American developments in the field of the mind."

"Scientology is psychology in a different form, or rather a step further ahead. The scientologist unlike the psychoanalyst needs to know nothing about you. Your problem is everybody's problem. His work is not primarily directed towards the mentally ill, but rather for those who are able in that they may become more able. After 14 hours processing the average normal person's IQ can be raised by anything up to 20 points."

"Scientology is used on life, and its forms and products. The chief uses of scientology are in the fields of education, organization, mental disability, social order and religion. Scientology is the first to give scientific meaning to these." "A Scientologist is a first cousin to the Buddhist, a distant relative to the Taoist, a feudal enemy of the enslaving priest and a bitter foe of the German, Viennese and Russian defamers of Man."

Evidence given before the Board has shown that these claims are not justified. However, the constant repetition of these assumed qualities and virtues has been so effective that the gullible have accepted their validity without reservation, question or criticism.

Two of Hubbard's books are *Scientology: 8-80* and *Scientology: 8-8008*; the "8-8", explains Hubbard, stands for "Infinity-Infinity" written upright, and the "0" represents the static, theta, 8.

Scientology was founded by Hubbard in 1951, though the word " scientology" was used by him probably as early as 1935. One claim made for Hubbard by Jack Horner in Summary of Scientology was that as early as 1935 Hubbard had already formulated some of the basic axioms of scientology. Hubbard himself in an article entitled "Scientology: A New Science", published in Scientology, issue 28-G about 1954, writes, "The basic science was named 'Scientology' in 1938. In 1947 L. Ron Hubbard changed its name to 'Dianetics' in order to make a social test of publication and popularity. The test completed, in 1952 he changed the science back to its original name, scientology. This was done to inhibit its being "monopolized for private purposes." There are several inconsistencies in references which the Board was given as to the early history of scientology and dianetics. The foregoing quotation from Hubbard himself clearly shows that dianetics and scientology are inseparably interwoven parts of the same subject. Whatever be the precise date on which names were changed, scientology from 1951 onwards developed out of what dianetics was at the time, and the two have remained inseparable and, as to substantial parts, indistinguishable. It will be necessary to consider at length various aspects of dianetics. At this stage it is perhaps sufficient to mention that fundamental to both dianetics and scientology is the proposition that a person has at least two minds, the conscious or analytical mind, and the reactive or irrational or subconscious mind. (Reference is sometimes made to a third, the somatic mind). It was Hubbard's contention both in dianetics and scientology that at the seat of every aberration was a basic cause, a basic-basic. In dianetics, this was said to be the engram, most frequently experienced in the prenatal period; in scientology, it has been said at various times to be an engram in an earlier lifetime, facsimile one, an implant, a non-cellular engram, the goalsproblems-mass, on the last-named of which newly discovered basic-basic Hubbard seems

currently to be working. These various causes were advanced in succession as being the new basic cause of aberrations and, as each such basic cause was discovered, it was proclaimed as the "ultimate" or a "breakthrough" or some other last word. Around aberrations of the mind and the causes which Hubbard says he has discovered for such aberrations, he has built a fantastic body of theory and has developed numerous practices or techniques for the resolving of such aberrations. In scientology, Hubbard claims to have found that the aberrative cause might have existed long before the prenatal period, that it could have occurred in some previous existence; and to explain how the past could be linked with the present he introduced the thetan.

The essence of scientology was said by Williams to be "the stating of an objective or goal along the lines of increased efficiency or well being, and the process used to obtain that object". By a process was meant a mental exercise designed to delete aberration from the human mind and to improve the performance of the innate faculties of the human mind. The "mental exercises" of scientology take many forms which, despite vehement protests to the contrary by scientologists, are, in a great number of cases, evidently hypnotic or based on hypnotic techniques. Some Of these "mental exercises" are of a repetitive nature, involving sustained questioning of the individual over many hours and even days on many topics, often of the most intimate kind. This processing frequently produces in the subject mental pictures which are in reality of a hallucinatory nature, but which are treated by the scientologist as being recollections of actual experiences of the thetan and as conclusive evidence of past lives and other scientology theories. These hallucinations frequently persist with the individual as realities and expert medical evidence makes it clear that they are potentially harmful.

The processes are conducted by scientologists who are called "auditors", and the individua1 who is undergoing processing is called the "preclear". There are numerous processes which are developed and changed from time to time, but whatever process is run in an "auditing session", the procedure "in session" is carried out on unvarying lines. Each session commences with a strict ritual and ends with a strict ritual. On occasions, an electrical instrument which is called the "Emeter" is used. Almost magical powers are attributed to the E-meter which is held in great reverence by scientologists and regarded with awe by preclears as being infallible in its determination of the truth or otherwise of statements made by them.

Extravagant and unjustified claims are made for scientology processing. Many of the processes are potentially detrimental to the mental health of the persons processed. In some cases harmful results are produced, and such apparent success as scientology processing may superficially appear to achieve is generally illusory. It is clear from the evidence that scientology involves the wholesale use of potentially harmful hypnotic procedures conducted by ignorant and sometimes mentally ill persons who are quite unaware of the potential dangers of the practices and the great harm which they may be doing. This Report deals with numerous features of scientology and its practices. These introductory remarks are designed to indicate broadly the general picture to be presented.

One difficulty which faces anyone concerned to obtain a comprehensive picture of scientology is that since 1956 no attempt has been made to produce a comprehensive and unified thesis a the theories and practices of scientology. From about 1956 the dissemination of scientology theory and practice has been by means of small books, periodical magazines, pamphlets, system of communications called variously bulletins, policy letters, &e., and tape recordings t Hubbard. Such publications as merit the description of books, as distinct from pamphlets, hat dealt only with limited aspects of scientology. In the result, there exists in an uncodified & unclassified form tape-recordings which are said to contain 30,000,000 words spoken by Hubbard and writings by Hubbard which run into tens of thousands of pages. The Herculean task reducing to manageable

size the content of these Hubbardian emanations must await the labour of a dedicated scientologist with the time, money, capacity, and compulsion to undertake such a useless exercise.

The Anderson Report CHAPTER 4 THE ORGANIZATION OF SCIENTOLOGY

On an International Basis.

Scientology is organized on a world-wide basis. Its "world", however, does not include communist countries, for Hubbard is very hostile to Communism.

Administrative international headquarters are at Saint Hill Manor, East Grinstead, Sussex, England, where Hubbard resides with his present wife and family. The headquarters, it is said, undertake research, provide technical information, advice, and managerial services, publish research material and conduct special courses. The organization of scientology is somewhat complex, for the parent company appears to be a company incorporated in Phoenix, Arizona, U.S.A., on the 19th May, 1954, under the name of Hubbard Association of Scientologists International. There is also a company incorporated in England under the name of Hubbard Association of Scientologists International Limited.

The articles of incorporation of Hubbard Association of Scientologists International, i.e., the American company, disclose that the company is in the nature of a company limited by guarantee to an amount of indebtedness not exceeding 250,000 American dollars. The company is empowered among other things to have and enjoy all the powers and privileges granted by law to non-profit corporations organized under the laws of the State of Arizona; to conduct and carry on any and all scientific research and more particularly the dissemination and advancement of knowledge of the human mind in mental, psychosomatic, and allied fields; to conduct schools and classes; to establish clinics; to publish and have published books, articles, letters, papers, magazines, and other periodicals; and to do and perform any and all acts not inconsistent with the laws of the State of Arizona.

There are central organizations of scientology in London, Washington, Los Angeles, Auckland, Melbourne, Cape Town, Sydney, and Perth. There are what are called city offices in New York City, San Diego, Port Elizabeth, Durban, Johannesburg, Berlin, Paris, Adelaide, Detroit, Dallas and Twin Cities, and Toronto. Hubbard claims there are franchise holders or field auditors in various parts of most of the continents and in various levels of governments and armed services in several countries.

Highly efficient lines of communication are maintained between Hubbard's headquarters at Saint Hill and the central organizations. The central organization in Melbourne is in direct telex communication with Hubbard and it is understood that other organizations are similarly linked. An English company, Hubbard Communications Office (World Wide) Ltd., is the channel through which all written communications from Hubbard are despatched. The function of HCO (WW) Ltd., is described in HCO Infm. Lr. of the 18th February, 1964, in these terms: "HCO (W.W.) Ltd., is concerned with the Organizations of Scientology on a world wide basis. It deals with Ron's personal communications to and from the HCOs, Orgs and Field, and with the Franchise Holders. It sends out Ron's policies and technical data. It has its own direct line to the HCOs, Orgs and City Offices all over the world." It keeps a close supervisory eye on local scientologists to ensure conformity to a "proper standard of Scientology ethics".

There is an almost constant stream of communications flowing from Hubbard in England, via the HCO, to the various branches throughout the world. These communications contain instructions on a great variety of matters and are variously described as bulletins, policy letters, information letters, technical data, newsletters and the like.

The Hubbard Communications Office, commonly referred to as HCO, is a highly important part of the scientology organization, not only in Victoria but generally throughout the world. The sole proprietor of the business name in Victoria is Hubbard, and the local HCO secretary, in the name of Hubbard, wields great power. Though it has the same address as the HASI in Melbourne, the HCO is a distinct organization.

One witness, a former scientologist, described the HCO thus, "It is a police force which governs the HASI. It is Hubbard's own police force which governs the HASI organizations throughout the universe. It is separate from the HASI and it ensures that the policy of the HASI is carried out, and it has power to hire and fire members of the HASI. Whether it was on paper this way, I do not know ". Allowing for such prejudice as the witness might have had against scientology, he appears to have described fairly well what the evidence, both oral and contained in scientology documents, showed the HCO to be.

Though as a matter of formality the HASI and the HCO are separate legal entities and Hubbard is merely a director of the American company, he is nevertheless the executive or governing director and in fact exercises complete and autocratic control over each organization and all phases of their activities and all members, His name it is that the organizations bear; his "discoveries" and techniques they exploit and practise; his books are their literature; his prestige amongst scientologists is such that he felt the need to direct that students be instructed that he is merely a human being; persistently to disagree with him brings punishment ruthless and severe, no matter how highly placed the offender.

Organization in Victoria.

The organization, direction, and control of scientology in Victoria is effected through and by the American company, Hubbard Association of Scientologists International, which is registered in Victoria under the Companies Act as a foreign company, with its registered office at 157-159 Spring Street, Melbourne.

The English company, Hubbard Association of Scientologists International Limited (formerly Hubbard Association of Scientologists Limited), is also registered in Victoria as a foreign company with its registered office at 157-159 Spring Street, Melbourne. This company appears to have been inactive in Victoria at all relevant times.

During the course of the Inquiry a company, Hubbard Association of Scientologists International (Victoria) Pty. Ltd., was incorporated in Victoria on the 6th February, 1964, with its registered of Ece at 157-159 Spring-street, Melbourne The apparent intention was that this company should take over the duties and functions of the American company but, so far as the Board is aware, it had not done so at the time the evidence was concluded.

It seems, therefore, that at all material times, of three companies bearing Hubbard's name, it has been only the American company which has been functioning.

There are, however, several registered business names which are associated with the Melbourne HASI. The businesses or firms to which these names refer are in some cases inactive. At the time

of the Inquiry at least the following business names were registered with their addresses as 157 and /or 159 Spring Street, Melbourne, though some may now have become de-registered pursuant to the operation of the Business Names Act 1962:

Melbourne Test Centre, proprietor, Peter Rogers Williams;

Test Bureau of Australia, proprietor, Peter Rogers Williams;

Academy of Personal Efficiency, proprietor, Lafayette Ronald Hubbard;

Hubbard Guidance Centre, proprietor, Lafayette Ronald Hubbard;

Academy of Scientology, proprietor, Lafayette Ronald Hubbard;

Hubbard Communications Office, proprietor, Lafayette Ronald Hubbard.

A business name, The College of Applied Philosophy, was registered as recently as the 15th February, 1965, by the Hubbard Association of Scientologists International.

Williams was the continental director of the HASI for Australia, New Zealand, and Oceania, and in that capacity he had general supervision of the offices of the HASI in those areas. Scientology appears to be active in Perth, Adelaide, and Melbourne. At the stage when the Board heard evidence, the HASI was not flourishing in Sydney; this was mainly due to the activities of one, Marcus Tooley, formerly a leading scientologist, who had broken away from the HASI and had set up in business on his own account, practising what he called "Dynamic Psychology" in the same premises and with the same business name, "The American College," as he had earlier used when a loyal scientologist. He took most of the HASI business with him, and, in the language of scientology, he had become a "squirrel". Scientology is active in New Zealand and has been since the early 1950's. "Oceania," it seems, is intended to describe "the northern islands," but its precise meaning in scientology is not material as there are no branches there at the present.

In Victoria, the HASI headquarters are at 157-159 Spring Street, Melbourne, and until some undefined date in 1964, the HASI also occupied premises at 2 Coates Lane, Melbourne, where the Academy of Scientology was conducted. The officer in charge of the Melbourne HASI is the Association Secretary (Assoc. Sec.) who was Gogerly at the beginning of 1964, but, in a subsequent reorganization, Williams assumed that office.

The Melbourne HASI, which is organized and conducted along lines indicated by Hubbard, is divided into two sections, the Technical Division with Tampion as Technical Director, and the Administrative Division of which Gogerly was formerly in charge. Each of these Divisions are further subdivided. The Technical Division is concerned, as its name suggests, with the practical side and deals with the training of students and the conducting of auditing. It is divided into two subdivisions, one of which deals with the courses conducted at the HASI. The elementary courses are conducted in the Department of Personal Efficiency of which Peter Gallop was the director, and the more advanced courses are conducted in the Academy of which Mrs. Tampion was the director.

Also in the Technical Division is the Hubbard Guidance Centre (generally referred to as "HGC") which was under the control of Ross Turnbull; this centre deals with preliminary assessments of preclears, professional auditing, and co-auditing.

The Administrative Division, as its name implies, deals with registration and correspondence, accounts, and materials.

To lend stature to the positions which various staff members hold, titles are freely distributed. Thus there are Directors of the various divisions and subdivisions, such as the Director of Processing, Director of Government Relations, Registrar, Letter Registrar, &c.

A highly complex and elaborate office system involving a tremendous amount of clerical labour is in operation. Files are commenced, it would seem, on every person of whose name and address the HASI can make a note. Thereafter, these persons are regular recipients of letters from various members of the staff, inviting and urging the addressees to come in to the scientology fold. This system of frequent letter writing is one of the "procurement" methods directed by Hubbard to be used as a means of eventually luring interested people into the HASI to undergo processing or training.

Once a person has embarked on a course of processing or training the most detailed records are kept concerning him. Report forms, containing even the most trivial details, are kept in respect of each session, and memoranda on every conceivable subject are filed away. Almost every communication is in writing, even complaints as to favouritism in tea making, and reprimands by a junior officer to a senior staff member for being late.

On the first day of the hearing the Board received as a bulk exhibit over 35 filing cabinets, as well as a large quantity of other files. These were left in the custody of Williams on his personal undertaking to produce them as required. Thereafter, counsel assisting the Board and the Board secretary had access to the files whenever required. These files, containing as they do every detail of every auditing session of every preclear, reveal a most alarming feature of scientology, namely, the tremendous power which the HASI is in a position to exercise over its preclears, for they contain the record of intimate disclosures made by thousands of preclears at a time when normal inhibitions and restraints were suspended and they were revealing their most secret hopes and fears, their shame and grief and guilt. This alarming aspect is dealt with in Chapter 24.

The system in operation involves sending to Hubbard in England each week a report on the processing of every preclear. These reports are retained by Hubbard in England and constitute a further disturbing feature of scientology.

At the time of the Inquiry there were two "district offices" functioning in Victoria. They were the Australian College of Personnel Efficiency at Geclong, of which Eric Bruce Lake, and his wife, June Margaret Lake, were the proprictors; and the Melbourne College of Personal Efficiency at Hawthorn, of which Peter Francis Gillham and his wife, Yvonne Doreen Gillham, were the proprietors.

The Anderson Report CHAPTER 5 FINANCIAL ASPECTS OF SCIENTOLOGY IN VICTORIA

At the request of the Board Mr. John Herbert Joseph Campbell, an officer of the Victorian State Audit Office, made an examination of financial records kept by and the general financial administration of the Hubbard Association of Scientologists International and the Hubbard Communications Office at 157-159 Spring Street, Melbourne. It was made in the early half of 1964 and presented in evidence on the 4th May, 1964, and subsequent dates. It did not, therefore, include the period of operations subsequent to 31st December, 1963, except on a few incidental matters. The investigation was thorough and the evidence given by Mr. Campbell was not challenged by the scientology organization.

As the Melbourne organization has kept separate books and accounts in respect of the transactions of the Hubbard Association of Scientologists International and the Hubbard Communications Office and has employed different public accountants to prepare and audit the financial statements of each business undertaking, Mr. Campbell's evidence, as far as practicable, dealt with each undertaking separately.

Hubbard Association of Scientologists, International.

1. ACCOUNTING.

The system of accounting appears to be based on instructions issued by the Hubbard Communications Office World Wide under the direction of Hubbard. These instructions set out the accounting records which are to be kept and the procedures to be carried out and contain various directions as to financial policy. A very large measure of financial control is exercised by HCO.WW.

The accounting system is largely manual with few mechanical aids. The system is stated by the HASI to be designed for the duties of the Director of Accounts to be carried out by unskilled clerical labour. A method of invoicing is used to account for both income and expenditure. In respect of income, serially numbered invoices are placed in an invoicing machine for issue as receipts and debit and credit notes. A separate invoicing machine is in use for disbursements. An invoice is issued for each payment.

The Director of Accounts is responsible for the receipts and disbursement of all moneys. The duties of the position include the issuing of receipts for cash collections; the posting of debit and credit entries to debtors' accounts; the banking of cash collections; the allocation of fixed proportions of cash collections to the appropriate bank accounts; the preparation of salaries sheets and the payment of salaries; the preparation of cheques for signature; the preparation of an invoice for each payment; the recording of weekly summaries of receipts and payments, and the opening of inwards mail.

Each week's transactions are treated as a single accounting period and copies of receipts and payments invoices together with the summary for each week are filed separately from the records applicable to other weeks' transactions. An orthodox cash book in a continuous form is not kept. A firm of public accountants and auditors is employed to collate the total of each week's transactions; to reconcile the bank accounts; to prepare periodical financial statements, and to carry out the audit

of the books and accounts of the organization.

The following matters indicate a weakness in internal control in relation to cash collections:-

- (a) Contracts entered into by individuals in respect of processing or training are not serially numbered and a contract register is not kept.
- (b) Although account cards are kept in respect of individual debtors, a sundry debtors control account is not in existence. It is, therefore, impracticable to reconcile the outstandings with a control figure and to verify that all of the individual debtors' cards are on hand.
- (c) Cash collections are not always banked in the form received.
- (d) It appears that the Director of Accounts usually opens the mail working alone. A remittance book to record remittances received in the mail is not kept.

2. BANKING.

Definitions of expressions used by the organization and adopted in this Chapter are :- "corrected gross income" is the weekly gross income less refunds or provision for refunds and incidental receipts such as international membership subscriptions, telephone recoups and proceeds from the sales of soft drinks; "proportionate amount" is the weekly corrected gross income less a 10 per cent. levy payable to HCO.WW.

The cash collections are required to be banked weekly.

Bank accounts are maintained at the Australia and New Zealand Bank Limited, Exhibition Street, Melbourne, in respect of the following accounts:-

(a) Hubbard Association of Scientologists International Main Account.

Apart from life membership subscriptions, cash collections are banked in this account in the first instance. After providing for refunds and remitting 10 per cent. of the corrected gross income to HCO.WW, the practice has been and is for all amounts standing to the credit of the account to be rebanked each week in other bank accounts.

The persons authorized to draw cheques on the account are either Lafayette Ronald Hubbard or Mary Sue Hubbard or any two of the following persons:- Peter Williams, Ronald Masters, Denzil Gogerly.

(b) Hubbard Association of Scientologists International Salary Account.

An amount equal to 55 per cent. of the "proportionate amount" is transferred to this account each week. The amount so credited is expended on staff salaries in respect of the Melbourne HASI and the Melbourne HCO. Prior to February, 1963, the weekly proportion so credited was 50 per cent. as the salaries of the HCO personnel were paid by that organization.

The persons authorized to draw cheques on the account are Lafayette Ronald Hubbard or Mary Sue Hubbard or any two of the following persons:- Peter Williams, Ronald Masters, Denzil Gogerly.

(c) Hubbard Association of Scientologists Building Account.

A sum equal to 15 per cent. of the "proportionate amount" is transferred to this account each week. The main purpose of the account is stated to be to provide funds for the purchase of buildings. However, expenditure incurred on the rental of premises and other expenditure at the discretion of Hubbard, including Christmas bonuses for the staff, have been defrayed from these funds.

It is of interest to note the only persons authorized to draw cheques on this account are either Lafayette Ronald Hubbard or Mary Sue Hubbard. Cheques are prepared in the Melbourne office, sent to Saint Hill Manor for signature, and subsequently returned to the Melbourne office for payment to the creditors concerned.

(d) Hubbard Association of Scientologists International Disbursement Account.

A sum equal to 30 per cent. of the "proportionate amount" plus the total of incidental receipts is credited to this account each week. The funds provided are used to defray expenditure not usually chargeable to any of the three previously mentioned accounts.

The persons authorized to draw cheques on the account are either Lafayette Ronald Hubbard or Mary Sue Hubbard, or any two of the following persons:- Peter Williams, Ronald Masters, Denzil Gogerly.

(e) Hubbard Association of Scientologists International Life Membership Account.

Life membership subscriptions of £31 5s. each were received during the financial years 1961-62 and 1962-63 for banking to this account. It appears that the intention was later to convert the subscriptions into shares in Hubbard Association of Scientologists International Limited. However, legal difficulties made this course impracticable and no such conversion has taken place.

The acceptance of life membership subscriptions has been discontinued as a result of an HCO Pol. Lr. of the 5th July, 1963. This letter states:- "Due to projected changes in the corporate status of Scientology organizations, no further sales of HASI Ltd. shares or lifetime memberships are permitted anywhere in the world, effective immediately".

Life membership entitles the holders thereof to receive discounts of 20 per cent. on the cost of processing, training courses, and the purchase of books.

The persons authorized to draw cheques on the account are Lafayette Ronald Hubbard or Mary Sue Hubbard.

Details of transactions within the account are given hereunder:-

1961-62

Receipts Subscriptions (270 at £31 5s. each) Disbursements	£ s. d. 8,437 10 0 £ s. d.
Loan to HASI, Melbourne	300 0 0
Loan to HASI, Sydney	231 0 0
Loan to HASI Ltd.	1,480 2 11

Refund of membership 31 5 0

Bank Charges 1 7 6
2,043 15 5

Cash at bank, 30th June, 1962 6,393 14 7

1962-63

Credits £ s. d.

Balance forward from 1961-62 6,393 14 7

Subscriptions (46 at £31 5s. each) •1,437 10 0

7,831 4 7

Disbursements

Transfers to main account re-

Refunds 62 10 0

Registration of HASI, Adelaide 44 2 0

106 12 0

Cash at bank, 30th June, 1963

7,724 12 7

Period ended 31st December, 1963:

Credits

Balance forward from 1962-63 7,724 12 7

Subscriptions

Disbursements

Transfers to Disbursement account re-

Cost of Instructor Training, E. Williams (inc. air-fare) 717 17 3

Legal expenses 1,000 0 0

Transfer to main account re refund 1,550 0 0

Refund of membership 31 5 0

3,299 2 3

Cash at bank, 31st December, 1963 4,425 10 4

3. FINANCE.

[•] Does not include an amount of £93 15s., which represents three subscriptions for life membership banked in the Disbursement account.

The following financial statements are attached to this Report:-

- (a) A consolidated statement showing the combined income and expenditure and the profit or loss of the Melbourne HASI and the Melbourne HCO for the period the 7th February to the 30th June, 1955, and for each completed financial year since then. (Appendix 7.)
- (b) A comparative statement of the Income and Expenditure of the Melbourne HASI for the financial years ended the 30th June, 1961, 30th June, 1962, and 30th June, 1963, together with Balance-sheets as at the close of those years set out in a comparative form (Appendices 8 and 9.)
- (c) A statement of receipts and payments for the six months ended the 31st December, 1963. (Appendix 10.)

Details and comments in respect of various items contained in the financial statements referred to above are given hereunder:-

Income.

Income is derived mainly from fees received either for processing or for training courses conducted by the organization. Amounts shown in the financial statements under these headings represent cash collections only and do not include amounts due from debtors.

Very substantial amounts have been received from many individuals in respect of services rendered by the organization. Examples are:-

- (a) £3,065 paid between 1957 and 1963
- (b) £1,801 paid between 1960 and 1963
- (c) (wife of (b)) £1,777 paid between 1960 and 1963
- (d) £1,631 paid between 1957 and 1963
- (e) £1,579 paid between 1961 and 1964
- (f) £1,014 paid between Oct., '62, and March, '63
- (g) (wife of (e)) £1,086 paid between Oct., '62, and March, '63
- (h) £1,275 paid between 1958 and 1963

The above amounts do not include sums, if any, paid to HCO, Melbourne, for books, E-meters, advanced clinical courses, &c.

- Mr. Campbell carried out an examination of various aspects of the procedures relating to cash receipts and a selective check was made of the transactions. These investigations revealed the following:-
- (a) Agreements are entered into between the organization and individuals for the services to be rendered. Agreements are also entered into in certain cases in which no charge is to be made for processing or training.
- (b) Persons may contract to pay for the services on either a cash or time payment basis. In some instances, payments are made in advance of the signing of agreements. Additional charges are payable in respect of contracts for certain services involving the payment of fees on a time payment basis, in the following instances:-

Processing. - Under the scales of fees dated 15th and 16th November, 1960, the fee for 25 hours processing was £105 cash or £124 on terms of £40 deposit and the balance payable over a period

of six months. The additional charges for terms represent a flat interest rate of approximately 45 per cent. per annum on the balance owing. However, the flat interest rate per annum would be approximately 12 per cent. in cases in which a rebate of £14 for prompt payment was applicable. Casual processing was available at various rates ranging from £4 14s. 6d. per hour downwards.

Hubbard Qualified Scientologist Course. - As from the 2nd January, 1964, the fee payable for this course was £40 cash or £44 on terms of £8 deposit and the balance payable over a period of ten weeks. The additional charges for terms represent a flat interest rate of approximately 58 per cent. per annum on the balance owing after payment of the deposit.

Hubbard Professional Auditor Course. - Under the scale of fees dated the 2nd January, 1964, the fee payable for this course was £140 cash or £160 on terms of £40 deposit and the balance payable over a period of six months. The additional charges for terms represent a flat interest rate of 33.3 per cent. on the balance owing after payment of the deposit. By November, 1964, the fees for training had changed; the following is the latest scale of which the Board has information:-

	£
HQS - (Hubbard Qualified Scientologist)	75
HCA - (Hubbard Certified Auditor)	75
HPA - (Hubbard Professional Auditor)	75
HCS - (Hubbard Clearing Scientologist, a course not	
being conducted but in contemplation)	100

- (c) Upon entering into an agreement for either processing or training under which payment is to be made on terms, the individual is usually required to sign a promissory note or promissory notes post-dated to the due date of the final installment.
- (d) Instances have occurred in which blank cheque forms have been provided by the organization and used by individuals to pay accounts (approximately 40 blank cheque forms were issued between April, 1963, and January, 1964). It was said by the HASI staff that these forms were held by the organization for the convenience of people who did not have their cheque books with them at the time of signing contracts. The forms were also in effect used as promissory notes, because, although certain of the cheques were not banked, cash equal to the value of the respective cheques was brought to account on the dates shown on the cheque butts.
- (e) The fees charged for processing and training were flexible though the prices paid for the services were generally in accordance with the scale of fees. Discounts were allowed to life members on processing and training, and to professional auditors, students and persons aged under 18 years in respect of processing. Staff were usually charged concession rates. In addition, a sliding scale of fees for processing, providing for a reduction in the hourly rate charged according to the number of hours processing contracted for, was operative from April, 1961, to October, 1963. Price reductions under the sliding scale were applicable not only in cases in which individuals contracted to undertake a large number of hours processing in the first instance, but also in instances in which they contracted to undertake additional processing at the time the previous contract expired.
- (f) Managerial Fees received represent cash received for managing the financial affairs of the Melbourne College of Personal Efficiency, Hawthorn, and the HAS1 Sydney, for the period February to June, 1963.
- (g) Miscellaneous Income includes amounts received in respect of sales of soft drinks, hire of E-

meters and sales of testing materials. Compared with the financial years 1960-61 and 1961-62, there was an increase in income under this heading in 1962-63. This was attributable to the receipt of international membership subscriptions amounting to £673 which, in former years, would have been payable to HCO, Melbourne. However, the income for that year does not include all amounts chargeable to other Australian branches of the organization in respect of testing materials, magazines and bulletins, &c., supplied. *Refunds of Income*.

The policy in respect of refunds of fees is set out in HCO Pol. Lr. of the 12th October, 1961 and subsequent policy letters. (See Appendix 11.) The manner in which this refund policy was applied appears from correspondence set out in Chapter 15.

Refunds made in 1962-63 included sums of £254 12s., £200 and six Life Memberships totalling £187 10s.

Refunds made for the period 1st July to 31st December, 1963, included sums of £175 7s., £1,517 3s., £58 15s., £90.

By letter dated the 28th November, 1963, the HASI undertook to refund £397 16s. to one individual and this was done on the 4th March, 1964.

Expenditure.

Details and comments in respect of certain of the items of expenditure are set out hereunder:-

Rent.

The premises leased by the organization in Melbourne and the respective rentals payable were:-

157 Spring Street £130 per calendar month.
159 Spring Street £130 per calendar month.

2 Coates Lane £108 6s. 8d. per calendar month.

Salaries.

An employment contract is entered into by each member of the staff prior to taking up employment in the organization. Each employee is allotted a certain number of "units" based upon the position to be occupied. Generally speaking, the respective salaries vary from week to week according to the weekly income of the organization and to the number of units allotted to the respective officers or employees. Officers may be employed on a full-time or part-time basis.

As mentioned previously, 55 per cent. of the weekly gross income is transferred to the Salary Account each week. After the deduction of fixed rates, for example, cleaner's wages and allowances to officers undertaking courses at Saint Hill, the balance of the weekly proportion is divided by the total number of staff units. The unit value so ascertained is multiplied by the number of units held by the respective officers or employees and the resultant amounts represent the salaries payable for the week.

Christmas bonuses financed from funds in the Building Account were paid to the staffs of both the HASI and HCO in the financial years 1961-62 and 1962-63. No such bonuses were paid in respect of Christmas 1963. The sums so paid amounted to £388 4s. and £295 in the respective years.

A perusal of the lists of the weekly salaries paid to the staff of both the HASI and HCO discloses that the earnings were moderate and, in very many instances, were less than the Basic Wage. For example, for the week ended the 17th May, 1962, where the unit appears to have been 5s. 6d., a person on 16 units received as little as £4 8s. for 40 hours' work. Other instances for that week were:-

No. of Units	Hours Worked	Gross Salary
		£ s. d.
25	40	6 17 6
19	421/2	5 4 6
28	421/2	7 14 0
20	40	5 10 0
13	321/2	3 1 16
3	71/2	0 16 6
33	47	9 1 6

The highest paid employee in the HASI in this week was the cleaner, who was paid at the rate of 10s. per hour. HCO staff fared a little better than the HASI, two of them getting £16 17s. 6d. and £16 respectively.

For the week ended the 26th September, 1963, where the unit was valued at 10s. 9d., the following are examples of the salaries paid:-

No. of Units	Hours Worked	Gross Salary
		£ s. d.
21	40	11 5 9
18	40	9 13 6
20	40	10 15 0
361/2	401/2	19 1 9
29	401/2	15 11 9
18	40	9 13 6
23	40	12 7 3
13	321/2	6 19 9
14	35	7 10 6
32	401/2	17 4 0
41	401/2	22 0 9

For the week ended the 23rd January, 1964, where the unit was valued at 7s. 3d. the following are examples of the salaries paid :-

No. of Units	Hours Worked	Gross Salary
		£ s. d.
18	40	6 10 0
16	40	5 16 0
26	401/2	986
23	40	8 6 9
29	401/2	10 10 3
Advertising.		

Expenditure under this heading includes amounts expended on:-

- (a) Renting of space for the display of posters in Melbourne trams for the period 4th December, 1961, to 3rd December, 1963.
- (b) Advertisements in suburban newspapers, magazines such as "Australasian Post" and "Pix", and sporting publications, e.g., the "Soccer Weekly".
- (c) Advertisements in the "pink pages" of the Melbourne Telephone Directory.
- (d) Screen advertising in picture theatres.
- (e) Distribution of leaflets and brochures in certain Melbourne suburbs. *Travelling Expenses*.

Apart from an amount of £52 included in 1961-62 in respect of travelling expenses incurred by Mr. P. Williams in U.S.A., the expenses were incurred by officers of the organization travelling within Australia. Overseas travelling expenses are included under the heading " Cost of Instructor Training ".

Tapes.

The expenditure is in respect of tape recordings supplied by HCO.WW. and the Washington Branch of the organization. All payments are made to HCO.WW.

Hire of Office Machines and Telex.

The sum expended in 1961-62 includes an amount of £422 5s. payable to HCO, Melbourne as a recoup of part of the cost of operating telex equipment. Expenditure borne by the organization in respect of telex equipment amounted to £450 11s. 1d. in 1962-63.

Cost of Instructor Training.

Details of the expenditure in each of the years are as follows:-

1960-61 Payable to HCO.WW.	£ s. d.
P. Williams -	
Special briefing course fees at Saint Hill	357 3 9
One E-meter	35 12 6
Special briefing from Hubbard at Johannesburg	312 10 0
Fares and expenses trip to Johannesburg Two instructors - special briefing course at Saint Hill (March, 1961)	339 0 7 625 0 0
	1,669 6 10

Payable to HASI, New Zealand-

Messrs.	Watters	and	Verity-
---------	---------	-----	---------

Two HPA retreads•	375 0 0
Hotel expenses	187 16 3
Fares	263 8 9
	826 5 0 (•-Refresher Course.)
1961-62	
P. Williams - Air fares (U.S.A. and London) B. Pesco -	£ s. d. 728 8 3
Air fares (London)	284 3 0
HPA retread	75 0 0
	359 3 0
D. Gogerly -	
Fares (Melb/London and return)	560 2 0
Travelers cheques	150 0 0
	710 2 0
Less Credit for P. Williams account 1960-61 expenditure	1,797 13 3 5 12 6
2000 Greate for 1. Williams account 1000 of experiations	1,792 0 9
1962-63	
D. Gogerly -	
Expenses	257 15 0
Course fees	358 12 3
B. Pesco -	616 7 3
Fares	126 10 1
Course fees	358 12 3
R. Turnbull -	485 2 4
Fares	428 6 9
Course fees	358 12 3
B. Painter -	786 19 0

Fares	460 19 10
Course fees	358 12 3 819 12 1
I. Tampion - Fares	58 0 0
Course fees	358 12 3 416 12 3
A. Tampion -	410 12 3
Fares	58 0 0
Course fees	358 12 3 416 12 3
E. Watson -	410 12 3
Fares	330 0 0
Course fees	358 12 3
P. Williams - expenses	688 12 3 769 8 2 4,999 5 7
Six Months ended 31st December, 1963.	
Saint Hill Course Fees - E. Williams Fares -	£ s. d. 358 12 0
E. Williams R. Turnbull and B. Painter E. Watson	598 11 0 375 0 0 30 0 0
Research Expenses	1,362 3 0

Research Expenses.

Details of the expenditure in each of the three years are as follows:-

1960-61

An account from HCO.WW. dated 16th February, 1961, states:-

	£ S.	а.	
Organizational share of expenses incurred in researches for Routine 3D	500	0	0
Researches at WW Horticultural labs (1960-61)	250	0	0
	750	0	0

(The amount of £750 Sterling converted to Australian £ at £100 Sterling = £125 Australian -- £937 10s.). 1961-62

No expenditure is shown under this item for the year. However, after the audited financial statements had been completed, an account for £7,191 5s. 4d. was received from the HCO.WW. in

respect of supplementary expenses incurred by that office on behalf of the Australian branches. This account sought to debit the Australian HASI with 30 per cent. of the cost of developing and research costs in respect of models IV and V of the E-meter, the cost of production of instructional and educational materials such as tapes, magazine costs and other items.

1962-63

At the time of Mr. Campbell's investigation, an account for supplementary expenses had not been received from HCO.WW. in respect of 1962-63.

Managerial Expenses and Service Charges HCO.WW.

A sum representing 10 per cent. of the weekly gross income less refunds and incidental receipts is payable to HCO.WW. each week.

HCO, Melbourne.

An amount representing 5 per cent. of the weekly "proportionate amount" was paid to HCO, Melbourne, each week during the years 1960-61, 1961-62, and in the period 1st July, 1962, to 7th February, 1963. HCO Pol. Lr. of the 1st February, 1963, directed that future payments be cancelled. The letter also directed that the percentage of the "proportionate amount" payable to the Salary Account was to be increased from 50 per cent. to 55 per cent. and that the salaries and certain other expenditure of each HCO Area Office were to be borne by the central organization.

Rates.

Under the lease of the premises at 157 Spring-street, Melbourne, the organization was required to pay municipal rates and water rates in respect of the period 1St March, 1958, to 1St June, 1961. The total amount, £478 17s. Id., was shown as expenditure in the financial year 1961-62.

E-meters Unsaleable due to Obsolescence.

The amount of £417 15s., written off in 1961-62, represents the value of eighteen E-meters which were unsaleable out of a stock on hand of 26 E-meters.

Assets.

Sundry Debtors. An amount of £56 12s. 6d. shown as outstanding as at the 30th June, 1963, was in respect of sales of E-meters. Substantial amounts outstanding in respect of processing and training fees, &c., were not included in the final accounts.

A list of debtors from the individual debtors accounts as at the 7th January, 1964, showed that the net amount outstanding at that date was £22,869, of which £3,347 was due from staff members.

The total of £22,869 included about £4,000 bad or doubtful debts.

Sundry Loan Accounts.

Details of the amounts outstanding in respect of loans made by the organization for the three years ended respectively the 30th June, 1961, 30th June, 1962 and 30th June, 1963, were £200, £238 8s. 8d. and £524 15s. 11d., granted in all but one instance to assist staff members while on training

courses at Saint Hill.

Amounts owed by other Branches and Associated Companies.

HCO, Melbourne - Loan Account.

Details are as follows:-

	£ s. d.
Net value of assets taken over from HASI, Melbourne	1,448 5 3
Payments on behalf of HCO re ACC course	106 14 0
Rebate due from Customs Dept. received by HCO	210 0 0
	1,764 19 3
Less Remittances-	
4000.04	400.44.0
1960-61	106 14 0
1960-61	1,448 5 3
	1,110 0 0
1961-62	36 0 0
	1,590 19 3
Amount outstanding at 30th June, 1963 HASI, Perth - Loan Account.	174 0 0
1,1,131, 1, 1,11, 1,11,11,11,11,11,11,11,11,11	

The outstandings are in respect of amounts due to S. Richards for the purchase of the business operated by him in Perth. The amounts were required to be paid by HASI Melbourne, and refunded by HASI, Perth.

HASI, Sydney - Current Account.

The amount outstanding includes the cost of materials supplied to that branch.

HASI Ltd.-Loan Account. Details are as follows:-

1961-62

	£ s. d.
Preliminary expenses paid on behalf of HASI Ltd	237 10 0
Advance to HASI Ltd., London	1,480 2 11
Balance outstanding at 30th June,1962	1,717 12 11

1962-63

Balance forward from 1961-62	1,717 12 11
Postage re share certificates	4 3 2
Registration fees and costs Adelaide and Canberra branches	55 8 9
Balance outstanding at 30th June, 1963	1,777 4 10

Liabilities.

Sundry Unsecured Loans.

The amount outstanding at the 30th June, 1963, represents the balance of sums loaned to the organization without security and not yet repaid.

The loans were given mainly during the financial year 1957-58, in which year amounts totalling £4,448 15s. were received. These loans came from some 25 scientologists and ranged from £30, £40, £50, £100, £255, to £410 and £620.

Amount Owed to S. Richards.

The amount outstanding at the 30th June, 1961, is in respect of the purchase of the American College, Perth, from S. Richards for the establishment of a branch of the organization in that city. The amounts paid to S. Richards have been shown in the accounts as a loan to HASI, Perth.

Amounts Owed to Other Branches-

HCO.WW.

Details are as follows:-

	£ s. d.
Managerial Expenses from date of commencing	
business in Melbourne to 30th June, 1959	4,021 3 1
Purchases of books to 30th June, 1959	3,404 9 7
A credit transferred to London (April, 1960)	41 5 7
Research Expenses, 1960-61	937 10 0
Cost of Instructor Training 1960-61	1,324 13 9
Additional Managerial Expenses 1960-61	625 0 0
Cost of Cables, 1961-62	27 2 2
•Supplementary Expenses 1961-62.	7,191 5 4
	17,572 9 6
Less Remittances -	
April, 1960	400 0 0
March, 1961	14 17 11
May, 1961	200 0 0
June, 1961	1,448 5 3
February, 1962	388 14 9
April, 1963	52 14 3
Amount outstanding at 30th June, 1963	2,504 12 2 15,067 17 4

[•] See footnote under Profit and Loss Account HCO, Washington.

Of the amount outstanding at the 30th June, 1963, the sum of £52 2s. 4d. is owed in respect of hotel expenses for Mr. P. Williams.

HCO, Melbourne - Current Account.

The amount outstanding includes a proportion of charges for rental of telex equipment, costs of *Communication* Magazine, cables and freight.

The sum outstanding at the 30th June, 1963, was arrived at as follows:-

	£ s. d.
Cost of Instructor Training - Watters and	
Verity (2 HPA retreads, air-fares and hotel expenses)	826 5 0
Debt incurred by J. Roberts in New Zealand	644 17 6
Test materials supplies	48 9 7
	1,519 12 1
Less Remittance	87 12 0
	1,432 0 1

Hubbard Communications Office.

1. ACCOUNTING.

In many respects the accounting system is similar to that of the HASI and is based also on instructions issued by HCO.WW. However, unlike the other Melbourne organization, cash books are kept to maintain a continuous record of the receipts and payments.

Financial policy is determined largely by HCO.WW. A return of income from all sources and a report on stocks of books and publications is required to be furnished each week to that body.

2. BANKING.

Although the organization is required to bank weekly, it was noted that in the financial year 1962-63 and in the six months ended 31st December, 1963, delays of up to one month had occurred in the banking of cash collections. The comparatively large sums of cash on hand at the 30th June, 1963, and the 31st December, 1963, reflect the position. Further, several instances of under- and over-bankings had occurred.

During the financial years 1960-61, 1961-62 and 1962-63, the following bank accounts in respect of the organization were maintained at the I.C.I. House Branch of the Bank of New South Wales:-

(a) Hubbard Communications Office Area Account.

(b) Hubbard Communications Office Book Account.

Generally, proceeds from sales of books, tapes and E-meters were banked in the Book Account, while cash collections in respect of franchises, congresses and membership subscriptions, &c., were deposited in the Area Account. Authorities held by the bank empowered either one of L. Ron Hubbard or Elizabeth Williams to draw cheques on the Accounts.

These bank accounts were closed on the 1st November, 1963, and balances at that date were merged in a new account opened under the name of Hubbard Communications Office Continental Account. Persons authorized to draw cheques on this Account were any two jointly of L. Ron Hubbard, Elizabeth Williams, Catherine Layton, Betty Miles and Lindsay Norman Bromley.

Another account in the name of Hubbard Association of Scientologists International A.C.C. Account is maintained at the Australia and New Zealand Bank Limited, Exhibition-street, Melbourne, for the purpose of accounting for the cash receipts and payments in respect of advanced clinical courses held in Melbourne. The persons authorized to draw cheques on the account are either Lafayette Ronald Hubbard or Mary Sue Hubbard.

3. FINANCE.

A comparative statement of the income and expenditure of the Hubbard Communications Office, Melbourne for the financial years ended 30th June, 1961, 30th June, 1962, and 30th June, 1963, together with Balance Sheets as at the close of those years set out in a comparative form are printed as Appendices 12 and 13.

The following details and comments in respect of various items contained in the financial statements are given under appropriate headings:-

Income.

Apart from income derived from the sale of books, tapes and E-meters, it was found that all income had been brought to account on a cash received basis.

Books, Tapes and E-meters.

Books, tapes and E-meters are purchased mainly from HCO.WW. for resale to persons and other branches of the organization within Australia. However, it was noted that 35 E-meters at a cost of £25 l0s. each were purchased from H. A. Wentworth Pty. Ltd., Melbourne, in August-September, 1961. Expenditure in respect of local purchases of books included a progress payment of £750 in December, 1963, to the Glenhuntly Printing Works Pty. Ltd. for the printing of 10,000 copies of the book, *This is Life*.

Excluding assembling and testing costs, the prices paid for a Mark IV and a Mark V E-meter were approximately £36 and £40 respectively. The selling prices after deducting discount were usually £56 8s. and £62 respectively. The current selling prices of books range from 1s. to £2 4s. per copy.

The holders of international membership of the HASI are allowed discounts on cash purchases of E-meters, tapes and books. Franchise holders (and district offices) have also been allowed discounts on such purchases.

In respect of the accounting for books, it would appear that there was a discrepancy in the value of the stock on hand as at the close of one or both of the financial years 1961-62 and 1962-63. However, as satisfactory records have not been kept to record stock movements, the reason for the apparent discrepancy cannot be readily ascertained. The following trading statement illustrates the position:-

	1960-61		1961-62		1962-63	
	£	£	£	£	£	£
Sales		3,378		3,245		2,509
Stock on hand at beginning of each year	1,869		2,571		2,728	
Purchases	3,262		1,419		2,138	
	5,131		3,990		4,866	
Stock on hand at close of each year	2,571	2,560	2,728	1,262	2,036	2,830
Gross Profit		818		1,983		
Gross Loss						321
Percentage of Gross Profit to Sales		24%		61%		

The apparent discrepancy could have been due to a number of reasons of which the most likely appear to be:-

- (a) Incorrect stocktaking.
- (b) Disposing of books at a price below cost.
- (c) Gifts of books.
- (d) Misappropriation either of books or of proceeds from the sale of books. Franchise Receipts.

This item includes a contribution of 5 per cent. of the "Proportionate amount" of the gross income of HASI, Melbourne and varying percentages of income of the branches of the organization and other persons carrying on the practice of scientology in Australia. However, in accordance with a direction received from HCO.WW., the payment of franchise fees by HASI, Melbourne was discontinued as from the 7th February, 1963. The payment of franchise fees by other branches and persons to the HCO, Melbourne also ceased during the financial year 1962-63.

In HCO Pol. Lr. of the 14th June, 1962, a direction was given to franchise holders that "no Franchise or Field Auditor should charge less for services than the Central Organization for that Continent. They may, however, charge more . . " (At least one franchise holder charged £10 10s. per hour.)

The following information has been collated from the financial records for 1961-62 to show the persons and organizations from whom franchise fees were receivable during that year together with the respective amounts shown to be received or receivable:-

£ s. d.
•2,069 6 10
732 1 9
638 17 9
23 15 0
34 14 3
86 17 8
177 2 10
1,683 16 10
156 13 6
575 7 10
45 7 6
1 1 9
6 3 8
33 10 0
85 0 0
472 3 10
14 1 0
5 0 0
1 16 0
202 12 6
7,045 10 6

• Does not include an amount of £29 10s. 10d. stated to be in transit. *Membership Fees.*

The fees comprise mainly annual international membership subscriptions (£5 5s. each) and associate membership subscriptions (5s. each).

Advanced Clinical Course Fees.

Advanced clinical courses conducted by HCO.WW. were held in Melbourne in 1959 and 1961. Fees were payable on either a cash or time payment basis.

Income from the courses has been accounted for on a cash basis.

Twenty-two undertook the 2nd Melbourne Clinical Course in April, 1961. The fees charged for this course varied; nine contracted to pay over £400 and three others over £300, with some discounts for prompt payment.

Congress Receipts.

The receipts are mainly in respect of proceeds from the sales of admission tickets and dinner dance tickets at various congresses conducted by the organization.

Sums equal to 50 per cent. of the gross receipts were payable to HCO.WW. in respect of congresses held prior to the 16th March, 1963. However, HCO Pol. Lr. of 16th April, 1963, directed that in future, in lieu of the previous levy, a payment of 10 per cent. of the gross receipts was to be remitted. Personal congresses given by Hubbard are not included in this directive.

Sundry Income.

Amounts received under this heading are mainly in respect of tape replays.

Expenditure. Salaries.

The weekly salaries paid to officers and employees of this organization are stated to be calculated on the unit value determined for HASI, Melbourne.

Congress Expenses.

The expenditure includes amounts paid to hotels for accommodation, &c., and printing, stationery and postage charges incurred in respect of scientology congresses.

Course Expenses.

The sum shown as expenditure in 1960-61, £814, is in respect of the cost of production of instructional films for scientology courses by HCO, Washington.

Expenditure of £195 14s. incurred in 1961-62 related to the purchase of books and stationery for use at an advanced clinical course.

Legal and Detective Agencies.

The expenditure in 1961-62 included an amount of £100 paid to a Mr. Huston on the 16th October, 1961.

Expenditure during the period 1St July to 31st December, 1963, included payments totalling £436 12s. 8d. to a W. D. Northcott, of Perth, for services rendered or to be rendered as a security consultant, and a payment of £59 14s. 10d. to Astoria Rent-a-Car on account of W. D. Northcott.

Managerial Expenses and Service Charges.

Amounts payable to HCO.WW. in each of the years under review were calculated as follows:-

1960-61

	Income £ s. d.	Amount due £ s. d.	to HCO.WW.
Franchise Receipts -			
10 per cent. levies	2,644 15 2-33.3	per cent. =	881 11 9
15 per cent. levies	1,747 4 1 - one-	third of two-third	ds (one third of £1,164 16s.) = 388 5 4
Book Sales - General	3,317 18 5- 10 p	er cent. =	331 15 10
Sales of tapes after 2.5.1961	-10 per cent	:. =	
Sales of E-meters after 2.5.1961 Administrative Expenses <i>re</i> Typing and Paper	,	5 0	-10 per cent. = 6 13 6 510 0 0

1961-62

	Income	Amount due to HCO.WW.				
	£ s. d.	£ s. d.				
Franchise Receipts -						
10 per cent. levies	3,051 4 5 -33.3	per cent. = 1,017 1 6				
15 per cent. levies	741 8 0 - one-	third of two-thirds = 16 13 0				
Book Sales	3,243 19 10	-10 per cent. = 324 8 0				
Sales of certain E-meters (ex	ccl. Sales Tax)	1,251 7 6 - 10 per cent. =	125 2 6			
Total 1,483 5 0						

1962-63

Income Amount due to HCO.WW.

£ s. d. £ s. d.

Franchise Receipts -

10 per cent. levies 1,001 5 10 - 33.3 per cent. = 333 15 3 Book Sales 2,403 5 5 - 10 per cent. = 240 6 6

Sales of E-meters (excl. Sales Tax) 2,829 7 6 - 10 per cent. = 282 18 9

Total 857 0 6

Rent.

The expenditure in 1962-63 is in respect of the rental of premises for HCO, Adelaide.

Telex and Telephone Charges.

The expenditure includes the rental of a teleprinter and the costs Of printergrams and cables.

Travelling Expenses.

Apart from expenses incurred by Mr. and Mrs. P. Williams, G. Jensen and E. Cooper on travel within Australia, a sum of £451 payable to HCO, Washington, representing fare for one Halpin, was charged as expenditure in 1960-61.

The expenditure in 1961--62 and 1962-63 appears to be in respect of executive officers travelling within Australia.

Assets.

Loans to Employees.

Advances to four employees amounting to £210 7s. 10d. were out-standing at the 30th June, 1963.

Sundry Debtors.

This item includes outstandings in respect of sales of E-meters, books, tapes and congress tickets.

Amounts due for advanced clinical courses were not included in the total outstandings, but a check of the individual debtors accounts disclosed that about £6,000 had not been collected at the 30th June, 1963.

Amounts owed by other Branches.

The balances represent the net amounts outstanding in respect of sums paid by the Melbourne organization on behalf of other branches together with amounts owed for materials, &c., supplied to the branches.

Liabilities.

Amounts owed to other Branches.

HCO. WW.

The following are details of the amounts payable to HCO.WW. as at the close of each of the financial years from 1959-60 to 1962-63:-

1959-60

General Account	£ s. d.
Balance owing at 1st July, 1959	197 12 7
Add Cash transfers from HASI	333 13 9
Royalties on E-meters (from B. Penberthy)	37 6 8

Books, tapes, &c., received during year	2,171 10 0
Percentage on books sales (10 per cent. on £2,846 8s. 6d.)	284 12 6
Managerial Expenses and Services Charges (33.3 per cent. on 10 per cent. Franchise receipts)	989 9 9 4,014 5 3
Less Cash remitted during year	3,207 18 5
Cost of Cables	17 4 4 3,225 2 9 789 2 6
Advanced Clinica1 Course	
Managerial Expenses and Administration Charges due to HCO WW	4,000 0 0
Cost of Cables	175 15 10 4,175 15 10
Less Cash remitted	2,000 0 0 2,175 15 10
Balance owing as at 30th June,1960	2,964 18 4
1960-61	
Balance owing at 1st July, 1960 Add - Administrative Expenses and Service Charges Books and tapes supplied for year E-meter Costs Administrative Expenses (typing) Postal (airmail) Cables and Telex Expenses Freight on books Magazine Printing and Distribution (PAB Magazine) Adjustment re 30th June, 1960, Credit (Previously treated as Aust	£ s. d. 2,964 18 4 1,608 6 5 1,697 0 0 187 0 0 593 15 0 281 5 0 246 15 0 128 5 0 307 0 0 tralian currency instead of Sterling)
Administrative Expenses	510 0 0
Cost of Cables	44 17 2 554 17 2 8,569 1 11
Less Cash remitted during year	2,945 1 6
Balance owing as at 30th June,1961 1961-62	5,624 0 5
Balance owing at 1st July, 1961 Add Meters, Books, Tapes, &c., supplied for year	5,624 0 5 3,987 0 0

Managerial Expenses and Service Charges for year	1,483 0 0
Share of Congress Receipts	715 1 6
	11,309 11 1
Less Cash remitted during year	4,515 7 3
Books transferred to HCO.WW	62 10 0
	4,577 17 3
Balance owing at 30th June, 1962	7,231 4 8

1962-63

	£ s. d.£ s.	d.£ s. d.	
Balance owing at 1st July, 1962	3,367 19 6	3,863 5 2	7,231 4 8
Add Meters, Books, Tapes &c. supplied for year	3,040 2 5	3,040 2 5	
Managerial Expenses and Service Charges for year	857 0 6	857 0	6
Share of Congress Receipts	423 2 2	423 2 2	
	4,648 2 2	6,903 7 7	11,551 9 9
Less Cash remitted during year	192 4 8	1,700 19 7	1,893 4 3
Balance owing at 30th June, 1963	4,455 17 6	5,202 8 0	9,658 5 6
HCO, Washington.			

The amount due and still unpaid as at 30th June, 1963, is in respect of

	£ S U
Fare <i>re</i> Halpin (1960-61)	451 0 0
Cost re Film (1960-61)	814 0 0
	1,265 0 0
Expenses re Tapes (1961-62)	101 5 0
	1,366 5 0

Personal Loans.

This item shown under receipts for the period 1st July, to 31st December, 1963, is in respect of sums received in December, 1963, to finance the printing of the book, This is Life. Sums ranging from £44, £50 to £350, £400 and totalling £1,494 were received from seven persons and the Australian College of Personnel Efficiency, Geelong.

Australian College of Personnel Efficiency at Geelong.

Mr. Campbell conducted in respect of the Australian College of Personnel Efficiency at Geelong a similar thorough investigation. The proprietors of this college are Eric Bruce Lake and his wife, June Margaret Lake, who earlier held a franchise to carry on the practice of scientology, and later had the status of a "district office".

It is unnecessary for the purposes of this report to give extensive details concerning this college.

For the three years ended the 30th July, 1964, the following income was obtained

Year ended 30th June, 1961 -

£ s d
Gross Income from-

Processing 2,822 16 0

Training 987 4 0 3,810 0 0

Year ended 30th June, 1962 -

Processing	3,718 3 6
Training	792 12 0 4,510 15 6
Year ended 30th June, 1963 -	
Processing Training	4,070 6 3 825 17 0 4,896 3 3

A levy of 10 per cent. on these gross receipts was payable by the Lakes to HCO.WW. via the Melbourne HCO.

The fees charged by this college for processing varied, ranging generally from 3 guineas to 41 guineas per hour.

Very substantial sums were received from several persons for processing and training. Between July, 1960, and March, 1964, £1,442 was paid in respect of three members of one family. Other amounts over the same or shorter periods were £1131 (for two persons), £712, £706 (for two persons), £662 (for two persons), £634, £583, £473, £387.

This college advertised over the radio through 3GL, Geelong, and in the "Geelong Advertiser".

The Melbourne College of Personal Efficiency at Hawthorn.

The Melbourne College of Personal Efficiency at Hawthorn, conducted by Peter Francis Gillham and his wife, Yvonne Doreen Gillham, was also investigated by Mr. Campbell. The Gillhams earlier held a franchise to practise scientology, and later had the status of a "district office".

Income is derived mainly from processing, co-auditing and training courses conducted by the college. The following table shows a classification of fees received or receivable in each of the completed financial years in which the college has carried on business:-

	1958-59		1959-60		1960-1	1961-62	1962-63
	£	£	£	£	£		
Processing	157	1,827	2,778	4,161	5,294		
Training Courses	292	827	1,022	329	195		
Co-Auditing			144	1,588	1,388		
Conventions, Recordings, &c.		15	18		26		
	449	2,669	3,962	6,078	6,853		

A franchise fee of 10 per cent. on the gross monthly takings was payable to HCO.WW. via the Melbourne HCO.

The fees charged varied from no charge for the personal efficiency course up to £4 14s. 6d. per hour for processing. There were intermediate fees for co-auditing at from 15s. to £1 6s. 8d. per hour later standardized at £1 per hour, and fees of £5, £10 and £40 for the HAS courses zero, level 1 and level 2 respectively.

Where fees were not paid in cash, processing or training was usually on a time payment basis, terms being arranged with Conquest Investments (Australia) Pty. Ltd., a company the nominal capital of which was £5,000, of which £2 was paid up, and of which the directors and shareholders were Noel Stanley Barton and Peter Francis Gillham.

Very substantial amounts were received from or on behalf of several persons in respect of services rendered by the college. They include amounts of £1,179, £936, £784, £703, £593, £562, £468, £479. The person paying the sum of £593 also paid an aggregate of £551 to the Melbourne HASI and the Australian College of Personnel Efficiency at Geelong, making a grand total of £1,144.

This college advertised in city and suburban newspapers, Melbourne trams, the "pink pages" of the Melbourne Telephone Directory, on the radio and by a notice board at the Hawthorn railway station.

General.

Though the system of accounts used by the Melbourne HASI and the Melbourne HCO left something to be desired

from an accountancy point of view, the investigation by Mr. Campbell was thorough and there is no reason to believe that the material available to him does not completely cover the financial aspect of these organizations' activities. Likewise, there is no reason to suppose that the organizations received any other income than from the sources indicated, or that there were any other amounts remitted abroad than those indicated above.,

Certain significant figures emerge from the investigation by Mr. Campbell, the most important being the large income (well over a quarter of a million pounds) of the scientology organizations in Melbourne for the six years ended the 30th June, 1963.

The following figures show such income and the rate of growth:-

Year ended £

30th June,1958 12,150 30th June,1959 30,500 30th June,1960 47,075 30th June,1961 57,640 30th June,1962 71,977 30th June,1963 54,071

273,373

The Melbourne College of Personal Efficiency, conducted by Gillham and his wife at Hawthorn, had a gross income from fees alone of nearly £17,000 for the three years ended the 30th June, 1963.

Year ended £

30th June, 1961 3,962

30th June, 1962 6,077

30th June, 1963 6,852

16,991

For the same three-year period, the Australian College of Personnel Efficiency, conducted at Geelong by Lake and his wife, had a gross income from fees alone of over £13,000.

Year ended £

30th June, 1961 3,809

30th June, 1962 4,510

30th June, 1963 4,895

13,214

The income of the Melbourne HASI for processing and training respectively for the three years ended the 30th June, 1963, amounted to nearly £130,000, being made up as follows:- Year ended 30th June, 1961- Processing26,837 Training14,284 Year ended 30th June, 1962- Processing29,023 Training17,267 Year ended 30th June, 1963- Processing32,610 Training9,342 129,363

The total income of the Melbourne HASI and the Hawthorn and Geelong organizations for the three years ended the 30lh June, 1963, was £213,753, as follows:-

Year ended £

30th June, 1961 65,411

30th June, 1962 82,564

30th June, 1963 65,778

213,753

The income of the three organizations from processing alone for the same period was £115,971 as follows:-

Year ended £

30th June, 1961 33,621

30th June, 1962 38,818

30th June, 1963 43,532

115,971

For the five-year period from the 1st July, 1958, to the 30th June, 1963, the total amount of money remitted to the HCO.WW. in England by the Melbourne HASI and the Melbourne HCO was £26,166, averaging £5,233 a year, or £101 a week. Included in this sum was £17,183 for managerial expenses and service charges, which in turn included most of the 10 per cent. on gross receipts.

The HCO.WW. claimed that a further £17,231 was owing to it as at the 30th June, 1963, in respect of managerial expenses, service charges and expenses said to be unpaid as at that date. This last mentioned item included an amount of £7,191 5s. 4d. in respect of a proportion of the cost of the development of research equipment abroad, the production costs of instructional and educational materials and some other items, the account for which was rendered by HCO.WW. after the Melbourne HASI's books for the financial year 1961-62 had been closed and a substantial profit for that year had been revealed. This aspect was an unresolved problem with the Federal taxation authorities at the time the Board heard evidence.

The Anderson Report CHAPTER 6 HUBBARD-THE FOUNDER OF SCIENTOLOGY

Biography.

Lafayette Ronald Hubbard, the founder of scientology, was born at Tilden, Nebraska, U.S.A., on the 13th March, 1911, the son of H. R. and Ledora May (nee Waterbury). He claims to be descended from Count de Loup "who entered England with the Norman invasion and became the founder of the English de Wolfe family which emigrated to America in the 17th century." His father's family came to America in the 19th century. His father's mother was a Scot; his maternal grandfather was a cattleman in Montana, on whose ranch Hubbard was raised till he was ten. In his later childhood and adolescence Hubbard travelled extensively in Northern China and India with his father whose duties as a United States naval officer took him to the Orient. His sojourn in the East aroused in him an interest in Eastern philosophies and he learned something of the Buddhist and other teachings from fakirs and yogis whom he met. His early formal education seems to have been sporadic. One of his friends and early tutors is claimed to have been a Commander Thompson, a doctor in the United States Navy, who is said to have been a student of Sigmund Freud in Vienna.

From 1930 to 1932 Hubbard was a student at the George Washington University where he claims to have studied engineering and to have been one of the first men to have studied nuclear physics. He has claimed, or allowed the claim to be made and repeated frequently without denial by him, that he is a graduate of that University in civil engineering, and he uses, and allows to be used in relation to himself, the letters "B.S." and "C.E.", intending to convey the impression that he has so graduated. In fact, he has no such qualification. He claims other academic distinctions also - "many degrees" it is said - but the only "university" degree which is identified is that of Doctor of Philosophy at the Sequoia University, Southern California. The Board caused inquiries to be made as to the identity of this university and was informed by the Australian Consul-General in San Francisco that the Sequoia University was a privately endowed institution which was not accredited, that is, not registered with the Western Association of Schools and colleges, which is the accrediting body for the west coast of America.

This somewhat suspect degree and a self-bestowed doctorate of scientology enable Hubbard to describe himself and be described as "Doctor" Hubbard. Though he writes extensively on medical matters, there is no basis for regarding him as a doctor in the medical sense.

Between 1932 and 1941 he travelled extensively in Central America and developed as a prolific writer of books of fiction, travel, science fiction and fantasy, finding time "to write seven million words of published fact and fiction". In dianetics and scientology it is evident that his output of science fiction has continued unabated. He finds his early reputation as a science fiction writer irksome, and in biographical information disseminated to scientologists his science fiction proclivities are sometimes discreetly omitted.

During World War II he served as an officer in the United States Navy and appears to have been seconded to the Royal Navy for a period.

As early as 1935 he appears to have been formulating some of the "axioms" which later bloomed into 57 scientology and nearly 200 dianetic "self evident truths". It is said that he wrote in 1938 a never-published work called by the code name *Excalibur*, on the subject of the basic principles of

human existence.

Both before and after World War II he studied hypnosis, and, as his writings indicate, he practised it for some time. Out of the hotchpotch of his experiences, his imagination and hypnosis he developed the theories which led to the publication in 1948 of his first book on dianetics, entitled *Dianetics, The Original Thesis*. His theories found little or no acceptance in orthodox medical or psychological circles, but his extravagant ideas found some targets on the fringe of learning and his writings appealed to some who accepted uncritically and without proof the astonishing claims which he made. In 1950, he wrote his first major book on dianetics, entitled *Dianetics: The Modern Science of Mental Health*. This book he claims to be of 300,000 words, but it is less than two-thirds that size, and his boast is that it was written in three weeks. He claims that it was an instant success, a best seller, which over the years has reached the million copy mark. The nature of dianetics is dealt with later in this Report. A definition of the word has crept into Funk and Wagnall's *New Standard Dictionary* in these terms, "A system for the analysis, control and development of human thought evolved from a set of co-ordinated axioms which also provide techniques for the treatment of a wide range of mental disorders and organic diseases: the term and doctrines introduced by L. Ron Hubbard, C.E., D.Scn., American engineer".

Other books on dianetics followed, the next substantial volume being *Science of Survival*, first published in 1951, which deals extensively with the "dynamics of behaviour" and "dianetics processing".

In about 1951 Hubbard established the Hubbard Dianetic Research Foundation in California, but he appears to have had disputes with its board of management, largely if not entirely because of his desire to explore advanced theories which he had developed concerning the thetan and its past existence. Accordingly, he severed his association with the Research Foundation in about 1952 and proceeded to found scientology which is the burden of this Report. The development of dianetics into scientology and the continued close association of both of these "sciences" with one another is dealt with later in this report.

The first scientology book appears to have been *Scientology: A History of Man*, published in manuscript in 1952 under the title of *What to Audit* and subsequently reprinted several times under the first mentioned title, *Scientology: 8-80* also appeared in 1952, *Scientology 8-8008* in 1953, *Scientology; The Fundamentals of Thought* in 1956. Several other small books appeared between 1952 and 1956, one of them being *Self Analysis in Scientology* in 1954. Since 1956 no substantial definitive book on scientology has come from Hubbard's pen though he has written many hundred of bulletins, letters and directives and dozens of books and booklets on specific aspects of scientology practice. He has further disseminated his ideas by a great number of tape recordings, containing an estimated 30 million words.

In addition to his books, he has produced or inspired the production of several scientology magazines in several countries, e.g., Great Britain, U.S.A., Australia, South Africa and New Zealand, with such titles as *Ability*, *Certainty*, *Reality*, *Affinity*, *Communication*. For several years these magazines have provided a very effective means of disseminating Hubbard's teachings amongst his followers and advertising his books, courses and congresses. His bulletins and other communications are frequently reproduced in these magazines. *Communication* magazine is "The Official Periodical of Dianetics and Scientology in Australia".

After Hubbard withdrew from the Hubbard Dianetic Research Foundation, it experienced certain difficulties, the nature of which did not clearly emerge, but seems to have been at least partly financial. In about 1954, Hubbard "acquired" the Foundation and in 1955 the "Unification" Congress

was held to celebrate the occasion. Now that dianetics was once more his own, he wrote a book, *Dianetics 1955*, described as "The Text Book of Human Communication," which dealt with dianetic techniques. Hubbard, however, had long been impatient with the slowness of dianetic techniques. Further, dianetics claimed to alleviate or cure man's mental and physical ailments. As a "science" claiming to cure illnesses, Hubbard feared that dianetics was likely to be vulnerable to scrutiny by civil authorities. Scientology, which officially did not claim to cure, ran no such risk, so Hubbard felt, and, furthermore, it covered a wider canvas and afforded better and more rapid methods of exploiting the gullible. Hubbard accordingly proceeded actively to promote scientology, ostensibly allowing dianetics to fade into the background. There is, however, a close and continuous link between the two "sciences," and, notwithstanding attempts at the Inquiry to assert the contrary, dianetics, if not in name, at least in reality is still very actively engaged in by scientologists. Many of their processes are virtually indistinguishable in essence. Dianetic and scientology techniques are frequently hypnotic and of a kind which are potentially harmful to mental health.

Hubbard had his headquarters in the U.S.A. for some years but about 1958 he transferred them to England and he now operates from Saint Hill Manor, East Grinstead, Sussex, England.

He has conducted congresses in various countries, visiting Victoria for that purpose in 1959. These congresses are an important source of revenue for Hubbard. A congress held in America in 1958 returned 800 dollars per head from at least 140 people. "The Games Congress," held in America in 1956, was attended by about 450 people.

By frequent and repeated reference to his qualities Hubbard has built up for himself in the eyes of his followers a mighty image. The adulation and obeisance which they offer him is almost unbelievable. Scientology students are specifically taught that he is not God, lest they think he is. He himself insists in *The Creation of Human Ability*, that preclears must be taught that he is only a man. He poses as a benign father to his scientology family. He stresses that he is always available to his flock. He is very approachable, and he encourages preclears to write directly to him, facilities for such communications being provided at the HASI. Frequently such communications are of a confessional nature, as Hubbard suggests they should be, and on this account his control over his followers becomes stronger.

In appearance he seems to be a solidly built man, perhaps a little over average height with sandy or red hair, thick but receding slightly, and a broad face and ruddy complexion. A larger than life bronze bust of Hubbard stands in the reception office of the Melbourne HASI, little the worse for the damage done by an assault on it by a disturbed scientologist.

An important requirement by Hubbard is that a weekly report on the processing of each preclear should be forwarded to him at Saint Hill. His condescension in considering each preclear's processing report is stressed as an indication of his magnanimity and the preclear is suitably impressed, not realizing until it is too late that he is very effectively acquiring domination over the awe-struck preclear.

He makes for himself, and allows others to make for him, the most exalted claims, He repeatedly asserts that he has perfected the techniques necessary to audit preclears to clear and beyond, and he complains that it is the fault of auditors who are unable to audit and use the E-meter that preclears have not yet become clear and OT. Jack Horner, the author of *Summary of Scientology*, writes of Hubbard, "I wish to state that L. Ron Hubbard is, in my opinion, one of the great geniuses of the 20th century, and that his discoveries will make possible a new era of living for Man." Such eulogies are not infrequent and are not discouraged by Hubbard. In the eyes of his followers he is infallible. Their attitude fervently expressed, is, "Ron is right," " What Ron says goes," whatever he

says or does. His mighty stature is heightened by permitting, and indeed encouraging, his followers to address him as "Ron". Everyone is on first-name terms with him.

His manor house at Saint Hill is the Mecca of his followers. The optimum state in scientology, that of being operating thetan (or OT), is said to be attainable only through the portals of Saint Hill or, at the very least, through processing by a graduate of Saint Hill.

He resides and presides at the manor house, a Georgian building of perhaps eighteen or so rooms, attended by a domestic staff consisting of a butler, housekeeper, two cleaners, a butler's assistant, tutor for the children, a nanny, and an estate staff which includes a chauffeur and maintenance staff.

He claims to be a man of substantial private means, making no charge, so he says, for the use of Saint Hill by the organization for conducting advanced courses and for experimental and administrative purposes. He claims that he does not benefit financially from the 10 per cent. of all gross takings which the HASI's throughout the world are required to pay as a levy to scientology headquarters. Nor does he benefit financially, so he claims, from the fees which advanced students pay for the courses they attend at Saint Hill. He says that his own means are provided by the proceeds of the sale of property in America and of a yacht.

He married his present wife, Mary Sue, in 1950. She is as enthusiastic about scientology as he is. It is said that when Hubbard works out a new auditing process, she first of all tries it out on him. It is certain that at least on several occasions she has been his auditor. Psychiatric evidence which the Board heard was to the effect that preclears tended to become dependent upon and under the domination of their auditor, because of certain undesirable features of scientology processing techniques. The extent to which Hubbard has become dependent on and under the domination of Mary Sue has not clearly appeared. That she is a person of considerable misdirected ability seems evident. Her contribution to scientology is substantial.

Hubbard is a man of restless energy, with tremendous enthusiasm in everything he undertakes. He is constantly experimenting and speculating, and equally constantly he confuses the two. He has acquired in a frenetic but superficial way a smattering Or knowledge of many subjects. In very many instances, however, his knowledge is fragmentary and inaccurate and sometimes positively incorrect, yet it serves as the foundation for pretentious and completely misleading pronouncements on scientific matters of which he is ignorant. In other instances where he has not even "enough of learning to misquote," he gives full rein to his imagination to produce incomprehensible and absurd results.

All that he writes and says is either accepted by his followers or, at the very least, it is not rejected. They are taught that they are entitled to question his pronouncements, but they are conditioned to the belief that whatever he says is right.

He has revised the calendar by renumbering the years, starting with the year 1950, the year in which he founded dianetics. Thus, "AD 10" means in the tenth year of dianetics and scientology, or 1960 to those who have not "cognited" on his "sciences".

To keep the illusion of greatness and condescension, he purports to establish what in effect is a Hubbardian dynasty by instituting the International Council of Dianetics and Scientology which is "legalized by the legal structure of HCO Ltd and by my will of December 26, 1960" to which (amongst others, including Williams and his wife), are appointed two quite young children, both surnamed Hubbard, as inheriting members for whom various properties may be held in trust.

On his birthday in 1963, in HCO Bull. of the 13th March, he aspired to almost regal dignity in his "Amnesty," which read -

"On my birthday and on achieving my own fourth goal in clearing, and in celebration of the first Eight first goal Clears by 3M, I hereby extend and direct all the organizations, officials and staff of Scientology Organizations to grant all Dianeticists and Scientologists penalized before this date a complete amnesty . . . Any and all offences of any kind before this date, discovered or undiscovered, are fully and completely forgiven. Directed at Saint Hill, on March the thirteenth, 1963, in the 13th year of Dianetics and Scientology.

L. Ron Hubbard".

His condescension is further evidenced by newsy circulars which he periodically distributes, telling of the busy life he leads, with barely sufficient time for his Pepsi-cola in the afternoon, and how he is improving on the organ which Mary Sue gave him as a present.

Despite his apparent graciousness, he is intolerant of opposition and is autocratic and harsh in his treatment of the dissident. By the imposition of many extra hours of auditing he disciplines the unfortunate who is out of step. He excommunicates the rebellious or threatens that they will be denied further auditing. These threats to an indoctrinated preclear are terrifying.

He is governing director of Hubbard Association of Scientologists International, the operative company of the HASI, over which he exercises complete and autocratic control.

In what passes for research in scientology circles, he is constantly claiming major "breakthroughs," each "discovery" being hailed by him as the ultimate and accompanied by the most extravagant promises (none of which is realized), only to be superseded by his next remarkable "discovery".

Some of his claims are that he has a cure for radiation burns, that he has been up in the Van Allen Belt, that he has been on the planet Venus where he inspected an implant station, and that he has been to Heaven. He even recommends a protein formula for feeding non-breast fed babies, stating that he "picked it up in Roman days".

He has an insensate hostility to psychiatrists and "medical doctors," psycho-analysts, psychologists, and those in other similar professions whose field of study and practice is the human mind. His writings about such professions are quite rabid at times. They are, in his estimation, frauds, charlatans, butchers, and neurotics, and their techniques are, he says, antiquated 19th century practices based on discredited continental procedures and not since developed. By contrast, he offers the pure lily that is scientology.

Hubbard is not concerned with logic as the following passage from *Dianetics: MSMH*, shows,

"Aristotle's syllogism in which two things equal to the same thing are equal to one another simply does not begin to work in logic . . . It is an abstract truth that two and two equal four. Two what and two what equal four? There is no scale made, no yardstick or caliper or microscope manufactured which would justify the actuality, for instance, that two apples plus two apples equal four apples. Two apples and two apples are four apples now if they are the same apples. They would not equal four other apples by any growth or manufacturing process ever imagined".

On his own admission he is not a philosopher. In *Scientology*, issue 15H, copyright 1953, he wrote,

"I am not, and never will pretend to be, a philosopher. The task of a philosopher is to go off and philosophise. Philosophers normally philosophise all the years of their lives, and in the books of philosophers all the absurdities and wisdom can be found. My entrance into this field of better minds was a forced one; I had a feeling that man ought to progress. It was with astonishment that I discovered that man, for all his prate of science, psychotherapy, all his yap of mysticism and philosophy in general, did not even vaguely know how to improve himself".

However Hubbard may appear to his devoted followers, the Board can form no other view than that Hubbard is a fraud and scientology fraudulent. In forming these conclusions, the Board has considered both the oral evidence and the great volume of Hubbard's own writings; the following pages of this Report will reveal that the conclusions at which the Board has arrived in relation to Hubbard are based almost entirely upon his own writings.

One further appraisal of Hubbard, based on his writings, has been made. Expert psychiatric opinion is that the many books, pamphlets. articles. bulletins, and the like which bear Hubbard's name indicate in their author symptoms of paranoid schizophrenia of long standing with delusions of grandeur. This aspect is more fully dealt with later in this Chapter.

At an early stage of the Inquiry, Hubbard indicated that he did not intend to come from England to give evidence before the Inquiry, being, as he said, quite satisfied with the way Williams was handling the matter. Later, an application was made to the Victorian Government and to the Board that Hubbard's expenses be paid to enable him to give evidence. Such application was made, knowing it would be refused - as it was. It was made for the purpose of providing a basis of criticism of the report of this Board, which Hubbard knew by that time must be hostile to him and to his "science". Hubbard had no intention of attending the Inquiry. His dilemma was that if he were to repudiate his writings his deceit would be confessed; and unless he repudiated them he stood condemned by their content. The Board has been conscious of the fact that Hubbard did not attend and give evidence but the decision not to attend lay with him, as also did the decision in the closing days of the Inquiry to withdraw the legal representation of the HASI before the Board.

The falsity of Hubbard's claims concerning himself and his "science" will be examined later in this Report.

Hubbard's Morbid Preoccupation with Perversion.

Much of Hubbard's writings show a morbid preoccupation with matters relating to abnormal behaviour of women, sex, rape, abortions, and similar topics. Especially in his books on dianetics, which are still recommended reading without which it is said one cannot succeed in scientology, he shows a prurient and distinctly unhealthy attachment to abortions, rape, perversion, and similar matters. This attachment, however, is by no means limited to his dianetic writings; there are several such references in his scientology writings also.

In *Dianetics: MSMH* and elsewhere, Hubbard declares that the engram and only the engram causes aberration and psychosomatic illness. An engram is a moment of "unconsciousness" containing physical pain or painful emotion. Most engrams were prenatal, so Hubbard taught, and he goes to fanciful and quite ridiculous lengths to illustrate how a prenatal engram is caused. There is no medical or other rational justification for his wild theories, and most of them are quite contrary to medical knowledge and to reason. Their ridiculous nature is perhaps best illustrated by using his own words, for to paraphrase his writings on this subject may give to them a suggestion of rationality which they certainly do not possess.

In *Dianetics: MSMH* he writes,

"Mama sneezes, baby gets knocked 'unconscious'. Mama runs lightly and blithely into a table and baby gets its head stoved in. Mama has constipation and baby, in the anxious effort, gets squashed. Papa becomes passionate and baby has the sensation of being put in a running washing machine. Mama gets hysterical, baby gets an engram. Papa hits Mama, baby gets an engram, Junior bounces on Mama's lap, baby gets an engram. And so it goes.

People have scores of prenatal engrams when they are normal. They can have more than two hundred. And each one is aberrative. Each contains pain and 'unconsciousness'.

Engrams received as a zygote are potentially the most aberrative, being wholly reactive. Those received as an embryo are intensely aberrative. Those received as the foetus are enough to send people to institutions all by themselves."

In the same book he writes:

"The prenatal child can, of course, experience terror. When the parents or the professional abortionist start after it and thrust it full of holes, it knows fear and pain . . . however . . . Being surrounded by amniotic fluid and dependent for nutrition on its mother, being in a state of growth and easily reformed physically, it can repair an enormous amount of damage and does."

The Board heard expert medical evidence to the effect that the foetus could not know fear and that, if thrust full of holes, it most certainly would not survive.

The following ridiculous statement appears in the same book:

"A large proportion of allegedly feeble-minded children are actually attempted abortion cases, whose engrams place them in rear paralysis or regressive palsy and which command them not to grow but to be where they are forever.

However many billions America spends yearly on institutions for the insane and jails for the criminals are spent primarily because of attempted abortions done by some sex-blocked mother to whom children are a curse, not a blessing of God . . .

The case of the child who, as this is read, is not yet born but upon whom abortion has been attempted, is not hopeless. If he is treated with decency after he is born and if he is not restimulated by witnessing quarrels he will wax. and grow fat until he is eight and can be cleared, at which time he will probably be much startled to learn the truth. But that statement and any antagonism included in it will vanish with the finishing of the clear and his love of his parents will be greater than before".

The chapter from which the above extracts are taken concludes:

"All these things are scientific facts, tested and rechecked and tested again. And with them can be produced a clear on whom our racial future depends."

All this, of course, is untrue and is nonsense. The Board heard evidence from an impressive body of expert medical witnesses which completely negatived Hubbard's pretentious nonsense, including the passages quoted and many others in *Dianetics: MSMH*. No good purpose would he served in painstakingly dealing separately with every incorrect statement of Hubbard and illustrating precisely how it was wrong. Such procedure would be too flattering to Hubbard's writings, which are largely self-condemnatory, and if they are to be commented upon, deserve no more comment than is sufficient to condemn them for the dangerous nonsense that they are.

In Dianetics: MSMH Hubbard writes:

"And in the case of the ulcers, here was baby poked full of holes (Mama is having a terrible time trying to abort him so she can pretend a miscarriage, and she uses assorted household instruments thrust into the cervix to do it) and some of the holes are through and through his baby's abdomen and stomach: he will live because he is surrounded by protein and has a food supply and because the sac is like one of these puncture-proof inner tubes that seals up every hole. (Nature has been smart about attempted abortion for a long, long time.) . . . Grandma lives next door and she comes over unexpectedly, shortly after the latest effort to make baby meet oblivion. Grandma may have been an attempted abortionist in her day but now she is old and highly moral."

In this book Hubbard tells of other cases:

"A standard class of prenatal engram has as its content the worry of the parents that the child will be feeble-minded if not now aborted in earnest. This adds an emotional overload to such engrams and it adds, as importantly, an aberrative condition in the now grown patient that he is 'not right,' 'all wrong,' 'feeble-minded,' and so forth. The difficulty of aborting a child is nearly always underestimated: the means used are often novel or bizarre: the worry

because the child has not come out of the womb after the abortion attempt is acute, and the concern that he is now damaged beyond repair all combine to make severely aberrative engrams."

Again:

"Fathers, for instance, suspicious of paternity, sometimes claim while trouncing and upsetting mothers that they will kill the child if it isn't like Father. This is a very unhappy type of token to say nothing of being, usually, a bad engram; it can go to the extent of remodelling structure, of making noses long or hair assent; it may compel an aberree into a profession he does not admire and all out of the engramic command that he must be like the parent."

And again:

"It is very noisy in the womb. A person may think he has sonic and yet hear no 'womb' sounds, which means that he does not have sonic but only 'dub-in'. Intestinal squeaks and groans, flowing water, belches, flatulation and other body activities of the mother produce a continual sound. It is also very tight in later prenatal life. In a high blood pressure case, it is extremely horrible in the womb. When mother takes quinine a high ringing noise may come into being in the foetal ears as well as her own - a ringing which will carry through a person's whole life.

Mother gets morning sickness, has hiccoughs, and gets colds, coughs and sneezes. This is prenatal life.

The only reason anybody 'wanted' to 'return to the womb' was because somebody hit mother and yelled 'Come back here!' so the person does."

And again Hubbard writes:

"The standard attempted abortion case nearly always has an infanthood and childhood full of Mama assuring him that he cannot remember anything when he was a baby. She doesn't want him to recall how handy she was, if unsuccessful, in her efforts with various instruments In the normal course of work the auditor will have his hands full of Mama screaming objections about her grown son's or daughter's entering into therapy because of what they might find out: Mama has been known, by auditors, to go into a complete nervous collapse at the thought of her child's recalling prenatal incidents. Not all of this, by the way, is based on attempted abortion. Mama often has had a couple of more men than Papa that Papa never knew about; and Mama would very often rather condemn her child to illness or insanity or merely unhappiness than let a child pursue the course of the preclear."

The interpretation which Hubbard seeks to put upon the mother's objection to her child's undergoing auditing is that she fears that her guilty secrets will be revealed; this is in line with his frequent assertion made in his pamphlet, Why some fight scientology (see Chapter 26) and elsewhere that families and friends of preclears oppose their interest in scientology because they have some shameful secret which they fear will be revealed to the preclear.

Again, in Dianetics: MSMH, Hubbard writes,

"Engrams, particularly in the prenatal area, are in chains"

and he lists a number of such chains which include coitus chain, father; coitus chain, lover; constipation chain; contraceptive chain; attempted abortion chain; masturbation chain. He explains that these chains are series of incidents of similar types, and the burden of his writing is that the unborn child records and remembers them, being able to distinguish each particular kind of event. It would seem, from Hubbard's notes, that the child can distinguish between the father and the lover because the father is either drunk or sober and the lover is always more enthusiastic.

In the same book, he writes;

"It may be that a patient is urgent in her insistence that her father raped her when she was nine and that this is the cause of all her misery. Large numbers of insane patients claim this. And it is perfectly true. Father did rape her, but it happened she was only nine days beyond conception at the time. The pressure and upset of coitus is very uncomfortable to the child and normally can be expected to give the child an engram which will have as its content the sexual act and everything that was said."

The Board heard expert medical evidence that an embryo nine days old, hardly visible to the eye, does not possess an ear, is not capable of being aware of anything and has no nervous system or brain capable of recording anything or of having an understanding, intelligent or otherwise, of anything said or done. Further, Hubbard's assertion that the embryo records and understands what has been said involves the embryo being capable of intelligent understanding of language which the child, after birth, would not attain to for several months or even years.

In *Scientology*, issue 28-G, 1954, Hubbard gives three case histories selected at random. In the first;

"The basic consisted of a severe quarrel between his mother and father with several abdominal blows being received by the mother. The mother was protesting that it would make her sick all of her life. At the same time the mother was coughing from a throat blow. The father was insisting that he was master in his own home and that people had to do what he told them. This quarrel occurred at about four and a half months after conception and resulted in the temporary paralysis of the preclear's right side."

The second case was of an eighteen-year-old girl in a condition of apathy "bordering upon a break and worsening."

"The basic proved to be a mutual abortion attempt by the mother and father. The mother said she would die if anyone found out.... The father said the baby was probably like her and he didn't want it. Eighteen penetrations of the head, throat and shoulders with a long orange-wood stick - probably in third month. Several similar incidents completed this chain. Coitus followed each attempt at abortions. Another incident proved to be basic without a chain and with innumerable locks: an attempted abortion by a professional abortionist who used some form of needle and scraper."

The third case related to a male negro who had convulsions when audited on prenatal life.

"The convulsion proved to be twenty engrams nearer birth than the basic which lay on another chain and which was discovered by dream technique. The convulsion was caused by the dramatization of an engram involving the injection of turpentine into the uterus by the mother in an attempted abortion. The main engramic chain consisted of the mother's efforts to abort herself. From engramic content it was gathered that the mother was a prostitute, for as many as twenty experiences of coitus succeeded two of these abortion attempts. . . The basic chain contained many quarrels about money between the mother and her customers. . . The basic incident. . . was found to lie about twenty days after conception, when the mother first discovered her pregnancy."

This man's IQ was stated to be "about eighty-five"; the auditor's was not stated.

Hubbard is particularly concerned to show that the act of sexual intercourse while the woman is pregnant is likely to cause an engram to the unborn child, and he considers that intercourse only a few days after conception could have a highly aberrative effect on the barely existing embryo. Hubbard's theories involve the embryo at that early stage being aware of and registering in an aberrative way conversation between the parties to an act of intercourse. On several occasions he actually sets out the conversation between the parties; not infrequently the conversation is between the mother and her paramour and the topic of conversation is in one case at least the conveniently absent husband.

In *Dianetics: MSMH*, Hubbard writes,

"Attempted abortion is very common. And remarkably lacking in success. The mother, every time she injures the child in such a fiendish fashion, is actually penalizing herself. Morning sickness is entirely engramic, so far as can be discovered, since clears have not so far experienced it during their own pregnancies. And the act of vomiting because of pregnancy is via contagion of aberration. Actual illness generally results only when mother has been interfering with the child either by douches or knitting needles or some such thing.... Morning sickness evidently gets into a society because of these interferences such as attempted abortion, and, of course injury."

Comment would be superfluous.

The Anderson Report CHAPTER 7 HUBBARD'S SCIENTIFIC DEFICIENCIES

Hubbard makes for dianetics and scientology a great many preposterous, ridiculous, and entirely untrue claims. He is a great propagandist and believes that, if he repeats sufficiently often his false claims for his "sciences", his pretence to knowledge in all the other sciences, his rabid denunciation of the medical profession, and the imagined villainy of those who criticise him, there will be sufficient people who will uncritically accept his nonsense as true.

In his books and other writings he frequently repeats the same weird idea in a multitude of alternative ways, often becoming quite didactic in an urgent and frenzied fashion. Any statement, even though initially only tentatively made, is thereafter asserted and repeated as true without any further proof as to its truth than Hubbard's repeated assertions. A common practice of Hubbard is to assert that something "could" be so, without any proof that it is, and upon such a foundation to build his false thesis. In most of his writings, as well as writing for the gullible and the anxious who may be mentally ill or on the verge of mental illness, he is writing for the uninformed and ill educated, who are unable to challenge his confident assertions of "fact", and who accept uncritically what he writes because its deceptive simplicity appeals and they think they are reading about facts, scientifically established, simply because Hubbard has said so. They are happy to read science fiction and to regard it as scientific fact.

It would be wasteful of time and energy to take each and every one of Hubbard's untrue or inaccurate statements and condemn it in specific terms, with detailed reasons in each case. Hubbard continually distorts and misrepresents, frequently asserting as fact propositions which are positively wrong. His writings and theories stand condemned en masse as being entirely contrary to conventional learning and experience in the many sciences in which he falsely claims to be knowledgeable. So far as his own theories are concerned, they are either contrary to existing scientific learning or are unsupported by any proof and are generally contrary to reason. If there is a scintilla which Hubbard has written or said that is justifiably excluded from the foregoing general denunciation of his works, it is of negligible content, and cannot serve as a foundation for the fabric of falsehood, fraud and fantasy which he has forged.

The Board heard many expert witnesses, of high integrity, holding the highest qualifications in their respective sciences. The roll call of such experts is imposing. Their names, qualifications, and appointments are listed in Appendix 3. They were uniformly of the opinion that Hubbard's writings revealed him as ignorant and ill-informed in those sciences in which they were expert. In varying degrees they considered that what he wrote was harmful to those who embarked upon Hubbardian treatment and to the community in general. They expressed, in various terms and with varying emphasis according to their personalities, their concern that such nonsense should circulate under the flattering cloak of science and that such pernicious techniques should be practised on the victims of Hubbard's deception. Again, it is unnecessary to repeat *in extenso* what each individual expert had to say concerning Hubbard and his works.

One remarkable feature of the expert evidence which the Board heard is that the scientology interests, though they were concerned to cross-examine the several experts searchingly and at length, did not call any, or any appreciable, evidence to controvert the opinions and conclusions stated by the experts. It is proper to say that, though the cross-examination was skilfully and

painstakingly done on instructions from the HASI, it served only to confirm, and indeed to strengthen, the several opinions expressed by the expert witnesses in their evidence in chief.

The only effect which the cogent and uncontradicted evidence of such an imposing gallery of expert witnesses could have on any tribunal of fact would be that the tribunal should accept it. Such evidence was inherently probable, and the material which it condemned so improbable, that the Board has no hesitation in accepting and acting upon it.

During the course of the Inquiry, the Board made repeated comments on the continued absence of any substantial scientific evidence in support of scientology teaching, in the hope that, if such evidence existed, it would eventually be tendered. The scientology interests evidently had no intention of tendering any such evidence - if, indeed, it existed - and, though they were given the opportunity and were invited to do so, they declined to lead any such evidence. A few scientology witnesses who gave evidence had scientific or technical qualifications and experience, but their evidence was not, except in the most tenuous way, directed towards justifying scientology on any scientific basis. These witnesses merely sought to justify scientology empirically and subjectively. Typical of their defence or justification of Hubbard's theories, which were contradicted by expert evidence, was the comment of one scientology witness, a medical practitioner who, in relation to Hubbard's theories as to engrams and that ulcers were due to attempted abortions, said, "I don't see how anybody could prove that it was not so. How can you prove it was not so?" Several witnesses sought to justify Hubbard's theories on the basis that they could not be proved wrong, and they took this stand, notwithstanding that they acknowledged that Hubbard's theories were contrary to generally accepted scientific principles. The qualifications and appointments, without names, of these scientology witnesses appear in Appendix 2.

As already indicated, although Hubbard claims that he has degrees, he has no worthwhile academic distinction. He seeks to remedy this deficiency by ridiculing those who possess academic degrees and by denouncing them as rogues and charlatans who perpetuate outmoded and evil ideas and disproved techniques of an earlier century. The Board finds it unnecessary to attempt a vindication of the several professions which Hubbard so abuses. The subject of this Report being scientology, this Chapter is devoted to some discussion of the standards of research said to be engaged in by the founder of scientology, and an examination of the validity of his claims to knowledge in the several sciences on which he claims to write with authority.

Hubbard and Medicine.

Hubbard's completely unjustified claims to speak with authority on medical matters are not confined to psychiatry and psychology. His writings range the gamut of medicine, and many of his theories are based on entirely erroneous ideas of several branches of medicine. A great part of this Report is concerned with the errors of Hubbard in relation to psychiatry and psychology. In this Chapter, other branches of medicine, particularly anatomy and gynaecology, regarding which Hubbard has many fanciful and incorrect ideas, are briefly dealt with. In relation to the matters dealt with in this part of this Chapter, the expert medical evidence which the Board heard included evidence from Professors of Anatomy and Gynaecology and Obstetrics and a Dean of Medicine.

As already mentioned, no good purpose would be served in dealing with the multitude of untrue or inaccurate statements which Hubbard has made in relation to medicine. For the purpose of illustrating the fallacies in his theories, it is sufficient to deal with a few matters which are fundamental to his whole teaching.

Of prime importance to both dianetics and scientology is the "engram" which Hubbard has defined

as a moment of unconsciousness containing pain or painful emotion. It was one of the fundamental axioms of dianetics that the engram was the single source of aberration and psychosomatic ills. the most prolific period for engrams was said to be the prenatal period, and Hubbard's writings contain many examples of the ways in which engrams are said to be received in this period. Several examples are set out in Chapter 6.

Briefly expressed, Hubbard's view was that, when a pregnant woman was struck in the abdomen or engaged in intercourse, the child she was carrying suffered pain, and aberratively heard words which accompanied such incidents, and so received an engram.

"Birth," wrote Hubbard, "is a very aberrative affair," and engrams were frequently received at birth because the child, in the process of being born, was experiencing pain, and, accordingly, words spoken at that stage were said to be aberratively heard and recorded.

Hubbard further considered that a person apparently unconscious under an anaesthetic was in such a condition as to be highly receptive of engrams because of the "injury" which the accompanying surgery inflicted upon him.

The Board heard expert medical evidence that the engram as developed in scientology is unknown to medical science, and that Hubbard's engram theories are based on assumptions which are contrary to or unknown to orthodox medical knowledge and principles. Of course, many of them are also contrary to reason. For instance, Hubbard wrote of engrams being received within a few days of conception. He even goes so far in *A History of Man* as to write,

"Pre-sperm recordings are quite ordinary Pre-ovum sequences are on record but are not common" the implication being that engrams may be received, even though conception has not taken place. The Board heard expert medical evidence that such things were not possible. Such assertions as Hubbard makes require that the sperm or the ovum before conception, or the embryo when it is barely visible to the naked eye, is capable of hearing without any aural equipment and of understanding and remembering without a nervous system or a brain. It involves, too, that the sperm or the ovum or the embryo, as the case may be, already knows a language which, when the child is eventually born, has slowly to be acquired over many months and even years.

The development of any equipment capable of receiving sounds does not take place till the foetus is substantially advanced, and the hearing of sounds by an unborn child, except very shortly prior to birth and then only if they are very loud, is not possible; any sort of intelligent understanding or appreciation of the meaning of words spoken prior to or at birth is, of course, impossible.

Hubbard further wrote that auditing procedures produce prenatal recollections of light. In A History of Man, he wrote that

"There is a 'visio' which is quite standard, of the race. There is quite often a light, a spark, in the sequence"

and that

"Actual prenatal visio, of course, is black except at such times as when light may be entered for surgical purposes, at which time light is sometimes recorded".

The Board heard expert medical evidence that such suggestions were nonsense.

Hubbard has written in many places that many aberrations are the result of injuries inflicted on the unborn child when abortions have been attempted. He writes of the child experiencing terror, fear,

and pain,

"when the parents or the professional abortionist start after it and thrust it full of holes"

and that ulcers in later life are caused by attempts at abortion in which household instruments are used and

"some of the holes are through and through the baby's abdomen and stomach"

and a lot more besides. He writes that attempted abortions are very frequent but generally fail, because

"the ability of the foetus to repair the damage is phenomenal,"

brain damage being ordinarily

"repaired perfectly regardless of how many foreign substances were introduced into it",

and other damage is also attended to, because the child

"is surrounded by protein and has a food supply and because the sac is like one of these puncture-proof inner tubes that seals up every hole".

The Board heard expert medical evidence that all these statements were incorrect. An unborn child does not know terror; a number of holes thrust into a foetus would certainly destroy it; once brain damage occurs it does not repair itself; the punctured sac does not seal itself, and a punctured sac is fatal to the child, unless it shortly precedes the birth of the child.

Hubbard's claim in Dianetics: MSMH that

"a large proportion of allegedly feeble-minded children are actually attempted abortion cases"

is medically absurd.

Hubbard has made many claims to the effect that dianetics and scientology can cure ailments. The broad claims are dealt with in Chapter 19. At this stage it is sufficient to mention some specific claims and to outline expert medical evidence which indicates that the claims to cure are quite unjustified.

Hubbard claimed that

"recording begins in the cells of the zygote - which is to say, with conception",

and that

"Cancer has been eradicated by auditing out conception and mitosis".

Expert evidence was that what was suggested was impossible. Cancer is not of psychological origin and is not cured by any type of psychological treatment. In Chapter 19 are the details of a tragic case where the Melbourne HASI attempted to cure, by auditing, a cancer sufferer. Hubbard's claims that scientology cures cancer, leukaemia, and radiation burns are entirely unjustified and dangerous.

Often Hubbard's ideas are so loosely expressed, and his terminology so inexact, that little meaning can be given to what he writes. In *Self-Analysis in Scientology*, he writes:

"Arthritis of the knee, for instance, is the accumulation of all knee injuries in the past ... The fluids of the body avoid the pain area. Hence a deposit which is called arthritis. The proof of this is that when the knee injuries of the past are located and discharged, the arthritis ceases, no other injury takes its place and the person is finished with arthritis of the knee. And this happens in ten cases out of ten - except in those cases where age and physical deterioration are so well advanced towards death that the point of no-return is passed."

The Board heard expert medical evidence that the assertions in this quotation, if intelligible at all, are quite contrary to accepted medical facts. The fanciful accumulation of "all knee injuries in the past" does not make sense: it is quite incorrect to say that in traumatic arthritis fluid avoids an injured knee, the reverse is the fact; where traumatic arthritis progresses to chronic arthritis due to repetition of injury, the osteo-arthritic condition which develops involves irreversible changes in the cartilage linings of the joint, producing a calcification or ossification of the cartilaginous joints. Such a condition is permanent, and spontaneous retrogression or cure by some mental process is not possible.

In the same book, Hubbard writes.

"Consider the bad heart. The person has a pain in his heart. He can take medicine or voodoo or another diet and still have a bad heart. Find and eradicate or nullify an actual physical injury to the heart and the heart ceases to hurt and gets well."

Expert medical evidence was that such a statement did not make sense, and that any actual heart condition, whether congenital or arising from injury or disease, could not be cured by psychological means.

The Board heard expert medical opinion at length on other medical claims made by Hubbard which were quite alien to accepted medical knowledge; but the foregoing selection is sufficient to illustrate the impossibility of Hubbard's pseudo-medical theories and knowledge.

Hubbard and Psycho-Analysis.

From time to time, Hubbard, in his denigration of the various professions concerned with the treating of the mentally ill, has singled out for particular attack and abuse the practice of psychoanalysis. In a long article entitled "A Critique of Psycho-Analysis", published in *Com. Mag.* Vol. 4, No. 10, October, 1962, Hubbard, while seeking to terrify his followers as to the imagined evils of psycho-analysis, in fact shows both his ignorance and his misconceptions of this particular branch of psychiatric medicine by the patently absurd and erroneous statements contained in the article. For instance, he writes:

"The first solid criticism of psycho-analysis is inherent in its failure to advance . . . Psycho-analysis did not advance or did not evolve. There is little, if any, difference between the writings of Freud in 1894 and the declarations of analysts today unless it is a deteriorated difference . . . Psycho-analysis has not [succeeded], and to-day is almost a lost subject. There are fewer analysts in the world to-day than there were twenty years ago despite the enormous wages which could have been earned by them."

All this is quite incorrect. Throughout the world there are several thousand practising psychoanalysts, and their number is growing. In selected cases, where psycho-analytical methods are indicated, psycho-analytical techniques are successfully practised, and they are being constantly developed. Hubbard shows ignorance of the general literature, learning and development of psycho-analysis. He does not know or understand the meaning of many technical terms which he uses. Though using psycho-analytical terms he gives to them a meaning different from their accepted meaning in psycho-analysis, indicating, that he has not understood the underlying concepts. Psycho-therapy has not stood still as at 1894 as Hubbard asserts but has made great advances. He has the mistaken idea that Freud alone is responsible for psycho-analysis as practised to-day, and he ignores or is unaware of the many famous psychiatrists who, since Freud's time, have assisted in developing psycho-analysis into what is practised to-day. He suffers from the popular but erroneous belief, formerly current amongst some to whom Freud was little more than a name, that, as he puts it, of the practitioner (i.e., the psycho-analyst) reads sexual significance into all discourse and evaluates it for the patient along sexual lines".

Such acquaintance as he has with the teachings of Freud seems to be limited to Freud's earlier writings. Many of Hubbard's theories as to the significance of recollections of "past lives" and "engrams", show that he is unfamiliar with Freud's later work.

His acquaintance with Freud's teachings is claimed to have been initiated by and come through a Commander Thompson, a doctor in the United States Navy, but not otherwise identified, who is said by Hubbard to have been a personal student of Freud and a source of inspiration to Hubbard. For no other reason than this attenuated connexion between Freud and himself, Hubbard writes:

"Better than others, then, some sixty-eight years after Freud's original declarations, I could be considered qualified to criticize not only the failure of the basic work of Freud but the later offshoots which, while following his original tenets, yet sought to expand information on psycho-analysis Very few living analysts today have as a direct connexion with the subject as I have. . . Having established then my possible qualifications to criticize and having compounded such right by having bettered the results of Freud, I feel it is necessary to overhaul rapidly the points of failure of psycho-analysis as we understand the mind today."

All this is patently false. He claims,

"I have used psycho-analysis as a practitioner",

but his writings show almost complete ignorance of what it really is, and such practice as he may have made of psycho-analysis must have been as twisted as his practice of other therapies, good and bad, which he has prostituted. How he, as a science fiction writer, who has spent years travelling the world investigating the customs of peoples in many continents and, during the last fifteen years promoting an impudent fraud, can claim to speak with authority on a subject of which he is almost completely ignorant, makes sense only to Hubbard and his deluded followers.

His critique of psycho-analysis contains such dangerous statements as

"The treatment of the insane today is far worse than it was two centuries ago and the brutality practised under the name of 'mental healing' cannot be regarded with equanimity by any civilized man."

Such a statement would be likely to cause a person in need of psychiatric care to avoid psychiatric treatments at all costs; and, if a mentally ill person, having read such a dangerous statement, later did undergo proper psychiatric treatment, the prospect of treating such a person successfully would be greatly diminished because of the grave obstacle which fear of the psychiatrist had presented.

The evidence of highly qualified psychiatrists and in particular of an expert psycho-analyst with high qualifications and long experience, makes it clear that far from having any real knowledge or appreciation of what psycho-analysis is and does, Hubbard's knowledge does not extend beyond a scrappy acquaintance with some of the popular misconceptions of Freud's early tentative writings.

Hubbard As a Scientist.

The editorial note to Scientology: 8-80 reads,

"The discovery and isolation of Life Energy in such a form as to revive the dead or dying has been an ambition as old as Man himself. In the last two thousand years a few individuals have claimed the ability without explaining it. With this book, the ability to make one's body old or young at will, the ability to heal the ill without physical contact, the ability to cure the insane and the incapacitated, is set forth for the physician, the layman, the mathematician and the physicist."

Scientology: 8-80, which has as its sub-title "The Discovery and Increase of Life Energy in the Genus Homo Sapiens", originally copyright in 1952 and subsequently reprinted at least in 1957, to the extent that it impinges upon anything vaguely resembling the province of the physician, the layman, the mathematician or the physicist, contains just so much nonsense. In this book, Hubbard devises wave lengths for aesthetics, analytical thought, emotion - there are high wave beauty and high wave ugliness - and he gives a "formula" of the energy of life source.

It would be wasteful of time and energy to attempt to discuss aspects of Hubbard's unashamed nonsense in this book. The Board heard evidence from a highly qualified physicist, a master of science and senior lecturer in physics at the University of Melbou rne, who said that much of the text of this book, if written as or claiming to be in any way scientific, was meaningless or just rubbish, and was the sort of nonsense a matriculation or first year student might "dream up" outside his formal study periods. He said none of the views set out in the book bore any resemblance to the theories of physics and that the "bits of science that crop up in this book are not the work of a competent nuclear scientist." Hubbard, this witness said, had not developed at all in physics; he had acquired some familiarity with the language of physics at a high school level - he knew some words existed in a genuine discipline that was physics - but he had not learned physics, and was quite unfamiliar with the scientific usage of the terms which he used.

One witness, called on behalf of the HASI, was also a master of science and had been a senior lecturer in mathematics at an interstate University. He was constrained to admit that as to some of Hubbard's writings in *Scientology:* 8-80 it would be necessary to ask the author just what he meant, and the witness was unable to reconcile other passages in the book with his own scientific knowledge.

Notwithstanding that the book purported to be written for scientists about a branch of science this witness endeavoured to justify those parts of it which were completely inconsistent with generally accepted principles of science on the basis that Hubbard was writing philosophically, and that "you can say something in the name of philosophy which is completely inconsistent with science." The witness was unable to reconcile Hubbard's writings in *Scientology 8-80* with his own scientific knowledge or anything scientific in the generally accepted sense, yet he sought to maintain the validity of Hubbard's writings because, as he said, being written in the framework of scientology there was nothing in such writings which could be "disproven by the science of physics" which was in a different "frame of reference." It is almost incredible that this witness, with such high academic qualifications, could voice such nonsense; yet solemnly he did so, and a careful re-reading of the evidence confirms the impression which the Board had at the time he gave evidence, namely, that he, in common with many other scientology witnesses, appeared to be overborne by some inescapable compulsion which conditioned him to give quite fantastic and incredible evidence on matters scientological, while at the same time appearing to be bright, alert and rational on other matters.

Some scientology witnesses had been processed only shortly before they gave evidence. and this

may have accounted for their peculiar attitude. It did not appear in evidence whether or not this particular witness had been processed shortly before he gave evidence.

Hubbard As a Nuclear Physicist.

One of the many claims made by Hubbard about himself, and oft repeated by his followers, is that he is a nuclear physicist, and his boast is that he was even one of the first nuclear physicists who, in 1932, were studying on lines which finally led to the atomic bomb. He claims to be an authority on the atom bomb and on radiation and he has written a book to prove it. This book, *All About Radiation*, is in two parts. The first part, being entitled "The Facts about the Atomic Bomb" and written by "a medical doctor", deals with certain medical aspects of atom bomb damage and is written in temperate language and, within limits, is reasonably accurate. The second part of the book is entitled, "Man's Inhumanity to Man", and Hubbard is its author. In the book he is described as "Dr. Hubbard" and as "A Nuclear Physicist".

The Board heard evidence from a highly qualified radiologist who has made a special study of radiation and its effects. He said that Hubbard's knowledge of radiation, as displayed by his writings in *All About Radiation*, was the "sort of knowledge that perhaps a boy who has read Intermediate Physics might, with a lot of misapprehensions and lack of understanding, demonstrate". One of Hubbard's assertions is that "a 16-foot wall cannot stop a gamma ray but a body, that is to say human, can". This witness said that this was a complete denial of physical truths and was a basic scientific fallacy which would lead to disastrous consequences in this community at this time if it were not resisted. He said that such a statement "shows complete and utter ignorance of physics, nuclear science and medicine". Speaking in general terms about Hubbard's contribution to *All About Radiation* which he described as "ravings and ramblings", the witness said:

"There is no attempt to present precise data or conclusions from precise data; no attempt to clearly present his views coherently and progressively in terms the reader can follow; colloquial and extraordinary terminology is used to confuse rather than enl ighten; the more basic fundamental established truths of science are ignored and replaced by imaginative fiction without a vestige of corroborative experiment designed to support such hypotheses."

In *All About Radiation*, Hubbard states that "the danger in the world today in my opinion is not the atomic radiation which may or may not be floating through the atmosphere but the hysteria occasioned by that question". This is in line with much of his teaching that most illnesses are caused by the mind, and he claims that man can be exposed to radiation and escape illness or damage, so long as no hysteria is associated with the event. This contention, the witness pointed out, is absolutely untrue and contrary to all authority on radiation which is uniformly of the view that damage done by radiation is quantitatively related to the magnitude of the physical dose received. "Radiation", writes Hubbard, "is more of a mental than a physical problem". This, said the witness, is completely false. Hubbard's ignorance and confusion of thought are further illustrated by his contention that sunburn is the same sort of burn as that caused by atomic radiation. This, of course, is not so, as the evidence made clear.

The witness gave many other examples as appearing in *All About Radiation* of Hubbard's lack of knowledge, misunderstanding and positively wrong views on matters of radiation and biology. It is unnecessary further to itemize instances of Hubbard's i gnorance on the subject of radiation except, perhaps, to refer to a formula which he claims to have evolved for a tonic called "dianazene", for which he claims almost magical powers. He wrote:

"Dianazene runs out radiation - or what appears to be radiation. It also proofs a person against radiation in some degree. It also turns on and runs out incipient cancer. I have seen it run out skin cancer. A man who didn't have

much liability to skin cancer (only had a few moles) took Dianazene. His whole jaw turned into a raw mass of cancer. He kept on taking Dianazene and it disappeared after a while. I was looking at a case of cancer that might have happened."

This, said the witness, was "utter nonsense". Hubbard gave the formula for dianazene in the book, but there was, as the witness said, nothing remarkable about it; it was much the same sort of formula which is found in many proprietary medicines which are given for vague ill health, though the dosage in dianazene may be a little higher than is generally the case in such medicines. The fact that many Melbourne scientologists have taken dianazene is indicative of the degree of influence exercised by Hubbard over them.

One cannot but agree with the witness that it is a tragedy that anything so ridiculous should be inflicted on the general public.

From this witness's evidence it is apparent that Hubbard is completely incompetent to deal with the subject of radiation and that his knowledge of nuclear physics is distorted, inaccurate, mistaken and negligible. No evidence was called which disputed in any way these conclusions.

The dissemination of such nonsense as Hubbard has written in *All About Radiation* may well have dangerous consequences, for people reading and accepting it may develop a sense of false security or neglect or mistreat a condition which might otherwise respond to appropriate treatment.

The Anderson Report CHAPTER 8 HUBBARD'S RESEARCH

If one excludes as "research" the fanciful imaginings of Hubbard, his research material appears to have been almost entirely what preclears have revealed in their processing. Almost if not entirely subjective and based on subjective standards, the "data" which Hubbard uses is that obtained from preclears who, while in a state of hypnosis induced by auditing, or otherwise mentally conditioned by processed which produce hallucinations, imagine fantastic, ludicrous and unusual experiences and thoughts, which Hubbard accepts as conclusive proof and on which he builds his theories. The evidence before the Board shows that processing reports of preclears from the several countries where scientology is practised are sent to Hubbard in England where they are exa mined and collated and the results reduced to some order, from which Hubbard then draws conclusions or gets ideas.

Williams described the development of a particular process called, "listing", wherein the goals of preclears were being sought. He said that preclears were required to provide a list of about 850 goals. These lists were compared and it was found that certain similarities began to emerge. There were many other goals, said Wil1iams, "which were not in common, but there were certain plots of those goals that were in common. The strange thing was, even though a person did not know the plot, with very little steering they could originate the plot. This was rather peculiar. Unless there was telepathic communication, how did the person know of the plot?" The question was rhetorical but the answer obvious. Preclears in auditing session are in a highly suggestible condition, and a "very little steering" would go a very long way. There was an abundance of evidence that preclears came up with particular past experiences only after they had become aware of Hubbard's theories on that particular topic. Furthermore, it is probable that in a list of 850 goals there would be at least some which would be found in another list of similar size. The thoughts, hopes and aspirations of mankind follow a fairly common pattern and their similarity in different people does not require a fanciful theory that they were electronically implanted in thetans trillions of years ago. Early in their sojourn in scientology, preclears become aware of Hubbard's ideas on a great variety of matters, and the evidence shows that when being audited they obligingly provide "data" on these matters, which "data" is taken as confirming Hubbard's theories.

The power of suggestion on preclears is strikingly illustrated by the disclosures by preclears as to helatrobus implants. During 1963 Hubbard promulgated information concerning the helatrobus implants, stating the periods at which these implants were laid in. Thereafter, when preclears were audited on the helatrobus implant of the goal "to forget", they told of the implanting of that goal at some date between those indicated by Hubbard, namely, 38 to 43 trillion years ago.

A further indication of the dependence of Hubbard on the evaluations of preclears during processing is Williams' evidence on "theta traps". In *A History of Man* Hubbard, in writing of "theta traps", states:

"There is no subject more interesting than that of theta traps. It is of vast interest to any invader, It is of vaster interest to your preclear. How can you trap a thetan? By curiosity, by giving him awards and prizes (of an implant), by retractor screens, by mock-ups, by ornate buildings which he will enter unsuspectingly only to be electroniced down; by many such means the thetan is reduced from knowing to a colonist, a slave, a mest body.

All theta traps have one thing in common: they use electronic force to knock the thetan into forgetting, into

unknowingness, into effect . . . The thetan feels himself, in some traps, being drawn up to a post. He fights it with his force. It cannot be suc cessfully fought. He succumbs. A day or a hundred years later, he is picked off and elsewhere used ... there was a theta trap called the fly trap. It was of a gummy material."

Williams said in evidence that Hubbard had no other way of acquiring such knowledge about thetans than from what preclears had revealed.

Hubbard's "research" is done at Saint Hill by himself and a small team of assistants. The facilities for "scientific research" into all the fields with which Hubbard claims great familiarity are very meagre.

Mrs. Williams, the most recently returned advanced Australian student to visit Saint Hill and to study there for about eight months, told the Board that "there is really not a laboratory at Saint Hill". He said that the manor house had three large halls and other rooms, a T.V. room, and a room with tape recording equipment, &c.; that the "experiments" comprised, in effect auditing experiments on the advanced students and a consideration of auditing results supplemented by reports from ,all over the world as to what has been produced by auditing in the way in which Hubbard has directed the auditing to be conducted; that Hubbard first puts forward to the students at Saint Hill whatever is to be audited and when it is successful with them he promulgates it for wider application, even though at that time research on it may not have been finalized; that the student groups which are the subject of such tests at Saint Hill may contain up to about 100 students from various parts of the world, and that the staff consists of from thirteen to twenty individuals, functioning as instructors, assistant instructors, case supervisors, clerks, auditors, and others and performing associated duties.

Gillham, who was also a Saint Hill graduate, said:

"We never saw Ron actually engaged in research, but then, because, as I understand, a lot of research was done in the early hours of the morning, but from the fact that the course was going, was also part of the research programme, as he would observe stu dents and see what they were doing, and then, in his own time, what was going wrong and correct it, and I was there for that." Gillham did not know of anyone else apart from Hubbard who was engaged in research and, as far as Gillham knew, Hubbard's "research" was done in his own private room.

Mrs. Gillham, another Saint Hill graduate, said that the only research she observed at Saint Hill was amongst the students, and that:

"Sometimes Ron would say if he wanted to do research on a certain process and there was at one time a number of students selected to run this process, but the majority of research he gets Mary Sue [his wife] to run. So whatever he worked out, Mary Sue runs on him before he uses it." Mrs. Tampion, who also was at Saint Hill, said that the research there consisted of "auditors, pc's and students running new processes on each other to see how they fared".

How Hubbard, without laboratory, equipment or scientific assistants, could carry out experiments in all the sciences in which he claims to speak as an expert remains unexplained.

Notwithstanding the absence of such indispensable aids, Hubbard claims that all his work in dianetics and scientology is validated by scientific proof and, early in the Inquiry, Williams went to great lengths in evidence in an endeavour to explain and pr ove that Hubbard and scientology proceeded by the scientific method to experiment, test, evaluate, prove and draw justifiable conclusions. However, his evidence was purely argumentative and mere empty words and no evidence in fact was called which remotel y suggested that there was anything scientific - or for that

matter any real method - in Hubbard's experiments and "research".

One witness, a former ardent scientologist, and now a bachelor of science, who languished in the slough of scientology for a few years during which his University course was halted, had the opportunity of observing at close quarters the validity of Hubbard's claims for scientology. His verdict was, "unproven, a lot of words, no evidence".

Hubbard makes claims that he has the proofs, that these proofs are inWashington and elsewhere, that his books and other writings tell of his experiments and their results. Judging the standard, value and extent of his research methods by what he has chos en to make public in his books and other writings, Hubbard fails ignominiously. Several expert scientific witnesses gave evidence to the effect that Hubbard's methods as revealed by his writings have not the slightest resemblance to scientific method.

Experts in many scientific fields criticized the so-called scientific method and research of Hubbard, and out of the evidence which the Board heard emerged the following criticisms, with which the Board wholly agrees: -

- 1. Hubbard is satisfied with, and regards as sufficient, subjective standards of proof;
- 2. He does not test the data obtained by subjective means even when there are means of objective testing; e.g., Hubbard's claims as to exteriorization could be tested objectively, but no such tests appear to have been undertaken. Hubbard and a scientolo gy witness claimed that scientology processing could add 30 pounds of weight to a body by mocking up a mass and bringing it into the body. The Board invited the scientology interests to arrange a demonstration, but the invitation was declined;
- 3. He does not give sufficient detail to establish the validity of his conclusions;
- 4. There is no confirmation of any of his findings by experiments carried out by other workers in the same field:
- The experiments are not described and there is no way of testing his conclusions;
- 6. The material which he uses is -
- (a) not capable of being tested,
- (b) obtained under conditions which do not admit of any control or check,
- (c) necessarily suspect for the reasons, amongst others, that it is generally obtained from persons who are hypnotized or who are in some illusory state, and to whom suggestions have frequently been made:
- 7. A great part of the material relied upon is no more than categorical pronouncements by Hubbard without any suggestion that there is any evidence to justify them;
- 8. An enormous proportion of Hubbard's so-called evidence, alleged results of experiments, findings and conclusions is totally at variance with orthodox theory and actual known facts which are the results of scientifically conducted experiments made under proper control conditions and capable of being duplicated, tested and confirmed.

Hubbard's attitude and that of his followers, as shown by several who gave evidence before the Board, is in effect, "I say such and such is possible, therefore it is. Now you prove me wrong, and since you can't, it must be right." Hubbard categorically asserts as a fact that "a few mornings ago" he was up in the Van Allen belt and that he then went on to Venus, where he inspected an implant station. Prove him wrong!

Each new "discovery" he hails as the ultimate, the final "breakthrough", the answer to all problems, requiring only that auditors develop sufficient skills in the techniques he has devised to "clear" the

world. Yet very soon that "breakthrough" is superseded by another, more wonderful and more promising than the last. And so it goes on.

Reliance on the data provided by preclears in auditing, where hallucinatory images are generated, does not of course bear any resemblance to scientific method. Yet it is on such a foundation that Hubbard has built the whole structure of his teaching. In *A History of Man*, which Hubbard claims is "a cold-blooded and factual account of your last sixty trillion years", Hubbard writes the most fanciful nonsense. Examples of the material contained in *A History of Man* and on which Hubbard based his "research" are set out in Chapter 11. Hubbard published this book in 1952, before the Piltdown man hoax was exposed. In it, Hubbard writes,

"The Piltdown contains freakish acts of strange 'logic', of demonstrating dangerous on one's fellows, of eating one's wife and other somewhat illogical activities. The Piltdown teeth were enormous and he was quite careless as to whom and what he bit and often very much surprised at the resulting damage. Obsessions about biting, efforts to hide the mouth and early familial troubles can be found in the Piltdown. It is a wonderful area in which to locate GE overt acts."

How preclears could recall "real" incidents which could not have happened has yet to be explained by scientology. Hubbard's research on the Piltdown man is surely a hoax upon a hoax.

It is the claim of scientology that it must be valid because it works, the test being "workability". In *Dianetics: MSMH*, Hubbard wrote -

"The only test is whether or not a fact works. If it does work and can be used, it is a scientific fact. And the prenatal engram is a scientific fact. Tested and checked for objective reality, it still stands firm. And as for subjective reality, the acceptance of the prenatal engram as a working fact alone makes possible the clear."

Hubbard finds his proof of workability in the claims made that preclears frequently say they feel better after auditing. Leaving aside for the time being the likelihood that post-hypnotic suggestion may well explain a preclear's statement that he feels better after auditing, workability of itself really proves nothing. This is evident from, for example, the research of Freud whom, incidentally, Hubbard acknowledges as one on whose work he has drawn. Early in the history of psychoanalysis his experiments led Freud to regard recent sexual trauma as being solely responsible for the production of hysterical, neurotic symptoms, He noted that, when his patients recalled a sexual trauma, they tended to lose their symptoms. However, Freud soon noticed that the "cures" were not permanent and he then assumed that the symptoms might have had their origins in more distant sexual traumas. When he investigated this theory he found that patients reported sexual traumas occurring in adolescence and that when they recalled those incidents their symptoms disappeared, but these gains likewise tended to be temporary. Further experiments which brought to light sexual trauma experienced in early childhood once more brought in its train some relief, often only temporary, of the patient's symptoms. However, Freud found that, in some cases where patients reported such incidents as having occurred in early childhood and they appeared to be benefiting from this recall, there was no possibility whatsoever that the incident could have occurred. This, "workability", the fact that a patient may benefit from "recall" of an incident which did not happen, was no proof that the incident happened.

As already mentioned, Hubbard's acquaintance with Freud's work appears to be very superficial and to be confined to Freud's early writings. If Hubbard was aware of Freud's later work and theories, in which Freud later considerably modified his earlier tentatively propounded theories, Hubbard entirely ignores them, for whereas Freud accepted the position that temporary relief was experienced by conjuring up incidents which did not happen, Hubbard wrongly treats the hallucinations of the preclear in relation to things that could not have happened as conclusive proof of such happenings, and on this entirely unwarranted assumption he bases the great bulk of his

theories and teachings on the thetan, the time track, past lives and many other fantasies.

The following extracts from *Dianetics: MSMH* illustrate the nature of Hubbard's "experiments" and the standard of his research. They are examples of how engrams are said to be acquired; one relates to engrams said to have been received at birth and the other relates to engrams said to have been received during a dental operation.

Hubbard considers that "birth is a very aberrative affair"; and to make his point tells of the case where, by processing a patient on his birth, it was established that

"his asthma had been caused by the doctor's enthusiasm in yanking him off the table just when he was fighting for his first breath. He had had conjunctivitis. That came from the eye drops. He had had sinusitis. That had come from the nose swabs used by the pretty nurse."

The second case is described in the following terms:

"Let us make this an example: a man is under nitrous oxide (the most vicious anaesthetic ever invented as it is actually not anaesthetic but a hypnotic) undergoing exodontistry. As usual everybody present around the 'unconscious' patient chatters and yaps about the patient, the weather, the most popular movie star, of baseball. The exodontist is a tough character, bossy to the nurse, apt to be angry about trifles; he is also very sympathetic toward the patient. The nurse is a blue-eyed blonde, who is sexually aberrated. The patient, actually in agony, receiving an engram amongst engrams which may ruin his life (terrible stuff, nitrous oxide; really hands out a fancy engram as any dianeticist can attest) is unanalytical. Everything said to him or around him is taken literally. He takes the valence of the exodontist as both the top valence present and the sympathetic valence. But every phrase uttered is aberrative and will be interpreted by that happy little moron, the reactive mind, on the order of Simple Simon who was told he had to be careful how he stepped in the pies, so he stepped in them carefully. These people may be talking about somebody else but every 'l' or 'he' or 'you' uttered is engramic and will be applied to others and himself by the patient in the most literal sense. 'He can't remember anything' says the exodontist. All right, when the engram keys-in, this patient will have an occlusion on memory in greater or lesser degree. 'He can't see or feel it': this means an occlusion on sight, pain and tactile. If the patient has his eyes watering in agony at the moment (though completely 'under') he may get actual bad vision as well as poor visual recall from this experience. Now they put him in the hands of this blonde nurse to let him sleep off the drug and recover. She is an aberree amongst aberrees. She knows patients do weird things when they are still 'out' so she pumps him for information about his life. And she knows they are hypnotic (yes, she sure does) so she gives him some positive suggestions. Amusing herself. She says he'll like her. That she'll be good to him. And stay there now for the present.

"So the poor patient, who has had two wisdom teeth, impacted, taken out, has a full anger-sympathy dramatization. The general tone he takes is the tone the exodontist showed to the others in the room. The exodontist was angry at the nurse. With his recalls all messed up, the patient a few years later meets a woman similar to this nurse. The nurse has given him compulsions towards her. The silly little moron, the reactive mind, sees in this entirely different person enough similarity to create an identity between the nurse and this new woman. So the patient divorces his wife and marries the pseudo-nurse. Only now that he has married the pseudo-nurse the dental engram begins to key-in in earnest. Physically he gets ill: the two molars adjacent to where the wisdom teeth came out develop large cavities and begin to rot (circulation shut down, pain in the area but can't be felt because there's a pain recall shut-out). His memory goes to pieces. His recalls become worse. He begins to develop eye trouble and a strange conjunctivitis. Further (because the dentist leaned on his chest and stomach with a sharp elbow from time to time) he has chest and stomach pains. This nitrous oxide hurt his lungs and this pain is also in chronic restimulation. But most horrible; he believes that this pseudo-nurse will take care of him and he stops to some degree taking care of himself in any way; his energy dissipates; and analytically he knows it is all wrong and that he is not himself. For he is now fixed in the valence of the exodontist who is angry with this nurse and so he beats the pseudo-nurse because he senses that from her all evil flows. The girl he married is not and was not the nurse, she sounds something like her and is a blonde. She has her own engrams and reacts. She attempts suicide.

"Then, one day, since this is one engram among many, the mental hospital gets our patient and the doctors there decide that all he needs is a good solid series of electric shocks to tear his brain up, and if that doesn't work, a nice ice-pick into each eyeball after and during electric shock, the ice-pick sweeping a wide arc to tear the analytical mind to pieces. His wife agrees. Our patient can't defend himself; he's insane and the insane have no rights, you know.

"Only the cavalry, in this one case, arrived in the form of dianetics and cleared the patient and the wife and they are happy today. This is an actual engram and an actual case history. It is a sympathy engram, pro-survival on the moronic reactive mind level".

The Anderson Report CHAPTER 9 DIANETICS

Dianetics is described by Hubbard as "the modern science of mental health". He explains that its derivation is from two Greek words, "dia" meaning "through" and "noos" meaning "mind". An examination of the content of dianetic teaching suggests that its derivation would more aptly be from Diana, the moon goddess, and that a word with a Latin root but much the same meaning would have been more appropriate.

Hubbard wrote a number of books about dianetics and it has been the subject of a substantial number of articles in scientological publications. Before discussing dianetics in the light of the evidence placed before the Board some excerpts from the books all articles written by Hubbard can conveniently be referred to for an idea of the fanciful claims which Hubbard made for it.

The word "dianetics" is said to be "a term employed to embrace the science of thought and including a family of subsciences by which the individual collective activities of mankind may be understood and bettered".

Hubbard writes that dianetics is "a sub-subject of scientology and covers the anatomy of the mind rather than the brain"; it is "that branch of scientology that covers mental anatomy"; it is "a form of science of thought applicable to psychosomatic ills and individual aberrations". Its purpose is "to pass man across the abyss of irrational, solely reactive thought and to enter a new stage of constructive progression to the ultimate goal."

The goal of dianetics is "a world without insanity, without criminals and without war". It looks for a simpler course to the achievement of its aims than through the conventional therapies. In *Science of Survival*, which was first published in 1951, Hubbard wrote

"It took four to twelve years to get an inkling of these [conventional] therapies and all evidence to hand, carefully compiled, shows that they do not work, that the problems of criminality, insanity and war still remained, with these systems of 'therapy' far out of control."

He then claimed that dianetics was a simpler solution, which did not require twelve years of schooling and practice to learn, but which provided "an understanding of life, man, and mind operation which could resolve the 19,000,000 insane, our millions of criminals, and international madness".

Dianetics is, wrote Hubbard in *Dianetics: MSMH*,

"actually a family of sciences With the techniques presented in this handbook the psychiatrist, psycho-analyst and intelligent layman can successfully and invariably treat all psychosomatic ills and inorganic aberrations. More importantly, the skills offered in this handbook will produce the dianetic clear, an optimum individual with intelligence considerably greater than the current normal, or a dianetic release, an individual who has been freed from his major anxieties or illnesses. The release c an be done in less than twenty hours of work and is a state superior to any produced by several years of psycho-analysis, since the release will not relapse."

The principal books on dianetics are *Dianetics: MSMH*, *Science of Survival*, and *Dianetics 1955*. These books contain an exposition of the principles of dianetics and instructions for the practice of dianetic therapy as well as a large number of instances of the alleged beneficial application of

dianetics. Fundamental to dianetics is "survival", towards which all activity should be directed. "The dynamic principle of existence is "survive!". The first axiom is: survive. "We do not know", writes Hubbard, "to what end we are surviving and in our field of the knowable and in our choice of only the workable axioms we do not know and have no immediate reason to ask why".

Survival, considered as a single and sole purpose, is said by Hubbard to subdivide into four dynamics.

"Dynamic One is the urge of the individual towards survival for the individual and his symbiotes. ['Symbiotes' means all energies and entities which aid survival.]

Dynamic Two is the urge of the individual towards survival through procreation; it includes both the sex act and the raising of progeny, the care of children and their symbiotes.

Dynamic Three is the urge of the individual towards survival for the group or the group for the group and includes the symbiotes of that group.

Dynamic Four is the urge of the individual towards survival for Mankind or the urge toward survival of Mankind for Mankind as well as the group for Mankind, &c., and includes the symbiotes of mankind."

The mind of an individual, so dianetics teaches, has really three parts. They are the analytical, the reactive and the somatic.

The analytical mind, sometimes referred to as the conscious mind, is that portion of the mind which perceives and retains experienced data to compose and resolve problems and directs the organism along the four dynamics. It thinks in differences and similarities.

The reactive mind is that portion of the mind which sites and retains pain and painful emotion and seems to direct the organism solely on a stimulus-response basis. It thinks only in identities.

The somatic mind is that mind which, directed by the analytical or reactive mind, places solutions into effect on the physical level.

Williams understood the "somatic mind" as a term used to describe

"that package of information or data appertaining to the automatic function of the body, and automatic responses of the body. The somatic mind, functionally, would include such things as the automatic heart beat and breathing and this sort of thing."

In Dianetics: MSMH. Hubbard wrote that

"the hidden source of all psychosomatic ills and human aberrations has been discovered and skills have been developed for their invariable cure."

The goal of dianetics being to cure psychosomatic illness and remove aberrations, dianetics was therefore dedicated to "running out" and so removing what Hubbard said was the cause of all the trouble, the engram. According to dianetics, psychosomatic ills are caused by engrams, the engram being the single source of psychosomatic ills and aberrations.

"The engram and only the engram causes aberrations and psychosomatic illnesses."

Aberrations include all deranged or irrational behaviour. Moments of "unconsciousness", when the analytical mind is attenuated in a greater or lesser degree, are the only moments when engrams can be received. The engram is a moment of "unconsciousness" containing physical pain or painful emotion and all perceptions and is not available to the analytical mind as experience.

Hubbard declared that all psychosomatic illnesses,

"which constitute 70 per cent. of man's illnesses, can be positively cured by dianetic processing".

This, he said, was effected by auditing the preclear or sufferer and thereby discovering and running out the engram. The source of the aberration was said to be the reactive mind which acted more or less irrationally and fed information to the conscious mind and body, without the knowledge or consent of the individual, save that sometimes the individual might perceive that he was acting irrationally in relation to some matter, but could not understand why. An individual, according to Hubbard, thus possessed two minds; the analytical mind, which is the conscious mind which becomes aware of things learned by actual experience and functions logically; and the reactive mind, which is the unconscious mind and which operates irrationally or irresponsibly and is the cause of all aberrations. The problems of life are complicated by the functioning of the reactive mind, and it is the cause of ill-health and mental troubles.

The reactive mind, so Hubbard claims, received engrams in the past. Initially, the "past" was earlier in the life of the individual concerned but quickly Hubbard propounded the theory that many engrams, perhaps most, were incurred in the prenatal period, i.e., during the pregnancy of the mother; and at this period the reactive mind of the unborn child became aware of events happening and words spoken at a time when it, the unborn child, received some injury, e.g. by the father hitting the mother in the stomach, or much more frequently by an injury received by the embryo or foetus during intercourse between the parents, the embryo being aware of and suffering injury and remembering such incidents and accompanying words from as early as eight days after conception. As these theories further developed, Hubbard claimed that engrams could be experienced by the individual even before conception! All of this, of course, is nonsense, and a large body of expert evidence was called to prove that it and much more fantastic theorising was nonsense, However, Hubbard, disdaining to admit such theories as fantasy, boldly asserts them to be scientifically proven facts.

An attempt was made by scientology witnesses at the Inquiry to show that dianetics is now, in effect, in abeyance, but that is clearly not so; it was the forerunner of scientology, and out of it scientology developed; and it is claimed by Hubbard that dianetics is part of scientology. Hubbard has consistently maintained that an understanding of his book, *Dianetics: MSMH*, is essential to an understanding of scientology, and he has repeatedly, right up to very recent times, directed that every effort be made to sell this book, a million copies, so he claims, having already been sold. Its present price is £2 3s. 6d.

In dianetics Hubbard considered that

"The entire physical pain and painful emotion of a lifetime, whether the individual 'knows' about it or not, is contained, recorded, in the engram bank. Nothing is forgotten. And all physical pain and painful emotion, no matter how the individual may think he has handled it, is capable of reinflicting itself upon him from this hidden level, unless that pain is removed by dianetic therapy."

Hubbard writes in *Dianetics: MSMH*,

" 'Dianetic therapy', may briefly be stated. Dianetics deletes all the pain from a lifetime. When the pain is erased in the engram bank and refiled as memory and experience in the memory banks, all aberrations and psychosomatic illnesses vanish, the dynamics are entirely rehabilitated and the physical and mental being regenerate".

Hubbard claimed that in dianetics had been discovered the method of refiling pain and that

"wide awake and without drugs an individual can return to any period of his entire life providing his passage is not

blocked by engrams".

The technique or therapy is done by what is called "dianetic reverie", and is applied by an auditor, who

"directs the attention of the patient to the patient's self and then begins to place the patient in various periods of the patient's life merely by telling him to go there rather than remember. All therapy is done, not by remembering or associating, but by travel on the time track. Every human being has a time truck. It begins with life and it ends with death, It is a sequence of events from portal to portal as recorded".

That many of the processes used in dianetics were hypnotic is evident both from their nature and their operation but Hubbard either ignores or denies this fact. In these processes the processing was towards clear, which was at the time said to be the optimum state attainable by man. In such a state, all illness and aberrations fell away. An intermediate stage was the release.

"In a release, the case is not progressed to the point of complete recall. In a clear, full memory exists throughout the lifetime, with the additional bonus that he has photographic recall in colour, motion, sound, &c., as well as optimum computational ability".

Dianetics taught that past experiences were recorded,

"that the engram recording was probably done on the cellular level, that the engram bank was contained in the cells", that "the engram is a recording like the ripples in the groove of a phonograph record".

Hubbard was emphatic in pointing out that what were impinged on the cell were not memories but actually prenatal engrams or recordings, and that these recording at a cellular level were capable of being made from the earliest stage of prenatal development onward; even at the zygote stage. The most prolific source of engrams was, so Hubbard considered, in the prenatal period, and his dianetic writings are cluttered to a degree which is obsessive with examples of how the prenatal engrams were experienced by the embryo, sometimes only a few days after conception, and by the foetus, as a result of violent assaults on or other physical injury to the mother, attempted abortions and intercourse which is often described as promiscuous. In Chapter 6, aspects of prenatal engrams and Hubbard's morbid preoccupation with abortion and sexual perversion are dealt with.

About 70 per cent. of the physician's "current roster of diseases", so Hubbard claims, falls into the category of psychosomatic illness. "Psychosomatic illnesses", Hubbard explained, "are those which have a mental origin but which are nevertheless organic". Hubbard writes,

"Arthritis, dermatitis, allergies, asthma, some coronary difficulties, eye trouble, bursitis, ulcers, sinusitis, &c., form a very small section of the psychosomatic catalogue. Bizarre aches and pains in various portions of the body are generally psychosomatic. Migraine headaches are psychosomatic and, with the others, are uniformly cured by dianetic therapy. (And the word cured is used in its fullest sense.)"

The Anderson Report CHAPTER 10 ASSOCIATION BETWEEN DIANETICS AND SCIENTOLOGY

Hubbard founded the Hubbard Dianetic Research Foundation in California in 1950. By 1951, his experiments in dianetic processing had reached the stage where he wished to develop the past lives theory beyond merely the prenatal period. He found in dianetic processing that preclears came up with incidents which could not have happened either in this lifetime or in the prenatal period. He accepted the incidents as real, though it appeared that they had occurred prior to the prenatal period, sometimes very many years earlier. In 1951, because of differences between himself and the board of control of the Hubbard Dianetic Research Foundation, he withdrew from that organization and continued alone his probing of the past. In this way, he founded scientology, which involved his "discovery", by what he claimed to be scientific methods, of the thetan or the soul or the spirit, which he claimed had existed for all time and had possessed countless bodies, over millions of years. There now opened before him the entrancing and limitless vista of the past with boundless scope for his imagination.

The Hubbard Dianetic Research Foundation seems to have fallen on hard times by about 1954, and Hubbard acquired it in some way which did not clearly emerge in the evidence. By 1955, dianetics was Hubbard's once more, and he held a congress to celebrate the reunion which he called the Unification Congress.

Scientology witnesses repeatedly asserted during the Inquiry that dianetics was not now practised. Their anxiety to establish this contention was understandable, because of the forthright claims made by dianetics to cure with certainty a great number of man's ills. Hubbard has not retracted any of the claims which he had earlier made for dianetics, and the scientology witnesses did not resile in any way from the claims which had been made as to the efficacy of dianetics to cure with certainty 70 per cent. of man's psychosomatic illnesses and aberrations. They said, however, that at the present time scientologists were not interested in curing ills but only in making the able more able. Tampion said that, though the highly efficacious skills of dianetics were available to cure the mentally ill, it was uneconomic to do so, as far quicker results were obtained "making the able more able" than by making the less able, able.

The scientology witnesses pointed out what they claimed were important distinctions between dianetics and scientology. For instance, Hubbard in HCO Bull. of 1st July, AD 13 (1963), wrote:

"With the advent of Scientology with its complete shift from Dianetic goals, healing went out as a reason for running engrams and concern about the body vanished as an auditing target Engram running has vanished as a healing process. Engram handling by chains has emerged as an entirely reoriented subject, not even vaguely connected with the body and with the target not of a human clear but of Operating Thetan."

It was said that since the discovery that the thetan had engrams the theory that the engram recording was cellular did not exist in scientology (notwithstanding the positive assertion in dianetics as a proven fact that the engram recording was cellular), because the recording was now of the engram of the thetan which had no cells on which the recording could be impinged.

Williams explained what he described as the "principal changeover" from dianetics to scientology thus: "The early dianetic work said that what aberrated the individual was what was done to the individual, whereas in scientology we see introduced the concept that it is what the individual

himself has done that can aberrate him," that is, the individual takes responsibility for his own condition. Williams further expressed the distinction between dianetics and scientology thus: "At the end of a dianetics se ssion he (the preclear) would say, 'I have recovered from something someone did to me.' At the end of the scientology session he would say, 'I have recovered from something as a result of what I have done.' So, it is contradictory".

One important aspect of scientology auditing is that a person should be "at cause," just as a clear is said to be at cause, and that a person should be responsible and should accept responsibility. The preclear is persuaded in auditing that, by probing the past and uncovering all aspects of whatever may be the situation being audited, he will understand how and why the situation came about. He is said then to have a clearer understanding of things and to accept responsibility, for auditing tends to explain whatever may have happened - or the preclear believes happened - as due to himself and to decisions made or acts done or thoughts entertained by himself. Nobody is sick, asserts Hubbard, unless he wants to be sick; even a soldier wounded by enemy action is said to be responsible for his wound because of his own non-survival thought or action. Anything which has happened to a person, so it is said, is because of some non-survival act which he himself is responsible for. The preclear thus examines his past in an accusative way, accepting responsibility and acknowledging guilt for happenings in respect of which he was in no way culpable. The quality of guilt is thereby given to the most innocent action, and the acceptance of responsibility in this distorted way, far from benefiting the preclear, leaves him prev to great anxieties and predisposed to undertake more auditing to find relief from the very anxieties that auditing produces.

Williams, in pointing out a distinction as to responsibility according to whether the preclear has been dianetically or scientologically processed, merely illustrates an immaterial distinction, for the techniques are essentially the same, and if scientology preclears soliloquize differently from dianetic preclears it is merely because they have been so indoctrinated.

Though there do exist these and other distinctions between dianetics and scientology, though the techniques which have developed in recent years are called by scientological names and are said to be directed towards the thetan or spirit rather than the body, there has continued to be a close and unbroken association between the two As already mentioned, dianetics is repeatedly referred to in scientological writings as a sub-subject of scientology - as that part of scientology which deals with mental anatomy - and scientology witnesses agreed that dianetics was still a part of scientology. Even during the Inquiry the two names appeared boldly across the front of the HASI premises at 157-9 Spring Street, Melbourne.

In HCO Pol. Lr. of the 26th February, 1961, Hubbard refers to the "International Council of Dianetics and Scientology", which "is legalized by the legal structure of HCO. Ltd. and by my will of December 26, 1960," to which Council Williams and Mrs. Williams were appointed, the purpose of the Council being stated to be, "to ensure the smooth running of Dianetics and Scientology throughout the world, to safeguard and increase their money and properties and to provide good administration, excellent service and justice".

Numerous bulletins contain references to dianetics which indicate that it is still very much alive.

Hubbard has been most insistent in numerous bulletins that the sales of his book, *Dianetics: MSMH* should be pushed virtually to the exclusion of other literature. In HCO Bull. of the 28th April, 1960, Hubbard writes,

"Dianetics: The Modern Science of Mental Health contains today a perfectly workable therapy. But more importantly it

contains a bridge between the uninformed and the informed public on the subject of scientology".

In the same bulletin, Hubbard writes,

"When people are asking you questions about Dianetics and Scientology, no matter how obtuse or abstruse the questions are, your best answer to these questions was my earliest answer and that was, Read *Dianetics: The Modern Science of Mental Health* and that will answer your question . . We have made a break-through. The moment of the break-through is recorded at public level with *Dianetics: The Modern Science of Mental Health*. If people do not read this book, they just will not have broken through To all Central Orgs. Push this book with every possible display and mention Play down all other PE books, display *Dianetics: The Modern Science of Mental Health* as the book they must now buy We've lost the people in a maze of many titles. Take down all your modern book displays. Concentrate on one, *Dianetics: The Modern Science of Mental Health*".

This direction makes it clear that dianetics is a current study and very much alive. Obediently to Hubbard's command the Melbourne HASI gave precedence to *Dianetics: MSMH* and prominently displayed the book in its window.

In HCO Bull. of the 6th October, 1960, Hubbard directs that professional auditors be written to telling them that there are unlimited supplies of *Dianetics: MSMH* available and that

"As we feel that this book is the best possible dissemination for Scientology, we want to get it out to people with the least possible delay". As an inducement to the auditors, the instruction continued, "For each and every copy of *Dianetics: The Modern Science of Mental Health* that you buy from us for cash you will get a credit of £1 in the HGC. This offer is only open to professional auditors - the public will buy the book from you in the usual way".

The Melbourne HASI monthly magazine *Communication* of which Mrs. Williams was at least until October, 1964, the editor, bore at that date the legend "The Official Periodical of Dianetics and Scientology in Australia".

The *Com. Mag.* of May, 1960, states in relation to *Dianetics: MSMH* that to say it was "invaluable both to the Student of Scientology and to anyone wishing to find his way in the labyrinth of human behaviour, is an understatement: the book is, in fact, essential reading". The *Com. Mag.* of June, 1961, reprints part of the first chapter of *Dianetics: MSMH*.

The Com. Mag. of November, 1962, refers to *Dianetics: MSMH* thus: "This book describes the states of a 'release' and 'clear' obtainable through Scientology processing". The *Com*. Mag. of February - March, 1963, describes *Dianetics: MSMH* as "the basic text book on the mind and clearing"

The *Com. Mag.* of May, 1965, is mainly an advertisement for *The Original Thesis* written by Hubbard in 1948, which is described as "The Original Text on Dianetics" and is said to be "an excellent beginning text for anyone starting in Scientology".

Many ideas developed in dianetics were adopted into scientology, e.g., the engram, the time track, the clear, the release. It may well be that emphasis and direction have changed, but dianetics has always remained and is still an integral part of scientology. Witnesses were constrained to admit that it was important to scientology. Williams conceded that "it has key importance", but the explanation of the HASI for the continued association of *Dianetics: MSMH* with scientology was that it provided helpful background material and enabled the history of scientology to be appreciated and understood, so that the inquirer "can read through Dianetics and can understand the rest of it".

The connexion between dianetics and scientology and the continued prominence given to *Dianetics: MSMH* were of much more significance and importance in this Inquiry than historical or background material. As will appear later in this Report, for a considerable number of years,

scientology, in a guarded and veiled manner yet nonetheless deliberately and consistently, has been purporting to treat illnesses and effect cures.

Protestations that scientology does not claim to cure are idle, because at the same time the HASI offers to inquirers its dianetic literature which proclaims the efficacy at wholesale healing of one of the branches of scientology. In the Melbourne HASI *Com. Mag.* Vol. 1, No. 10, there appears in bold type the announcement that "All you have read about in scientology and dianetics can be achieved by you with processing from the Hubbard Guidance Centre". The two subjects are deliberately woven together in this confusing way in order to deceive the uncritical who are led to believe that scientology has healing techniques, and at the same time to enable the HASI, when challenged by government authority, to reply that is does not claim to heal. The scientological attitude is, in effect, "Scientology does not claim to cure, but in the past it has done the wonderful things set out in *Dianetics: MSMH* and other books. Of course it does not claim to cure but" The success of this technique is measured by the number of mentally and physically ill people who have patronised the Melbourne HASI. Such number does not admit a precise definition, but it is evident from an examination of numerous files that the technique adopted by the HASI of disclaiming to heal but pointing to past achievements was effective in many cases.

The Anderson Report CHAPTER 11 SCIENTOLOGY THEORIES

In Scientology: The Fundamentals of Thought, written by Hubbard in 1956 and described as "The Basic Book of the Theory and Practice of Scientology for Beginners", Hubbard seeks to discuss and explain the basic principles of scientology. In quoting from this and other writings of Hubbard, the Board has not followed Hubbard's practice of printing in capitals the many words, phrases and even sentences which he dignifies with capital letters for the purpose of emphasis.

It is also to be noted that, in many of his writings, Hubbard puts in parenthesis after certain words other words with the same or substantially the same meaning for the assistance, so he explains, of translators. In quoting Hubbard's actual texts this Report includes the words which Hubbard has put in parenthesis.

One further preliminary observation should be made: the attempted codification or description of scientology theories in this Chapter - and, indeed, in any other part of this Report - should not be taken as indicating in any way an acceptance of their validity.

Cycle of Action.

Hubbard writes in *Scientology: The Fundamentals of Thought* that fundamental to scientology is what he calls the "Cycle of Action" which is "an apparency as follows: Create, then survive; then destroy; or Creation, Survival, Destruction". Hubbard illustrates the apparent cycle of action by this example: "A child is born, he grows, he reaches manhood, he grows old, he dies". However, he stresses that this is only the apparent cycle of action and that "The actual cycle of action is as follows: Create, create-create, create-counter-create, no creation, nothingness": "create-counter-create" being to "create something against a creation", which equals "to create one thing and then create something else against it", which equals to "destroy".

Hubbard says that an actual cycle of action consists of various activities, each of which is creative. By way of illustration he takes a wall which is seen to be standing.

"To be apparent it is necessary that the wall be constantly created. The act of 'destruction' is to exert against the wall another creativeness that of the action or activity of knocking the wall down. Both the wall standing there and the action of knocking it down are 'creative' actions. Because we may object (to argue against, dislike) to a wall being knocked down, we vilify (swear at, scorn) the creativeness involved in knocking it down with the word 'destructive'. Actuality tells us there is no such thing as destruction. There is only creation against a creation. There is another 'type of destruction' and this is no more creation. By no longer being a party to (a member of) the wall's creation, the wall, in theory, can cease to exist for one. This is true in actual practice in Scientology ".

Reality.

From this line of "reasoning", which is reminiscent of discarded oddities of earlier philosophic speculation, Hubbard then propounds the proposition that what is real for a person is real for that person and what is unreal for that person is unreal for that person, and his followers carry this

proposition to extreme lengths. On this basis, several witnesses were prepared to accept Hubbard's claim to have gone to Venus on a particular occasion in this lifetime, because Hubbard had said he had been there a few days before and if it was real to him then it was real to him.

"Reality," writes Hubbard, "is the way things appear. Reality is apparency Reality is fundamentally agreement. What we agree to be real is real". This statement and the confusion of the use of the word "create" illustrates the wholesale distortion of the meaning of words in which Hubbard engages. His writings abound in instances of words being assigned meanings different from their accepted or conventional meanings and he frequently invents new words for his purposes. Of course, each branch of learning has its own vocabulary, and a person, whether he be a lay person or the founder of scientology, may make his own dictionary. But Hubbard confuses his followers by the use of common words in a quite artificial and contradictory way, and obscures the meaning which he is endeavouring to convey.

Throughout the Inquiry, the Board has been aware of the difficulty caused by Hubbard's misuse and abuse of common English words, and it has endeavoured to bear in mind the distorted meanings which Hubbard has given to such words. However, allowing for Hubbard's peculiar deviation from standard English meanings, the Board finds no more acceptable to reason the great body of Hubbard's teaching on scientology theory.

The cycle of action thesis which he propounded may appeal to some as an interesting view of life, but one cannot ignore the fallacies which it contains, such as reality being an apparency and the claim that for a thing to exist it must be constantly created, and that in scientology a thing may cease to exist if one ceases to think it exists.

There are three conditions, writes Hubbard, which comprise life, and they are "be, do and have": the conditions of "being" or "beingness", "doing" and "havingness". Hubbard teaches that people's beingness, doing and havingness may vary greatly according to their mental conditions and that scientology processing can produce in a person a greater degree of beingness and havingness, as well as raise his ability to do things.

"This is Life".

One of the highly recommended books on scientology described as standard reading for beginners, is a volume of about 100 pages entitled, *This is Life, An Introduction to Scientology* by Reg Sharpe, an American. This book was written in 1961 and reprinted on several occasions, including once in Australia. Sharpe somewhat simplifies the cycle of action, stating it to be simply "Create-Survive-Destroy (Decay)" or, as applied to a human body" Birth-Life-Death" and he quotes axiom 13 of the "Axioms of Scientology" in which Hubbard has written "The cycle of action of the physical universe is Create, Survive (persist), Destroy".

This is Life is a simply written book, in which the author takes simple examples to illustrate his points, but lacks logic and assumes a multitude of premises without proving them - in effect, leaving it until later when the student will become more advanced in scientology thought and able then to "cognite" on them without having to be satisfied that such premises are valid.

Stable Datum.

Sharpe stresses the need for a person to have "stable data" in all matters, and he quotes axiom 52 as follows, "A Stable Datum is necessary to the alignment of data". It emerges from a consideration of Hubbard's and Sharpe's writings on stable data that so long as the stable datum

is true for you, all is well, even though it may not be true for others; and, further, if a stable datum which you have chosen turns out to be untrue for you, you start off with a new stable datum and again all will be well. Sharpe readily assumes that the reader will cognite on scientology and requests him to read the book uncritically by the following exhortation

"the reason I have drawn your attention to all this is because in studying this book and, indeed, anything Scientological, I want you to be sure that the data presented is true. If you sit and try to prove it wrong you will be in trouble. On the other hand I do not wish to try to prove it right. I just want you to read it and make up your mind as to its truth or otherwise. I do hope you will neither accept nor reject anything just because you do not understand Above all I want you to discover for yourself what is true".

He then deals with people who are sure of themselves and who, he writes, have "certainty", because they have the most stable data.

"There are so many things on which they are certain. In consequence they are better able to sum up situations and act to the best advantage. At worst we have the insane. There is little if anything that they are certain about. Certainty you will observe is closely allied to sanity. The more certainties a man has the more sane he is."

In this insidious way, the reader of *This is Life* is led to read the rest of the book, always with the subconscious thought that what he is reading is right and that his own deficiencies are the explanation for his failure to understand and accept, and the further subconscious thought that to lack certainty about any proposition in the book is, to that extent, to be insane.

The Eight Dynamics.

In Scientology: The Fundamentals of Thought, Hubbard writes, "As one looks out across the confusion which is life or existence for most people, one can discover eight main divisions to each one of which applies the conditions of existence. Each division contains a cycle of action." These divisions are called "dynamics" and there are eight dynamics which, according to Hubbard, are: -

"THE FIRST DYNAMIC - is the urge towards existence as one's self. Here we have individuality expressed fully. This can be called the self dynamic.

THE SECOND DYNAMIC - is the urge toward existence as a sexual or bisexual activity. This dynamic actually has two divisions. Second Dynamic (a) is the sexual act itself and the Second Dynamic (b) is the family unit, including the rearing of children. This can be called the sex dynamic.

"THETHIRDDYNAMIC is the urge toward existence in groups of individuals. Any group or part of an entire class could be considered to be a part of the Third Dynamic. The school, the society, the town, the nation are each part of the Third Dynamic, and each one is a Third Dynamic. This can be called the group dynamic.

THEFOURTHDYNAMIC is the urge toward mankind whereas the white race would be considered a Third Dynamic. All the races would be considered the Fourth Dynamic. This can be called the mankind dynamic.

THEFIFTHDYNAMIC is the urge toward existence of the animal kingdom. This includes all living things whether vegetable or animal. The fish in the sea, the beasts of the field, or of the forest, grass, trees, flowers or anything directly and intimately motivated by life. This can be called the animal dynamic.

THESIXTHDYNAMIC is the urge toward existence as the physical universe. The physical universe is composed of matter, energy, space and time. In Scientology we take the first of each of these words and coin a word MEST. This can be called the universe dynamic.

THESEVENTHDYNAMIC is the urge toward existence as or of spirits. Anything spiritual, with or without identity, would come under the heading of the Seventh Dynamic. This can be called the spiritual dynamic.

THEEIGHTHDYNAMIC is the urge towards existence as Infinity. This is also identified as the Supreme Being. It is carefully observed here that the science of Scientology does not intrude into the Dynamic of the Supreme Being. This is called the Eighth Dynamic because the symbol of infinity stood upright makes the numeral '8'. This can be called the infinity or God dynamic." It will be noted that dianetics had the first four dynamics only. Scientology, so Hubbard writes, "embraces dynamics one through seven as known territory, scientifically demonstrated and classified". This claim is not borne out by any material placed before the Board, whether orally or as written material. Such written material as dealt with dynamics beyond the third dynamic was patently speculative. Indeed, in practice, scientology has rarely, and barely, gone as far as the third dynamic. The self dynamic and the sex dynamic have been the dominant themes, if one wishes to classify scientology processing in Victoria in terms of dynamics.

Of the eighth dynamic, Sharpe writes, "The 8th Dynamic. This with all humility I'll leave to you. God. The Supreme Being. The Creator. The Infinite. Nature. But you are certainly interested in It's survival. Nothing would be here without It".

Though Hubbard claims that "the science of scientology does not intrude into the Dynamic of the Supreme Being", this is not so, as Chapter 27, which deals with Hubbard's disparagement of religion, makes clear.

The dynamics are constantly referred to in scientology practice, preclears being processed on, and their aberrations being described as being in respect of, one or more of the dynamics.

ARC Triangle.

In Scientology: The Fundamentals of Thought Hubbard writes,

"There is a triangle of considerable importance in Scientology and understanding of it gives a much greater understanding of life, and an ability to use it.

The ARC triangle is the keystone of living associations. This triangle is the common denominator to all of life's activities. The first corner of the triangle is called Affinity The word 'affinity' is here used to mean love, liking or any other emotional attitude

The second corner of the triangle is Reality. Reality could be defined as 'that which appears to be. Reality is fundamentally agreement. What we agree to be real is real'.

The third corner of the triangle is Communication. In human relationships this is more important than the other two corners of the triangle in understanding the composition of human relations in this universe. Communication is the solvent for all things. It dissolves all things

ARC are understanding. If you would continue a strong and able communication with someone there must be some basis for agreement. There must be some liking for the person and then

communication can exist.... Communication is essentially something which is sent and which is received. The intention to send and the intention to receive must both be present in some degree before an actual communication can take place"

Original with Scientology, as are all these concepts, the ARC triangle understood is an extremely useful tool or weapon in human relationships. For instance, among the ARC triangle laws, a communication to be received must approximate the affinity level of the person to whom it is directed.

As people descend the tone scale they became more and more difficult to communication with, and things with which they will agree, become more and more solid. Thus we have friendly discourse high on the scale and war at the bottom. Where the affinity level is hate, the agreement is solid matter, and the communication . . . bullets."

ARC is stated to be "the component parts of understanding," and in scientology practice great emphasis is placed on the ARC triangle, especially on communication. Several highly placed HASI witnesses explained at great lengths the intricacies of the ARC triangle. Indeed, so detailed and protracted was their evidence on this particular phase that the Board was at first puzzled by the repeated insistence of these witnesses as to the importance of this aspect. It was explained that unless there was affinity, a readiness or willingness to receive the communication, there could be no communication; and, further, unless both the sender and receiver had a reality on what was being communicated the triangle was not functioning and there would be no communication and, of course, if there was no communication there was no triangle. It was said, further, that for the triangle to function effectively there had to be an acknowledgement of the communication by the recipient, and an acknowledgment of the acknowledgment by the sender. For how many stages such acknowledgments should go back and forth was not determined. The acknowledgment could take various forms, and witnesses explained the way in which auditors acknowledged the acknowledgement of preclears to whom they had communicated a reality. Each individual auditor, it was said, had his own pet phrases, such as "good", "right", "O.K.", "beauty", or some other similar pleasantry.

Summary of Scientology by Jack Horner confirms the teaching concerning the travelling back and forth of acknowledgments. He writes, "There are actually five lags in one-half of the cycles; between origination and receipt, between receipt and answer, between answer and its receipt by originator, between receipt of answer and acknowledgment, between acknowledgment and its receipt". In Scientology: The Fundamentals of Thought, Hubbard stipulates the necessity of repetition and of acknowledgments. He writes that auditing ability includes "an ability to place one question worded exactly the same way over and over again to the preclear no matter how many times the preclear has answered the question. It should include the ability to acknowledge with a 'good' and 'all right' every time a preclear executes or completes the execution of a command."

As the Inquiry proceeded it became evident that the scientology witnesses were proceeding on the basis that scientology was real to them but at that stage not real to the Board, and that for the Board to "cognite" on and so understand scientology, it was necessary to establish affinity with the Board and to communicate to the Board the reality which they had on scientology. And so the witnesses sought to indoctrinate the Board, and, as far as circumstances allowed, they applied techniques appropriate to ARC which, however, were merely techniques of salesmanship, whereby they sought to obtain the Board's assent to each statement they made or proposition they advanced. Witnesses even went so far as to seek the Board's actual acknowledgment of each proposition advanced. In this preoccupation to receive the Board's assent to whatever might be said lay the explanation, in part at least, for the thoroughness of their treatment of the ARC triangle.

The Board believes that the processes so practised on it did not extend beyond the elementary personal efficiency course level, probably because graduation or promotion to higher levels in scientology requires a "cognition" on all that has gone before.

When one is in disagreement with the HASI or a person, there is said to have been an "ARC break". A person who endeavours to break away from scientology finds it very difficult to do so, because he has been conditioned to believe that any hostile attitude he develops towards scientology is due to an ARC break for which he must accept responsibility and that it is in his own interests to have the break repaired. If the ARC break is with the HASI, it makes great efforts to repair the break. Frequently many hours of processing are directed towards repairing an ARC break. One very common method by which the preclear is lured back is by the letter writing technique. This technique, devised and directed by Hubbard, involves the writing of letters, sometimes dozens of them to the one person, requesting the recipient to come to the HASI and talk things over and so repair the ARC break.

If a preclear during a course of auditing has an ARC break with HASI it is said that it is because of a "by-passed charge", the auditing being faulty, as for example where the auditor has wrongly found as the "basic" in a chain of engrams an engram later in time than the real basic. In such a case Hubbard directs that the preclear be telephoned and told that there is a more basic incident or RI (reliable item) or GPM (goals-problems-mass), and the rebellious but worried preclear is induced to return for further processing.

Axioms of Scientology.

"Scientology as a science", writes Hubbard in *Scientology: The Fundamentals of Thought*, "is composed of many axioms (self-evident truths, as in geometry)". There are 57 axioms in scientology. In dianetics there are 190 axioms. There are also 7 prelogics and 24 logics. Hubbard described the first 10 axioms in scientology as the most fundamental "truths", by which he means "commonly held considerations". In propounding the axioms of scientology Hubbard makes entirely unfounded assumptions and is quite illogical. The whole list of scientology and dianetic axioms and the prelogics and logics appear in various scientology publications, including *Ability*, issue 80, and the New Zealand *Affinity* Magazine, issue No. 8: and no good purpose would be served in setting out at length in this Report the text of the axioms, prelogics and logics. It will be sufficient to refer in detail to the first 10 axioms of scientology which Hubbard regards as fundamental, and which he sets out as follows in *Scientology: The Fundamentals of Thought*:

"Axiom 1. . . Life is basically a static. (Definition: A life static has no mass, no motion, no wavelength, no location in space or in time. It has the ability to postulate and to perceive.)

Definition: In Scientology, the word 'postulate 'means to cause a thinkingness or consideration. It is a specially applied word and is defined as causative thinkingness.

- Axiom 2 . . . The static is capable of considerations, postulates, and opinions.
- Axiom 3 . . . Space, energy, objects, form and time are the result of considerations made and/or agreed upon or not by the static, and are perceived solely because the static considers that it can perceive them.
- Axiom 4 . . . Space is a viewpoint of dimensions. (Space is caused by looking out from a point. The only actuality of space is the agreed upon consideration that one perceives through something and this we call space.)

- Axiom 5. . . Energy consists of postulated particles in space. (One considers that energy exists and that he can perceive energy. One also considers that energy behaves according to certain agreed upon laws. These assumptions or considerations are the totality of energy.)
- Axiom 6 . . . Objects consist of grouped particles.
- Axiom 7 . . . Time is basically a postulate that space and particles will persist. (The rate of their persistence is what we measure with clocks and the motion of heavenly bodies.)
- Axiom 8 . . . The apparency of time is the change of position of particles in space.
- Axiom 9 . . . Change is the primary manifestation of time.

Axiom 10 . . . The highest purpose in the universe is the creation of an effect."

The Board heard by way of final address an analysis of the axioms of scientology by Mr. Phillip Wearne. He submitted that a cursory glance at the axioms may give the impression that they are aligned somewhat with early Christian gnostic faith, and the axiomatic thetan-theory may appear to be a statement in Greek symbols of spirit as created of the physical universe, so that axioms 1 and 2 describe God as Spirit and axioms 2 to 10 describe the creation of the physical universe by God and subsequent axioms state the separation of souls from God by the interposition of MEST, the human soul being then finally designated as the "thetan" in axiom 50. Mr. Wearne submitted, however, that Hubbard really means that a life static is an entity which partakes of the quality of thought and not matter, and is not one Supreme Being by which all life and all things were created but that the physical universe was created jointly and severally by a conglomeration of thetans, that is, by individual souls, each of which had the capacity of infinite and eternal intelligence as distinct from temporal and spatial existence of material substance. The concept of a thetan or thetans creating the physical universe is, as Mr. Wearne pointed out, quite the reverse of Christian belief. Mr. Wearne developed this thesis in a way which would be of interest to the philosopher or theologian minded to embark upon controversy with Hubbard. For the purpose of this Report, it is sufficient to say that Hubbard's dogmatic assertions and assumptions as expressed in the axioms, are unsupported by reasons or proof. As axioms they claim to be self evident truths, but they are neither self evident, nor are they true. One scientologist was quite enthusiastic about Hubbard's axioms of scientology which he regarded as "a gem of literary expression", and as being "the finest piece of writing in the English language", but, to use his own words, "I have found that the axioms were very confusing to me for a start and I will say that it was a couple of years before they began to make sense". So much for their self evident qualities! Gillham, co-principal of the Melbourne College of Personal Efficiency, said that after being a scientologist for many years he still did not understand the axioms.

Not only do some of the axioms contain startling and unproved assertions, but some of the terms used by Hubbard in the axioms are in effect defined in the axioms themselves, so that one goes round in a circle and finds axioms proving each other but little else and with no acceptable premises as a starting point. Hubbard further confuses himself and the reader by equating concepts to substances. For instance, in axiom 3 Hubbard states that space, energy and time, all of which are concepts, may be perceived, presumably in the same way that objects are perceived, and that all or any of them are the result of considerations made and/or agreed upon or not by the static by which presumably he means they exist, only because they are in any given case considered or agreed upon by the static to exist. In axiom 7 he defines "time" as being basically a postulate that space, which is a concept, and particles, which are substances, will persist. So too, in axiom 5, he writes that energy, which is a concept, consists of postulated particles, which are substances, in space, which is a concept.

In the axioms, Hubbard thus confuses abstract concepts with concrete objects and treats the existence of each as purely subjective to and dependent on the considerations and/or agreements of the individual static.

The Tone Scale.

In scientology, constant reference is made to the tone scale. The tone scale was first developed by Hubbard in dianetics and its purpose was to classify a variety of human emotions in a graduated order from the most dismal or lowest, which is said to be apathy, to the highest which is said to be enthusiasm. Each emotional level is given a number and the emotions are arranged in a descending order of desirability. The tone scale has been produced in various forms and representations of this tone scale have not been entirely consistent one with the other. So far as the numbering is concerned it is said that the numerical gradation of the emotions is incidental and any other set of numbers could equally well have been chosen; the significant feature is that the emotions are in a descending order of desirability. Some representations of the tone scale have been very detailed and comprehensive, one such representation or a variant of it being the "Hubbard Chart of Human Evaluations" which is a chart with 12 columns vertically and about 50 columns horizontally, and reproducing what is, in effect, a detailed table of much of the contents of Science of Survival. In this book of several hundred pages, which is "built around a chart", Hubbard has set out what he considers the significance of the tone scale and the way in which persons rise up the tone scale in response to dianetic processing, and, as he would now have it, in response to scientology processing.

For the purpose of this Report the reproduction of Hubbard's highly elaborate Chart of Human Evaluations is unnecessary. It is sufficient to produce here two variations of the tone scale, one appearing in *Scientology:* 8-8008, and the other appearing in *This is Life*.

The thetan tone scale appearing in *Scientology:* 8-8008 is as follows:

- 40.0 Serenity of beingness
- 8.0 Exhilaration
- 4.0 Enthusiasm
- 3.0 Conservatism
- 2.5 Boredom
- 2.0 Antagonism
- 1.8 Pain
- 1.5 Anger
- 1.2 No-sympathy
- 1.0 Fear
- 0.9 Sympathy
- 0.8 Propitiation
- 0.5 Grief
- 0.375 Making amends
- 0.05 Apathy
- 0.0 Being a body
- 0.2 Being other bodies
- 1.0 Punishing other bodies
- 1.3 Responsibility as blame
- 1.5 Controlling bodies
- 2.2 Protecting bodies
- 3.0 Owning bodies
- 3.5 Approval from bodies
- 4.0 Needing bodies
- 8.0 Hiding

The tone scale appearing in *This is Life* is in the following form:

4.0 Enthusiasm	
3.5 Strong Interest	
Mild Interest	
3.0 Contentment	
Indifference	
2.5 Boredom	
2.0 Expressed Hostility	
1.5 Anger	
1.1 Unexpressed Resentmer	١t
1.0 Fear	
0.5 Grief	
0.1 Apathy	

It will be observed that the tone scale as shown in Scientology: 8-8008 is rather more extensive than that shown in *This is Life*. The reason for this is that the former takes in the "thetan scale range" well below body death at '0' down to complete unbeingness as a thetan". Hubbard explains this extension of the tone scale by saying that "This sub-zero tone scale shows that the thetan is several bands below knowingness as a body and so he will be found in the majority of cases. In our homo sapiens he will be discovered to be below zero on the tone scale". This remarkable statement is consistent with Hubbard's proposition that most persons are in such a low state that it is not until they have been processed for a substantial time that they are lifted up the tone scale sufficiently to feel apathy. And Hubbard writes in Scientology: 8-8008 that the auditor must not be dismayed that the preclear, who may be otherwise acting normally, becomes irrational in the course of auditing, but should "persevere until he has the thetan up into rational scale. A raving mad thetan is far more sane than a normal human being". Hubbard used this particular argument to explain and justify what one would normally regard as the reduction of a person's condition to what is sometimes referred to by Hubbard as the "sad effect". This is a condition in which the preclear is in such a depressed state after scientology processing as to be likely to contemplate suicide, yet the preclear has been so conditioned by scientology as to believe that, in reaching this state, he is in fact progressing.

Confusion frequently arises because of the loose way in which the word "preclear" is used. The preclear is the real person, *is* the thetan, which for the time being is in association with the body which it has picked up. The thetan functions through the reactive mind, not the analytical mind which is the adjunct of the "meat body", the mechanism of the physical object which is visible to the eye. Thus the purist does not say "my thetan", but "the thetan which is I", and such refinement of language is to be kept in mind when considering the adventures of a thetan as it journeys up and down the tone scale and ranges beyond emotions experienced when in the state of body-plusthetan.

The whole purpose of scientology auditing is said to be to raise a person up the tone scale, and as one rises up the tone scale one is said to shed undesirable qualities and to assume more desirable qualities, not only mental but also physical.

Thus a person at 4.0, which is the conventional number given to enthusiasm on the tone scale, would be, writes Sharpe, "excellent at anything he undertakes", be rarely ill, have a high concept of truth, be a truly courageous man, be very much aware of the existence of any danger in a set of circumstances, have a very high sense of responsibility, be excellent as a friend and liked by many, understand and be understood very well.

A person on 2.5 (boredom), so Sharpe writes, whilst capable of some action, would be relatively inactive, bored or indifferent, be occasionally ill and susceptible to the usual diseases, be insincere and careless of facts, just could not be bothered about the truth, have neither courage nor cowardice, be too careless to be trusted with more responsibility, have only fair friendship value

and be liked only by a few, misunderstand and be often misunderstood.

In these days of personal hygiene, Hubbard's note in *Science of Survival* is of passing interest: "The body is normally sweet-smelling down to 2.0 but begins to exude chronically certain unpleasant effluvia from 2.0 down. Individuals from 2.0 down commonly have bad breath. Their feet may have a considerable odour . . . in the orient wives are commonly selected by the sweetness of their perspiration. This is apparently a very reliable test for position on the tone scale."

As to the unfortunate person who is 1.1 (the quality of unexpressed resentment and fear on Sharpe's scale-fear being shown on Hubbard's scale as 1.0), Sharpe writes that he is "only capable of minor executions," is sulky and has a chip on his shoulder, has frequent illnesses and gives way to them and goes to the doctor for the most trivial reasons, is a plausible liar, is scared of the truth and fearful of the consequences, is definitely cowardly, has an incapacity for responsibility, is in fact utterly irresponsible, is almost certainly unable to cause anything, on the score of friendship is a dangerous liability and usually despised and has no real understanding at all.

The tone scale is a great "conditioner" of preclears: it conditions them to listen to the claims by Hubbard that scientology processing effects personality improvements. The expanded tone scale really incorporates almost all human qualities and defects and is sure to contain some reference to a trait as to which an individual is self-conscious and has a secret fear of inadequacy. The friendless person finds himself classified as perhaps 1.1; he desperately wants to be loved, to be appreciated, to have friends; the scientology practitioner finds him very ready to grasp at the rewards which processing is said to achieve with certainty.

There have been other classifications of human emotions in the past and there will no doubt be others in the future; Hubbard's chart does little more than set out in an expansive way his particular classification of recognized emotions and a variety of other human qualities, real and assumed. The existence of emotions and their graduated classification is one thing. The completely unfounded claims by Hubbard that scientology auditing is capable of producing improvements in any real sense is quite another.

The Thetan.

In *Science of Survival*, Hubbard tells about a life force to which he has assigned the name theta, which is the eighth letter of the Greek alphabet, and is symbolized by 8. Theta, he writes, is "thought, life force, *elan vital*, the spirit, the soul, or any other of the numerous definitions it has had for some thousands of years". Hubbard elsewhere has described theta as life, and it may be one of his quirks that the letter, theta, in ancient Greece was synonymous with death, being the symbol by which the people marked ballots when voting for a verdict of death upon an accused person.

In scientology, the thetan, or theta being, is an individual entity. It is said that there is a thetan in possession of each human being and that the thetan is infinitely more important than the "meat body" with which it is associated in this lifetime. The meat body is frequently referred to as "mest" body, the letters M, E, S, T, standing respectively for matter, energy, space, and time.

When a preclear is audited, the thetan, according to scientology theory, is contacted and it reveals many things which are hidden from the conscious mind. The individual man, so Hubbard claims, is divisible into three parts, "The first of these is the spirit called in Scientology, the *Thetan*. The second of these parts is the *Mind*. The third of these parts is the *Body*".

The thetan is fundamental to scientology. It is described in scientology as having "no mass, no

wave-length, no energy and no time or location in space except by consideration or postulate". In *Scientology: The Fundamentals of Thought*, Hubbard wrote in 1956:

"Probably the greatest discovery of Scientology and its most forceful contribution to the knowledge of mankind has been the isolation, description and handling of the human spirit. Accomplished in July, 1951, in Phoenix, Arizona, I established along scientific rather than religious or humanitarian lines that that thing which is the person, the personality, is separable from the body and the mind at will and without causing bodily death or mental derangement."

This thing, the person, the spirit, is the theta being or thetan. Though said by Hubbard to be the spirit, the soul or something like it, it is *sui generis* and finds no parallel in any other discipline or teaching, sacred or profane. It is said to possess immortality, to have existed from all time and to be destined to exist for ever in the future.

In Scientology: 8-80, first published in 1952, and reprinted as recently as 1957, Hubbard writes:

"Thetan is the word given to the awareness of awareness unit, the life source, the personality, and the beingness of homo sapiens It is the person The thetan is a glowing unit of energy source. He seems to himself to be anything from a quarter of an inch to two inches in diameter. His capability is knowing and being. He exudes and uses energy in many forms. He can perceive and handle energy flows easily. The thetan enters sometime in early infancy. This may be before. during or following birth. He comes in a state of personal unknowingness, desiring to have an identity which he considers he has not without a body. He throws capping beams at the genetic entity, takes over the body The thetan in most preclears is within the skull. It shifts on auditing (which is to say the pc shifts) from behind to in front of the head Standing behind anybody the thetan can adjust and change any error in the body at will. He sees these as black spots. To get rid of them one has only to get the flows necessary to make and keep them white The thetan collapses into the body when the body feels pain. This was how he got trapped A thetan can get partly out of himself on a ridge. Then the preclear seems to be inside himself and yet outside The thetan is the preclear. The body-plus-thetan is no increase of personality. The body is a sort of vegetable run by the genetic entity. The thetan can clean up and heal his own body and those of others at will."

The thetan "is the 'I', it is WHO the preclear is is both outside and inside the mest body". Though the usual residence of the thetan is said to be in the skull or near, it can, so Hubbard explains in *Scientology: The Fundamentals of Thought*, be in one of four positions: entirely separate from the body or even from this universe, near the body and knowingly controlling the body, in the body, and compulsively away from the body and unable 0 to approach it.

In A History of Man (originally called What to Audit), Hubbard writes that the "very best place" for the thetan is "out of contact with the mest body entirely and fully alive as 'I'. His worst place is inside the mest body." He further writes that "his second best place is just outside the mest body" and that "the thetan in most cases, is behind and above the mest body". Sometimes, however, he may be in front of the body, and this causes a

"direction reversal on the part of the person so that he doesn't know right from left-people can teach him continually but he will still say his right is his left and his left is his right, and so it is, for 'I' in this person is the thetan and the thetan is in front, facing the mest body and right is the thetan's right, of course."

The thetan has, so Hubbard suggests, remarkable capabilities, all of which, in 1952, he hesitated to enumerate, leaving it to the preclear, who is the thetan, to find out for himself. Some of its skills and attributes he did mention. Thetans communicate by telepathy, they can move material objects

by throwing an energy flow at them. They emit a considerable electronic flow, can be rendered unconscious by wave action, can be hypnotized, can be made to sleep. They can quarrel with each other, feel pain, live on some planets.

"A theta body with its alertness restored is capable of remoulding the human body within its field, taking off weight here, restoring it there, changing appearances and even height."

Of equal importance to scientology theory and practice is the doctrine of "past lives," in which the thetan had weird experiences, the influences of which are of great significance to the thetan in its present sojourn on earth. In the early stages of the Inquiry the scientologists endeavoured to create the impression that, at the present time, the concept of past lives played very little part in scientology. However, as the Inquiry proceeded, it became evident that past lives were today as significant and important in scientology as they have always been.

In its long existence through the ages, the thetan has been on earth many times before this lifetime and has had many bodies. It has been picking up bodies and dropping bodies from time immemorial. In scientology it is generally thought that the thetan picks up a body at or about birth but it can be after birth and it can be before birth; in some cases it can even be many months before birth. In his writings Hubbard tells of auditing bringing to light incidents which occurred eight days after conception and even before conception. When one witness, a former scientologist, gave evidence before the Inquiry that in scientology some took the view that sometimes a thetan, looking for a body, would follow around a woman who looked as though she was likely shortly to become pregnant, the Board at first doubted the veracity of the witness; but this witness was later vindicated, for Lake, one of the principals of the Geelong scientology centre, gave evidence that he had experienced two well authenticated cases of this phenomenon.

It is said that a thetan, when it newly picks up a body, is in a state of unknowingness and may be confused because, though it is aware of the body it has taken over operating, it has never been told the identity of its body, and, while there are quite a few adults around, it has not been told that there are specific adults who will care for its body until it can manoeuvre the body thoroughly. So in the ceremonies of the American founding church of scientology there is a "christening" ceremony whereby the thetan is introduced to the human parents of its body. An account of Hubbard's performance of the "christening" ceremony is a travesty.

Apparently, in Hubbard's view, a thetan can take over a body already occupied by another thetan. The extent to which thetans had any choice of bodies did not clearly emerge, though one witness threatened counsel assisting the Board that she would return as his grandson's thetan thirty years hence.

The origin of thetans, whether they reproduce and how many thetans there are, witnesses could not say. Whether there would be sufficient thetans to go around when the population explosion occurs is not known. A scientologist in Hong Kong, corresponding with a highly placed staff member of the Melbourne HASI, has adverted to a gloomy prospect for those thetans which have bodies in this present lifetime. He writes,

"I do know that the population of China is 650 million, and that babies are being born at the rate of 17 million per year. Work this out for the next hundred years keeping in mind that three-quarters of the world's population at present is Chinese The chances are that next lifetime one would be a Chinese body, be indoctrinated with Communism from birth, have a minimum of education, be overworked and underfed and eventually die of malnutrition when one should be in the prime of life."

with the prospect, it seems, of picking up yet another Chinese body and repeating the same

procedure forever.

When a thetan drops off a body, it goes to an implant station, where it is implanted with various goals, the goal to forget being, if not invariable, very common. The thetan waits until it can pick up another body, for which there is sometimes competition amongst disembodied thetans, for to be without a body can be a some what uncomfortable experience for a thetan. In one of his taped lectures, Hubbard states that the minimum period spent on an implant station is sixty-nine days, though the period was earlier thought to be shorter. Hubbard claims to have visited Venus and to have inspected an implant station there. There are other implant stations throughout the universe to which thetans report. In *A History of Man* Hubbard writes,

"The report area for most has been Mars. Some women report to stations elsewhere in the Solar System. There are occasional incidents about Earth report stations. The report stations are protected by screens. The last Martian report station on Earth was established in the Pyrenees." Hubbard has worked out processes which he claims will rehabilitate the thetan by removing all aberrations. If this is achieved in this lifetime, the thetan will be able to enjoy future lives without being troubled by problems. if the aberrations are only "keyed out" the death of the present meat body will still leave the thetan subject to a recurrence of the aberrations; but, if a thetan is properly rehabilitated by scientology processing, Hubbard's teaching is that it will not have to go back to an implant station, where those who have not been properly rehabilitated are required to go and where there awaits the dreadful fate of having goals implanted by electric shocks, the passing of currents, "raw electricity" and a variety of other terrifying procedures. It is to escape such experiences as these that preclears, believing completely in Hubbard's teachings, seek desperately to be processed to clear and beyond, so as to avoid such experiences in the "between lives" periods.

Auditing makes contact with the thetan, and its past experiences are brought to light. A great amount of theory has been propounded concerning the thetan, much of which is not material to this Inquiry. It is sufficient to mention some of the more significant aspects of the thetan.

In scientology auditing, an attempt is made to locate and remove the aberration which is preventing the enjoyment of the optimum quality of being clear. The search is for the "basic-basic" which in dianetics and early scientology was the engram. However, in *Scientology: 8-80*, first published in 1951, "facsimile one" was stated to be

"the one basic engram, on top of which all this-life engrams are mere locks. It was received by the human race many, many centuries ago, and probably was a supersonic shot in the forehead, chest, and stomach, incapacitating, and reducing, the size and function of the pineal gland."

Implants.

Subsequently, Hubbard made a dramatic "breakthrough," and the basic cause of aberrations was said to be implants. Thetans, it seemed, had a variety of goals implanted in previous lives. A goal common to many thetans was found to be the goal "to forget." The unhappy lot of the thetan is that it does not remember its past, because in a past age it has been implanted with this goal to forget. This appears to have been done on the planet Helatrobus between 38 trillion and 43 trillion years ago. The hostile work of implanting in thetans the goal "to forget" and other goals was done by implanters, generally referred to as "they," who by electronic means and with great ferocity assailed the unfortunate thetans. Other implants have been inflicted on thetans, such as the gorilla implants, the aircraft door implants, the train implants, the bear implants and the glade implants. These implants were inflicted upon thetans at points earlier in time than the helatrobus implants. In HCO Bull. of the 24th July, AD 13 (1963), Hubbard set out a table of periods as "verified" by his

"research," showing the times between which the different types of implants were implanted:-

Helatrobus Implants 382 trillion years ago to 52 trillion years ago. Aircraft Door Implants 216 trillion years ago to 315 trillion years ago.

The Gorilla Goals 319 trillion years ago to 83 trillion trillion years ago.

The Bear Goals 83 trillion trillion trillion years ago to about 40.7 trillion trillion trillion years ago.

The Glade Implants (formerly called Black Thetan)

40.7 trillion trillion trillion trillion years ago to 5.9 trillion trillion trillion trillion years ago.

The Invisible Picture Goals 5.9 trillion trillion trillion trillion years to a date not fully determined.

The Minion Implants Not yet determined.

The Story of Creation

Implants 70 trillion trillion

The last mentioned, writes Hubbard, is the earliest date yet found, "but an earlier incident is known to exist."

In HCO Bull. of the 14th July, AD 13 (1963), Hubbard gives intriguing details of the manner in which the various goals were "laid in." The "aircraft door" goal "was given in the mocked up fuselage of an aircraft with the thetan fixed before an aircraft door. (There are also two or more aircraft fuselages used in the Helatrobus Implants, but the preclear moved through them, was not fixed in them)." The "gorilla goals" were

"given in an amusement park with a single tunnel, a roller coaster and a ferris wheel The symbol of a Gorilla was always present in the place the goal was given. Sometimes a large gorilla, black, was seen elsewhere than the park. A Mechanical or a live gorilla was always seen in the park. This activity was conducted by the Hoipolloi, a group of operators in meat body societies. They were typical carnival people. They let out Concessions for these Implant 'Amusement Parks.' A pink-striped white shirt with sleeve garters was the uniform of the Hoipolloi. Such a figure often rode on the roller coaster cars. Monkeys were also used on the cars. Elephants sometimes formed part of the equipment. The Hoipolloi or Gorilla goals were laid in with fantastic motion. Blasts of raw electricity and explosions were both used to lay the Items in."

The bear goals were handled by a group called " 'The Brothers of the Bear' and were the ancestors of the Hoipolloi." The black thetan goals "were given in a glade surrounded by the stone heads of 'black thetans' who spat white energy at the trapped thetan. The trapped thetan was motionless."

The experiences of thetans for countless ages in the past have the quality of the ultimate in science fiction, with wars between galaxies, machines with minds, and celestial travel between universes whose existence was not even suspected before Hubbard's time.

A History of Man was one of the earliest books on scientology; it was first published in 1952 in America with subsequent editions in 1954 and 1957. Its author is Hubbard in whom the copyright of the book resides. It is still recommended reading for scientologists, but HASI witnesses endeavoured, while not disowning the book, to create the impression that it was really not now of any significance. This attempt was belied by the evidence of many scientology witnesses which showed that they were well acquainted with and accepted many of the theories propounded in A History of Man, many of which Hubbard has repeated and endorsed in more recent writings.

To say it is an astonishing publication does not convey adequately the peculiar qualities or contents of *A History of Man*. Its style is slapdash and its contents bizarre. It is in the tradition of science fiction and for compressed nonsense and fantasy it must surpass anything theretofore written. Yet in the first sentence of the Foreword the claim is made that "This is a cold-blooded and factual account of your last sixty trillion years." The third paragraph proclaims,

"This is useful knowledge. With it the blind again see, the lame walk, the ill recover, the insane become sane and the sane become saner. By its use the thousand abilities Man has sought to recover become his once more."

This may seem laughable nonsense, but the tragedy is that so many people, reading uncritically and wanting to believe in something, have accepted the validity of these fantastic claims and the contents of the book and a great mass of similar and, if it were possible, more ludicrous nonsense. None of this nonsense which Hubbard has propounded has been repealed or repudiated. A very large part of it has been repeated in later writings and on tapes, much of it quite recently.

This book is concerned particularly with the development of the "whole track" and past life concepts. The contents of the book and the date of the first edition suggest that it was written about the time Hubbard broke with the Hubbard Dianetic Research Foundation which looked with disfavour on his theory of past lives. Hubbard claimed that by auditing the time track over the whole span of the track-at that time considered to be about sixty trillion years-he obtained excellent results, whereas in auditing only the current lifetime one could only obtain slow and mediocre results. "Further," he writes,

"it is very hard to argue with a miracle. Today, Eleanor has arthritis. She is audited 'whole track' with 1952 techniques. Tonight she doesn't have arthritis. Miracles, using 'whole track' are plentiful. By using this data an auditor can obtain a Mest clear rather easily."

Hubbard revealed then that with the invention of the Electropsychometer, commonly called the Emeter, great advances in processing were made. He greatly extended the time track, and sixty trillion years is but the yesterday of the time track which now, seemingly, extends back to the beginning of time and, if possible, even beyond that - one trillion tri

The Genetic Entity.

In *A History of Man*, Hubbard tells of the "genetic entity" or GE. A human body has both a thetan and a GE. The GE is the entity which carries forward from the earliest formation of the mest body and is located more or less in the centre of the body, the stomach. It was formerly referred to as the somatic mind in dianetics.

"The GE facsimiles include a transfer of somatics from past theta beings, for it is not common for a GE to have the same theta being twice A GE departs from the body much later than the theta beings abandon one, sees it through the death to the end and only then leaves to join the line once more some two or three days before conception."

Hubbard then develops the thesis that the GE "carries on through the evolutionary line, parallel with the protoplasmic line, generation to generation, usually on the same planet," whereas the theta being comes into the line from various quarters, with the result that a theta being lives each life with a different GE.

"The genetic entity apparently enters the protoplasm line some two days or a week prior to conception. There is some evidence that the GE is actually double, one entering on the sperm side, one entering on the ovum side

Pre-sperm recordings are quite ordinary Pre-ovum sequences are on record but are not common Conception and the impulses generated answer the conditions for one type of cancer, embryonic Cancer has been eradicated by auditing out conception and mitosis. The theta being apparently joins the track immediately prior to birth. Its sequence, for itself, is death, between-lives,

birth, all in a few minutes according to some findings, a sequence which is quite aberrative Birth is a very aberrative affair 'Facsimile one' had as many as eighty-thousand births on it as locks. Birth presents itself to be audited much of the time because Mother, complaining of how difficult the preclear's birth was, makes it an overt act. This overt act had birth as a motivator The genetic line consists of the total of incidents which have occurred during the evolution of the Mest body itself. The composite of these facsimiles has the semblance of a being," the GE. The discovery of the GE, writes Hubbard,

"makes it possible at last to vindicate the theory of evolution proposed by Darwin You as a theta being, may or may not have seen Greece or Rome. Your Mest GE has probably activated a body there, just as it has been an anthropoid in the deep forests of forgotten continents or a mollusc seeking to survive on the shore of some lost sea."

Hubbard then tells of various things the auditor will meet in processing. For instance, there is the clam. Millions of years ago the clam had many troubles, the first of which was the double hinge problem. One hinge wanted to stay open, the other wanted to close and so conflict occurred which was resolved by one hinge overcoming the other with the result that the latter suffered pain. In present time, so Hubbard claims, all you need do with certain people to restimulate the tragic defeat of the weaker hinge is to ask such a person, "Can you imagine a clam sitting on the beach, opening and closing its shell very rapidly?" at the same time making a motion with thumb and forefinger of a rapid opening and closing, whereupon the victim may grip his jaw with his hands and feel quite upset. He may, writes Hubbard, even have a few teeth pulled mistaking for toothache the restimulation of the engram suffered by the vanquished clam shell hinge.

How silly can people be? Yet, one woman, who told of benefits which scientology had given her, said in evidence that when counsel assisting the Board mentioned "clam" to her, she had a pain in her jaw.

Then there is the Weeper, originally, so Hubbard writes, called the "Grim Weeper," or the "Boohoo," a truly pathetic case. "After leaving the sea," writes Hubbard, "the GE spent a half a million years on the beach." In this state it needed food from the sea, but also air to breathe. It would open up to get food and get a wave in the shell. But as it had to breathe it had to pump sea water out, hence the name "the weeper." "The inability of a preclear to cry is partly a hang-up in the Weeper," Hubbard writes. "He is about to be hit by a wave, has his eyes full of sand or is frightened about opening his shell because he may be hit. An occluded case is sometimes merely a case of 'shell-shut' ". Hubbard tells, too, of engrams about birds and bats, of being eaten and of the sloth, the latter being responsible for fear of snakes and falling, and of the caveman and the Piltdown man, the latter being responsible so writes Hubbard, for "Obsessions about biting, efforts to hide the mouth and early familial troubles," for "the Piltdown teeth were ENORMOUS and he was quite careless as to whom and what he bit and often very surprised at the resulting damage".

The meeting of the GE and the theta body often led to great confusion because, "A theta being takes a picture of the memory of a GE and carries it as a record. A GE takes a picture of the memories of theta being and carries those." A consequence of these developments is that in auditing it is quite a problem whether it is the experiences of the theta being then being audited that come up or those of a theta being whose memories were carried by a GE, whose memory was being carried by a theta being, whose memory in turn was being carried by a GE, from which the theta being then being audited acquired it.

The GE also finds prominent and important mention in *Scientology: 8-80* and other scientology publications by Hubbard. Williams said that the GE is not now the subject of scientology teaching,

is not now known and practised, and if revived as an active study would require some modification. The GE, however, appears still to be part of the actual structure of scientology theory; it is referred to in axiom 55 of scientology by Hubbard, and has not been abandoned, repudiated or substantially modified since the first occasion when Hubbard wrote in the early fifties of the thetan "playing games" with it. Hubbard's practice is not to acknowledge that anything in his pronouncements ever requires modification or recantation. Earlier inconsistencies and contradictions are glossed over or ignored.

The Clear and The Preclear.

The concept of "clear" is one of the fundamentals of scientology. The word "clear", used as a verb, an adjective and a noun, is the mainspring of scientology; for the principal purpose of scientology auditing is to clear the "preclear" so that he attains to the state of being "clear", and is thus a "clear", shedding the prefix "pre" as he sheds the last of the aberrations from which auditing is said to free him. Everyone who is not a "complete clear" is said to be a "preclear".

On the way to clear is said to be the "release"; this is a person, according to *Scientology: 8-80*, "who has reached a point in processing where he no longer is suffering from a psychosomatic illness, or who has been freed of his chronic mental and physical difficulties and painful emotions. While he is far from being a 'clear', he is above normal, has good stability, and can enjoy life." A release is said to be a case which will not get worse.

The concept of clear was not developed in Hubbard's first work on dianetics, *Dianetics: The Original Thesis*, published in 1948, though reference was made therein to an individual who had been "cleared of engrams and chains and who has achieved a general tone four."

The first extensive references to "clear" appeared in *Dianetics: MSMH*, published in 1950, and it has been constantly referred to in dianetic and scientology publications since that time.

The "clear" is derived from the language of computers and adding machines. Hubbard used the calculating machine as an analogy to the mind. The action of "clearing" was said to be concerned with the clearing of various zones of thinking and to be the removing of some old unwanted computations from the mind, so that the mind could think and act freshly and clearly on some particular problem.

Originally, "clear" as a noun attempted to describe someone who was said to be "cleared of neurotic, psychotic and suboptimum patterns and thoughts and actions that would arise from these."

In *Dianetics: MSMH*, a "clear" was said to be "an individual who, as a result of dianetic therapy, has neither active nor potential psychosomatic illness or aberration", and "to clear" meant "to release all the physical pain and painful emotion from the life of an individual or, as in Political Dianetics, a society."

Williams accepted as an adequate current description of clear the following: "A clear can be tested for any and all psychoses, neuroses, compulsions and repress ions (all aberrations) and can be examined for any autogenic (self-generated) diseases referred to as psychosomatic aberrations." An almost identical definition appears in *Dianetics: MSMH*. Hubbard wrote in *Dianetics: MSMH* that tests on such a person "confirm the clear to be entirely without such ills or aberrations." Williams claimed that he was a clear, and he has in fact been awarded a silver bracelet by Hubbard to prove it. Hubbard has even said that Williams is well beyond clear, but attaining to such an exalted and

supposedly sickness-free level did not protect him from a common cold which he sustained during the Inquiry.

Williams said that by clearing in the mental field scientologists meant the removal of some unwanted fixed thought, idea, consideration, motivation or goal that was interfering with the individual's ability to think clearly and rationally about a particular subject. However, the prospect of attaining to clear seems depressingly remote, in spite of Hubbard's promises, for Williams pointed out that "a large mental difficulty is often composed of several difficulties, or several different subjects which have to be cleared separately." Processing is so manipulated that the final difficulty is never reached and cleared; fresh difficulties are constantly discovered as part of the "large mental difficulty", and the preclear is kept always expectant but never attaining to the goal of clear.

A clear was said to be produced by erasing some aberrative materials and keying out others. This state, when all the aberration has not been erased, is called, a "keyed out clear" or "mest clear" or just "clear". There were also "theta clears" and "cleared theta clears".

Theta clears were said to be superior to mest or keyed out or just plain clears. At one stage those beyond theta clear were described as "cleared theta clears." Theta clears are an intermediate step between mest clears and operating thetan. Williams told the Board that Hubbard's teaching was that theta clears could operate outside their bodies, and in *The Creation of Human Ability*, Hubbard defines "theta clear" as being "An individual who, as a thetan, is certain of his identity apart from that of his body, and who habitually operates the body from outside." Williams said he had witnessed Hubbard read a plaque on a wall some distance away. Williams considered that Hubbard might well be clear, but HCO has no doubt about Hubbard's state, for HCO Infm. Lr. of the 24th November, 1963, states, "He [Hubbard] was clear before 1962, and is now, at this writing, four goals down from present time toward OT, scientology's highest state. He badly needs a vacation." In *A History of Man*, Hubbard wrote that the first stable clears, produced in 1947, were in reality theta clears, not mest clears. In 1963, the terms "first goal clear", "second goal clear", &c., were used to indicate that that particular number of goals were cleared away from the person's mind.

Over the years the concept of clear has developed, and though the general nature of the concept is indicated in many writings, some of the expositions are contradictory and the boundaries between the various kinds of clears are confused. In some advertisements and elsewhere Hubbard has defined a clear as "a person at willing and knowing cause over his own life, his body and his surroundings and without a reactive or subconscious mind". A clear is credited with having a happy well-balanced personality, with efficiency and intelligence above the average and with an IQ of 135 plus.

Some of the descriptions of "clear" confuse the lower grade clear with the operating thetan, who, however, is also a type of clear. A person who has been "cleared" but is still being audited is still regarded as a preclear and is properly described as "pre-OT (pre-operating thetan)", according to *Certainty* Magazine, Vol. 5, No. 7. The suggestion that there may even be a "super clear" does not make for clarity.

Operating Thetan.

In *Com.* Mag. Vol. 5. No. 5, May, 1963, it is said that there are three grades of clears, namely, mest clear, theta clear, and operating thetan, the operating thetan being "a rather esoteric level, hard to reach and hard to describe in full."

An operating thetan, frequently referred to as "OT", is said to be a thetan with restored ability to

control or "be at cause" over matter, energy, space, time, life and form. An operating thetan has the ability to knock off hats at 50 yards, to lift telephone receivers from a distance and to read a book some distance away. The ultimate goal in scientology is to become OT. There is an award of a gold bracelet to anyone who becomes OT. Williams hopes to achieve this state and he has been undergoing processing to that end. Remarkable attributes are claimed for the state of OT. In *Certainty* Magazine, Vol. 5. No. 10, appears the statement that,

"Every operating thetan has outsight. Some of the qualities prominent are:-High ethical level; enthusiasm; accident-proof; fast reaction times; theta communication; ability to move at will throughout all universes; high creative ability; self mastery, grants beingness; high truth and courage level; strong sense of responsibility on all dynamics."

Hubbard's enthusiasm about the state of OT is equalled only by the extravagance of his boasts and predictions. In *Scientology*, issue 15G, Hubbard writes, "There is no excuse now not to be clear. A good thorough-going operating thetan should not take more than fifty hours of auditing."..

In HCO Infm. Lr. of the 24th November, 1963, it is written, "He [Hubbard] considers the research part of his task concluded in August of 1963, since all targets ever envisioned were realized as of that date and only codification and recording remained Now that the skills for OT are achieved they are found to be very precise."

Ron's Journal, No. 7, which is HCO Infm. Lr. of the 17th December AD 13 (1963), reads, "Well, here we go into AD 14, with all our technology assembled, with a complete bridge, with OT's emerging."

HCO Infm. Lr. of the 5th February, 1964, reads,

"OT course open to all auditors who have good basic training. I have kicked the doors open on Class VI. training at Saint Hill. All auditors who have good basic training are eligible for enrolment at once in running Actual GPM's. I am just completing a full training course for the Instructors and we are making OT's at Saint Mill smoothly after three years of hard and arduous research This news should tell you quite obviously that we have won all the way. And it's waiting for you at Saint Hill."

Though, as earlier indicated, Hubbard considered that 50 hours auditing could produce operating thetans he has now somewhat revised that estimate, and in HCO Bull. of the 9th July, 1963, he states that he himself is "definitely on the easy last half to OT" and that he considers "that OT lies on the sunny side of 1,000 hours of processing now for cases that can be audited"

Mrs. Williams was present in the 1964 classes at Saint Hill as one of the advanced students, but she did not become an OT, not even clear, and she did not make the acquaintance of any OT's and did not know of any OT's being produced. But that OT is a desirable state is evident, for in *Certainty* Magazine, Vol. 5, No. 10, Hubbard writes,

"the state of operating thetan is about 50 million miles higher than clear, if we put it into symbolic form. There is actually no comparison. A clear can still be invalidated; a clear can still make effect in the face of overwhelming odds. An OT, well, I don't know: for OT, read 'God' and that's just about it. But one thing is certain you cannot be OT without passing through clear first Faster routes to OT are envisaged."

Hubbard attributes his failure to produce OT's, after promising for so many years to do so, to the incapacity of auditors to audit properly. His repeated complaint is that he has devised techniques which are certain to produce clears and OTs, and the fault lies not with his techniques but with auditors who are unable to apply the techniques and operate the E-meter.

In HCO Bull. of the 9th July, 1963, Hubbard typically complains,

"It is no longer a question of whether Scientology works, it is only a question of whether the auditor can work Scientology. If he or she can't, then the trouble lies in one or more of these basics [i.e., the auditor's basic skills]. The trouble does *not* lie with the procedure or with the pc."

Hubbard is endeavouring to produce an OT meter, superior to any E-meter, which will, he claims, make possible auditing to OT.

From time to time Hubbard has made claims implying that OTs were being made or had been made. In reply to a pointed inquiry by a rebellious scientologist, Hubbard wrote that, "Our theta clears and operating thetans are for the present remaining incognito." The Board is therefore unable to report upon the identity of any person claimed as OT. It would seem, however, that if Hubbard is to be believed, an OT is "just around the corner", like the fulfilment of so many of his other extravagant promises.

Overt Motivator.

The Overt Act-Motivator in scientology is a principle or working hypothesis, not invariably applicable, that if a person does harm to an area, he will be likely to receive harm from that area.

Williams explained that the principle is that one does an overt act, something bad, harms some part of life, breaches some moral code, and thereafter, because of regret, guilt, &c., and variously associated phenomena, lays oneself open to be hurt by whatever it is against which the overt was committed. The response is called the "motivator," but opinions fluctuate as to which comes first, the overt act or the motivator. People may even be "motivator hungry," and an example was given by Williams in these words: "If you had a lawyer who kept on doing things which caused the judge to scream at him, you would have a lawyer, who in the field of jurisprudence (if that is the word) would be 'motivator hungry' and the assumption would be that he had committed an overt against the judge."

Reference is sometimes made in scientology to "victim," the victim being a person suffering from a run of bad luck or persistent misfortune of any kind, including illness, who, however, is entirely responsible for his own misfortunes, even though it be deliberately inflicted on him by another. As it is a scientology principle that man is good and can be made better, the fault is said to lie with the individual himself, the emphasis in scientology being to make the individual conscious of his complete responsibility for his own condition, whatever it is, and to find out by auditing just what it is in the preclear's own behaviour which has laid the foundation for his present aberrations and misfortune.

Goals-Problems-Mass.

Probably the latest "breakthrough" by Hubbard is the "goals-problems-mass" or GPM, which he now appears to regard as the "basic-basic" cause of all aberration. The GPM is associated with the time track, implants, past lives and OT's.

In scientology it is said that a person's problems have weight which can be measured. As early as *Dianetics:* 1955, Hubbard wrote,

"If there were no energy being created by the awareness of awareness unit, then one would be at a loss to account for mental energy pictures, for these things, being made at a tremendously rapid rate, have considerable mass in them-mass which is measurable on a thing which is as common

and everyday as a pair of bathroom scales."

In evidence, Williams defined GPM as being "the mass of stored mental energy which has accumulated following the setting of a goal." He explained how a person

"sets himself a goal, and in his attempt to succeed in that goal, he acquires the problem, and the problem is trying to achieve the goal, on the one side, and forces opposing that, on the other, and there is conflict giving us the idea of mass."

It means, he said, "a mental mass which certainly has mental weight, but whether it is discernible or not, that could be debatable." "Mentally speaking," he said, "it has thickness and other dimensions like that." There was further evidence that, according to scientology, the mind could mock up a mass - a black mass floating in front of the individual but really invisible - which he could then bring into his body and thereby increase his weight by up to thirty pounds, and, by further thinking, he could expel this same weight. Whether a person could, by thought, permanently expel weight from the body was not clearly determined, and scientology has not yet emerged as an alternative to dieting.

During the Inquiry the scientology interests were invited on several occasions to demonstrate how a mass could be mocked up and brought into the body and how the increase of the body weight could be measured on a scales, which was the claim Hubbard made, and which Williams asserted could be relatively easily done. The invitation was not accepted, although it was made clear that the Board would be greatly impressed by any such demonstration.

However, strange physical developments are attributed to processing. In *Ability* Magazine, issue major 2, is a testimonial from a young lady who writes, "Before starting on these sessions my breasts were unusually small. In fact I wore a size 32A brassiere I am now wearing a size 34c, and from all indications will wear still larger." Possibly inspired by such a report, two prominent female members of the Melbourne HASI staff, each with the same male auditor, set goals which were quite opposite. One set as her goal to have a smaller bust, and the other to have a larger bust. Whether the first attained her goal through auditing did not clearly appear in evidence. However, the second woman considered that she had attained her goal through auditing, because formerly she walked round-shouldered, but now she stands up straight and holds her shoulders back.

In HCO Bull. of the 23rd September, 1963, Hubbard tells of implant GPM's and writes that these

"have only passing importance as a pc's Actual Goals and GPM's are a thousand thousand times more aberrative and important than Implants. But one has to know the extent and nature of Implant GPM's in order not to get them confused with Actual GPM's."

In HCO Bull. of the 28th September 1963, Hubbard writes, "Confusion between Implants and Actual GPM's occurs because the implanters used types of goals and patterns found in actual GPM's. Implanters obviously had a knowledge, from historical record or even research, of what a thetan's own goals look like but obviously they never developed the data to a workable therapy or they probably would not have continued to be driven to such costly expedients as continuous implanting, between lives installations, &c.

The highest level of treatment technology known to exist in the universe before Scientology was Pictureology wherein, at a signal from the therapist, the thetan crunched up the engram. This is currently in use (and has been for many trillions of years) in the Galactic Confederation."

The latest information available to the Board is that work on GPMs is still in the developmental stage. In HCO Infm. Lr. of the 5th February, 1964, Hubbard writes,

"You can't run an Actual GPM unless you're trained to Class VI. You'd kill somebody. The new technology has not been released and will not be. It is too exacting. We can do it safely at Saint Hill".

Though Hubbard in the same information letter stated in effect that by the running of Actual GPM's "we are making OT's at Saint Hill smoothly" the scientology interests did not produce to the Board an OT or anyone who had seen an OT, and in this respect Mrs. Williams, who was a Saint Hill student during the first half of 1964, disappointed many who had hoped to see an OT when she gave evidence before the Board in September, 1964.

The foregoing summary does not exhaust the content of scientology theories but is sufficient to illustrate their general nature.

The Anderson Report CHAPTER 12 THE TEACHING OF SCIENTOLOGY

The formal teaching of scientology in Victoria is conducted at the Melbourne HASI in two separate divisions. A limited amount of instruction has been given at both the Geelong centre and the Hawthorn centre. At the HASI the elementary instruction is done in the personal efficiency department, and the more advanced teaching is done at the Academy of Scientology, which at the commencement of the Inquiry was in premises at 2 Coates Lane, Melbourne, but has since vacated those premises and is now at 157-159 Spring Street, Melbourne.

Great efforts are made to get persons "on course" at the HASI. Though individuals may initially attend the HASI only to be tested or for the free lectures, their thoughts and inclinations are skilfully channelled towards the various courses which the HASI conducts. These courses train them to be auditors, at the same time indoctrinating them in scientology theory and principles.

The HASI takes care to ensure that any advertising which is addressed generally to the public contains no reference to scientology theory or any of the advanced techniques. It is almost entirely confined to the "We can improve your condition," "We can make you more able," type of advertisement, in which the fact that the initial lectures are free is stressed and IQ and personality tests are recommended. Such advertisements appeared frequently in the daily metropolitan newspapers and in suburban newspapers, and also in some periodicals. A substantial proportion of the advertisements were inserted in newspapers by "The Melbourne Test Centre," an adjunct of the HAS1 and one of many business names used by the scientologists at 157 Spring-street, Melbourne. In 1961 the daily metropolitan press refused to accept scientology advertisements. Posters have also been exhibited in trams, and slides shown on theatre screens. Pamphlets have been distributed and are available at the HAST to any who care to take them. The Geelong and Hawthorn centres have advertised over the radio. The aspects of advertising are dealt with elsewhere in this Report (See Chapter 15). Advertisements with a scientology content are addressed to a more limited class, namely, persons already in scientology, or merely on mailing lists kept by the HASI. A name may find its way onto such a mailing list on the slightest pretext; its inclusion does not necessarily indicate any association with scientology. A regular means of disseminating scientology material and of advertising more advanced courses is the monthly magazine published by the Melbourne HASI, entitled Communication.

Personal Efficiency Courses.

The public advertising seeks to encourage people to attend the free lectures, which constitute an initiation into scientology and are part of the personal efficiency course which is run free of charge and is one of the main avenues for introducing people to scientology. Another important avenue is through recommendation by people who are already in scientology. Those attending the lectures are made to feel welcome, called by their Christian names and addressed by carefully selected instructors whose competence lies in their ability to arouse interest in scientology and procure submission to auditing and enrolment on paying courses. The PE course is considered of prime importance as, in Hubbard's words, it procures "bodies for the shop." The personal efficiency department has a director, a registrar and other staff members.

The audience generally contains a large proportion of people who are anxious and inadequate,

who feel insecure or are unable effectively to cope with their environment, and the lectures are directed towards instilling self-confidence and an ability to face their environment without fear. The audience often includes persons who are anxious to improve themselves or are desirous of undergoing some form of business training.

About 4,000 persons have taken the PE course at the Melbourne HASI, and it is said that at the end of the course only 1 per cent. have filled in report forms which were unfavourable. The Board is unimpressed by this figure, which does not deal with equivocal answers and, in any event, relates to alleged benefit received from what are probably cleverly delivered lectures with conventional but elementary psychological content and negligible scientology content. Such benefits are comparable with benefits derived from lectures given in the friendly atmosphere of a club or similar institution to people who want to be cheered up and encouraged.

During the PE course the instructor explains that the principles and practices of scientology are aimed at making a person's environment less confused, and if those attending feel they have received benefit from the course and want to benefit still further, they are informed there are courses in scientology which they can embark upon, the next course being the Hubbard Apprentice Scientologist (HAS) Course.

In the PE course, "scientology" as a word, and something more, is cleverly introduced, the E-meter is referred to, perhaps even produced for inspection, and there is talk of IQ tests and personality tests which earlier formed part of the free PE course. (These tests were formerly free, but, from August 1963, they have been part of the HAS course, and fees have been paid for tests). These IQ and personality tests figure prominently in scientology, and are often used to deceive a person as to his state of mental health and to create a feeling of inadequacy and anxiety. The evidence shows that these initial tests are deliberately used to make the individual uncertain and anxious and predispose him to undertake scientology processing or study on the promise that scientology techniques can improve some undesirable trait supposed to have been revealed. These tests are conducted by inexperienced persons, who neither understand what they are doing nor appreciate the significance of the results which they obtain.

The second course is the HAS course of two weeks duration, three nights a week, and the fee is £5. It includes lectures on listen style auditing, the students being told that listening is helpful, and they practise listen style auditing on each other. They are introduced to Sharpe's book *This is Life*. As appears in Chapter 23, "listen style auditing" can be accompanied by grave consequences to individuals in certain conditions of mental health.

The HAS course is followed by a further course, HAS Course 1, which is conducted three nights a week for four weeks, and the fee is £10. During this course the student is taught the scientology principles of affinity, reality and communication, the eight dynamics and the auditor's code, and he practises more advanced auditing techniques on his fellow students. The successful student obtains a certificate stating that he has passed the HAS Course at Class 1 level.

Academy of Scientology.

Courses beyond the HAS Course 1 are conducted in the Academy of Scientology, which also has its separate director and staff. The courses normally conducted at the Academy are the HQS (Hubbard Qualified Scientologist), HCA (Hubbard Certified Auditor) and HPA (Hubbard Professional Auditor.) As from the 2nd July, 1964, the fees for the HQS Course were £40 cash or £44 on terms, and for the HPA Course £140 cash or £160 on terms. A further course, to be known as HCS (Hubbard Clearing Scientologist), is in contemplation. This will be more advanced than the

HPA Course.

Before a student is accepted for a course at the Academy he is given a security check. Likewise, a security check is given to a preclear before he is accepted at the Hubbard Guidance Centre for processing. These security checks are designed to ensure a favourable predisposition towards Hubbard, scientology and its imagined efficiency. For such security checks the E-meter is used. (See Chapter 24 as to some types of security checks.)

At the Academy, a student is trained in various drills which Hubbard claims to have developed. The first of these is TR 0, entitled "Confronting Preclear," this drill being developed by Hubbard "to train students to confront preclears in the absence of social tricks or conversation and to overcome obsessive compulsions to be 'interesting'." Instructions for conducting this drill are as follows:

"Have student and coach sit facing each other, neither making any conversation or effort to be interesting. Have them sit and look at each other and say and do nothing for some hours. Students must not speak, fidget, giggle or be embarrassed or anaten. It will be found the student tends to confront with a body part, rather than just confront, or to use a system of confronting rather than just be there. The drill is misnamed if Confront means to do something to the pc. The whole action is to accustom an auditor to being there three feet in front of a preclear without apologising or moving or being startled or embarrassed or defending self. After a student has become able to just sit there for two hours 'bull baiting' can begin. Anything added to being there is queried by the coach with a 'What happened?' Twitches, blinks, sighs, fidgets, anything except just being there is promptly queried with the reason why, if necessary. TR 0 has been divided into four parts. Each part is drilled for about 15 minutes in turn and then begun over again and again."

"Bull baiting" is, as its name suggests, an activity designed to provoke the student to display some resentment at the manner in which the instructor is treating him. Training drill TR 0 is divided into a number of parts, and in one of them directions are given that "the coach may say anything or do anything except leave the chair. The student's 'buttons' can be found and tromped on hard." In this context, "buttons" means something about which the student is sensitive.

The second drill is TR 1, and is entitled "Dear Alice." The purpose of this drill is to "train the student to deliver a command newly and in a new unit of time to a preclear without flinching or trying to overwhelm or using a via." The procedure in this drill is to require the student to read to the coach a phrase from *Alice in Wonderland*. This book is chosen because it offers the greatest choice of nonsensical statements. The student is then required to repeat the phrase until such time as it is his own, that is, until it appears to emanate entirely from him. Part of this drill is a command by the coach to "create the space of the coaching session by locating 4 points in front of you and four points behind you."

The third drill, TR 2, entitled "Acknowledgements," is conducted with an E-meter. This drill was developed by Hubbard to teach the student that "an acknowledgement ends a communication cycle and a period of time, that a new command begins a new period of time." In this drill, the coach reads lines from *Alice in Wonderland* to the student and "the student thoroughly acknowledges them."

The fourth drill is TR 3, entitled "Duplicative question," and is designed to teach the student "to duplicate without variation an auditing question, each time newly, in its own unit of time, not as a blur with other questions, and to acknowledge it. To teach that one never asks a second question until he has received an answer to the one asked." This drill trains the student to repeat the same question for hours at a time, without any variation. These repetitive processes in auditing are one of

the many disturbing features of scientology techniques . The student practises this drill with such questions as "Do fish swim?" or "Do birds fly?"

In the fifth drill, TR 4, which is a development of TR 3, the coach endeavours to digress, and the student is required to handle the situation as he would with an obstructive preclear in an auditing session.

The foregoing information on the training drills is contained in HCO Bull. of the 29th April, 1963, entitled "Modernized Training Drills." There are other drills, numbered 6 to 9. These are repetitive drills which condition the student to audit preclears with similar repetitive processes.

These drills have produced in students who have undergone training some astonishing quantities which were later apparent in some of the witnesses who gave evidence before the Board. Scientology-trained witnesses generally "confronted" the Board and spoke directly and clearly. With an air of apparently complete sincerity and frankness they were able on occasions to talk utter rubbish, to persist in it and to seek by illogical argument to justify nonsensical statements. What they were saying on many occasions was untrue, though it may be that they believed it, and one hesitates to brand as perjurers a number of individuals who, while apparently normal in other respects, did not appear to know the meaning of truth when they came to speak on scientology. As mentioned elsewhere in this Report, their capacity for almost ecstatic sincerity was heightened in a number of cases by the fact that they had undergone processing only shortly before giving evidence.

The HQS course is either a full-time day course of a month on five days a week, or an evening course of thirteen weeks, three nights a week. In this course, which is at Class 2 level, the student learns verbatim the Auditor's Code and the Code of the Scientologist. These codes are respectively Appendices 14 and 15. He receives a general background knowledge of the theory of scientology and is able to apply the easier scientology procedures.

The Hubbard Professional Auditor (HPA) Course is an intensive course which often takes students a year or longer to complete, though a student may qualify in three months. Even when the course has been completed and the student has been successful, he does not receive his certificate entitling him to practise in his own right as an auditor until he has been security checked and is regarded as satisfactory from a security point of view. One requirement, before a preclear can advance beyond a particular level, is that he should "cognite" on everything scientological up to that stage, and a failure to satisfy the organization that one has a reality on all relevant scientology theory to that stage delays the issue of the HPA certificate.

The HPA course is designed to instruct the student in all aspects of auditing; a large part of the course is devoted to instructing the student in the operation of the E-meter.

Various processes in which the student is instructed are dealt with later in this Report. (See Chapter 13.)

Saint Hill Training.

The ambition of the dedicated scientologist is to visit Saint Hill Manor, and there do an advanced course conducted by Hubbard himself. A variety of claims are made for the efficacy of a Saint Hill course. It is said Saint Hill trained auditors are clear or nearly so, they alone can audit to OT, they alone can be entrusted with the most advanced techniques. Elsewhere in this Report are other references to Saint Hill and the activities which witnesses who had been there were able to talk

about. It seems that at least nine scientologists have gone from Victoria to Saint Hill for such advanced courses, at a cost of £358 12s. each. The cost of the course, which did not include fares and living expenses, has been borne by the student in question, or partly or wholly by the Melbourne HASI. In cases where the HAST has contributed to the cost of the course, the student has been required to enter into a bond to serve on the staff of the HASI for a specified period on his return from Saint Hill. (See Chapter 8, as to research at Saint Hill.)

These advanced courses, termed "special briefing courses", are attended by as many as ninety students at a time. They come from various parts of the world, including U.S.A., South Africa, South America, Greece, France, East Africa, Australia, New Zealand, Germany and England.

Indoctrination in Scientology.

Training as an auditor involves quite intensive study of scientology theory and a large amount of practical work in which students audit one another in co-audit sessions and undergo other individual auditing. The discipline is firm, almost harsh, with heavy "infractions" for slight transgressions. As the student aspires to be HPA (Hubbard Professional Auditor) it is impressed on him that he is now a superior type of person who is leaving behind mundane considerations which preoccupy the despised homo sapiens. Advancement and success are dependent on the complete acceptance of scientology theory. Before commencing the professional courses (i.e., above HAS) one necessary qualification is "Complete subjective and objective reality on the entire scope of the Science of Scientology." This involves accepting without question or reservation-or what is called "cogniting" upon-not only those items of scientology theory mentioned in this Report but much similar additional material.

The oppressive processes of auditing, with their repetitive and searching procedures, produce, in the case of a person who is already mentally or neurotically disturbed, at least some psychotic condition. The Board heard expert psychiatric evidence that repetitive questions and repetitive commands increase suggestibility and, if continued long enough, may reach the point where indoctrination could be effected, and a reversal of opinions and ideas previously held could be obtained. Many of the processes are hypnotic, and there is a fruitful field for post hypnotic suggestion. A preclear who "cognites" unreservedly upon scientology theory at the HPA level may well be paranoid, though apparently normal in other respects. A person suffering from a paranoid disorder can often disguise the symptoms of his disease by an impressive facade of reasonableness and normality; he may even be in an advanced state of paranoia, but, provided that he comports himself in a creditable manner and speaks lucidly and reasonably, a layman - even an informed layman - may not realize that the individual is mentally ill. A quality of imperturbability and an air of reason and normality on most matters tend to give to the nonsensical utterances of such a person a spurious quality of reason. Herein may lie an explanation for the dedication of several of the witnesses who gave evidence in support of scientology. Several spoke with the same engaging frankness and clarity about normal matters and the nonsense of scientology. Their speech and attitude were equally as open and direct when they talked about thetans, engrams, theta bops on the E-meter, time tracks and past lives, as when they gave their names and their addresses appropriate to this lifetime. One began to understand how staff members, confronting the curious enquirer and lecturing to those attending the free lectures, could impress the gullible and the anxious and the mentally ill with an admixture of common sense and nonsense.

The effect of prolonged processing and training is that the critical faculties and common sense of the individual are destroyed, so that he comes to believe that Hubbard is right, scientology is right and everything else is wrong, and that the greatest thing that he can do is to work for Hubbard and thereby assist in saving the world.

Brainwashing.

During the Inquiry reference was made from time to time to brainwashing. This expression usually connotes techniques applied to an unwilling subject. Where it is so practised on a subject in normal health the objective of the operator may be achieved, but generally only temporarily, for once the processing is halted and the subject is returned to normal conditions and surroundings, reason tends to reassert itself, and the subject usually tends to resume his former way of thinking. In scientology, however, the auditor has a very willing victim, who submits to procedures which are in the main hypnotic and have been devised in a calculated way to achieve domination over the mind of the preclear. These procedures constitute a kind of brainwashing, though their pattern may not precisely be found in other techniques which are conventionally termed brainwashing. The Board is not concerned to find that the scientology techniques are brainwashing techniques as practised, so it is understood, in some communist-controlled countries. Scientology techniques are, nevertheless, a kind of brainwashing.

Expert psychiatric evidence was to the effect that in the early stages of brainwashing, a psychotic state, very much like schizophrenia, can be achieved. A person who "cognites" unreservedly upon scientology theory at the HPA course level has been scientologically brainwashed and might well be schizophrenic.

The astonishing feature of scientology is that its techniques and propagation resemble very closely those set out in a book entitled *Brain-washing*, advertised and sold by the HASI. This book purports to be "A synthesis of the Communist Textbook on Psychopolitics," "Psychopolitics" being defined as "the art and science of asserting and maintaining dominion over the thoughts and loyalties of individuals, officers, bureaus and masses, and effecting the conquest of enemy nations through 'mental healing'." (One edition of *Brainwashing* is described as "A synthesis of the Russian Textbook on Psychopolitics".)

The Board heard astonishing evidence about the *Brain-washing* Manual, as this book was usually called. Though purporting to be a synthesis of the communist text-book, with an introduction which purports to be the text of a lecture said to have been delivered by "Beria", to American students at the Lenin University, the English version of the manual bears a startling resemblance to Hubbard's own literary style. Whether he is the author, as was suggested by a witness hostile to scientology, is probably immaterial. What is of some significance is that his organization assiduously sold and distributed this manual. The Board heard evidence to the effect that Hubbard or his American organization, desiring to draw the attention of the "authorities" to the contents of the manual and to expose the craftiness of the Russian psychopoliticians, posted from America to the Melbourne HASI an envelope containing a copy of the manual, and a similar envelope but with no copy of the manual in it. When the two envelopes arrived at their destination, the Melbourne HASI then complained to the "authorities" that the contents of one envelope were missing, the suggestion being that the manual had been removed en route by communists, and the other envelope containing the manual was produced to the "authorities," so that they could see the nature of the material involved, and in this way the manual was brought to the notice of the "authorities."

It was a fanciful story, but it was consistent with Hubbard's policies of deceit and may very well be true. Certainly, a great part of the manual is almost a blue print for the propagation of scientology. One remarkable exhibit tendered to the Board was a series of extracts from the *Brain-washing Manual*, with, however, the substitution of a number of words with a scientology connotation for certain words with a communist connotation. With these substitutions effected, the extracts were in the main startlingly applicable to scientology as operating in Victoria. This exhibit, with the substitutions made, is Appendix 16.

The Anderson Report CHAPTER 13 SCIENTOLOGY AUDITING PROCESSES

Over the past fifteen years Hubbard has developed a great variety of auditing processes. These have been tried for a time and then put aside when Hubbard has claimed some new dramatic ultimate breakthrough and has enthusiastically announced techniques considered appropriate to his new ideas, But seldom, if ever, have any of his processes been declared to be wrong or defective. Quite the reverse. Mrs. Williams was unable to recall any process which had been abandoned, though some may have been put into cold storage. In HCO Bull. of the 15th May AD 13 (1963), Hubbard wrote in relation to engram running that "all lectures and books on it are true", though subsequent HCO bulletins took precedence over earlier material. In HCO Bull. of the 9th July, 1963, Hubbard wrote, "Today we assume that every successful process we ever had is and was a valid process." And as late as the 26th November, 1963, in a policy letter of that date, he wrote, "It is evident that 13 years of research developed many processes and styles of auditing and that these are all useful and necessary to the successful progress of cases." Thus there are many processes developed by Hubbard which may be drawn upon according as his whim or that of the director of processing or an auditor directs. Both in relation to Hubbard's theories and his techniques, scientology witnesses experienced embarrassment. Notwithstanding that during 1963 Hubbard wrote a number of bulletins which in sheer fantasy at least equal or even surpass the contents of A History of Man, Williams and other witnesses sought to explain away the earlier fantastic theories by saying that, though not jettisoned, they did not play an active part in current scientology practice. This is not the case. The most fanciful nonsense about helatrobus implants and gorilla goals and much more besides dominated Hubbard's effusions of 1963 and made it a vintage year for absurdity.

The scientology witnesses also said that many of the techniques appearing in Hubbard's writings were not now used, at least, not in Victoria; but it was admitted that they could be reintroduced at any time. The evidence of scientology witnesses as to practices and processes used at the ETASI was generally unsatisfactory in that they endeavoured to play down the nature and extent of practices in use, the intensity with which they were practised and the apparently disturbing effects they had on preclears.

The HASI sought to present scientology as a benign regimen, broadcasting health, happiness and sanity in the most pleasant of ways, kindly, helpful, simple, easy, noble. It was none of these. The evidence the Board heard and the files it examined give the lie to such claims. The attempt to play down scientology became more apparent as the Inquiry proceeded, efforts being made to disguise or modify the various features and effects of processes, which were shown by expert psychiatric evidence to be detrimental to mental health. The Board was told that various procedures were not now used, that others were not as long or as intense as had earlier appeared, that exteriorization was not now an objective, that past lives were not of importance, even that the demonstration sessions which had been conducted at the beginning of the Inquiry were not now as fully representative as they had been because the emphasis was now more on "overt withholds" than on "straight wire ARC". The Board was not deceived by these attempts to present a "watered down" picture of scientology. Just around the corner are more of Hubbard's "breakthroughs", and more techniques and theories. The bulletins and other written material of 1963 and 1964 abound in details of engram running, goals-problems-mass and a multitude of other past life matters. The Board voiced the view during the Inquiry, as it does now, that once this Inquiry is finished there will

be a resurgence of all the pernicious activity which marked the progress of scientology up to the appointment of this Board, if only it can find sufficient victims to exploit.

Exteriorization.

Many of Hubbard's processes, especially those developed in the 1950's, were deliberately designed to produce "exteriorization", a state or condition known to psychiatrists as "dissociation", which is produced by hypnotic means. Dissociation is an hallucinatory experience in which the subject believes that he is outside his body. Hubbard claimed that the thetan did in fact leave the body during "exteriorization" and go to some place outside it in obedience to the command of "Be three feet back of your head" and similar commands. One of Hubbard's books, *The Creation of Human Ability*, published in 1955, is largely devoted to techniques designed to produce exteriorization. In this book the command "Be three feet back of your head" is then followed by other commands to the preclear to be in various places alternatively pleasant and unpleasant, and "then in more and more dangerous places until he can sit in the centre of the Sun." These techniques are essentially hypnotic and repetitive.

Routine R1-13 directs that the preclear should "duplicate his body many, many times and push the mock-up into it." The auditor is then directed,

"Have him interiorize and exteriorize many times from his present body. Then have him mock-up the electronic structure of his body until he can see it easily. Now ask him 'Do you see any gold balls in your head?' If he does not, have him mock-up gold balls until his head's golden balls appear. Now have him move any golden balls he finds which are out of position in his head until they are in proper position."

And so it goes on. These are dangerous hypnotic procedures in unskilled hands. One of the routines, given the number and name "R2-17: Opening procedure by duplication until preclear feels good about it", involves the selection of two dissimilar objects such as a book and an ash tray or a book and a bottle. These are placed several feet apart. The auditor then commands the preclear, "Go over to that book", "Look at it", "Pick it up", "What is its colour?", "What is its weight?", "Put it down in exactly the same place". The same commands are given in respect of the ash tray or the bottle. Hubbard then writes, "The same word and the same formula are used over and over again until the preclear has had a sufficient number of hours of Opening Procedure by Duplication to enable him to do it without communication lag, without protest, without apathy, but only cheerfulness, each time seeing the items newly. This is a process which is done by the hour. The process is better when done consecutively for many hours rather than done an hour apiece each day for several days."

This process is in fact dangerous to the mental health of the preclear, who may be reduced to a state of insensible automation or may become deranged under the persistent commands of the auditor who is controlling him by hypnotic procedures. This process has been used in the Melbourne HASI until relatively recent times. One witness who for a time had been director of processing said that he had been run on this process for nine and a half hours without any breaks at all.

The Creation of Human Ability, a book of over 250 pages, is full of details of similar techniques for producing exteriorization. No good purpose would be served in detailing the other techniques, for all are condemned for the same reason.

Demonstration Sessions.

Certain processes admittedly in current use were demonstrated for the Board in specially arranged sessions. These demonstrations were presented for the information of the Board at the HGC premises in Coates Lane, Melbourne. The Board witnessed the sessions on a closed television circuit; in each of the demonstration sessions Tampion was the auditor. The preclears were selected by lot from a small group of scientologists who had volunteered for the purpose. The demonstrations, seven in number, were intended to be a cross section or sample of the processes then being currently used at the HASI. They were described as being processes from level 1 to level 4. The demonstrations were not full length auditing sessions, for the normal session is about 21/2 hours in length and sessions are often conducted twice daily for several days, whereas most of the demonstrations took only from half an hour to an hour. Some were even shorter. Furthermore, the demonstrations suffered from the handicap that the preclears concerned were aware that they were being observed on television, and, while both the auditor and the preclears no doubt endeavoured to simulate an orthodox session, they may have been inhibited to some extent by the consciousness that they were on show.

These demonstrations, which were given at an early stage of the Inquiry, were probably conducted along efficient scientology lines as far as circumstances permitted. However, the scientologists themselves conceded that certain limitations should be placed upon their value and adequacy.

One important aspect of all the demonstrations was the strict routine followed by the auditor when he and the preclear went into session and when they came out of session.

Listen Style Auditing.

The first demonstration was of a process entitled "Listen Style auditing". This is a Class I process and engaged in at quite an elementary stage in scientology experience. It involved the auditor asking the preclear, "Tell me about the right decisions you have made", whereupon the preclear began to talk about her affairs without pause or interruption until after about half an hour the auditor stopped her as the allotted time had run out. At the beginning of the session the preclear had stated her goals as being that the session would be fantastic, that she should gets lots of wins, feel more positive about things, feel more relaxed, and feel more positive that she could handle things, situations at home "so that everybody is happy". At the end of the session she had her goals read out to her and enthusiastically said that she had had gains in them all. Nine days later she was admitted to hospital as a mental patient. Her tragic case is referred to in Chapter 23, where the fact that this apparently simple process can have dangerous consequences is discussed.

A.R.C. Straight Wire.

The second process demonstrated was called "ARC Straight wire". The auditor commanded the preclear repeatedly to "Recall an emotion", "Recall something real", "Recall a communication". The preclear complied with these commands by giving details of his experiences as a sailor, as a soldier and in other roles. The session was of nearly an hour's duration, but a normal session would be much longer. In this demonstration an E-meter was used, though it was said that this is often not the case. By this process the preclear was supposed to have gains of increased affinity, reality and communication, to feel happier, have more agreement with people and communicate more easily and effectively. This, said Williams, is brought about by the preclear's re-evaluation of past experiences and changed attitudes towards people in the environment as well as the "release of emotional charge".

As will be mentioned later in this Report, the repetitive commands which are designed to probe the preclear's past may have adverse effects on the mental health of particular individuals. In the demonstration session, the preclear, conscious of the audience, recalled innocuous emotions. There is clear evidence in the many processing files examined that, in the privacy of an ordinary session, emotionally disturbed preclears readily confess to their auditors matters of an intimate and shameful nature.

CCH 's.

The third demonstration session, a common form of process in which the preclear does not necessarily speak, was of the CCH's, meaning "Control-Communication-Havingness". The instructions for training students in the CCH's are contained in HCO Bull. of the 5th July, 1963. There appear to be four CCH's. Students are trained in these techniques by coaches. Thereafter, as auditors, they use them on preclears.

According to the bulletin, in CCH I and CCH 3 the student and the coach sit facing each other, the student's knees on the outside of both of the coach's knees.

The four CCH's were demonstrated to the Board. In CCH I the auditor asked the preclear for her hands alternately, repeating the command a number of times. Each time the preclear tendered a hand. In the demonstration session, the command was given three times for each hand, but in session and in training this process, either alone or in conjunction with others, can go on for hours.

In CCH 2, the auditor commanded the preclear "You look at that wall", "You walk over to that wall", "You touch that wall", "Turn around". The preclear did all these things, the auditor saying "Thank you" after each command was carried out and also asking the preclear some questions. The series of commands was repeated only three times, but in full session they may be repeated many times.

The third, CCH 3, is called "Hand Space Mimicry". In training, the student and coach sit facing each other. In the demonstration session the auditor asked the preclear to place her hands against his and to follow them and contribute to their motion. The auditor moved his hands around and the preclear's hands followed. This was repeated six times, but in full session may be repeated many times over a much longer period. Hubbard's instructions in relation to CCH 3 are that after the exercise as described above is done in training the command is then given, "Put your hands facing mine about a half an inch away, follow them and contribute to their motion." This exercise is then continued until it is "flat", that is, satisfactory, and then the "student does it with a wider space and so on until coach is able to follow motion a yard away."

In CCH 4, called "Book Mimicry", the auditor in the demonstration took up a book and made several movements with the book which the preclear imitated. The series of movements was repeated only a few times in the demonstration session but in full session the process is continued for much longer periods, until the process is "flat".

The purpose of these processes is said to be to develop in the preclear control of his body, to develop his reality on things and to improve his communication.

These processes are frequently very protracted in full session. When continued for a substantial time, the emotions of preclears may be taxed beyond endurance, so that they become hysterical, and symptoms of mental imbalance develop or are intensified. They frequently lead to discussion between the preclear and the auditor of the preclear's most intimate sexual secrets and behaviour,

and at times to the preclear developing sexual urges towards the auditor, if of the opposite sex. These happenings are explained by the fact that these processes when practised for hours at a time almost invariably produce a state of hypnosis.

ARC straight wire and the CCH's are Class 2 processes, and proficiency in these is necessary to qualify for the HQS certificate.

Slow Assessment by Dynamics, Rudiments, Havingness.

Demonstration 4 was of a process entitled "R2C - slow assessment by dynamics plus rudiments, questions and havingness." This was a composite session of a number of processes. The auditor asked the preclear, a woman, "Since you were last audited has anything been suppressed?" She had been audited the previous evening and found some difficulty in finding anything which had been suppressed since then, but as the E-meter showed a reaction each time the question was asked, she struggled to find an instance. She was then asked, "Since the last time you were audited is there anything you have been careful of?" to which she gave an oblique and rambling answer about care in taking her children to school that morning.

The circumstance that her last auditing session had ended at about ten o'clock the previous night probably inhibited this demonstration. The preclear was herself an experienced scientologist, who was evidently aware of the unseen audience in the room outside. In a full session, the questioning would continue for much longer and an already disturbed preclear would tend to see in everything which he had done a possible suppression or occasion to take care. This particular technique tends to mobilize guilt in the preclear, and, if continued for any appreciable length of time, could have harmful effects in particular cases. The auditor in this case asked and repeated only two questions. "Since the last time you were audited has anything been suppressed?" and "Since the last time you were audited is there anything you have been careful of?" However, in some processing when "an 18 Prep check" is embarked upon, the "Prep check buttons" are used, and the question, "Has anything been ?", or a similarly phrased question, is applied not only in respect of "suppressed" and "careful of", but also "failed to reveal", "invalidated", "suggested", "mistake been made", "protested", "anxious about", "decided", "withdrawn from", "reached", "ignored", "a failure", "helped", "hidden", "revealed", "asserted", "solved". In certain cases, Hubbard directs that the question must be preceded by "In this lifetime ", otherwise the preclear will become "over-restimulated".

Hours and even days could be spent on this phase of this process, for the questions are repeated until the E-meter does not "read". Though the question put in the demonstration was "Since you were last audited ", the "time limiter" may be some other point in time, thereby greatly widening the area which the preclear has to traverse in searching for all that he has suppressed, been careful of, failed to reveal, been anxious about, withdrawn from, or as the case may be. Sometimes, in running rudiments the preclear is checked on the E-meter with such questions as "Are you willing to talk to me about your difficulties?" Expert evidence was that this is a question designed to worry the preclear because it presupposes that he is experiencing difficulties and it may stir up feelings of guilt and anxiety.

Returning to the fourth demonstration session, the auditor then proceeded with the routine of slow assessment by dynamics. In this process, the preclear is given the choice of the dynamic to be investigated. This preclear chose the first, or "self", dynamic. The auditor asked the preclear her name and then asked her to tell him, "What consideration have you had about your name?" The preclear then talked at length about thoughts she had on her name and other names by which she had been called. Most of this seemed innocuous, but the preclear had not liked her first Christian

name and her maiden surname and the E-meter gave various reads. The conversation with the auditor drifted to various employments the preclear had had, and other similar matters. Each aspect was probed by questions till there was no reaction on the meter. This part of the session resembled in some ways the "listen style" process, but in this instance the auditor was positively in control and was directing and questioning and prompting the preclear.

This demonstration took only 35 minutes and bore little resemblance to a full session of perhaps 21/2 hours (which in turn would normally be only one of several sessions in a course of processing). The particular process is well named "Slow assessment by dynamics", for, in respect of whatever dynamic the preclear may choose, there have been listed a great variety of topics to be discussed, and, if conducted in the manner and at the speed indicated in the demonstration session, would take months and even years to complete.

The final process demonstrated on this preclear in this session was "havingness", the preclear being asked to squeeze the cans of the E-meter and then to touch various objects such as the table, the top of her suit, her skirt, the table again. and her chair. She was then asked to squeeze the cans again, and was considered to have had some gain because the E-meter reading was then "up about half a dial".

In this session the auditor followed the strict route by ending the body of the session, but continuing for a short time discussing with the preclear the goals she had attained in the session, after which the "havingness" process was embarked upon. This was evidently a session in which the preclear in question had been subjected to a hypnotic process, and the "havingness" process was characteristic of hypnotic techniques wherein the patient is slowly brought back from the past to the present. Again, following strict routine, the auditor then loudly announced, "End of Session".

In HCO Technical Form of the 17th October AD 13 (1963), are set out "the breakdown of the first four dynamics", being the items or areas that may be dealt with when the preclear is being processed on the first four dynamics. HCO Bull. of the 31st October AD 13 (1963), sets out "breakdown of Dynamics Five through Eight". It is unnecessary to set out these "breakdowns" in full detail, but some of the particular matters dealt with are significant.

In relation to the first, or the self, dynamic, such things as the following are dealt with in detail: names, positions, pc's identity (what pc is mainly being, what pc would rather be. what pc would rather not be, other identities pc has been and is being, &c.), education, interests obsessive activity (things pc feels compelled to do, must prevent himself from doing, fears, bad habits, &c.), pc's doingness, body (genetic line, body defects, illness, initial care, relationship to body, ARC with body, birth, death, &c.), location, time sense, ownership (personal effects, property, business interests, stocks and bonds, handling and control over the dynamics, handling and control of mest, &c.), pc's havingness, unusual mental treatment (mental condition, mental defects, medical-psychiatric treatment, squirrel auditing, &c.), scientology processing.

In relation to the second or sex dynamic, the following matters (some of which are dealt with in considerable detail) are listed: parents, parents' family, own family, sexual relations (both marital and extra-marital), other sexual activity (types, homosexuality, sex with animals, fetishes, unusual sex, masturbation), procuration (contraception, sex for pleasure, abortion).

The third or group dynamic has listed, with many subdivisions, friends, enemies, groups, government.

The fourth or mankind dynamic deals with such areas as international relations, extra-terrestrial

relations (intelligent life on other planets, Markab Federation, Galactic Federation, space travel, flying saucers), mass communication, homo sapiens (survival of the species, over-population, under-population, the New Man), scientology five (world dissemination, world clearing, mental healing methods, healing, operating thetans.).

Beyond the fourth dynamic there is almost limitless scope for auditing. The fifth or animal dynamic deals with all aspects of plants, animals, living things, nature, nature study, organic material, genetic line, interrelations of life forms; the sixth or universe dynamic, with possessions, matter, energy, the mind, areas of livingness, schedules, measurement and description, manufacture, property, transportation, communication media, national forces, geographical areas, the universe, other universes; the seventh or spiritual dynamic, with spirits, theta, entheta, affinity, reality, communication, spiritualism, aesthetics, ethics, philosophy, art, creativeness, magic, scientology; the eighth or God dynamic, with the Supreme Being, religious entities, religion, religions, church, mysticism, anti-religion, after death, creation, religious philosophy, religious practices.

A variety of subheadings is indicated in respect of dynamics five to eight as being appropriate topics or areas on which to audit the preclear should he select any of these dynamics. However, the dynamic most likely to be selected would appear to be the first or the second, and with either of these there is almost unlimited scope for disclosures which are of a personal, secret, shameful nature, and which, when recorded on the preclear's file virtually ensure the domination of the HASI over the preclear.

Routine 2H Assessment.

The fifth process demonstrated was described as "Routine 2H Assessment". The purpose of this routine was "to put a person into better communication with other persons in his life by finding moments in his life when he went out of communication with certain persons", or to repair an ARC break. The preclear in this session was a young man in his early twenties who stated as his goal for the session, "To clean up an ARC break I had with a girlfriend". The auditor proceeded then to ask him seven times in succession the question "Since the last time you were audited has anything been suppressed?" The preclear, who had been audited about eight days previously, answered variously to this question until the auditor pronounced the meter to be "clear". He then asked the preclear to "Recall an ARC break", which the preclear did, and the auditor questioned him until the details of this incident were exhausted. He was again asked to "Recall an ARC break", which he did and he was also questioned on this at length.

This session finished in the same way as the previous one, with squeezing of cans and "havingness" routines, whereby the preclear was slowly brought back to present time.

This session lasted only 35 minutes, and the ARC breaks were not dramatic and were only lightly touched upon. In a secret session of 21/2 hours, a worried person could be mentally tortured by the probing of the auditor. *Problems Intensive*. The sixth demonstration was of a process named "Problems Intensive". Hubbard regards the "problems intensive" as a highly effective technique. In HCO Bull. of the 27th September AD 12 (1962), he writes,

"The only fully valid lower level process today that achieves enormously effective results, is the Problems Intensive. It does the following: Eradicates feelings of illness; Adds years to life; Subtracts years from appearance; Increases I.Q. People are suddenly losing all manner of things they thought were illnesses and were calling arthritis and ulcers and what not. They weren't sick. They were just suppressed."

His instructions are, "Find on the meter the most important most reactive change in the pc's life, and then locate the prior confusion to that change, which confusion may have occurred from 2 weeks to 5 minutes before the actual decision to change."

The process involves the auditor asking the preclear, "What changes have you decided to make in your life?" In this process, the preclear's answers are written down; these are checked with the preclear on the F-meter, and the particular decision which reads most on the meter is selected as the topic for auditing. The auditor then asks the preclear a series of questions called "prep-check" questions (see demonstration 4, ante) such as "suppressed", "careful of", &c., and he phrases the question, "Since [a month before the decision to change] has anything been suppressed?" The theory behind this process is that some decisions to change are made as a solution to some problem and confusion that have existed in the preclear's mind. The supposed benefits of this process are said to be that the preclear will have a feeling of increased well-being and will find the resolution of a problem which he has had since the period of confusion and the decision to change.

The preclear in this session was asked many times the precise question, "What other changes have you decided to make in your life?", and he gave a variety of answers, together with appropriate dates. After eight decisions were detailed, the auditor called a halt because of limitation as to time, and then checked the eight decisions on the E-meter, and got a" blow-down" on the decision "To build up my body a bit" at a particular date.

The auditor then asked the preclear many times, "Since [the date] has anything been suppressed?" and the preclear was still telling of things suppressed when the hour allotted to this session elapsed.

In an ordinary session the seeking of decisions goes on until the preclear exhausts his recollection of decisions and things suppressed. As the usual course for a "Problems intensive" is 25 hours as a minimum, but may be much longer, it requires little imagination to picture the mental exhaustion of a preclear after five hours questioning each day for five days in succession. When a preclear has finished telling of all the things suppressed, he may be asked to tell all the things of which he has been careful, and then of all the things he has "failed to reveal", made mistakes about, been "anxious about", "ignored", failed at, &c., as listed in demonstration 4, ante. In this way hundreds of hours of auditing can be, and are, aggregated.

R3SC - A Service Facsimile.

The seventh demonstration session was of a process called "R3SC-a service facsimile." In scientology, a "service facsimile" is "an aberrated, non-survival solution the preclear uses to make others wrong, self right, to aid the survival of self, hinder the survival of others, help self dominate others and help escape domination." There are various means, so it was said, by which an auditor can locate a service facsimile. In this particular process, the procedure is for the auditor to ask the preclear, "What method have you used to make others wrong during your life?" The auditor writes down the answers and then with the aid of the E-meter gets a "blow-down" on one of the answers which is then selected as the item to be audited to determine the facsimile.

The benefits claimed for this type of auditing are "a considerable rise towards optimum behaviour level of the person and a cessation of the use of that method used to make others wrong as well as a general increase in well-being, confidence, happiness and ARC."

The preclear in this demonstration session was a woman who had been audited the previous

week, and she was asked by the auditor the customary preliminary question, "Since the last time you were audited has anything been suppressed?" A "read" was obtained on the meter, and in explanation the preclear said, "Well, all auditors' questions always read at first. 1 am protesting so much and I sort of hate them, you know, they make me feel guilty."

The auditor then asked repeatedly, "What methods have you used during your life to make others wrong?" Then she was asked such questions as, "Is there anything you have done to make anybody else guilty during your life?" "Have you done anything during your life to make yourself right?"

It emerged from this questioning that the preclear considered that she had been critical in making others wrong and herself right, whereupon the auditor embarked upon a series of questions such as, "Will you tell me how being critical will make you right?" "How would being critical make others wrong?" "How would being critical aid your survival?" "How would being critical hinder the survival of others?" "How would being critical help you to escape domination?" "How would being critical help you to dominate others?" and many other questions to the same effect, in relation to being critical.

Finally, the auditor asked "Concerning being critical has anything been suppressed?" He immediately announced there was nothing on the meter, the preclear said she felt "Okay", and the auditor decided then to end the session earlier than intended because the service facsimile had been found and run. Tampion admitted that this session might have been better had some other person been the preclear, as the preclear in question had had "this sort of thing handled quite a bit in past auditing."

Characteristics of an Auditing Session.

In the light of later evidence which the Board heard from expert psychiatrists, several features of these demonstration sessions are of particular importance and significance. The scientologists considered that these sessions were fair representative samples of the various processes demonstrated, having regard to the limitations imposed by the time and circumstances, and Hubbard actually bestowed upon Tampion a doctorate of scientology for the part he played in conducting the sessions. From what has been set out above and from other aspects of these demonstrations the following features emerge:

- 1. Sessions are always commenced with a strict routine. The preclear is asked if it is all right to audit in the place where they are, and is it all right to start the session? The auditor then utters a loud "start of session" and the session is then under way.
- 2. The auditing questions are repeated in identical or almost identical terms, and are continued until the meter does not read.
- 3. Goals are stated and recorded at the beginning of the session, and at the end of the session the preclear is asked what gains he has made. Sometimes the preclear was vague about the goals set, but when reminded of them enthusiastically stated there had been gains.
- 4. The questions as to whether goals have been obtained are asked in such a way as to induce favourable answers.
- 5. The auditor assumes a pose of being obligated to the preclear for his (the preclear's) help in making the session a success. The auditor thanks the preclear for whatever gains are said to have been made and expresses regret that he (the auditor) has, in effect, failed in cases where a gain has not been obtained, and the preclear is made to feel that he must help the auditor make it a good session. There is evident in the preclear a desire to please the auditor.

- 6. The preclear searches deeply into his memory for incidents which may be made to fit the subject matter of the question, and frequently matters with only the most tenuous connexion with the matter being probed, and even with no connexion, are mentioned.
- 7. A suggestion of guilt is implied in each question, and the answer is generally in the nature of a guilty admission, an abasement, a confession. Alternatively, there is an attempt to justify what the preclear is admitting.
- 8. Notwithstanding the innocuous nature of most of the areas dealt with, the truncated nature of the session and the presence of an audience watching, the preclears were speaking in what, on normal standards of conversation, would have been an uninhibited level.
- 9. Each session is brought to an end by a strict routine. There is a slowing down of the tempo of the session, and the auditor ends the body of the session which, however, still continues while the auditor inquires as to gains made and conducts processes designed to bring the preclear from the introspective past up to the present. The session is finally ended by a sharp "end of session", and in the last two more advanced demonstrations the auditor commanded, "Tell me I am no longer auditing you", to which the preclear in each instance replied, "You are no longer auditing me."
- 10. At least two of the preclears had symptoms which would have alerted a psychiatrist to the likelihood that the preclear was suffering from mental or physical illness. In one case the process being run, short though it was, was positively harmful to the preclear's condition. As to the other case, the Board subsequently heard expert psychiatric evidence that auras the preclear said he was seeing could be part of an organic illness like epilepsy and that auras may be one of a series of premonitory symptoms which precede an epileptic fit. Neither the auditor, Tampion, nor Williams was apparently aware of the possible medical significance of the phenomenon which this preclear described. Tampion, a man still in his twenties, a former school teacher who gave up his University course in Commerce in about 1961 to become a full-time professional scientologist, considers that his knowledge of the human mind is superior to that of psychiatrists, and that rather than recommend psychotic persons to a psychiatrist for treatment he would advise "that they have a good holiday and get away from the things that upset them."

The foregoing features assume great significance when considered in conjunction with evidence which the Board later heard from expert psychiatrists and psychologists as to what was the nature of psycho-analysis and hypnosis. Almost every feature of an auditing session has its parallel in command or authoritative hypnotic procedures. Grave consequences flow from the indiscriminate use of such hypnotic procedures. Chapter 18 deals with the hypnotic aspects of scientology processing.

S.C.S.

One process in respect of which the Board heard somewhat contradictory evidence was called "S.C.S.", or "Start-Change-Stop." Williams in his evidence rather suggested that this process was a simple process, the purpose of which was to increase a person's familiarity with something, and he illustrated this by the example of the inefficient typist whose ability was increased by her being made more familiar with her typewriter by being told to touch various parts of it, and to start it moving by pushing it from one place to another. Williams made the claim that the British Pentathlon Team in the 1956 Olympic Games improved its skill in pistol shooting by being given such exercises which made the team more familiar with its weapons.

However, other evidence was given as to the real nature of the S.C.S. process, which shows certain of its features to be quite different and of a harmful nature. The S.C.S., like the CCH processes, is a control process, and though "control" is said to connote control by the preclear over his body, the actual effect is to place the auditor in a position of complete domination over the preclear and, in

scientology, S.C.S. is classified as "8-C" and represents "a very strong control to make sure that a person does what he is required to do."

It was said at the Inquiry that S.C.S. was not now practised at the HASI. However, on a chart entitled "Processes, Levels and Theoretical Material as currently used and as expected to be used in the future", prepared for the Inquiry and tendered in evidence by Williams, processes described as "S.C.S. on object" and "S.C.S. General" were shown as being used or expected to be used respectively at levels 1 and 2 which were levels at which the HASI was then functioning. Files examined by the Board make it clear that well into 1963 wholesale use was being made of the more dangerous S.C.S. process.

The Board heard evidence that the S.C.S. process involves the auditor instructing the preclear to start his body moving, change the position of his body from one place to another, and stop his body moving, upon the command of the auditor to "start", "change" and "stop", respectively. It was said in evidence that S.C.S. was sometimes used as a punishment for recalcitrant preclears, the witness in guestion stating that he had been "sentenced" to 25 hours of S.C.S. for some infraction.

That S.C.S. is not merely the simple process that Williams suggested is apparent from Hubbard's own writings. In Scientology: Clear Procedure, issue one, Hubbard directs that the command in S.C.S. is to be, "Now I want you to get your body moving in that direction and when I say Stop, I want you to stop the body." Almost precisely the same commands are set out by Hubbard in Control and the Mechanics of S. C.S. The significance of the S.C.S. commands to start "your body" moving, &c., is that the thetan is considered to be distinct from its body, and it is the thetan which is compelling or requiring the body to move. This concept is drilled into the preclear during hours of this processing, and he is thereby reduced to a condition in which he is ready to believe that the thetan does just that.

In Scientology: Clear Procedure, issue one, Hubbard writes,

"These S.C.S. processes produced the greatest control changes that have been produced with any control process An apparent drop in havingness is occasionally experienced by the preclear as he does these processes. This is because of compulsive exteriorization. If a preclear is about to fly out of his head he'll fly out of his head on S.C.S. If he does fly out of his head on S.C.S., or any other process, you, of course, continue the process In running S.C.S., however, the preclear flying in and out of his head will experience various changes of havingness which are quite upsetting . Don't be surprised in running S.C.S. if the preclear suddenly flies to pieces, goes into flip-flopping, has to be picked up off the floor and put over on the couch and left aghast, but do be very surprised at yourself if you fail to get the preclear back up on his feet and into session again at once."

In the same publication Hubbard explains "flip-flopping" in this way:

"Flip-flopping was a process by which the preclear's excess motion was taken off It is extremely interesting that a preclear exteriorizing from his own body which is out of control, flip-flopping, writhing, convulsing and going into epileptiform seizures was at a distance from a flip-flopping body. One day while in his own body he causes some other body to go out of control, he shoots somebody or hits somebody, and has this person go into a flip-flop."

In Control and the Mechanics of S.C.S. Hubbard states in relation to S.C.S., "The end result of this process is exteriorization."

The S.C.S. process is a hypnotic process which is dangerous to mental health and produces

distressing and alarming symptoms in preclears, who may exteriorize while being processed, and who develop almost the qualities of an automaton.

A former scientologist volunteered to demonstrate S.C.S. The manner in which he demonstrated was criticized by the scientologists, who, however, though invited to demonstrate S.C.S., declined to avail themselves of the opportunity.

R2-12 (Listing).

A further process which was not demonstrated to the Board was one called "R2-12, Listing". A substantial part of scientology theory involves thinking in opposites. In Scientology 8-8008, Hubbard refers to "The Dichotomies", amongst which he lists "Survive, succumb", "Affinity, no affinity", "Start, stop", "Be, be not", "Win, lose", "Right, wrong", "Sane, insane", and several others. In subsequent teaching he developed terminals and opposite or opposition terminals, or "terms" and "opterms" for short. The idea of opposition developed into the R2-12 process which involves asking the preclear the questions, "Who or what would oppose?" and "Who or what would oppose?" The object being opposed or opposing is initially determined by reference to a long list of items enumerated by Hubbard, generally with a scientology connotation, such as "scientology", "auditors", "minds", "mental health", "hats", mental doctors", "psychiatrists", &c. In this process the preclear is on the E-meter, and the auditor asks and repeats the questions as necessary, writing down the preclear's answers and noting the reaction on the meter. The asking of one of these questions may take many hours, spread over several days; hundreds, and sometimes thousands, of answers may be obtained to the repeated asking of each question. One preclear on 23 almost consecutive days gave over 9,000 answers to a series of questions which included "Who or what would somebody with a goal not to be the centre of attraction oppose?", "Who or what would the most reasonable being (a being of integrity), (a servile being), (certainty), (birth), (the planet) oppose?".

After several hundred answers to a particular question have been obtained, they are checked on the meter in search of a "reliable item", which is one of the answers which reads the best (or worst) on the meter. The reliable item is then used as the subject or the object of the questions in following sessions, when the "Who or what" questions proceed for several more days.

This process has been very common at the HASI. It was said in evidence by HASI witnesses that it was not now being used, but it was in constant use well into 1963. It has highly undesirable features, and one can well understand the anxiety of the HASI neither to demonstrate it nor to admit that it was in current use.

This process produces in preclears a condition from which all moral restraints are absent. In many of the files examined by the Board this process was used, the preclear's answers frequently relating to very disgusting matters of sex, depravity and perversion. The initial subject of the "Who or what" question is a simple word, but in the course of finding 600 to a 1,000 answers to such a question as "Who or what would scientology (or an E-meter) (or psychiatrists) oppose?". the preclear will often give a sexual or depraved answer which then suggests the next answer which will be more disgusting and thereafter the floodgates of filth and depravity are open. Preclears whose questioning has started off with some simple subject finish up giving the most debased answers, striving obsessively to outdo in vulgarity and depravity the previous answers by the addition of vulgar adjectives.

When the preclear is checked for a reliable item, one of the depraved answers is often selected, and the preclear may be questioned for days on "Who or what" would oppose that particular

reliable item. An example, by no means isolated, is a question asked for days of a young married woman, namely, "Who or what would a sexy little piece oppose?". Many of her answers show an abandonment of any restraint.

Mere declamation does not sufficiently describe this shockingly immoral feature of scientology processing; included in Appendix 19 is a small selection of the questions asked and some of the answers given in R2-12 processing.

The Anderson Report CHAPTER 14 THE E-METER

The only apparatus used in scientology is a small electric meter, housed in a wooden box measuring about 10 in. by 6 in. by 2 in. On its top side is a dial with a moving needle, some control knobs and a rheostat control termed the "tone arm." It has two terminals, to each of which is attached, at the end of a lead, an electrode which is a steel or tin can, resembling, and sometimes actually being, a soup can. Within the box are small batteries and a transistorized circuit. The circuit is simple and is appropriate for apparatus designed to record or register electrical resistance, It is a variant of the Wheatstone bridge.

The meter is called an "E-meter", which is an abbreviation of the "electropsychometer" or "electrometer". The E-meter has been improved and developed over the past fifteen years, and there have been five models, ranging from Mark I to Mark V. There is some suggestion that shortly there is to be a Mark VI model, and even another model which is to be used for auditing to OT.

When used in scientology auditing, the cans are held by the preclear, one in each hand, and the auditor, sitting opposite, faces the meter and records and interprets the readings on the dial, making various adjustments with the tone arm as the general level of resistance recorded by the meter rises or falls. The auditor writes down the various readings of the meter which are said to be correlated to whatever is being discussed by the auditor and preclear. The detail of such discussions is also written down.

It is claimed that emotions and emotionally charged thought can register electrically on the E-meter. When a preclear is "in session" and is holding the cans, questions which stir his emotions are said to produce a "read" on the meter. Hubbard and scientologists generally regard the E-meter as a most important adjunct to auditing, an almost indispensible tool.

In a book entitled *The Hubbard Electrometer*, said to be based on research and development by Hubbard, written by one John Sanborn and copyrighted by Hubbard in 1959, the operation of the Emeter in scientology is explained. It is there stated that

"The E-meter works on a very easily understood principle. It measures the relative density of the body. The relative density is changed as the facsimiles change. The E-meter then measures shift in thought. It registers in particular shifts in thought relating closely to the questions asked by the E-meter operator. The operator asks, the facsimiles shift under his asking. The E-meter measures the shift and thus the mind is read."

A facsimile is defined in scientology as a recording in energy of an incident or part of an incident from the past. It is "picture" in motion. When the picture comes into play again, it produces motion," writes Hubbard.

"What we are reading, or measuring", writes Sanborn, "is specifically the impingement of the individual himself (the spirit) upon the body by the direct action of thought. It is to be emphasized that this does not mean the brain We mean that thought is direct and that thought is produced only by the spirit, and that in the aberrated individual (one who lives by mental image pictures seen or unseen, reactively) it just happens that whatever thought he thinks, he accompanies by the automatic production of mental image pictures Mental image pictures impinge upon the body. Thetans are quite obsessive about this."

It is then explained that the ohms resistance in the body always registers in one of several ways.

"Thus we could say simply that the E-meter measures what and how much the spirit is doing to the body. It is

therefore a precision aid to clearing."

In his handbook *E-meter Essentials*, Hubbard writes, "the meter tells you what the preclear's mind is doing when the preclear is made to think of something", and that the "current is influenced by the mental masses, pictures, circuits and machinery. When the unclear pc thinks of something these mental items shift and this registers on the meter."

The thetan is said to produce a change in resistance this way: the auditor asks a question which produces a facsimile or a mental image picture, which is said to contain pain or heavy emotion. The "heavier" the emotion the "heavier" and "more dense" is the facsimile. The facsimile or mental image picture impinges on the body, the relative density of which is changed as the facsimile changes, and such impingement produces changes in the resistance of the body. The idea of energy and thought having physical weight and physical density is quite foreign to science, but finds ready acceptance in scientology. Williams spoke at some length, purely speculatively, about the possibility of the thetan having an electrical field somewhere around or in association with the body and he said that the effect of emotional activity flowing from the thetan through its electrical field and through the nervous system could produce a change in resistance.

Williams, likewise without any evidence to support it, toyed with the idea that the thetan produced a change in voltage in some way whereby the E-meter recorded not only a change in resistance but also a change in voltage. His attitude was consistently scientological, but quite illogical, and was to the effect that if and when a sufficiently sensitive meter was developed it might record a change in voltage and, if it did, the change was due to the activity of the thetan; and therefore, since a thetan produced a change in voltage which was recorded on the E-meter, the theories of scientology as to the thetan were thereby confirmed. It is standard technique with scientologists not only in relation to the E-meter, but generally, to make unfounded or illogical claims and then to challenge those who criticize such claims to prove them wrong. In effect, their attitude is, "We say a thetan causes a change in resistance, prove us wrong."

A great air of mystery and reverence surrounds the E-meter. It is accorded almost magical powers. It is said to be infallible. Of it Hubbard writes, "The E-meter is never wrong. It sees all; it knows all. It tells everything."

In fact, it is no more than a powerful gimmick for controlling preclears and developing in them a sense of awe and of submission to and dependency on the HASI. It is merely an electrical measuring instrument, with no other capacity than to detect and record the degree of resistance to the flow of electricity of any medium placed between its terminals to complete the circuit.

It is unnecessary, of course, to concoct the fantastic theories of scientology to explain the functioning of the E-meter. The principles involved and the reasons for the changes in resistance recorded by the E-meter are well known to science. This was made clear by evidence the Board heard from a number of expert witnesses, including one who was Reader in Biophysics at the University of Melbourne, and had the degrees of Doctor of Philosophy, Master of Science and Bachelor of Arts; and another who was a research student, a Bachelor of Science, who was completing his Master of Science degree in Physiology. These witnesses examined and tested two E-meters, one of which was a Mark IV model supplied by the HASI.

The E-meter is not a new type of instrument. It is one which is well known to science and has been in use in one form or another for many years. As early as the 1920's, experiments were conducted in psychological research with what was then called an electro-galvanometer or psychogalvanometer. Such meters had and have specific but limited use in psychology, but their use in no

way relates to such fanciful things as electronic fields of the thetan and the measuring of the weight and mass of thought. The E-meter, like one form of "lie detector", is merely a development or modification of these earlier instruments. The current models of the E-meter are well constructed, accurate, sensitive - almost too sensitive - instruments capable of doing what their specifications claim for them, namely, indicate "variations in the electrical resistance of the human body." There is no reason at all to suppose that they can do anything more then register resistance. The scientology rejoinder that the thetan caused the change in resistance in the manner described above was not supported by any evidence. On the contrary, the Board heard abundant expert evidence which explained the phenomena which produced the variations in resistance.

Expert evidence which the Board heard showed that once the outer layer of skin of the human body was penetrated the resistance of the body between two specific points is low and uniform. However, should the human body be introduced into the circuit by means of the hands holding electrodes, a condition occurs which may well produce apparent changes in the conductivity of the body, for the skin and associated physiological factors have to be taken into account, and the resistance of the whole of the body may vary because of variations in conductivity between the electrodes and the hands. These variations are due to such factors as the dryness or dampness of the hand, the amount of salt present in any moisture on the hand and minute sweat secretions. Furthermore, blood circulating in the tissue layers immediately beneath the skin may also cause changes in resistance. An important cause of variation in conductivity is due to involuntary muscle movement resulting in a change in the contact between the hand and the can. Once the hand is held in a posture, such as that assumed when the can is grasped, there occur minute changes in the grip, thereby altering the degree of contact with the can and producing a change in conductivity. Variations in the dampness of the hand by sweat, the changes in blood circulating in the skin tissues and the muscle variations altering the grip may be caused by emotional stress; they are generally involuntary though attempts may be made to control them (especially the muscles of the hand) and they may be minute and unknown to the subject. However, the more sensitive the instrument the more likely it is that it will detect slight changes in resistance caused by these factors.

Though Williams said in evidence that hand contact had nothing to do with variation in resistance, the Board is satisfied that hand contact is an important factor in the operation of the E-meter, and it witnessed an experiment by an expert witness, a psychiatrist, who demonstrated that by the application of a contact cream on the hands greatly increased conductivity was obtained. In the early days of scientology the meter was used with cans as the electrodes but as time went on other types of electrodes were introduced. At first Hubbard considered that these were equally effective but eventually he discovered that only cans were reliable and effective and he banned the use of other devices as electrodes. This emphasizes the significance of the good contact given by the wide surface of the cans which lessens the resistance.

Scientologists, without any proof that thought causes a facsimile which alters the density of the body or the electrical field of the thetan or the voltage passing through the E-meter, proceed on the assumption that it is the thetan which causes the E-meter to register.

Hubbard claims in *E-meter Essentials* and elsewhere that what the E-meter records is the reaction of the reactive mind, that is, the thetan, which quite independently of any answer which the conscious preclear may give, responds accordingly as it knows the answer to be by reference to the incidents recorded on its time track. The conscious mind, so it is said, may be quite unaware of the effect of the question upon the reactive mind and a negative answer by the preclear may be accompanied by a read on the meter which is equivalent to a positive answer.

The E-meter is said to provide conclusive evidence of the existence of thetans and past lives and the validity of scientology theories generally. Scientologists contend that if a preclear is asked such a question as, say, "Have you ever committed adultery", and there is a "read" on the meter then, though the preclear may answer, "No", adultery of which the thetan is aware has nevertheless been committed. Then, if the same question is asked, but preceded by the words, "In this lifetime...", and there is no "read" on the meter, the scientology explanation is that, though there may have been no adultery in the lifetime of the thetan's present mest body, the thetan was aware of adultery in the lifetime of some other body to which it had earlier been assigned, and it had irresponsibly caused the meter to read in answer to the first question because it had been imprecise. The reactive mind is quite unanalytical, writes Hubbard, and it does not reason. "A equals A equals A equals A" is how the reactive mind works, and adultery to the thetan is adultery whenever it occurred, and it answers the question asked irrespective of the experiences of the mest body which it occupies for the time being.

Amongst the various uses to which the E-meter has been put in scientology is the determination of the dates of incidents in the distant past on the time track of the preclear. This is done by time spotting, the auditor and the preclear co-operating to ascertain by the reaction of the meter when the incident in question happened. Frequently, while on the E-meter, preclears have described events trillions of years ago, even though the preclear in some cases did not know whether it was the English trillion or the American trillion to which he was referring, Hubbard, however, cleared this matter up by indicating that he meant the American trillion when he used the term in his writings. It was said that the E-meter is capable of great precision when it comes to fixing a date in the past. Scientology witnesses gave evidence and files showed that dates had been fixed down to the second, such precise fixings as 43,851,924,801 years, 94 days, 12 hours, 1 minute and 3 seconds, being quite common.

Hubbard, however, had set the fashion for such precision, as indicated in the Heaven Bulletin, which is dealt with at length in Chapter 27. In this bulletin of the 11th May, 1963, Hubbard tells of one implant he received "43,891,832,611,177 years, 344 days, 10 hours, 20 minutes and 40 seconds from 10:02.30 P.M. Daylight Greenwich Time May 9, 1963", and of another implant which he received "42,681,459,477,315 years, 132 days, 18 hours, 20 minutes and 15 seconds from 11:02 P.M. Daylight Greenwich Time May 9, 1963."

Such is the reverence in which the E-meter is held that it is almost axiomatic that when the E-meter reads it is recording something. Hubbard, however, is constrained to make a few modifications of such superlative claims. In HCO Bull. of the 30th November, 1961, he makes the concession that "An E-meter has a frailty I have just discovered. It operates only if the auditor has some, even small, command value over the pc, and operates hardly at all when the auditor has no command value over the pc". This is a remarkable, though possibly unintended, admission that in order to achieve results in auditing the auditor has to be in a dominating position; in other words, at least some hypnotic or other similar control over the preclear is necessary.

Hubbard repeatedly berates auditors for failure to operate the meter efficiently. He even claims that, with his perfected techniques and good auditing with an E-meter, preclears can be audited to clear and even to OT. The trouble, he complains, is that auditors are incompetent. He maintains that it is impossible to audit to clear without the E-meter. It tells an auditor immediately, says Hubbard, what eventually he would otherwise find out only by long and arduous processing.

In 1962 Hubbard was becoming concerned about the embarrassing interest which the United States Food and Drug Administration was taking in the E-meter and claims made for it. In Pol. Ltr. of the 29th October, 1962 he endeavoured to change the role which the E-meter played in

scientology, and he wrote that "regardless of any earlier uses of psychogalvanometers in Dianetics or Psychology or in early Scientology publications when research was in progress, the Electrometer in Scientology today has no other use than" to "disclose truth to the individual who is being processed and thus free him spiritually", and that "the Electrometer is a valid religious instrument, used in Confessionals, and is in no way diagnostic and does not treat." Such a change of attitude, by Hubbard was evidently dictated by the circumstance that the Food and Drug Administration was making investigations and did subsequently institute proceedings in respect of the E-meter on the basis that unfounded and illegal claims to treat illnesses were being made for it. These proceedings, which began towards the end of 1963, were still pending when the Board finished hearing evidence. Whether or not the use that is claimed for the E-meter in the United States constitutes an offence against the laws of that country is immaterial. Whether or not its use is in breach of any existing law, the use to which the E-meter is put in scientology is dastardly.

None of the scientology theories associated with, or claims made for, the E-meter is justified. They are contrary to expert evidence which the Board heard and are quite fantastic and inherently improbable. Nothing even remotely resembling credible evidence was placed before the Board in attempted justification.

The E-meter is an instrument which efficiently registers electrical resistance and nothing more. But, when used in scientology auditing, its impact is alarming. Its assumed infallibility and the theories founded on it are not questioned in any way by scientologists. The preclear is introduced to the meter at an early stage; it is often used in the personal efficiency stage; and, throughout his whole association with the HASI, it is kept prominently in the preclear's mind. The fear or the threat of being put on the E-meter because of some ARC break with the organization deters the incipient rebel.

The E-meter enables the HASI to assume, intensify and retain control over the minds and wills of preclears. Fears of its abilities keep them in constant subjection. Its use can be so manipulated by cunningly phrased questions that almost any desired result can be obtained, and it is used unscrupulously to dominate students and staff alike. All the evil features of scientology are intensified where the E-meter is involved. When used in conjunction with hypnotic techniques, its evil impact is greatly increased.

This simple electrical device is not, of course, the sole basis for the condemnation of scientology, but without the E-meter scientology would be partly disarmed.

The Anderson Report CHAPTER 15 PROCUREMENT AND DISSEMINATION

In a remarkably skilful way, Hubbard has developed methods whereby people are lured into scientology and, once ensnared, are kept in subjection. So diabolical are the methods he has devised for procuring "bodies in the shop" that, but for the fact that the procedures are fully documented by his own written instructions, one would be pardoned for doubting that anyone could be capable of planning such villainy.

The particular techniques dealt with in this Chapter are those methods of deception designed to entice people into scientology by creating and exploiting anxieties and fears which constrain them to embark upon scientology processing to cure their real or imagined ills.

In the middle 1950's Hubbard had already developed means whereby unsuspecting and unfortunate people were being exploited by deceitful practices. In PAB 73, what Hubbard calls three plans or methods of dissemination are set out and of them he writes, " Out of these three plans . . . can come large and vital practices". These three methods are:

- 1. "I will talk to anyone."
- 2. Illness researches.
- 3. Casualty contact.

It was said in evidence that the third method was not tried in Victoria, but that the first two were tried with varying success. The significance of these instructions is not the degree of success which attended their use or the extent to which they were used but the fact that they were methods promulgated by Hubbard and were designed to deceive and ensnare; and they serve as illustrations of Hubbard's remarkable propensity for calculated deception.

The first method, "I will talk to anyone," required the insertion of advertisements in newspapers stating, "Personal counselling - I will talk to anyone for you about anything. Phone Rev. so-and-so between hour and hour". When the victim telephoned, the "minister" would not discuss the caller's problem on the phone since the purpose was to get him into a weekly processing unit, from which opportunities would stem for individual auditing. The whole plan was to "get people into scientology", which, however, was not mentioned in the advertisement lest it attract press attention.

The second method, "Illness Researches", was of the most callous kind. This method involved the insertion in the press of an advertisement to the following effect: "Polio Victims. A research foundation, investigating polio, desires volunteers suffering from the after effects of that illness to call for examination at " When people arrived they were immediately given about three hours processing. This type of conduct Hubbard seeks to justify by writing, "It was given under the guise of investigation and was in actuality a research project". He gave instructions in these terms: "Any auditor can constitute himself as a minister or an auditor, a research worker in the field of any illness. In that he is not offering to treat or cure the illness but is strictly investigating it, the laws concerning medicine do not obtain to him. Anybody, even a ditch-digger, can look over polio or arthritis or asthma or anything else. It is best that a minister representing himself as a 'charitable

organization' which is what he is, do the research so that the ad. would then read: 'Polio victims - a charitable organization investigating polio desires to examine several victims of the after effects of this illness. Phone so and so.' " One of the side plans of this method, writes Hubbard, "was to have another person good at finance go around to all those who had been helped by the investigation and tell them that their investigation which helped them was paid for by another person, and ask the present person whether or not he wouldn't like to pay for somebody else's recovery, but this was never put into effect, although it may be workable".

The third method is the inspiration of a ghoul, and is based on exploiting grief. Of this technique Hubbard writes,

"One takes every daily paper he can get his hands on and cuts from it every story whereby he might have a preclear. He either has the address in the story itself or he gets the address as a minister from the newspaper. As speedily as possible he makes a call on the bereaved or injured person He should represent himself to the person or the person's family as a minister whose compassion was compelled by the newspaper story concerning the person. He should then enter the presence of the person and give a nominal assist, leave his card which states exactly where church services are held every Sunday and with the statement that a much fuller recovery is possible by coming to these free services takes his departure. A great many miracles will follow in his wake and he is later to become a subject of the press himself. However, in handling the press we should simply say that it is a mission of the Church to assist those who are in need of assistance". Hubbard's instructions for liaising with the press are set out in Appendix 17.

In the early days of scientology in Victoria the impersonation of ministers of religion was practised at least by John Farrell and Frank Turnbull, the latter assuming the title of "bishop", though of what he was a bishop did not appear. On occasions other scientologists wore collars reversed. This was commonly practised by Farrell, who it was said by a witness "used to tell us if we had a collar reversed we could walk into places where we may not otherwise be admitted, and if any questions were asked about what we were doing we would raise a finger and say, 'Would you interfere with the work of a man of God?' And this would shatter anyone and they would open doors and let us through".

Early scientology advertisements in Melbourne, about 1955, included press advertisements which stated in effect, "Asthma sufferers wanted. We may be able to help you".

In more recent times, press advertisements have been somewhat guarded, being directed mainly, if not entirely, towards publicity for the personal efficiency course and matters incidental thereto. Such advertisements generally do not deal with any of the more advanced aspects of scientology, the impression sought to be given by such advertisements being that the HASI and other scientology organizations are engaged in only elementary psychological practices.

Samples of advertisements published in the press were tendered in evidence. From 1957 onwards care seems to have been taken in press notices to disguise the real nature of scientology, its techniques and its practices. The text of such notices varied from time to time, but the pattern of deception dictated by Hubbard in the PAB's already mentioned and in subsequent directives, some of which are set out later, is discernible. Typical of such advertisements is one appearing in the "Professional" column of one of the daily newspapers which was repeated from time to time with variations:

PEOPLE NEEDED FOR RESEARCH PURPOSES IN PERSONAL EFFICIENCY

You give us your time for six evenings. We train you and observe progress. You give testimonial on

completion of course. No financial obligation Hubbard Foundation of Personal Efficiency, 157 Spring Street, C. 1. (Incorp. in the HASI)"

This type of advertisement developed in content and size during 1958 and panel advertisements measuring about 3 in. by 3 in. regularly appeared in the morning daily press. The advertisements still stated, "People needed [or wanted] for research purposes in personal efficiency", and told how personal relations and concentration would be improved. These advertisements sometimes stated, "Before and after tests are given. Your only obligation is a frank report outlining how you find the course. Reports are being collected on similar courses in London, Washington, Auckland, Brisbane, Sydney, Perth, Johannesburg, &c., and are needed for research purposes". Sometimes the advertisement read, "Your only obligations are punctual attendance for the six classes and a frank report outlining how you find the course".

Advertisements stressed the apparent magnanimity of the HASI: "The course is free", "Time is your only obligation", "No obligation", "No financial obligation".

Sometimes the advertisement would propose the undertaking of a course of study with HASI and state that a text book, *Problems of Work* or *Fundamentals of Thought*, would be available. Neither of these books gives any indication of the high-powered pernicious practices and dangerous nonsense that awaited the preclear who embarked upon a course in scientology Hubbard shows a peculiar skill in disguising the real nature of scientology in these introductory books of which he is the author.

Some of the advertisements, inserted by the "Scientology Centre", boldly headed "The Clear", read,

"Definition: A clear is a person at willing and knowing cause over his own life, his body and his surroundings and without a reactive or subconscious mind (L. Ron Hubbard C.E., Ph.D., Author of *Dianetics: The Modern Science of Mental Health*, Scientology Fundamentals of Thought and other books). Testing shows this person to have a happy, well-balanced personality; efficiency and intelligence above average (IQ 135 plus). Your first step towards Clear: Enrol on Monday, 7.15 p.m. sharp, for the Free Persona1 Efficiency Course.

"Purpose: To make a better worker of the worker, a better executive of the executive, and a better homo sapiens on all dynamics. Course Runs for Five Consecutive Evening Classes." Some of the advertisements which offered professional training in scientology stated that the goal of such training was "validated cleared auditor (you will be cleared as part of the course)". Sometimes the word "scientology" was prominently displayed, as in the advertisement inserted by the "Scientology Centre", an alternative name for the HASI at 157 Spring Street, Melbourne, which read:

"The primary aim of SCIENTOLOGY is to make able people more able.

"A clear analytical mind and a stable understanding of life are THE obvious necessities for success, notification and stability in any and every trade and profession and ALL life activity.' Free Personal Efficiency Course."

The unsuspecting reader of these and similar advertisements had no idea what a "clear" was, in spite of the definition which was sometimes included in the advertisement, but was attracted by the promises of greater efficiency. He early learned that a clear was some vaguely defined but desirable state which, when reached, ensured solution of all one's problems. Its attainment, though offered during the course, was never achieved and Hubbard and his followers, in spite of

claims that clears have been and are being regularly produced, and in spite of silver bracelets and certificates issued to those who are said to have attained the state of clear, have done no more than to certify those who have become completely obsessed by the dangerous theories and practices of scientology.

An appeal was made to the ambitious, but dissatisfied, in advertisements which invited them to

"A free adult evening course to attain:

Ability to earn more

Ability to lead

Ability to organize

Ability to converse

Ability to concentrate

Ability to understand

Ability to be interested

Ability to accept responsibility

Ability to make friends and keep them."

In the advertisements, such as the following which appeared in the Victorian Public Service Journal on behalf of the Melbourne Test Centre of 159 Spring Street, Melbourne, there was the appeal to the student-coupled with the hint that the free offer might shortly be withdrawn:

"IQ Tested

The Melbourne Test Centre offers for a limited time FREE Intelligence and Personality Tests. Do you know what is stopping you. . . from Achieving good exam results. Achieving long-sought-after promotion. Overcoming confusion in your work. Handling others around you. These and many other things which are stopping you can be shown and explained on the evaluation of these modern IQ (intelligence quotient) and Personality Tests. Your IQ and Personality determine your future. Know them - no obligation."

The Academy of Personal Efficiency, also of 159 Spring Street, Melbourne, made an appeal in its advertisements to "Housewives and mothers", exhorting them to -

"Learn More:

To make you happier and less tired in the Home. A series of lectures has been especially arranged for Housewives and Mothers by the Academy of Personal Efficiency.

You'll learn:

How to save on your Family Budget.

How to relax at home.

Dealing with the Children.

How to have more Friends.

Understanding your Husband.

Cost of whole course is £1."

The irony of this advertisement is that savings from the family budget, if they were effected, were channelled to the scientology organization to pay for processing. A number of housewives did give evidence on behalf of scientology, stating that they had derived various benefits from it. Their evidence was of very doubtful weight for reasons mentioned elsewhere in this Report; however, none of them had limited their expenditure to the One pound which was the "Cost of the whole

course," almost all of them had spent hundreds of pounds on processing, and several were still receiving processing at the time they gave their evidence.

These and similar advertisements appeared in the daily metropolitan press, in suburban newspapers and in the journals of various private associations. The daily metropolitan press and at least some suburban newspapers declined after October, 1961, to receive scientology advertisements, though an occasional advertisement slipped through; in November, 1963, suburban newspapers carried an advertisement for the Scientology Centre which read,

"SCIENTOLOGY

The common people's science of Life and Betterment.

A free introductory evening course is available.

Over 20,000 Australians have taken this course."

As late as February, 1965, a suburban newspaper carried an advertisement for the Melbourne College of Personal Efficiency at Camberwell Junction. These and similar advertisements contained in dodgers, pamphlets and circulars were directed to be used by Hubbard in HCO Pol. Lr. of the 2nd December, 1960, in which he explained how to promote a PE Foundation by the use of "Physical Ailment type" advertisements, such as "Sick all the time? Have your IQ and Personality checked as you may be disease-prone."

Displays of scientology and dianetics books and other publications and of posters eulogising scientology and dianetics in general terms still appeared in the windows of the Melbourne HASI at least till the end of the Inquiry; boxes of pamphlets, which did not disclose the true nature of scientology and its practices, but praised its achievements and invited customers, were available to anyone passing by who accepted the "Please Take One" offer.

Some of the newspaper advertisements offered free literature and "free information packages." Pamphlets in current use were tendered as exhibits. These pamphlets offered popularity, aids to successful living, and "the directest and easiest route to wisdom," the last named route starting with the free personal efficiency course, the next course being the HAS course for six weeks at £15 for the course, which "will prepare you for further courses where even greater wisdom can be acquired."

One pamphlet stated that "The Scientology Centre is conducted by the HASI (Hubbard Association of Scientologists International) a world-wide non-profit organization with no political or religious affiliations."

The "Information package" stated that the free course was to "actually start you well on the road to achieving" more popularity, more self-confidence, more relaxation, solution of more problems, increased earning power, the faster attainment of ambitions. The package informed the inquirer that scientology was the common people's science of life and betterment, that it held that people know best about themselves, and that it stood for freedom, freedom from mystery, freedom from fear. The reader was further informed that "Doctor" L. Ron. Hubbard, described as C.E., Ph.D., an American scientist, author and philosopher, who was early trained in nuclear physics and had as "his fellow physicists, Einstein, Oppenheimer, Fermi and others," discovered scientology and "studied in the field of the mind, to learn, if possible, how to make Man capable of handling and controlling such a giant as the Atomic Bomb and the machine age. That Doctor Hubbard

succeeded is attested by the popularity of his books which have sold over ten million copies, and the wide use of his work throughout the world today."

The package concludes with purported extracts from comments from people at the end of the weeks's course, and an advertisement for Hubbard's book, Problems of Work.

Another form of advertisement read, "Worried all the time? Have your Intelligence Quotient and personality checked, as you may be just suppressed. Where? At the Melbourne Test Clinic, 159 Spring Street." Such an advertisement is in the terms directed by Hubbard, and makes an insidious appeal to anxious and worried people who are impressed by the suggestion that a probable solution exists to their problems, because they are "just suppressed", and that this can be ascertained and presumably successfully treated at the "Clinic."

Hubbard went to great pains to ensure that any person who showed the slightest degree of interest in scientology was not thereafter able to escape the HASI's importuning. Numerous directives from Hubbard dealing with every step to be taken instruct the HASI staffs in the art of procurement. HCO Pol. Lr. of the 15th November, 1960, entitled "Modern Procurement Letters," emphasises the importance of keeping on writing and writing to people until they eventually respond.

HCO Bull. of the 9th April, 1960, dealing with letter-writing technique, stresses the value and efficacy of writing letters: "Letters written over four years to one prospect without response in innumerable cases has resulted in the person turning up to be trained or processed." It was necessary, Hubbard wrote in HCO Bull. of 12th April, 1960, to write many letters and write them frequently, that quantity was more important then quality and that income was reflected by the number of letters written.

Evidence was given of numerous cases in which 50, 60, 70, and more letters were written by members of the HASI staff to persons who had ceased to visit or communicate with the HASI. One member of the staff had the title "Letter Registrar". These letters were generally short and casual, designed to re-establish communication between the individual and the HASI. Various members of the staff would take it in turn to write. If a person wrote asking that the letters cease, the reply was to the effect that his meaning was not understood, and in fact letters would continue to be sent to him. Persistence in this letter-writing technique, sometimes after as many as 70 letters, brought results. It was characteristic of these letters that reference was usually made to the recipients' having "problems" and being in need of help and that scientology would assist them.

Though written in friendly vein, these letters often had a disturbing undertone, as appears from the following series of letters written to a university student who visited the HASI, ostensibly out of curiosity, and submitted to an IQ test and a personality assessment. He was told that, though he had a high IQ and was a genius and could do anything he wanted to, his character, as the graph showed, was defective, that he was mentally unstable and that he was going to have a mental breakdown in eighteen months' time unless he had scientology help, and it was ado suggested to him that he had homosexual tendencies. He was put on the E-meter and, when asked the question, "Do you have problems?" deliberately squeezed the cans which made the needle jump and caused the interviewer "to write notes furiously". He was urged to return for treatment, but did not do so.

Thereafter he received a series of letters extending over twelve months, of which the following extracts are samples:

"Would you write me out a list of your goals and ambitions for life and if you think scientology can

help you obtain them."

"To help us continue to send you copies of *Communication* magazine would you take out an Associate Membership and become a member in the world's largest health organization in the field of mental health?"

"From the look of your file you were a pretty worried boy last year. Most of your points on the graph are in urgent attention required. So I suggest you call in this year for a new Case Assessment and find out what can be done to help your mind."

"With an IQ like yours you shouldn't have failures on exams! But of course when you're not happy, you certainly can't put your mind on study, can you?"

"When are you coming in to see us again?"

"How did you go with your exams last year. We can help you become more able regarding study."

"I notice you intended to do the rest of our Personal Efficiency Course after your exams in 1961. What happened that you didn't do this? I'd like to know."

"When will you be able to do the Personal Efficiency Course you paid a deposit on last September?"

"You can talk to us about your failures in life. You need to have someone to communicate to about your difficulties so why not now."

"It seems to me that you had quite serious problems when you did your Personality Graph last year. Come in and do another and see how you feel now."

This student, who was a mentally normal individual, was considerably disturbed by what the HASI had told him and by these letters, as is clear from the evidence of the University Student Counsellor, an experienced psychologist, whom he consulted.

In HCO Pol. Lr. of the 31st December, 1963, under the heading "Handling Incoming People", Hubbard writes:

"Reception must regard any people that walk in, except trades people and business callers, as potential preclears or students. Snap them onto our lines fast. Sign them up for something and get them wheeling along our very efficient lines of Process and Training when they walk up the front steps. Get the person's name, address, and phone number. Make out a green slip used for this purpose, and for change of address.

" Anyone wishing general information on Dianetics and Scientology should be routed to the Registrar For new people, always recommend the Personal Efficiency Course, and get the person's name on the mailing list Sell the person a book."

In HCO newsletter of the 7th May, 1962, the following instructions are given:

"Register every new person walking in the door, even the postman. No matter what they say, if they are there, they have come in for help of some sort. Sell them a book. Don't let them leave without something Sign every new person up for testing and an interview. Put them on the meter and pull their missed withholds - What people should have found out about them and didn't Sign up students for a specific period of time with payment in advance. Sign up a preclear for an intensive of

the length necessary to get a major case change that is real to him. If the guy needs 100 hours, audit him for 100 hours. Let the preclear finance his own auditing You're not in the credit business." In HCO Bull. of 9th April, 1960, Hubbard gives these instructions:-

"When the prospect comes in, see him or her at once (No waiting). Be courteous, friendly, businesslike. Rise when they enter and leave. Call reception to show them out if they stay too long. Be willing to take their money. Always prefer cash to notes. We are not a credit company. Always see the student or the pc before they leave the place after service. You can often sell more training or processing It is a maxim that unless you have bodies in the shop you get no income. So on any pretext get bodies in the place and provide ingress to the Registrar when they're there." Hubbard has shown remarkable acumen as a high pressure salesman of the product of his meretricious labours. He recognizes the need for creating an interest in the prospective buyer and then of stimulating and developing that interest with tantalizing but incomplete peeps at the next stage, for which those persons with interest now aroused in his wares feel a need or curiosity to explore. He has marked down as his particular victims the more gullible, and he has devised sedulous means whereby the victim's interest, once aroused, is not allowed to flag until he has come effectively under scientology's domination, and then it is too late.

Hubbard appreciates the need for creating such an interest, and he has developed the personal eff1ciency approach as being the most effective to ensnare In numerous directives, policy letters, bulletins, and the like, he has laid down precise techniques to be used to arouse and sustain interest and effectively to capture raw meat for the "org". The bait is the offer of something free, namely, the three free lectures; and the personal efficiency course, though initially costly to the HASI, is, in Hubbard's eyes, more than justified for it gets the victim into the "org" and so on the path which leads to scientology bondage.

The detailed and precise instructions of Hubbard in respect of the personal efficiency course appear from the following substantial extracts from the bulletins - In HCO Bull. of the 29th September, 1959, entitled "The Organization of a PE Foundation", Hubbard writes,

"A PE Foundation is a programmed drill calculated to introduce people to Scientology and to bring their cases up to a high level of reality both on Scientology and on life. This is best accomplished not by giving them samples and bits and trying to lead them into auditing but by giving them gen and serious results as heavily and rapidly as possible. A PE Foundation in its attitude goes for broke on the newcomers, builds up their interest with lectures and knocks their cases apart with comm course and upper indoc. The student does not get out of the comm course until he can be trusted to show up well in a muzzled co-audit NEVER let anyone simply walk out. Convince him he's loony if he doesn't gain on it because that's the truth The whole dream of a PE Foundation is to get the people in fast, get them invoiced in a congress type assembly line, no waiting, give them hot, excited, positive service and boot them through to their HAS [Hubbard Apprentice Scientologist course] and then worry about doing something else with them. And never let a student leave or quit - introvert him like a bullet and get him to get audited. If he gets no reality don't let him wander out. If he walks in that door for a free PE, that's it. He doesn't get out except into an individual auditor's hands in the real tough cases, until he's an HAS."

The sequel to the free lectures is the offer of a free IQ test and of personality tests. (It was said that since about August, 1963, the tests were not free) The ignorance and the lack of skill of the HASI staff who conduct these tests, a subject dealt with later in this Chapter, is sufficiently serious, but Hubbard makes it abundantly clear that the real use of such tests is not so much to analyse the victim's IQ and personality for the purpose of benefiting the victim's condition as to exploit the victim still further and to tigl1ten the net in which he is caught. The sustained villainy of Hubbard's

methods is almost unbelievable, but his own writings document each step which he directs to be taken. Thus, in HCO Pol. Lr. of the 28th October, 1960, Hubbard, in dealing with the use of these tests, writes,

"Remarks that 'Scientology can influence this or that characteristic' or 'auditing can remedy that' or 'Processing can change this' or 'Training can stabilize that' should be repeatedly used during the evaluation for the sake of impingement Remember low cases want only to escape the consequences of life Certain traits showing difficulty in handling people should be stressed as most easily remedied and kept remedied by academy training. Graphs showing the 'therapeutic' value of training should be in the display book and on walls We will take full advantage of the superstitions of people at the level of prediction."

A month after the last mentioned policy letter, Hubbard, in Pol. Lr. of the 24th November, 1960, entitled "Testing Promotion Revised", gave precise instructions to staff members of the HASI as to the manner in which an "Incomer" should be dealt with by the "Evaluator" to ensure success in procuring the Incomer. The suggestion of levity which may seem to be in Hubbard's style in this policy letter should not deceive. The instructions contained in it were serious and were intended to be carried out, and were carried out, by the HASI staffs. By such techniques, Hubbard ensured absolute control to the last detail of everything that happened in the Melbourne HASI, and his directions and routines were faithfully carried out under the ever watchful eye of Hubbard's own security organization, the HCO in Melbourne. In this letter Hubbard wrote,

"Evaluator takes Incomer off meter without explanation and turns to graph. Evaluator now explains each point of graph. But it is vital that at each low point, where explained he adds, 'Scientology can help that.' This is said directly to make an impingement. The wording can be varied but the sense must be the same. Do not precede this statement with 'Don't worry' or the like as this cancels impingement. Graph done, Evaluator explains IQ. If low he says 'Scientology training can raise that.' He explains levels of IQ; tells person even if it's high that IQ means little unless person, knows something with it. Evaluator now takes up the Meter Case Assessment sheet. Here he tells of pc's future. It is done by looking at pc's statement of his past and by rephrasing saying it is going to happen, (without Scientology fates don't change much. Accidents, divorces, &c., happen again). This is all rapidly done. Factually, expertly. . . The Evaluator now leans back and says 'That's it.' Incomer is hanging on ropes. If Incomer says anything like 'What can I do about it?' Evaluator says, 'That is very commendable. A good point in your favour, wanting to do something about it. I'm a technical person not a sales personnel. Confidentially, though, I'll give you a tip. Don't spend money foolishly until you know what you're spending it for. Psychiatrists and so forth could cost you thousands. You'd buy anything they said because you know little about the mind. So why don't you take an Anatomy Course and learn something about the mind. That's just a tip. It's cheap and you'll be wiser about what to do about yourself. The person over there is in the Service Department. Ask him.'.... If the Incomer walks out without buying, the PrR man (even if he is interviewing someone else and even if Incomer has not approached him) rushes over and gives Incomer a copy of Problems of Work and Dianetics. Evolution of a Science, and says 'Here are two books that might help you,' and without waiting for an answer goes back to his desk. The above routine is at this time a set, fixed activity. As it works further it may be improved."

The Board heard expert evidence to the effect that a mentally normal person, who happened to do the PE course and have IQ and personality tests or who merely had these tests, might quite readily be convinced that he had serious faults and failings and pronounced personality defects, if the person conducting the tests had a sufficient air of confidence and efficiency and gave the impression of validity to the spurious practices being engaged in.

Hubbard, and the HASI under his direction, have practised calculated and deliberate deceit in the routine followed for signing up preclears for processing. Hubbard laid down the procedure for signing up preclears who, having been disturbed and upset by the fraudulent advice of the "Evaluator" as to their mental or other condition and told of the benefits which scientology could produce, were now in such a state or condition that they more readily accepted the suggestion that they should have auditing. In HCO Bull. of the 9th April, 1960, Hubbard gave the following instructions,

"Once the prospect is in the office the old routine of signing prospect up, getting technical acceptance from D of P [Director of Processing] and resigning prospect for increased hours must be done."

This particular instruction has been carried out at the Melbourne HASI with marked efficiency. Almost invariably, the unsuspecting prospect has found himself trapped into a commitment far greater than he originally intended. The evidence of numerous witnesses, who gave evidence in support of scientology, made it clear that the following almost standard routine was regularly followed. The prospect was interviewed by the Registrar who, after taking particulars, would then obtain the signature of the prospect on a printed form, on which the prospect undertook to receive a relatively small amount of auditing, generally 25 hours or thereabouts. Then the Registrar would escort the prospect to the Director of Processing, who would go through the motions of considering the application. The Director of Processing would then sign a printed certificate on the back of the application form to the effect that he had considered the application but, being of opinion that in the case of the applicant a "stable result" could not be obtained in less than a specified number of hours, generally from 250 to 300, he was prohibited by "the laws of the Board of Trustees" from accepting the applicant, and he would then return him to the Registrar. One is almost aghast at the deliberate callousness of this routine, which involved the generating of worry and distress in the mind of the prospect and then virtually the refusal of assistance claimed to be efficacious, unless he committed himself for some hundreds of pounds of auditing, which was said to be the least that could be expected to produce the mysterious "stable result." The fate of the applicant upon being returned to the Registrar can readily be pictured. The prospect, by now thoroughly upset, and whether or not he was really aware of what he was doing, regularly signed for the much greater number of hours stipulated by the Director of Processing - with the qualification that he only paid for such auditing as he had. Many witnesses were surprised when shown their signatures to an agreement to take the greater number of hours. All of them, however, had taken many more hours of auditing than the modest 25 hours or so for which they had initially signed, and many of them. after hundreds of hours of processing, were still committed to further auditing by the terms of the contract.

That the signing up of prospects in the way described is a deliberate trap is undoubted. In this way the HASI preys upon the anxieties of its victims. The prospect having had a slight acquaintance with scientology and having attended the free lectures and believing that some benefits might derive from association with scientology, must receive a very disturbing shock to learn that the exalted Director of Processing reports so gloomily on his case. He is then easy prey for the much larger commitment.

The Melbourne HASI faithfully carried out Hubbard's instruction in this respect. The printed form of application, together with the disturbing certificate on the back of it awaiting the Director of Processing's signature rejecting the applicant unless he signs up for a greatly increased commitment, testifies to the almost invariable routine which Hubbard directed must be carried out.

The value to HASI of the prospect signing for the larger number of hours becomes evident should a

preclear, after taking processing but not the full amount signed for, realise that he has been duped and seek to obtain a refund, which, according to the rules of HASI, he is entitled to receive if not satisfied with the results of processing. The promise to refund to dissatisfied preclears is as specious as many other Hubbardian promises. An actual example, almost without comment, is sufficiently eloquent. A woman signed up for 300 hours auditing with the HASI. She took 175 hours, and then wrote to HASI in the following terms:

"At the conclusion of the last 25 hours auditing which I had you promised to write to me. That is now five weeks ago and I have not heard from you.

My last auditor asked me why I did not ask for my money back. I have been thinking this over and have come to the conclusion that I am quite justified in doing so. If I have received any benefit from the HPA course and from auditing it is so vague that I cannot even be certain that any good has been achieved. I feel there is something in Scientology, but I don't even feel certain of this. If you assessed this result as being worth £50 it would still be no bargain. It is certainly not worth the hundreds of pounds I have paid you. Scientology has abjectly failed to make good its claims in my case, and I would like the money I have paid, less £50 for the doubtful benefits received to be refunded."

Ron Masters, then Chief Registrar at the Melbourne HASI, replied as follows:

"Thank you for your letter....

About refunds of money I will quote the Policy on this: 'Processing refunds may never be given until the person has completed all the hours for which he has signed up, even if he has not paid for them'. You signed for 300 hrs and have only taken 175 hrs of this amount, so it is not within my power to make a refund.

We understand what you are going through at this time, but it is a truism to state that 'the only way out, is the way through'.

Call in for a chat when you're free, as I'd like to see you."

She then wrote:

"I received your letter this evening. The position regarding refunds is most unsatisfactory. As far as I know it is not possible to receive any auditing without paying for it in advance. Therefore, I would have to pay approximately £250 more, as well as wasting further time, in order to claim a refund. What guarantee would I have, if I did this, that any refund would be made? Has any money ever, in fact, been refunded by any Scientology organization? I consider 275 hours - and this is not counting the 300 or more hours I had before coming to the HGC - an ample test of your claims."

Masters' reply was:

"Thank you for your letter.

The position regarding refunds remains the same.

You have a guarantee by Policy that a refund would be given after taking all the hours, and refunds have only been on this policy.

The policy safeguards the PC who thinks he or she has had enough, you see. I repeat, the only way out is the way through."

She then wrote:

"I received your last letter and in reply I claim that my situation has in every way worsened under the influence of Scientology, while the last 25 hours of auditing has been absolutely disastrous. Now there are times when I have to struggle to even stay awake, and as a consequence I fear to lose my job and the little security that gives me. It takes a frightful effort not to go to sleep and is a terrible battle to keep my eyes focusing. I have to shake myself awake to read a figure, and before I have transferred it to another sheet I am nearly asleep and have to struggle again to keep awake and forget what the figure was. I become nearly unconscious even in the middle of making an entry and the figures trail out and become indecipherable. I am slow in my work and make mistakes. There is a period nearly every day, lasting between 1 and 2 hours when every ounce of energy I possess is needed just to keep awake. I am not exaggerating the misery of this situation. I dread these periods of battling to keep from just giving in and going to sleep. I am always exhausted and sometimes can hardly walk along. This the result I have obtained from spending all my money on something that is claimed to increase alertness and intelligence and generally benefit people. I have received no benefit at all, have come not one whit nearer to the solution of my problems, but have lost all that made my life at all happy.

Under these circumstances it is no help to be told 'the way out is the way through'. I have no money left for further auditing and no chance of saving any since I barely make ends meet. That, after 600 hours of auditing, Scientology has failed to bring the PC 'through' is evidence that it is unable to do so. I feel that I have been let down again and again, and - except by [two names omitted] - no attempt to understand or help my case has been made."

Her correspondence with the Melbourne HASI produced no refund, and thereafter she wrote direct to Hubbard in these terms:

"The Scientology centre in Melbourne states in various leaflets that all cases are personally reviewed by you. If that is true you should know something of me. I have had 300 hours auditing at this centre, and done an HPA course there, and have achieved nothing of any benefit at all. At the commencement I had a job I liked, which paid me fairly well, and enough money put away to feel reasonably secure. Now 1 have a job I don't care for, which does not pay so well, my money is largely gone, and instead of being reasonably content I feel that nothing is worth doing, have periods of absolute exhaustion, and look forward to the remaining years of my life with complete hopelessness, as just a dreary round of work, work, work at something I don't care for to earn enough to exist to go on working.

Prior to going to HASI I had at least 300 hrs going once a fortnight to a private auditor for a number of years, and an improvement in my health and spirits did slowly take place, though there were no results such as would warrant me having a high opinion of Scientology, or regarding you as anything but a blowhard and liar with the claims you have always made.

You repeatedly claim that Scientology can help all cases, that only bad auditing errors can hold up cases now, and even, I seem to remember reading in one of the last PAB's I received, that the initial case level does not affect the wonderful benefit to be gained from new processes.

The last 25 hours intensive was given to me free, as I had finished up before that, after 275 hours which I had paid for, fearing even to lose my job as I had long periods at work when I could hardly

keep awake and didn't know what I was doing. I don't feel in any way grateful for this free auditing, as my own feeling is that it is the very least they can do. If they repeated the whole 275 hours it would not be any more than should be done.

I would like your comments on this situation."

Hubbard referred this letter to the Melbourne HCO, where Mrs. Williams "diagnosed" it as a case of "missed W/Hs [withholds]".

Shortly afterwards, this preclear was constrained to visit the HASI, and was persuaded to undergo 25 hours' free processing for her "missed withholds". Two years later she was still in the clutches of the Melbourne HASI, having resumed auditing on the basis of her paying for it.

Once the prospect has signed up, he is subjected to scientology processing which enslaves his mind and dominates his will and extracts from him hundreds, sometimes thousands, of pounds. On the one hand, the extravagant promises of clear and beyond, of a new world, of the homo novus, of happiness, serenity, immortality, and, on the other hand, the fear engendered as to what terrible fate lies outside scientology, and the dependency on scientology which is carefully cultivated, are effective means whereby the preclear is kept in continuous subjection.

Any incipient defection or any act or thought hostile to scientology is specifically dealt with by exploiting fear and particularly the fear of death. The procedure for countering such defection is more processing which further shackles the preclear. This is abundantly clear from the HCO Bull. of the 31st December, 1959, where Hubbard writes of the terrible plight of people leaving scientology without clearing, and who have overts against scientology:

"I assure you, and with some sorrow, people have not often recovered from overts against Scientology, its Organization and related persons. They don't recover because they know in their hearts even while they lie that they are wronging people who have done and are doing enormous amounts of good in the world and who definitely do not deserve libel and slander. Literally, it kills them and if you don't believe it I can show you the long death list."

In this bulletin he tells of a man with uncleared overts and withholds who contracted TB within a few weeks and was dying in a London hospital. The bulletin continues:

"I once told a bill collector what and who we were and that he had wronged a good person and a half-hour later he threw a hundred grains of veronal down his throat and was lugged off to hospital, a suicide."

In the same bulletin, Hubbard gives instructions designed to ensure an increase in his domination over the incipient defector. He writes:

"Uneasy lies the head that has a bad conscience. Clean it up and run responsibility on it and you have another better person, and if anyone feels like leaving just examine the record and sit down and list everything done to and withheld from me and the Organization and send it along. We'll save a lot of people that way."

IQ Tests and Personality Assessments.

The HASI and its satellite organizations, such as the Melbourne Test Centre, by their advertisements, invited the public to attend to have their IQs tested and their personalities

assessed, promising to increase people's IQs and to improve their personalities and their enjoyment of life in various ways. Examples of these advertisements are set out earlier in this Chapter.

Until August, 1963, the testing and assessment often formed part of the personal efficiency course and were free. Thereafter a small charge was made for these services.

The IQ test and the personality assessment are associated tests normally undertaken by psychologists. The claims to be able to increase IQ and improve personality are unjustified. The IQ of a person is, in broad terms, a standard of measurement of the degree or quality of intelligence inherent in a particular person, and a numerical scale is used as a convenient method of classifying the intelligence of persons tested. Experiments and wide experience have produced a scale which broadly grades intellectual and other potential capacity. This scale measures things quite distinct from knowledge and skill, though these qualities may be reflected in testing and various allowances are made for them. A person's IQ, however, is inherent and does not admit of being increased, though tests made at different times on the same person may produce apparently different results. Such variations may be attributable to familiarity with the surroundings, previous tests, present condition of health, lack of concentration or interest, perversity and other matters. Broadly speaking a person's IQ does not rise and fall, and it is generally accepted by psychologists that, for practical purposes, at least after the middle teens, a person's IQ cannot be increased.

The ascertainment of an IQ is not an end in itself, but is an assessment of the intellectual potential of an individual, in much the same way as a person's height may be a consideration if he were to be employed where tall men were needed.

The personality testing at the HASI is done by a system called the Oxford Capacity Analysis, which incorporates a graph on which various personality traits are listed. The Oxford Capacity Analysis is based almost entirely on a particular American system of personality analysis, which, according to expert psychological evidence, is not highly regarded. The copyright of the Oxford Capacity Analysis is in Hubbard, and it is a peculiar circumstance that it is one of the very few pieces of written scientology material which does not prominently carry either his name or the word, "scientology".

An expert psychologist, after examining the instruction manual for conducting an Oxford Capacity Analysis, gave evidence that "the overall impression one gets from reading this manual is that it has been prepared by somebody with a smattering of psychometrics rather than by someone who is really competent in the field."

The Board heard evidence from a number of expert psychologists concerning IQ and personality tests. It appears that the Australian Council for Educational Research, which has the agency for several intelligence and personality tests, restricts their sales in order to prevent their misuse by unqualified persons. There are various kinds of tests for various purposes and varying qualifications are required in those who desire to conduct such tests, but the Australian

Titles in Scientology.

Part of the deception practised by scientology is the misleading use of titles, qualifications and insignia for the purpose of conveying the false impression that scientologists so described have valuable qualifications. As mentioned elsewhere, Hubbard makes false claims to a degree in engineering from the George Washington University, and his "Ph.D." appears to be from an unaccredited institution. The title "Doctor" which he sometimes assumes is based on this Ph.D., or on a doctorate of divinity, virtually self bestowed, and granted, it seems, by one of his own

institutions. The founding churches of scientology have awards entitled, "Doctor of Divinity", "Minister of the Church" and "Spiritual Counsellor".

The scientology qualifications are designated by various letters which likewise deceive. There are "degrees" of "Doctor of Scientology" and "Bachelor of Scientology", respectively designated by "D.Scn." and "B.Scn.", both intended to suggest an affinity with "D.Sc." and "B.Sc." which are appropriate to Doctor of Science and Bachelor of Science, both well known University degrees. The "D.Scn." carries, of course, the appellation, "Doctor", which readily deceives even the alert person who may well believe that in the HASI, where the emphasis is on health, the title denotes a doctor of medicine.

The lesser scientology qualifications also carry letters after the names of those so qualified. Hubbard Professional Auditor is denoted by "HPA", Hubbard Qualified Scientologist by "HQS", Hubbard Certificated Auditor by "HCA".

One scientology witness falsely claimed to be entitled to have the letters "B.A." after her name, and even managed to deceive the HASI that she was "B.A. London, 1923". In evidence she admitted the deception, but claimed to be entitled to use the letters "B.A." as representing, successively, "Basic Auditor" and "Book Administrator".

The Likely Victims.

It is common psychiatric knowledge that one type of person who would be likely to respond to the attractive offers and promises contained in the personal efficiency course advertisements and invitations to have IQ and personality tests would be those who would be likely to feel insecure, inadequate, inferior, and anxious-the person who is usually a poor mixer, feels lonely or unloved, feels he is not recognized for what he really is, finds the business of living too burdensome. The class of persons attracted is not limited to such unfortunate people. The appeal is effective with normal people, and also with the greedy and the opportunist.

The free offer makes a triple appeal: first, what is offered is more than a bargain, it is free; secondly, it is offering just the sort of release or escape or opportunity the worried person thinks he needs; and, thirdly, it engenders in the mind of such a person the thought that the organization offering such wonderful things free must be of superior stature and quality to be able to achieve the wonderful results promised.

Great efforts are made by the HASI to impress the inquirer at the earliest stage. An atmosphere of efficiency prevails; a particular point is made of punctuality, the instructors and staff are outwardly confident and certain of themselves; at the first lecture the individual is addressed by his Christian name and so flattered that he begins to feel that scientology is already doing him some good. One witness said that, at the very first free lecture, he had been taught to do something he had never been able to do before, namely, to look people straight in the eyes; now he could do that and, so he said, he had never looked back. He was, instead, spending all he could afford, and probably more, on scientology processing with no further observable results.

At the end of the lecture, the time of the next lecture is mentioned and the instructor makes sure everyone intends to attend. Any waverer is questioned, given some suggestion as to how to overcome the difficulty of attending, and then told that he will be present at the next lecture. "No", is not taken as an answer, and the impressed beginner is somehow constrained to attend.

The inadequate and inferior person, having experienced the pleasure of being accepted and being

shown affection by the organization, feels a desire to reciprocate; he wants to prove his worthiness to be accepted and loved and he enthusiastically seeks to please in whatever way he can. The way expected of him is to show continuing interest, and later to participate at his own expense in the activities of the organization.

At the stage when the preclear is entering into scientology he is sometimes security checked; that is, he is interrogated at great length according to a set formula, and on these occasions he has sometimes been asked such questions as "Do you deserve to be helped by scientology?". This heightens his desire to demonstrate his worthiness and at the same time it increases his dependence on scientology.

In auditing, the preclear, with his inhibitions in suspense, reveals his secret faults and recalls and acknowledges his inadequacies and transgressions. The mental distress attendant upon such feelings of guilt makes him wish to obtain some relief from his discomfort by atonement, confession or redress; it renders him more likely and ready to subject himself to the confessional, sacrificial and self-abasing process of auditing whereby he can purge himself of his shortcomings; and it develops in him almost a missionary zeal whereby he has the urge to render the same service to others by being an auditor himself. He is thus seeking to emulate the auditor on whom he has developed an increasing dependency, and in the role of an auditor himself he hopes to enjoy the same sense of power over his own preclears.

The development in the preclear of a desire to be an auditor facilitates the HASI's task of persuading preclears to become students. The preclear regards his auditor as a superior person who is on the way to clear, and who conducts himself in auditing sessions with confidence and authority. Scientology seems to the preclear to be a short cut to success, power and authority. It makes unnecessary, so it seems to the preclear, persistence with one's secular studies or interests and it enables him to ignore the reality of the world outside scientology and its problems and responsibilities. All these halcyon concepts of scientology and its practice are the outcome of a pot pourri of contributing factors - the individual's own inadequacy and gullibility, the desire to be thought worthy, the insidious advertising, the free lectures, the fantastic promises, the confident but shallow attitude of the instructors and staff, the awe of the E-meter, the assurances of success the lure of the clear, the hypnotic effect of techniques even at this early stage-and the calculated springing of the trap baited so temptingly for the unsuspecting victim.

These are no alarmist observations, for the Inquiry has revealed that scientology has made its conquests with anxious people: the worried and inadequate, the uncertain, the groping, the seeking, the uncritical, the unfortunate who grasps at a promise of betterment eagerly and without question. In the realms of the mind there are investors and speculators, worried and anxious people, eager and greedy people, the little and the large investor, all too ready and willing to rise to the bait of an entrancing future of sanity and mental serenity, of freedom from worry and care, of increased intelligence and greater ability and any other seductive pot of gold that the charlatan concocts.

Courses in auditing are carefully organized and conducted to induce in a preclear the desire for further auditing. Almost without exception the files show that at the end of a course of processing the preclear is desperately eager for further processing. At the end of a typical 25-hour intensive, the preclear is either depressed or elated; each of these conditions is treated as an indication that further processing is desirable; a depressed condition is said to indicate a need for more processing to raise the preclear up the tone scale; an elated condition is readily accepted by the preclear as a promising prelude to more processing. Preclears are processed into a strange world of illusion which offers an escape from reality, and they are conditioned to desire more and yet

more processing until a state of euphoria is reached and maintained. They thus develop a pitiful dependency upon the HASI, which is a direct consequence of the hypnotic processes used, and this dependency is sedulously nurtured by the HASI. In the practice of medicine the likely development of dependency by the patient upon the psychiatrist and psycho-analyst is well known, and responsible practitioners are alert to ensure that it is minimized and guarded against. Indeed, the object of orthodox treatment is to restore the patient to a state where he can face reality. The conduct of HASI is thus the negation of orthodox psychiatric and psychological practice, and constitutes a significantly evil feature of scientology processing.

Part of the scientology technique is not to allow the preclear ever to achieve complete fulfilment of his aims, and the preclear who has obtained good results is told that he could do better, and is left with the guilty feeling that he has not done his best. The feelings of guilt are thus accentuated by encouraging the preclear to find fault with and further abase himself. Such a procedure is common in brainwashing techniques.

The Anderson Report CHAPTER 16 THE PRECLEAR'S PROGRESS

Almost invariably, a person who enters scientology has been told by the HASI that he is "low toned" and that scientology can raise him on the tone scale. In nearly every case he will have had an IQ test and a personality assessment, and he will have been told that, whatever his IQ may be, scientology can increase it and that certain personality traits shown on the assessment as urgently needing attention can be improved by scientology processing. He will have heard of the fantastic successes claimed to have been achieved by scientology in the fields of mental or physical health or both and in general well-being. He has read, perhaps uncritically and without fully appreciating what he read, some of the introductory literature of scientology. His preoccupation with his own particular ambitions, condition and problems leaves him a ready prey to extravagant but confidently asserted claims, and he sees the rosy prospect of improvement, recovery, rehabilitation, success, and happiness almost within his grasp. His initial experience with scientology in the personal efficiency course may have cheered and encouraged him. He has been made welcome, greeted by his Christian name, made to feel important by being told in a competently delivered "pep" talk that he is as good as the next man, and so he is prepared, as a number of witnesses said, to "go along" with scientology.

So many "go along" to their ultimate confusion and ruin, financial and mental. It is this "going along" which plunges the unfortunate deeper and deeper into the mire of scientology. Scientology, with a specious display of candour, parades as something most reasonable. In an often quoted passage Hubbard has written:

"We are not interested in scientology in getting people to accept what we say without question. We ask them to question it. We ask them to please look at the physical universe around about. To please look at people, at their own minds, and understand thereby what we are talking about happens to be actual. I am not giving you new things. I am giving you old things. By understanding these old things which we have rediscovered, you become free."

The significant part of this statement is that the person looking around has already been conditioned to accept "that what we are talking about happens to be actual", and that by understanding, "these old things which we have rediscovered", he will become free.

Other scientology writers, e.g., Sharpe in This is Life and Horner in Summary of Scientology, loyally copy Hubbard's technique and tell their readers not to test but to look and see for themselves that scientology is right. Implicit in such writings is the suggestion that, if the preclear does not agree, it must be because of his own deficiencies, for Hubbard has said it is true and the preclear's inability to see the truth has been explained to him as being due to his low tone which eventually scientology will remedy.

Williams gave evidence that Hubbard's teaching was that:

"Nothing is true for you unless it is true for you and that which is true is that which is true for you and that which is not true is that which is not true for you. That which is real to you, you can accept. Nothing in dianetics or scientology is true unless it is true to you, and you do not have to accept it unless it is true to you."

Such utterances as this, and others in similar vein, may give to the individual entering scientology the impression that he will be self-determined in his association with scientology. But, very quickly, processing robs him of any real will. Coupled with such statements are statements to the effect that, "You may not accept what we say but it's all true, you know," and assurances that when, by processing, the individual has been brought higher up the tone scale he will have a better understanding of the things which are now beyond his understanding. The subtle approach works. The individual, in "going along", is persuaded not to reject anything, not even the most fanciful and fantastic nonsense, which, when he has been only a short time in scientology, he hears and reads.

Several scientology witnesses, who stated that they still had no "reality" on one or more of the fantastic ideas of scientology said, however, that they did not reject such ideas - they merely had no reality on them but were confident that eventually they would have a reality on them. They think that eventually it will all "add up" to something on which they will "cognite".

Since nothing has been rejected the mind and will of the preclear remain potentially receptive of the more advanced and fantastic teachings of scientology, when he has become better conditioned to accept such teachings, and he is assured that he will eventually have a reality on them when he is sufficiently high on the tone scale.

This technique involves a resolution on the part of the individual to exercise the mind and will, if they are to be exercised, only in the direction of acceptance of scientology teaching when in a befuddled state he gets a "reality" upon the particular thing. Acceptance of scientology "truths" may occur when the preclear, in a process-conditioned state, accepts as true the fantasies and hallucinations he has experienced during the course of auditing.

The unfortunate preclear is further conditioned towards acceptance or at least non-rejection-by the "code of a scientologist" which requires him "to hear or speak no word of disparagement to the press, the public or preclears concerning any of my fellow scientologists, or professional organization or those whose names are closely connected to this science, nor to place in danger any such persons". He is warned about the evil motives of those who might attempt to persuade him to give up scientology, and the "Training Course Rules and Regulations", published in HCO Pol. Lr. of the 22nd November, 1961, contain the following requirement: "25. Do not discuss your case - or that of any other student - your auditor, your Instructors, your classmates, L. Ron Hubbard, Organization personnel or the Organization with anyone". He is persuaded to withdraw from communication with his associates of his pre-scientology days, and to move only in the vacuum of the HASI. In this way he is proofed against the efforts and entreaties of his friends and family. He is assailed by scientology literature such as, "Why some fight scientology", the burden of which is that those who oppose scientology have something to hide, and that the spouse who opposes scientology has been unfaithful or has some guilty secret.

This "going along" panders to the vanity of the individual who shortly feels of increasing intellectual stature, being, as he believes, on the threshold of a new body of learning which claims to have rediscovered long forgotten truths; and of being transformed from homo sapiens to homo novus. He feels that to question or deny what the organization has stated to be a scientific fact would reveal him as unprogressive, of low intellect, as unfitted to associate with members of what he imagines to be a very learned, progressive, and efficient organization with which he has become acquainted. He is congratulated on being chosen - the implication is, chosen by Hubbard - as one of the ten thousand to be clear, and on being, in effect, part of the avant-garde of an organization which is going to perform such wonders for the world as to clear it of all its aberrations. He aspires to the delights of being clear and to the wonderful vistas beyond, though just what they are does not appear very clearly. But that does not matter at present because he will know when he gets there.

One very common feature of scientology indoctrination is the loss of the preclear's critical faculties and his abandonment of reason. One witness said that before he entered scientology he was critical but now he could control his ability to criticize and did not now use it.

Of course, during all this time, the preclear is undergoing processing and paying for it, and something is being achieved, though whether for good or evil he does not know, except that after processing, as after a drug, he may experience for a while an elation and some sense of ecstasy before he plunges, as after a drug, into a state of depression. When this happens, he is told it is a good sign, because it is considered in scientology that "the way out is the way through", and so more processing is prescribed, undertaken and paid for.

This continued association with scientology may result in some preclears who have done courses with the Academy becoming staff members, giving their full-time or part-time services for very low financial reward. In that way they continue to have auditing either in return for services or at reduced rates. Other preclears who do not go on staff maintain active association with the HASI, taking further processing, attending various functions, e.g., tape nights and social gatherings, and receiving scientology periodicals and other literature.

They have no opportunity of being free of scientology influence. Their associates tend to become exclusively their scientology acquaintances, and scientology continues to be an escape from reality, making it seem unnecessary and foolish to go "outside scientology" in an attempt to cope with the problems of life. And, pitifully, they squander their money on processing on goals which have changed from goals personal to the individual - to be liked, to have confidence, to be able to confront people, to be less shy, to improve one's work or study, to get promotion, to get well physically, to make more money, to be rich, and goals of a similar kind - to such scientology ambitions as to be clear, to be OT, to go to Saint Hill or to obtain other distinctions in scientology. So that the transition from personal to scientology goals is not too drastic, the preclear is encouraged to state his goal or goals for each processing session, which may be of a selfish or personal nature, and also to state his ultimate goals which generally have a scientology association. In time, the pattern is for scientology goals to predominate, and the preclear is concerned only to advance in scientology.

Preclears move in a strange dream world where normal standards and family and other conventional responsibilities are ignored. To them the very fact that normality is different from scientology makes normality insane. It is all so Alice in Wonderlandish, and yet it is what the evidence showed to be the case.

The Anderson Report CHAPTER 17 DEDICATION TO SCIENTOLOGY

One striking feature of scientology is the extent to which preclears are prepared to make great personal and economic sacrifices to advance themselves in scientology and to advance the interests of scientology. A dedication to scientology, amounting to fanaticism, was observed in several of the witnesses who gave evidence.

This dedication was fairly general amongst scientologists undergoing processing and for a period after processing had ceased, even though a perusal of HASI processing files showed that they were complaining that they were being impoverished and that their conditions were worsening. The development of this dedication took various forms. A pattern common to many scientologists was that they entered scientology because of the promises it made and the hopes that it held out that it would in a most dramatic and effective way improve them mentally and physically, that it would help them obtain their ambitions or goals by making them more able, and in general that it would greatly improve their happiness and position in the world. The promises of scientology appeal variously to the hopes, aspirations, vanity, anxiety, greed of people, who see in scientology the prospect of obtaining their desires or improving their mental or physical or intellectual or social or economic positions by newly discovered short cuts which are said to be invariably successful.

This dedication to scientology resulted in several persons who became students of scientology giving up particular courses of study at university or technical or high school level, and taking employment at the HASI where they could, with less interruption, pursue their studies of scientology; it involved others forsaking employment where they enjoyed good conditions and prospects of advancement and working in the HASI at greatly reduced wages and with very few prospects of material advancement, with long hours, on six and even seven days a week. Amongst highly placed scientologists who devoted their whole time to scientology were several former school teachers, a former surveyor, a former policeman, a former senior bank official, a former trained nurse.

There were others, too, who were employed in various ways as auditors, clerks, receptionists, &c., who were sometimes unremunerated or whose wages were merely nominal. Several such individuals worked at the HASI part time only, giving their labours freely as though for a worthy cause or in return for auditing either free or at reduced rates. In May, 1962, for example, for 40 hours or more of work per week members of the HASI staff earned variously £5 4s.6d., £7.14s., £6.6s.6d., £4.8s. For 50 hours, one married woman received £5.15s.6d.; for 45 hours a woman received £5.IOs. In September, 1963, for 40 hours or more of work the wage was variously £9.2s.9d., £10.15s., £12.18s. For 40 hours, some married women received £9.13s. 6d. In January, 1964, 40 hours of work returned as low as £6.10s.6d., £6.17s.6d., £7.5s.; and in the highest posts, £10.10s.3d., £11.12s., £12.13s.9d. There were instances when the remuneration was so low that economic necessity forced some members of the HASI staff to forego full-time employment at the HASI and obtain more remunerative employment outside, while still, however, devoting whatever spare time they could, at weekends and at other times, to their employment at the HASI. In several instances, both husband and wife were on the HASI staff and their joint earnings barely provided a living wage.

The standard of living of the Melbourne HASI staff is in marked contrast with the luxury in which

Hubbard and his entourage live at Saint Hill. Hubbard appears to have adopted a deliberate policy of keeping the HASI staff on meagre salaries; and it was a matter of complaint amongst staff members that whenever the revenue of the HASI increased Hubbard directed that the staff be enlarged, thereby keeping the individuals' remuneration at a low level.

Generally, most of the staff received less than the basic wage; frequently, as the above figures and the records of the HASI show, the weekly payment was grossly inadequate. The weekly salary of staff members depended entirely on the amount of the cash receipts of the organization in any particular week. Each week the 10 per cent. levy was deducted from the gross receipts and, of the remaining amount, 55 per cent. was allocated to staff salaries, each member receiving a sum appropriate to the number of "units" which he or she had, and which had been allotted according to positions held.

The Board had the assistance of evidence from Mr. John Clifford Thomas, Chief Industrial Officer of the Victorian Department of Labour and Industry. It appears that there is no wages board appointed in respect of scientology employees as such. Whether such a board could be constituted under the Labour and Industry Act is at least arguable, since a wages board is appointed to determine conditions in respect of any process, trade, business or occupation. It is a question yet to be determined whether scientology is a process, trade, business or occupation. The word "process" in the Act does not connote anything in the nature of a scientology "process". Though scientology may be said to be a business or occupation, it is not the function of this Board to investigate the eligibility of scientology employees to make application for a wages board to determine their conditions of employment and their remuneration.

From Mr. Thomas's evidence it appears that some of the employees of the HASI may be covered by certain determinations, such as the Commercial Clerks Determination and the Cleaners Determination, but the duties of many of the staff are of such a nature that they do not fall into any recognized category of work, and no conditions are prescribed for them under the Labour and Industry Act.

This Board does not recommend that scientology be given some entirely unjustified semblance of respectability by the appointment of a wages board to determine conditions of employment and remuneration of its employees.

Various scientology witnesses with orthodox scientific training showed their loyalty to scientology and to Hubbard, when, being unable to reconcile Hubbard's "scientific" writings with what they had learned in the course of their university and other formal studies, explained that Hubbard was nevertheless right because he was writing in a scientology "frame of reference".

One scientology witness, a bachelor of electrical engineering, considered that a thetan could generate electricity and produce energy and that, while a thetan could not now produce matter (whereas previously, "a tremendously long time ago", he could, because Hubbard had said so), scientology auditing and training could restore to the thetan his former ability to create matter. "Ron" had said it, so it must be true. Another highly placed scientologist, Gillham, the co-proprietor of the Melbourne College of Personal Efficiency, had no such reservations. He accepted as correct the scientology axiom that persons could create matter by postulation and agreement. All that was needed, he said, was that there should be a reality on it; and at the Inquiry one morning, an occasion memorable both for its comedy and tragedy, this witness solemnly swore that he and counsel assisting could create by postulation and agreement a log of wood which could even be ignited, and that if they had a reality on it, it would be there on the floor of the Board room burning and giving off heat, even though other persons present saw no log of wood and felt no heat.

Another scientology witness, a master of science, could not reconcile Hubbard's writings on science with his orthodox knowledge of science but he justified Hubbard's writings as being correct because Hubbard was writing in a different "frame of reference". Another scientology witness, with some pretensions to scientific training, said that the proper approach to scientology was to realize that there may be a statement true in physics and tested as to its validity which in scientology would be utter nonsense, and vice versa. This was because Hubbard was writing in a different "term of reference" from other scientists.

This "frame of reference" or "term of reference" was a grand refuge for scientology witnesses who repeatedly sought to explain Hubbard's nonsensical and incomprehensible writings on subjects on which he claimed to be an expert by saying that they were easily understandable if only one viewed them from a scientology viewpoint and in a scientology frame of reference. When asked to explain some of Hubbard's writings, they were unable to make any sense out of them. One scientology witness said that Hubbard was "the most clued up" man in the last two centuries. He admitted, however, that it took him two years to understand Hubbard's axioms. Gillham, after many years, said he still did not understand them.

The dedication of such people, particularly those on the HASI staff, is a disturbing matter. Many of the HASI staff members are undoubtedly mentally ill to a greater or lesser degree. Generally, their introduction into scientology has come about because they possessed some social or emotional or intellectual deficiency, in much the same degree and kind as some of those whom they now seek to entice into scientology. The extravagant promises of success and advancement, without having to undergo the conventional travail of study and hard work and waiting, made to them the same deceptive appeal as it makes to those who today turn aside from their struggle with the realities of life and are ready to choose what is said to be the easy and quick scientology way.

This dedication was apparent in the manner in which scientology witnesses gave their evidence. The HASI presented its material to the Board competently and clearly. Its witnesses were fluent and were undeterred by cross-examination which revealed much of what they said as nonsensical and quite contrary to accepted norms of reason. They paid more than lip service to the "science of certainty", the science of "knowing how to know"; they immolated their reason in defence of Hubbard and his teachings. They proceeded on the basis that "Ron is right" and that scientology was "true". They believed that in order to "cognite" on the great truths of scientology it was necessary that one should be "in scientology", and that those outside did not comprehend, because being outside made them incapable of cogniting. Such outsiders, they believed, were aberrated and required the balm of scientology teaching and processing to raise them up the tone scale and so condition them to the reception and realization of the truths of scientology.

But for the fact that pity, sympathy, and allied charitable attitudes were anathema to scientology, the HASI witnesses would have pitied those outside their fold. As it was, they were cheerful in their exclusiveness.

The Anderson Report CHAPTER 18 SCIENTOLOGY AND HYPNOSIS

The Board heard expert evidence in relation to hypnosis and hypnotic techniques. There are, broadly speaking, two types of hypnosis: passive hypnosis, and command or authoritative hypnosis. Hypnosis, properly administered by skilled practitioners, has its proper place in psychiatric practice, and, in suitable cases, produces beneficial results. In the practice of medicine the type of hypnosis generally used is passive; the patient is allowed to, and helped to, go into hypnosis entirely voluntarily, and the hypnotist plays a completely passive role. This technique is quite the reverse of authoritative or command hypnosis, where the hypnotist assumes positive authoritative control over the patient who, though he may or may not be aware of what techniques the practitioner is practising on him, is nevertheless under the domination of the hypnotist pursuant to positive commands.

Because of the serious risk of harm to the patient, it is only in rare and exceptional circumstances that authoritative hypnosis is resorted to in medical practice. The distinction between the two types of hypnosis, the one beneficial and the other potentially harmful, is to be borne in mind when considering the hypnotic qualities of scientology techniques. Hubbard is very familiar with hypnosis, called by him, "hypnotism", a term which, even at the time of his early writings, was being superseded in professional use. Hubbard earlier practised hypnosis as shown abundantly by his writings. In *Scientology*, issue 15-G, he writes, "I was schooled in hypnotism and mysticism". Elsewhere he writes that he tried hypnotism, but gave it away because of a number of undesirable features which he said it possessed. In *Dianetics: MSMH*, *Science of Survival*, and other writings, Hubbard is highly critical of hypnosis, but what he is criticizing and denouncing is authoritative or command hypnosis, and to the extent that his criticisms relate to, and are limited to, authoritative hypnosis, they have considerable validity. Such criticisms, however, do not apply to passive hypnosis as practised by skilled and experienced psychiatrists with benefit to their patients in selected cases.

In the skilled practice of hypnosis the practitioner is well aware of the dangers which may arise from the tendency to develop in the patient a degree of dependency upon the practitioner, who is concerned to ensure that this and other dangers inherent in hypnosis do not develop. It is the firm conclusion of this Board that most scientology and dianetic techniques are those of authoritative hypnosis and as such are dangerous. Hubbard and his adherents strongly protest that his techniques are neither hypnotic nor dangerous. However, the scientific evidence which the Board heard from several expert witnesses of the highest repute and possessed of the highest qualifications in their professions of medicine, psychology, and other sciences - and which was virtually unchallenged - leads to the inescapable conclusion that it is only in name that there is any difference between authoritative hypnosis and most of the techniques of scientology Many scientology techniques are in fact hypnotic techniques, and Hubbard has not changed their nature by changing their names. Hubbard seems quite capable of thinking that if he postulates that scientology techniques are different from hypnotic techniques then they are different. Whether or not Hubbard realizes that the only differences are in name, his followers loyally and uncritically accept his word and believe that scientology techniques are distinctively Hubbardian and that hypnosis is something quite different and evil and to be avoided. A number of scientology witnesses, when asked what they believed hypnosis to be, answered vaguely that it was some sort of stage technique for mesmerizing people by the waving of hands in front of them, or some such thing. It

may, of course, be that, but it is many other things also and of its real nature Hubbard's followers seem generally to be unaware.

The common practice of Hubbard is to change the names of hypnotic phenomena to names of his own invention, purporting thereby to change the nature and significance of such phenomena. Thus, a form of unconsciousness experienced in hypnosis he has renamed variously "anaten", "boil-off" and "dope-off"; hypnotic hallucinations he has called "mental image pictures"; and "dissociation" he has called "exteriorization".

Though in hypnosis there is no E-meter as there is in scientology, at almost every stage there is a parallel between scientology auditing and hypnosis, and it is even to be observed in the initial stages when the auditing session is about to commence. It is well recognized amongst psychiatrists that persons who desire to be hypnotized, or are expecting to be hypnotized, more readily succumb to hypnotic processes. It is not necessary that the subject should be expecting to be hypnotized, he may not be aware of the meaning of hypnosis or of what is involved in it. It is sufficient that he expects to receive treatment and he makes himself ready and available to the practitioner for the treatment which is to be applied and is ready to accept direction from the practitioner and the consequences or the results of such treatment. In authoritative hypnosis where the subject is a willing subject and is more or less consciously under the domination of the practitioner, it is found that the subject will readily go into hypnosis, even though he may be unaware of the technical name of the treatment he is receiving or the fact that he is, or is about to be, in hypnosis. Scientology techniques begin with a preclear who is well aware that he is to be "processed", and the circumstance that he does not know that the process, which is called by a non-hypnotic name, is in reality a hypnotic process is quite immaterial. The name has no significance to the preclear, but the process remains hypnotic by whatever name it is called. The preclear then, expecting to be "processed", finds that his processing commences, as Hubbard directs it shall commence, with solemn and strict ritual. After some standard preliminary questions such as, "Is it alright if I audit you in this room?" and "Is it alright if we start the session now?" an auditing session in scientology processing starts with an unvarying routine. When the assent of the preclear has been received to these preliminary questions, the auditor then commences the session with a loudly uttered, "Start of session." In the demonstration auditing sessions which the Board witnessed, the statement, "Start of session", was spoken in a loud sharp tone, quite at variance with the rest of the speech of the auditor and was evidently designed to impress upon the preclear that now he and the auditor had embarked on the really serious part of the business.

Such a dramatic and startling procedure conditions the already expectant preclear for the exercises or events which follow and is incontrovertibly that of authoritative hypnosis. The preclear is expecting to be "taken in hand", to use a neutral expression, and that is just what the auditor does, and that is what a hypnotist practising authoritative hypnosis would likewise do. The Board heard expert psychiatric evidence to the effect that a person who is so expectant is a very ready subject for hypnosis; it was said that a hypnotic condition could be induced in some patients merely by telling them to lie on the couch on which they had lain on a previous occasion when under hypnosis, and that even the entry into a room in which previously they had been under. hypnosis may be sufficient to return some people to hypnosis. In scientology, where processing goes on day after day, the return to the same auditor and to the same place and to the same ritual would readily predispose the expectant preclear to submission to scientology's hypnotic techniques and to a return to the hypnotic state which these circumstances commemorate.

In hypnosis, there is a condition of rapport between the subject and the hypnotist, loosely and variously described as a bond of sympathy, confidence, confidingness, trust. Hypnosis is a state of mind in the subject which is frequently induced by interaction with the hypnotist with whom the

subject is said to he in rapport. In scientology processing there is established between the preclear and the auditor such a bond or understanding, and scientology techniques are developed and designed to maintain this bond during the whole of the session; it is considered bad auditing if this bond is broken and techniques are prescribed for remedying the break.

In hypnosis, a degree of dependency develops and the expert practitioner is on guard against, and realises the potential danger of, this condition. In authoritative hypnosis this dependency is allowed to develop, often with harmful results. In scientology there is this same dependence which is allowed to develop without restraint. It persists after the auditing has finished and has significance in the desire of the preclear to return again and again to the HASI for further auditing. In the case of some witnesses a quality of almost desperate dependency on the HASI was observable.

A significant characteristic of hypnosis is what is referred to as the atavastic regression of the subject, "regression" signifying the going back to some previous event or circumstance, and "atavistic" connoting and pertaining to ancestry and referring to the losing or dulling of more recently acquired biological activities, so that the subject becomes less alert, less critical, and may become almost childlike, with heightened respect for the hypnotist, the development or intensification of rapport and a desire on the part of the subject to identify himself more closely with the hypnotist. Hypnosis may be induced by a great number of different procedures which initiate some degree of regression in the subject. Very many scientology procedures are designed to initiate this regression.

If command hypnosis is unskilfully practised, hallucinations which have been created during hypnosis persist later as reality. In scientology, "mental image pictures" experienced during an auditing session persist thereafter as reality and the preclear comes to believe that the past experiences and activities conjured up during these hallucinatory periods really took place, and so there is engendered a readiness to subscribe to the various scientology theories about past lives, the thetan and similar beliefs.

Frequently a preclear who in auditing has experienced hallucinations concerning murder, rape, homosexuality and other criminal and disgraceful behaviour comes to believe that such behaviour actually occurred during his present lifetime. This results in feelings of anxiety, guilt and self-loathing and a desire for confession and self-abasement, all of which increase dependency on and domination of the HASI. This position is to be contrasted with what obtains where passive hypnosis is used by skilled practitioners; in such cases, though the patient under hypnosis may be uninhibited and may experience distressing hallucinations, they are handled by the practitioner in such a way that, if recollected at all, they do not persist as realities, and beneficial results are obtained from competently administered hypnotic techniques. Furthermore, whereas in the professional use of hypnosis the objective is to bring to a conclusion a course of such treatment as speedily as possible, scientology practice is to prescribe more processing to deal with the hallucinations already experienced and bring to light fresh ones.

One characteristic feature of hypnosis is the increased suggestibility of the subject, of which advantage can be taken by the hypnotist In the state of regression found in hypnosis fantasies may be experienced which may be spontaneous or as the result of suggestion. To the subject these fantasies are apparently real and true experiences, and if authoritative hypnosis is used, these fantasies persist as reality. Preclears are highly suggestible and readily conjure up past life experiences of a kind and along lines suggested by the auditor and by what Hubbard has written. Hubbard finds much of the material for his "research" in these hallucinations which are quite fanciful and often contain details of "past lives". A striking illustration of the increased suggestibility of persons undergoing scientology processing is the helatrobus implant on which Hubbard worked

in 1963, and which was the cause of great excitement amongst scientologists. Hubbard wanted preclears to be run on the helatrobus implanting of the goal "to forget" between 38 and 43 trillion years ago, and many preclears in auditing sessions readily imagined weird things happening to them in the period stipulated, some of them giving to the second how many trillions of years ago they had had past life experiences. This, however, was after Hubbard himself had issued a bulletin stating that he had had such experiences. Afterwards some of the preclears were quite satisfied that they had been recalling true experiences in past lives.

It is recognized in hypnosis that repetitive commands and the exercise of other hypnotic techniques are likely to induce regression in which the psychological mechanism of repression is less effective; when this relaxing or lessening of repression occurs, matters in the unconscious mind are allowed into consciousness, and the subject may be very ready to discuss quite freely many intimate and shameful matters in respect of which the subject would be greatly or entirely inhibited if not under hypnosis. In scientology auditing, all these features are present.

In scientology there are many processes, including those which involve repetitive commands, which produce a lowering of barriers of restraint, a lessening of reticence, a readiness to talk unreservedly about the most intimate and secret things and past shameful experiences, and there are even scientology techniques designed to overcome, in an almost sadistic way, any reluctance on the part of the preclear to "withhold" anything. In scientology processing there is the same relaxing of repression and the same regression that is found in hypnosis.

One of the features of hypnosis is that various psychological mechanisms operate in a more florid form; thus, while in ordinary life a person may show little manifestation of hysteric behaviour, under hypnosis he is far more likely to show hysteric behaviour in a gross form. In scientology processing it is almost standard practice for the preclear to manifest some heightened hysterical features; a great many of the HASI files indicate that preclears have highly developed bouts of hysterical manifestation.

Post-hypnotic suggestion, which is an important feature of hypnosis, is the name given to the implanting during hypnosis of a command, belief or idea which is subsequently given effect to. Post-hypnotic suggestions may be made in relation to ideas, beliefs, attitudes of mind and the like which the patient is to assume after coming out of hypnosis. In scientology auditing, the auditor, in following the prescribed strict procedure for closing the session, inquires of the preclear whether the preclear has achieved his goals set for the session and any other gains and whether he is satisfied with the session. The auditor is still very much in control of the situation, for the preclear, being in a state of hypnotic rapport with the auditor whose wishes are in effect his, is more likely to answer that the goals or some of them have been obtained and that the session has been a success. This is a form of post-hypnotic suggestion, and after the session the suggestion that the session was a success may persist. This feature of auditing may well account for the " statistics" which Williams produced as the percentage of preclears which it was claimed had received benefit from scientology processing (see Chapter 20.)

Dangerous consequences may follow some post-hypnotic suggestions. If a post-hypnotic suggestion be given in hypnosis that the subject would not experience a particular symptom, e.g. a headache, after a session had ended, the subject might not experience a headache which normally he would have experienced, and thereby not be alerted to a possible medical condition, such as a brain tumour, of which the headache would have been a warning sign.

In hypnosis, a condition which is described by psychiatrists as "dissociation" may be experienced by the subject. This is a feeling or sensation or belief on the part of the subject that for the time

being he is outside his body. This is a complete delusion though it seems real enough to the subject who is experiencing it. If the processing is authoritative hypnosis, then the hallucination of having been outside one's body may persist after the session has concluded, and this may be dangerous to the mental health of the subject. In scientology auditing, a state which the scientologists call "exteriorization" is sometimes deliberately sought; in fact, exercises and procedures for exteriorization are the subject of a large part of Hubbard's instructional writings. This exteriorization, according to scientologists, is the actual departure of the thetan from the physical body to some position remote from the body. "Dissociation" and "exteriorization" are the same thing, produced by essentially the same means. Whereas in hypnosis, dissociation or exteriorization is recognized for what it is, namely, a feeling or sensation or belief on the part of the subject that he is outside his body, in scientology the preclear is specifically told that the hallucination which he experienced did in fact occur as a reality and that the thetan has been exteriorized. In such circumstances the harmful effects of scientology processing persist by inculcating in the mind of the preclear an entirely fallacious belief. A preoccupation with such beliefs, involving a refusal to face up to reality, may be dangerous to the mental health of the subject.

A command to "mock up" some object is a standard technique for the induction of hypnosis. Hubbard's writings, both in books and pamphlets, abound with descriptions of procedures which involve mocking up objects. A very great part of *The Creation of Human Ability*, a book of nearly 300 pages, and recommended reading, is devoted to the explanation of procedures which involve mocking up objects and / or exteriorization. Bulletins, policy letters, and other literature from Hubbard repeatedly deal with these two topics.

In hypnosis, it is not uncommon for the subject to experience disturbing hallucinations that relate to repressed things in his mind, such as hallucinatory homosexual experiences which a subject in his normal existence may never have experienced or entertained. Because of loss of repression these thoughts become known to him in a hallucinatory form, and he is likely to experience extremely severe anxiety even to the extent of panic and self-loathing. A subject who, in passive hypnosis, has experienced these or similar thoughts and may have had feelings of revulsion while under hypnosis, may safely be brought out of hypnosis and no ill effects will follow; on the contrary, benefit may result and feelings of shame will not persist. However, if similar hallucinatory and shameful thoughts are conjured up in authoritative hypnosis there may be dangerous consequences. In scientology, preclears have frequently complained of morbid feelings of guilt and depression persisting after auditing.

A further similarity between hypnotic processing and scientology auditing is the attention which both pay to the terminating of a processing or auditing session. A skilled hypnotist exercises great care in terminating a hypnotic session; he has to be satisfied that the subject is ready to be returned to a normal state from the hypnotized state. Too rapid a transition from one state to the other may have harmful effects. In scientology, there is as much strictness applied to terminating an auditing session as there is to the starting of such a session. The auditor brings the preclear up to "present time", usually running a "havingness" process for this purpose; he then enquires whether the session can be ended, and when he has the preclear's assent, he loudly proclaims "End of session" in much the same ritualistic way as he commenced the session.

The foregoing illustrations are sufficient to show that at almost every point there is a similarity, amounting almost to identity, between features of authoritative hypnosis and parallel features of scientology techniques. The dangers of the wholesale practice of these pernicious techniques cannot be over-emphasized. Not only does it constitute a very grave threat to the mental health of those already in scientology and in need of psychiatric help but it menaces persons who by

ordinary standards are quite normal but may find their way into the scientology centre merely out of ambition, curiosity or adventure. Being unaware of what is in store for them, they may easily succumb to the lure of being made more able, and shortly find themselves mentally crippled by the dangerous practices of ignorant operatives. This has already been the tragic fate of many.

The Anderson Report CHAPTER 19 THE HEALING CLAIMS OF SCIENTOLOGY

A very important phase of the Inquiry was whether scientology -

- (a) claimed to heal or cure ills or ailments,
- (b) practised the healing or curing of ills or ailments.

Williams and other executive scientologists from time to time asserted in evidence that scientology neither claimed to cure nor practised curing, that the practising of scientology to cure any ailment was forbidden and that auditors would be decertificated if they were found doing so or promising to cure any ailment. It was said that the curing of ailments had been a dianetic activity and that, because of troubles associated with non-qualified persons purporting to cure ailments, dianetics was put aside and scientology developed in such a way that specifically no claim was made to heal.

Certain bulletins and other documents emanating from Hubbard likewise state that the curing of ailments is not the role of scientology, but these are quite at variance with many other bulletins and similar publications, which describe scientology in terms which are consistent only with a claim that scientology can and does effect cures of the most remarkable kind.

The official attitude advanced at the Inquiry that scientology did not claim to heal was, and is, only a camouflage. The real intention of scientology is to inculcate in the minds of anyone who becomes interested in it the impression or belief that, as well as being a panacea for all problems, worries and aberrations, it is a gateway to sure cures for a great variety of mental and physical ills. And it is at the vary basis of scientology teaching that mental and physical well-being is assured to those who have sufficient scientology processing.

A fundamental principle of scientology is that a rehabilitated thetan is capable of caring for the body and the mind of the human being which it possesses and, therefore, if the thetan is rehabilitated and all its abilities are restored, it will be able to cure the ills of the body and mind. The only way to rehabilitate a thetan is said to be by scientology processing.

The belief that scientology is a cure for many illnesses, both mental and physical, is propagated by Hubbard consistently and deliberately. No opportunity is missed of claiming for scientology the credit of a cure, and in his books and other writings repeated claims are made and cases quoted of alleged cures, many of them said to be miraculous.

Campaigns are planned for enticing people into scientology because of what it offers in the way of alleviation and cure of illnesses. From time to time there is a protest by Hubbard and the HASI that they are not concerned with cures of bodily ills, and that their only goal is to make the able more able. But these protestations are hollow-sounding in the light of the calculated deception which is consistently practised in the drive for more and still more adherents to scientology. Sometimes the attitude adopted is, in effect, "We don't claim to cure, curing is not our concern. Of course, if you look at our record, look at the miraculous cures we have effected, look what dianetics, which is a branch

of scientology, can do and has done, you will see that scientology does cure; but we don't claim to cure". This is a cunning and effective pose, its effectiveness being best judged by the great number of ardent scientologists who firmly believe, as their evidence before the Board and a perusal of their files show, that scientology possesses remarkable curative powers, and that in many cases they are "in scientology" partly at least because of such beliefs.

Mindful of this technique, one well understands why Hubbard insists that his million copy best-seller, *Dianetics: MSMH*, is essential reading at all stages of scientology study, even though the book was written in 1950, before scientology was founded, and notwithstanding that dianetics was said by witnesses not now to be practised.

The claims for dianetics, as appearing on the dustcover of *Dianetics: MSMH* and in the text, include assertions that cases of "Psychosomatic ills such as arthritis, migraine, ulcers, allergies, asthma, coronary difficulties (psychosomatic - about one-third of all heart trouble cases), tendonitis, bursitis, paralysis (hysterical), eye trouble (non-pathological) have all responded without failure"; in fact, the claim made is that with the techniques mentioned in *Dianetics: MSMH*, "The psychiatrist, psycho-analyst and intelligent layman can successfully and invariably treat all psychosomatic ills and inorganic aberrations," and that it can cure 70 per cent. of all man's illnesses and aberrations.

In various places Hubbard has written to the effect that arthritis, eye conditions, heart conditions, cancer, all psychosomatic illnesses, morning sickness, ulcers, tuberculosis, the common cold, the common cough, illness from bacterial or virus infections, alcoholism and a multitude of other complaints and conditions are engramic and respond to processing.

In Com. Mag. Vol. 1, No. 10, the official publication of the Melbourne HASI, the statement is made that " All that you have read about in scientology and dianetics can be achieved by you with processing from the Hubbard Guidance Centre".

In Com. Mag. of December, 1963, Hubbard writes,

"The psychologist could not change intelligence quotient or personality at will. The Scientologist can. The psychologist could not restore sanity and happiness to the insane. The Scientologist can Scientology will inherit the hospitals, the clinics, the asylums, the halls of learning where humanity was abused. Scientology will inherit the task of sign-posting Man upon a better road. There are only two reasons why this is so: they had their chance and did not do their job; we have our chance and are doing ours".

That Hubbard and scientology specifically claim to heal, and attract people in the belief that they do heal, is obvious from HCO Pol. Lr. of the 1st September, AD 12 (1962), entitled "Healing Promotion". That letter reads:

"By healing you can graduate a pc up to clearing interest and thus we have a lower level feeder line, capable of successful accomplishment with normal HCA/HPA training. That programme has the following thought major: Maybe you're not sick. Maybe you're j ust suppressed. See us and find out.

"The phrasing can be more elegant, the message remains the same.

"Legally, this permits us to heal without engaging in healing as, in actual fact, we address no illnesses and indeed, deny people are ill - they are only suppressed. Sickness occurs, we say, where suppression has been too great. The argument is - have you been sick? Did you go to doctors to be cured? Did they cure it? Then (as they didn't) maybe you're not sick, maybe you're just suppressed. So take some processing and find out. And the person gets well! We use on him the exact button he came to us on. S o he's never dismayed at any change of tack on our part. Then we interest him in clearing.

"This, I am sure, is the long sought gradient. This, used right, will build our new buildings, use our Academy Graduates and give us a chance to train up auditors to clearing.

"The legal argument is simple, we don't believe in sickness, we do not address illness, we do not diagnose, we believe that freeing the human spirit also incidentally prevents sickness. We are doing prevention. We also find people do not have to be crazy to be suppressed, that nearly everybody is suppressed. We do send acutely ill people to doctors. We advertise to cure no diseases! That last is important legally. We only infer that people who think they are sick are really not, but only suppressed."

A circular issued by the Melbourne College of Personal Efficiency, the Hawthorn scientology organization, claimed

"Tens of thousands of case histories (reports on patients' individual records) all sworn to (attested before public officials) are in the possession of the organization of scientology Scientology in the words of an expert can cure 70 per cent. of Man's illnesses It is valid. It has been tested Scientology does things for people which have not been done before. It makes them well from illnesses which were once considered hopeless. It increases their intelligence by actual measurement, it changes their competence, and betters their behaviour. In addition to these things it brings them a better understanding of life. One outstanding thing which it does - it alleviates burns received from atomic bombs. Scientology is the only specific cure for atomic bomb radiation flash burns. Scientology processing given to persons burned by radiation can eliminate the majority of difficulty."

HCO Pol. Lr. of the 12th October, 1962, states that the purpose of the Hubbard Guidance Centre is to "do more for people's health and ability than has ever before been possible."

A letter dated the 6th October, 1962, from the Melbourne HASI to a preclear contained a specific offer to treat physical complaints, in the following terms, "If you suffer from physical complaints which have not been cured by usual medical methods--why not call in and let us find out if you are just suppressed? We have special methods of handling such difficulties.

In HCO Pol. Lr. of the 2nd August, 1963, Hubbard wrote,

"Scientology for use in spiritual healing. This is a healing strata The target is human illness. We have never entered this field but as we are not thanked for staying out of it, we might as well dominate it. It is a good procurement area."

The claims made for curing are so numerous and consistent that it is impracticable to refer to all of such instances but a selection will indicate the general nature of them.

In *A History of Man*, Hubbard wrote: "Cancer has been eradicated by auditing out conception and mitosis". In *Scientology: The Fundamentals of Thought*, Hubbard claimed that "Scientology is the only specific (cure) for radiation (atomic bomb) burns". In *Scientology*: issue 15-G, Hubbard writes,

"Leukaemia is evidently psychosomatic in origin and at least eight cases of leukaemia had been treated successfully by dianetics after medicine had traditionally given up. The source of leukaemia has been reported to be an engram containing the phrase 'it turns my blood to water'."

The 1957 edition of Hubbard's *Scientology: 8-80* contained a new introduction by a "doctor of scientology", stating, "how an auditor trained in this material by the Academy of Scientology can handle with precision even the insane or a few days' old baby"; and the editorial note reads,

"The discovery and isolation of Life Energy in such a form as to revive the dead or dying has been an ambition as old as Man himself. In the last two thousand years a few individuals have claimed the ability without explaining it. With this book, the ability to make one's body old or young at will, the ability to heal the ill without physical contact, the ability to cure the insane and the incapacitated, is set forth for the physician, the layman, the mathematician, and the physicist."

In the early days of scientology in Victoria, advertisements by the HASI appealing for asthma sufferers were published in the press. That scientologists generally believed in the healing powers of scientology is evident from the facts

- (a) that many preclears set as their goals, to be obtained through processing, physical health and well being, and
- (b) that auditors consistently audited preclears in an effort to improve their health.

One scientology witness who was for a time the director of processing at the HASI believed scientology would aid his deafness. Various scientologists believed that scientology would improve their eyesight or their hearing. Others believed that it would cure stomach pains. One dedicated scientologist did an HPA course in order to be able to audit and, so he believed, cure his father who was suffering from Parkinson's disease. Hubbard wrote in *Science of Survival* that a process called "straight memory" had alleviated Parkinson's disease. One practising scientologist woman, whose daughter was a highly placed scientologist, believed that had she been processed in scientology for a bad arm, a cancer for which she had received medical attention would not have developed.

Individual scientology witnesses left no doubt that they believe that scientology could cure physical ills. Highly placed scientologists believed that well trained auditors could handle all psychosomatic illnesses and that processing could proof a person against polio, hepatitis, malaria, &c. Mrs Williams believed processing cured a cancer of the abdomen. Gillham thought that processing could cure blood pressure cases. His wife thought that illnesses such as colds, 'flu and heart trouble and some mental disorders could be cured by scientology, and that it was possible to proof against the strain of physical and mental illness.

During 1959-1960, over a period of about five months, one unfortunate man was audited at the Melbourne HASI for more than 200 hours in an endeavour to cure him of cancer from which he was dying and did die. At a time when it was known at the HASI that he had been under medical treatment and was suffering from a malignant growth in his lower abdomen, the HASI quoted him 200 hours' auditing for a stable case gain. He embarked on such a course, and processing ceased only a few weeks before he died, one of the last auditor's reports on his case being that the preclear was about to drop the body. This case is typical of the callous disregard which the scientologist practitioners are trained to have for their preclears. It is a sign of weakness in an auditor to feel pity for a victim, and the auditing processes in this case, as in very many others, were applied guite brutally. During the five months over which this preclear's processing was spread, he had such pitiful goals as "to be certain that I have lost This tumour", "to get rid of this tumour", to "drop off this tumour-cancer-growth", "to make my stomach a bit more comfortable", "to reduce this growth in size", "my health to start to improve, to find the cause of this growth quickly". With these and similar goals, the processes on which he was run included the following and other similar commands, which would be repeated time and time again, for periods of 2 to 3 hours at a sitting, sometimes for six hours a day, and sometimes extending over several days: "What stomach could you confront?", "What stomach would you rather not confront?", "Think of a stomach you could confront, think of a stomach you would rather not confront", "What part of a stomach could you be responsible for?", "What about a victim could you be responsible for?", "What could you admit causing a victim?", "What could you withhold from a victim?". It is morally certain that this preclear was accepted for this processing only after the HASI had consulted Hubbard or Mr. or Mrs. Halperin (or similar name), all of whom were then touring Australia, the Halperins being extremely close associates of Hubbard.

During this preclear's auditing, he suffered severely from "somatics" or pains in various parts of the body, frequently "doped off" and had hallucinations about murder, rape and other acts of violence, as well as recalls of death and hanging in past lives. At one stage towards the end of this unhappy life, his condition inspired the endorsement of his file: "Good, pc is coming up to apathy", the

significance of which is appreciated by reference to the tone scale, apathy being the lowest of the emotions on the scale at 0.05, and only slightly above death.

HCO Bull. of the 24th July, AD 10 (1960), contained "Special Project Australia", which was reprinted in various scientology magazines. In a campaign to publicise scientology and discredit the medical profession, Hubbard wrote,

"This is our offered programme:

- "(1) In every letter, communication and broad statement, insert as well the fact that Scientology works. Scientology is the only fully validated science of human behaviour. Scientology is a good, reliable science. Any positive statement. Always add it to anything else we say.
- "(2) No longer wait for permission from the government or anyone else to take a positive action to help our fellow man overcome his susceptibility & sickness;
- "(3) Let us make of every man, woman and child in Australia a 'disease-proof person'
- "... It is within our power to proof Australia against mental and physical illness... You can advertise all you want to 'eradicates disease proneness', to 'proof Australians against illness' since all law applies to healing sicknesses, and could never be extended to preventing prevention."

And the form of advertisement he recommends is: "Prevent illness. Scientologists are seldom sick. Join a Scientology group and be able".

The HASI repeatedly claims to be the "World's largest organization in the field of mental health". Receipts issued by the Melbourne HASI carry the statement "Scientology proofs people against mental and physical illness".

The Anderson Report CHAPTER 20 ALLEGED BENEFITS OF SCIENTOLOGY PROCESSING

Scientology claims to make the able more able. In Chapter 3 are set out many similar claims made by Hubbard and his organization in book, pamphlet and advertisement. Scientology claims to be the science of knowing how to know; in very recent times it claims to be the science of learning how to learn. Though it was asserted by several scientology witnesses that scientology made no claims to heal, its literature is crammed with claims that it and its companion "science", dianetics, can positively cure a multiplicity of physical and mental ailments. The healing claims made by scientology and dianetics are dealt with in Chapter 19.

Many active scientologists believe they have obtained benefit from scientology processing. This is not proof that in fact they have obtained benefit, because processing and training result in the dulling of one's critical faculties, and are specifically designed and conducted to create in the mind of the preclear the very delusion that benefit has been obtained. Apt parallels are the drug addict and the inebriate who, in their bemused state, have feelings of well-being.

Early in the Inquiry, Williams gave evidence that, on the HASI statistics, 96 per cent. of those who had received processing at the HGC had obtained benefit therefrom. The manner in which this figure was arrived at seemed unsatisfactory at the time the evidence was given; after hearing evidence of several witnesses who claimed to have benefited and after studying files of preclears who have had processing, the Board is satisfied that almost all the benefits claimed to have been received from processing are quite illusory and this optimistic figure cannot be accepted. The degree and the nature of the benefits do not appear in the statistics, and as the basis of determining whether there has been a benefit is primarily the preclear's subjective appraisal at or shortly after the end of a session, any statement which, at that stage the preclear may make as to benefits should be heavily discounted.

One of the characteristics of scientology processing and training is that a preclear develops a great degree of dependency on the auditor and the HASI. This result is deliberately sought by Hubbard. At the end of a processing session a preclear may be in a highly emotional and exalted state and temporarily so enthusiastic and dependent on the auditor that his own assessment of his own "case gain" affords no guide to the true position. The assessment of case gains is a self-deluding process, for the auditor is aiming for a gain in the preclear and it requires but little skill on the auditor's part to suggest to the highly susceptible preclear the auditor's desire for a gain, with which suggestion the preclear obligingly complies. A preclear's confidence in his auditor generally enables the auditor to audit until he gets a gain before he ends the session. The duration of the gain does not matter. In the demonstration sessions which were staged for the Board's information, one marked feature was the evident desire of the preclear to please the auditor. Even if a preclear did not feel there had been a gain, the innate politeness of a patient to a practitioner would generally prompt the patient to concede some benefit. The figure of 96 per cent. was not compiled on any scientific basis, and was not capable of being checked or verified. It was an inherently improbable figure and the subsequent evidence made it quite obvious that the figure could not be accepted.

The Board is satisfied that, if there are any cases in which a preclear has obtained any real benefit from scientology processing, the number of such cases is negligible and that such benefit has

occurred in spite of scientology processing rather than because of it. When speaking of "real benefit", the Board does not include within that expression any temporary apparent benefit which requires continued processing to maintain the preclear in such a state of "benefit".

The feeling of benefit, of being better and happier, is akin to the elation following the taking of a drug. It is common experience in psychotherapy for a patient readily to acknowledge that there has been a gain, even though the practitioner is satisfied that such a gain has not been obtained or is illusory. The psychotherapist, if properly trained and aware of the danger and delicacy of the position, is at pains to guard against the too ready "improvement" which results from the patient's dependency on the practitioner and his desire to please. The degree of suggestibility is high in all forms of psychotherapy; in scientology auditing the same situation obtains, probably to a greatly accentuated degree because the auditor is not trained in any real sense, is unaware of the implications and pitfalls, is blindly confident that the preclear will experience a gain by being processed and is, unwittingly perhaps but nevertheless wholeheartedly, using pernicious hypnotic techniques where all the risks of doing harm are greatly intensified.

A number of witnesses gave evidence of benefits they had received from scientology processing, and it seemed they believed it was scientology that was responsible for such improvements in their lot. Yet, in many cases, there was a natural and normal explanation for a change in their outlook on life. Thus, there was the University student who was so shy that she used to eat her lunch in the wash room; she took up scientology, married and thereafter found life far more pleasant and less frightening. Another witness, a railway employee stationed in an outback Western Australian settlement, on the advice of the HASI in Perth, came East and found that his talent lay in salesmanship in which he prospered. A worried factory foreman found that promotion brought with it less worry; he attributed the lessening of worry to scientology. A man with a poisoned leg attributed its cure to scientology processing and ignored the fact that he was currently receiving medical treatment. A man who was a University graduate claimed that it was scientology that cured his "golden staph" infection because he had taken 20 hours' auditing at the end of which he had had a "cognition" to see a doctor, which he then did, and was cured. A farmer prospered on taking over his parents' firm which he worked more successfully by using modern methods but he attributed success to scientology. Several witnesses gave evidence of improvement in such general terms as being able to get on better with people, being able to handle problems more confidently and successfully, being able to do housework more quickly, being able to do one's job better, feeling more confident, &c. The benefits claimed to have been experienced were subjective to the witness concerned and incapable of proof by a demonstration.

The attitude of these witnesses varied. Some were almost defiant in their assertions that scientology had benefited them; others had a brittle and, it seemed, forced cheerfulness. They all answered with assurance the questions asked them, even though their answers were sometimes nonsensical; they showed considerable mental agility in parrying cross-examination and they made repeated use of scientology phrases to their own satisfaction, but often their answers lacked logic and were sometimes inane. Gillham, for instance, said that time could be created in scientology by a person, using scientology, doing in half an hour a task which had formerly taken him an hour to do: in other words, he said, half an hour of time had been created. These witnesses seemed to think that the stock scientology answer necessarily proved their viewpoint. Some were currently undergoing auditing or had been recently audited, and many had received scientology training. All were still in communication with the HASI, and were eager to vouch for scientology, even though their evidence in several instances was directly in conflict with the contents of their processing files.

Many of these files indicated that auditing for them had not been an unalloyed joy: on the contrary, it

appeared that many had had more than one period of processing, the second and subsequent periods being sometimes designed to repair the effect of certain earlier processing which had produced undesirable results, or to restore the "benefits" of earlier processing which had worn off. Of course, the aim of processing was to produce in the preclear a feeling of benefit, but therein lay its great danger and potential evil, for the effect of processing was often to produce in the preclear a feeling or quality of elation with nothing but delusion to sustain it and when the effect of the auditing wore off the preclear lost the belief that he was " able" and became the victim of disillusion and despair.

The danger of relying on the opinion of a preclear or his auditor that he has obtained benefit from auditing was emphasized by evidence in three specific instances. In one case, a male preclear was processed to a state of great enthusiasm, and three days later was apprehended as a violent and raving lunatic. In another case, the Board witnessed the processing of a female preclear in a demonstration session, at the close of which she stated she had had gains, yet nine days later she was admitted to a mental hospital. The third case was that of a male preclear, obsessed by the mystery of his parentage, who was processed to such a stage that half way through the course he suddenly went interstate possessed of an almost ecstatic resolve to do something about clearing up the mystery, and on returning to Victoria had to be taken into custody and was committed to a mental hospital.

At an elementary stage the personal efficiency course may well have given to some preclears a degree of confidence in themselves, thereby enabling them to confront and cope more successfully with their everyday problems by the realization that their own ability and initiative, properly directed, were adequate to deal with everyday tasks. Had the tuition stopped at that point, where the preclears had been made aware of their own capacity to cope with ordinary problems, there would be perhaps little criticism. But the purpose of the personal efficiency course was not to give people benefit as such: it was, as Hubbard had directed, to lure the unsuspecting person into the web of scientology which was being spun around him by a combination of conventional techniques and scientology mystique.

There exist many establishments, commercial as well as voluntary, where people are encouraged to develop their dormant or ill-developed skills, and where confidence in oneself and how to overcome "complexes" and adjustment to surroundings and to people generally are taught. Such establishments are teaching people to overcome inhibitions which are handicapping them socially or in business or employment. They use well recognized means which acknowledge that the abilities which are being developed are innate to, and involve the development in, the individuals concerned of an awareness of their potential powers and a justifiable confidence in themselves. In the personal efficiency course, something of this nature is dabbled in, not, however, as the end, but as the means of deliberately trapping people for further processing and training.

It was apparent that several of the witnesses who claimed to have received benefits from scientology were looking forward to further processing to sustain the illusion of well-being. Repeated bouts of processing were necessary to maintain the attitude of blindness to reality which preclears appeared to have. They were sustained in their elated condition by the acceptance of the theories which Hubbard taught as to thetans, past lives, implants, immortal destiny and the ultimate goals of being clear and OT. They were as uncritical of the unreality of these things as they were of the weird and fanciful world in which scientology allowed them to move. Indeed, they are the tragic figures of scientology, these unfortunate creatures whose apparent happiness and confidence are resting on such uncertain props, for when the props go, as eventually they must, the realization of the extent and nature of their deception will be quite shattering.

Scientology offers a ready refuge for those who are uncertain of their future and who are afraid to face reality and cope with the problems of life. Psychiatrists recognize that there is such a thing as a healthy anxiety about one's future, for such an anxiety keeps a person face to face with reality and spurs him on to the attainment of goals normal to any member of the community according to his environment, be he student, artisan, executive or professional. Scientology introduces the preclear to a make-believe world where the sense of urgency is lost and where he develops the delusion that, because he is in scientology, he will be able to achieve by postulation what otherwise he could acquire only by study or hard work outside the fold of scientology. Along this pathway lies disaster.

Typical of the abandonment of reality is the case of one preclear who was seriously worried about certain aspects of his business and who undertook processing for the purpose of finding the solution to his problem. His attitude of mind when he emerged from processing was that the problem was not worth worrying about. This attitude resulted in his neglecting the problem.

Prolonged and anxious consideration of the evidence as to whether any real benefits result from processing by scientology leads the Board to the firm conclusion that virtually no benefits attend upon scientology processing and that such benefits as are claimed are in the large majority of cases, illusory and of short duration. On the contrary, the harm done by scientology processing is frequently serious and at times tragic.

The Anderson Report CHAPTER 21 GENERAL EFFECTS OF PROCESSING

Short Term Effects.

During an auditing session, a preclear is under the control of the auditor whose objective is to be in command of the situation and keep the preclear's mind rigidly on the subject matter which is being probed by his repetitive questions and commands. In Chapter 18, reference has been made to the hypnotic aspects of auditing.

Preclears react in various ways to auditing. Though the effect of processing is to develop in the preclear a degree of dependency upon scientology and the auditor, that dependency is on occasions allied to fear and hate of the auditor because of the strain which the repetitive questioning on matters of a harrowing nature imposes on the preclear. Sometimes during processing there are periods of ecstasy and exhilaration but these quickly wear off. Sometimes the preclear exteriorizes; frequently the preclear has hallucinations of the most fanciful kind which may persist as realities after the session has concluded.

Scientology auditing is of a confessional nature. The preclear is encouraged to reveal his innermost secrets and thoughts and to describe in detail all features of anything in his past life - not necessarily a previous life - which is troubling or has troubled him. Such a procedure involves the dredging up from the almost forgotten and ill remembered past of shameful thoughts and experiences. The questions are designed to ensure that no phase or feature of any guilty circumstance in the past is left unexplored: processes such as the "overt withhold" scour the preclear's mind with questions like, "Are you withholding anything?" Each aspect is savoured, appraised and dwelt upon.

There is inculcated in the mind of the preclear a belief in the infallibility of the E-meter with the result that, should there be a "read" on the meter, it is assumed that all has not been told, and the preclear is required to probe and dredge his recollection to find some explanation for the "read". Such a course often results in the preclear developing feelings of guilt in respect of matters which are guite innocent.

The evidence and the processing files reveal that, during this time, the preclear is very frequently experiencing mental torture, which shows itself in contorted and flushed features, tears, moaning, inability to speak, apparent deafness, nausea, dizziness, sensations of pain, coma and unconsciousness. One witness said that he almost killed his auditor, a close personal friend, who was questioning him about withholds he had as to "sexy thoughts" concerning a female staff member; the witness said he aimed a judo blow at the auditor and checked himself just in time to avoid inflicting injury. This evidence was corroborated by the contents of his processing file. Sometimes preclears are so distraught that they scream, develop murderous feelings, have bouts of anger, grief and morbid feelings and thoughts; their sexual passions are aroused, they act insanely, laugh hysterically and engage in other hysterical behaviour; they become violent and try to escape and have to be restrained. Such behaviour is not unexpected by the scientologists; see Chapter 13, dealing with the S.C.S. process.

In scientology parlance, when such manifestations as these occur, the preclear is being "restimulated"; in fact, he is being debased and mentally crippled. The scientologists have their

own names for these phenomena: for instance, coma and unconsciousness are variously described as "anaten" or "analytical attenuation", "dope-off" and "boil-off", the last-mentioned being described in *Scientology: 8-80* thus:

"It is not uncommon for the preclear to go into periods which resemble sleep but boil-offs are not sleep. They can last for hours The sudden wave of unconsciousness occurs because the preclear has run an outflow or inflow of energy beyond the limit of elasticity of the flow".

In fact, the intensity of emotion produced by the auditing has resulted in a condition of mental exhaustion.

Scientology auditing mobilizes guilt and can produce dangerous mental attitudes in persons who are suffering from anxiety. A distorted desire to debase and torture themselves develops in some preclears and they assume attitudes of guilt over conduct which is often innocent and often imagined. They are sometimes euphoric and excoriate their minds. They develop a morbid desire to dwell upon their guilty past, like the witness of sheltered upbringing who was seduced and thereafter found solace in scientology by being repeatedly processed on aspects of her shame; and like the witness, a young man, whose acts of masturbation subsequently weighed on his conscience, who was repeatedly processed at about six monthly intervals about the habit long since discontinued and far better forgotten. The files of the HASI contain very many cases where a guilt complex in respect of the past actions of preclears has not been allowed to lapse but has been kept alive and the incidents relived in courses of auditing which are sometimes spread over many months and even years.

At the end of a session or a course of auditing the preclear may feel depressed, fatigued or ill; or he may feel elated with a completely unfounded belief that he is possessed of increased abilities of a very high order and that he is capable of doing almost anything if he really wants to - of course, he does not want to, but he is buoyed up by the belief that he has only to decide to do such things and he would be able to do them, for, so he thinks, he has been made more able.

Should the preclear appear to be depressed or not have gains or the processing not produce the desired result, the verdict of the director of processing would be that more auditing was required to improve the tone of the preclear. On the other hand, if the processing appeared to have been successful, the verdict would be that the preclear was in good shape which was a good sign, indicating that he ought to have more processing to put him in better shape.

Some preclears have said that they have been "clear" for a time after auditing. However, the feeling of being clear wears off, as do other feelings of benefit and exhilaration. On the other hand, Hubbard admits in HCO Pol. Lr. of the 10th October (year not given but probably 1960), that there have been several instances of persons taking processing who had "nervous breakdown" and were sent to an asylum for "observation".

Long Term Effects.

Beyond the elementary stages of scientology, almost invariably processing and training go hand in hand, and their combined effect usually conditions preclears to believe that scientology possesses almost magical powers. The bizarre idea develops that only in scientology is the truth to be found, that the rest of the world is a place of insanity and that the HASI is a little island of sanity; that outside scientology is a dangerous environment, that in scientology it is always safe. Scientology practitioners become imbued with the idea that they are creating a new race of men, the homo novus, which is to take the place of the homo sapiens, for whom they have great contempt. The phrase "homo sapiens" is sometimes abridged to "homo sap", with emphasis on the "sap".

Preclears are flattered by being informed by the HASI that they have been chosen - the implication is by Hubbard - to be amongst the ten thousand to be cleared, and they respond to the summons to attend for more processing by being told such things as "Ron wants a clear in your area". The service of scientology becomes an end in itself; some business men who have espoused scientology have endeavoured to introduce its practice to their staff and to require that their executives and others undertake processing; scientologists endeavour to procure members of their families to take up scientology, and family opposition is interpreted as an indication that the objector has some guilty secret which he is afraid scientology will find out, and so unhappiness and mistrust are engendered. Any degree of mental imbalance is accentuated, and even the scientologists themselves recognize a condition in which a preclear, after auditing, may exhibit exaggerated symptoms of cognition, as where he claims a too ready understanding of scientology principles in order to be thought more acceptable. A person so conditioned is said to be "theety-tweety". Preclears become even more suggestible, their critical faculties are suspended and they readily accept - or at least do not reject - nonsensical ideas because they emanate from Hubbard.

Many scientologists impoverish themselves by their exclusive preoccupation with scientology either on the staff of the HASI at submarginal remuneration or by spending large sums of money, far beyond what they can reasonably afford, on deleterious processing.

This first part of Chapter 30 deals in detail with the likelihood of professional scientologists and other persons who have undergone sustained scientology processing and training being permanently affected.

Individual Cases.

To the outsider there are aspects of scientology which may be highly amusing and diverting. The outsider laughs at the thought of people believing in past lives and in the fantastic experiences of trillions of years ago. However, the Board has heard evidence which satisfies it that almost every case is one of unrelieved tragedy, for these aspects demonstrate that the preclear has been "brainwashed" and conditioned to the extent of accepting such experiences as true.

A few of the very many cases which illustrate this may be briefly mentioned. A constable of police with many years' service and good prospects of promotion became a scientologist and was given the role of liaison officer with the Victoria Police Force; certain of his suggestions that scientology methods be adopted by the Force were rejected, and he resigned from the Force. With his wife and family he is now living in a country town, on the verge of destitution, doggedly seeking to preach and practise scientology. One woman who successfully conducted her own school introduced scientology into the curriculum; the evidence makes it clear that the parents objected to the techniques which she was applying to the children and the school was forced to close. Another woman, a de facto wife, who had had a somewhat colorful earlier life, sought solace in scientology processing but without success, and in the course of her endeavour spent £800 on scientology and is further in debt to the HASI to the extent of £250; she had a nervous breakdown. There was the young married couple, the husband believing he was so high on the tone scale that he did not need sexual relations with his wife whom he considered too low on the tone scale for such relations, anyway. There were several cases of families where one spouse was a scientologist and money needed for household purposes was being squandered on scientology to the detriment of the household. One of the few professional people who gave evidence on behalf of scientology was a woman medical practitioner, who, in 1957 declared that she intended to be clear the following week, but six years later and £2,000 lighter, in 1964, she was being processed for three half days a week, her goal still being to be clear. A University student, who after two visits to the HASI was not further interested in scientology, was pestered by letters warning him that the problem which the HASI imagined him to have was an acute one and that he needed scientology processing. A young woman, highly placed in scientology, believes that her future is a bleak one because, due to her devotion to her 4-year-old son whom she will not leave, she is denied what she believes to be the sublime bliss of going to Saint Hill. A man who has had 700 hours of auditing for which he paid £3,000, his goal being to be clear, is still not clear. A young man aged about 23 years, a dedicated scientologist who claimed some improvement to his eyes through scientology, who was earning between £6 and &;163;7 per week, was paying £5 per week back and hoped that he would be able to go to Saint Hill. There were several cases of persons who had sacrificed prospects of good careers to become auditors or members of the staff of the HASI. These included a number of school teachers, a surveyor and people with various technical qualifications. Several preclears abandoned their studies to concentrate on scientology. A former University student, while doing his course, became a scientologist; he spent a few years in scientology to the detriment of his course, which, however, he ultimately resumed when he cast off scientology and he obtained his degree. Scientologists seek to claim credit for his having obtained the degree, but he now considers that scientology was for him an unqualified disaster. It is not improbable that scientology contributed to his matrimonial trouble. A witness before the Board, a young country girl who was lured to Melbourne by the promises of scientology, was endeavouring to break with scientology but was constrained to stay for reasons, amongst others, that she feared that those who opposed scientology would die. A young man in auditing had a hallucination of having killed a woman and buried the body and he was greatly worried and puzzled subsequently because he had not been guestioned and the police were not making any enquiries. This young man was quite certain that he had been Mark Antony in a previous life. He remembered Cleopatra, but not the battles of Phillippi or Actium because, as he said, he had not read much history and had had an "occlusion" on some parts of his past lives. He had also been, so he claimed, Chopin, and though born in the 1930's, he claimed to have met a man in this lifetime whom he had known in Paris in 1924. The lifetime which he was then experiencing came to a sudden and dramatic end in 1936, when he was executed in Germany by a firing squad. He was born into this lifetime in the following year.

One scientology witness said that she had met a man in this lifetime whom she recognized as Simon Legree, whom she had met in a previous lifetime when she had been John Brown. She said, however, that she had not met Uncle Tom, as he was merely a fictional character.

One man had been a pirate chief whose crew had mutinied and killed him in 1813. Another man recalled being on a ship which was involved in a collision in the North Sea in February, 1917. He was on the bridge at the time, dressed in tropical rig. He went down with the ship and was lost. He was born into this lifetime the following year. In a previous lifetime another young man had been a space jockey engaged upon a journey in a space craft. He had a doll body at the time, "a body very similar to the humanoid body but not made of flesh, sort of mechanical". He was on a journey of 6,000 light years from Euripodes to Helatrobus, both of these being planets, but in what constellation it did not appear. On this journey he went elsewhere and got lost. He reported to a space controller who happened to be the space controller of the planet Helatrobus and he was sent to an implant station of Helatrobus which was Heaven, where he was stuck on a pole for five days and then entered a round vestibule and climbed up marble stairs and was implanted, apparently by electronic charges emitted by little boxes at ankle height. At one stage, about 75 trillion trillion years ago, this preclear's thetan had a goal to be sexual. The explanation for such a goal was that at that time his thetan had as its body a doll body which was sexless because it had not been reproduced as human bodies are, but had been manufactured by a machine. This same preclear had had experiences 48 trillion trillion trillion trillion trillion vears previously and had experienced bear implants and black thetan implants in a glade.

A married woman, with two children to support with only occasional help from her husband, for several years was herself working five days a week and was then occupied by scientology study, auditing and doing training drills all day Saturday and frequently Sundays. This woman over the years had goals which included "to find courage to ask for money for [name deleted]'s auditing, to be clear, to have a slim body, to increase havingness, to get a better reality on figures", "to get mum and dad into scientology", "have a clear outlook", "have gains in session", "win confidence", "lose mass". One auditor's note stated that the preclear complained of mass in her head, and was concerned about "decayed thetans". This woman spent some hundreds of pounds on auditing, on occasions borrowing from her parents.

They were people with "problems", anxieties about particular matters, uncertain, introspective, seeking some solace or escape, groping for something certain on which to rely. They were unable to cope with their problems as the normal person bears and copes with his problems. They were looking for something which would protect them from the realities of life. Some had tried a succession of religions. In scientology they now found the answer to everything, for they could avoid reality for the time being, as scientology allowed them to believe with conviction anything they wished. Others who had sought to lose themselves in study found in scientology a respite from formal study which offered a short cut to learning and was fascinatingly diverting and seemed to be so far in advance of conventional learning and so easy to cognite upon. Some were people with specific mental illnesses. One was a manic depressive who for years refused to have psychiatric treatment and took drugs instead; he found that he could, at the cost of many hundreds of pounds and economic embarrassment, for short periods be processed out of his depressed state. He considered that scientology solved his problems at work for when he had a problem which seemed insoluble he had on occasions looked through the window at a bright light like an illuminated sheaf of wheat which then exploded and he found the problem solved.

There were some instances where benefit seems to have been obtained but these were generally cases in which such benefit had been obtained at the personal efficiency level, where the procedures were mainly of a conventional psychological nature and efforts were made to give to the HASI the atmosphere of a club or association for mutual and friendly uplift and co-operation.

The purpose of scientology processing is to keep the preclear permanently enthralled. The brainwashed, hypnotized preclear does not advert to Hubbard's broken promises of infallible processes. Hubbard attributes to his auditors failure to process preclears to clear; but auditing makes a preclear feel that the deficiency is in him, and his bewildered mind is a shuttlecock between the despair which follows one course of auditing and the ephemeral elation which follows the next.

The circumstance that those who fall victim to scientology's lure are often the mentally disturbed or weak or greedy or curious is no occasion for complacency. That such victims act voluntarily when they embark upon scientology is no reason for polite non-interference in other people's affairs. Indeed, the fact that so often the weak and mentally disturbed are so vulnerable affords adequate reason for taking steps to guard them and others from the dangers inherent in scientology processes.

The Anderson Report CHAPTER 22 HOSTILITY TO MEDICAL PROFESSION

One of the great dangers of scientology is that it poisons the minds of its followers against the medical profession and generates an abhorrence of medical treatment generally, and psychiatric and psychological treatment in particular.

Hubbard seldom refers to the medical profession without using the occasion to attack and vilify it in the most intemperate manner. He has an irrational obsession about the techniques of psychiatric treatment. He is completely intolerant of opposition or criticism, and he resorts to almost incoherent, hysterical, low grade abuse whenever he believes himself or scientology to have been attacked and often when no such occasion exists. His special targets are psychiatrists and psychologists, whose realm is the mind. Concerning psycho-surgery and ECT, which have their proper use in the successful treatment of the mentally ill, Hubbard makes a number of completely untrue and unjustifiable statements.

Typically Hubbardian inaccuracy and falsity abound in his attacks upon the medical profession and its techniques. Examples of Hubbard's hostility which follow are given to illustrate the nature and degree of his intemperate abuse, and the imbalance of the mind which produces it; these examples are not to be taken as being statements which are even vaguely related to fact.

In *Dianetics: MSMH*, in an introductory chapter, Hubbard seeks to frighten the reader away from orthodox medical treatment by the following extravagant passage: -

"According to a modern writer, the single advance of psycho-therapy was clean quarters for the madman. In terms of brutality in treatment of the insane, the methods of the shaman or Bedlam have been exceeded by the 'civilized' techniques of destroying nerve tissue with the violence of shock and surgery, treatments which were not warranted by the results obtained and which would not have been tolerated in the meanest primitive society, since they reduce the victim to mere zombyism, destroying most of his personality and ambition and leaving him nothing more than a manageable animal. Far from an indictment of the practices of the 'neuro-surgeon' and the ice-pick which he thrusts and twists into insane minds, they are brought forth only to demonstrate the depths of desperation man can reach when confronted with the seemingly unsolvable problem of deranged minds."

The following passages from a later chapter in the same book are in similar vein:

"The auditor should be extremely cautious, at least for the next twenty years, about any case which has been institutionalised, for he may be getting a case with iatrogenic psychosis - caused by doctors - in addition to the patient's other engrams. Dianetics may help a mind a little in which the brain had been 'ice-picked' or 'apple-cored', but it cannot cure such insanity until some clever biologist finds a way to grow a new brain. Electric shock cases are equivocal: they may or may not respond to treatment, for brain tissue may have been burned away to a point where the brain cannot function normally."

"The 'tests' and 'experiments' with human brain vivisection in institutions are not, unfortunately, valid, For all the pain and trouble and destruction caused by these 'experiments,' they were done without a proper knowledge of aberration and mental derangement."

"Then one day, since this is one engram among many, the mental hospital gets our patient and the doctors there decide that all he needs is a good solid series of electric shocks to tear his brain up, and if that doesn't work, a nice ice-pick into each eyeball after and during electric shock, the ice-pick sweeping a wide arc to tear the analytical mind to pieces. The wife agrees. Our patient can't defend himself: he's insane and the insane have no rights, you know."

In Dianetics: The Evolution of a Science, Hubbard writes: "Pre-frontal lobotomy is such certain and

complete mind-murder that one cannot be certain thereafter of anything in the patient except Zombie-ism." An almost identical sentence appears in *Scientology: The Evolution of a Science*.

In Science of Survival, Hubbard writes:

"In treating psychotics, always remember that one is working with minimal theta present and maximal entheta No mixture of Dianetics with old treatments or practices of any kind are recommended to the auditor. Electric shock has been known to lay a severe engram into an already overcrowded reactive mind and is not successful in any way, other than making a few patients so apathetic that they are barely acceptable to society. Psycho-surgery, removing pieces of the brain, has long been acknowledged an entire failure so far as any actual 'cure' is concerned. Electric shock and psycho-surgery may alter the behaviour pattern of the individual and may suppress him into some tractable condition, but the result is inevitably harmful to the ability, efficiency and general worth of the subject, with the further detraction that they cause damage to the brain from which the individual never completely recovers.

Now there are three ways of handling delusion. The first way is to electric-shock, or prefrontal-lobotomize, or sedate the preclear into utter apathy and uselessness in the society and wreck him completely. This is not recommended."

In Scientology: The Fundamentals of Thought, Hubbard writes:

"The use of electrical shocks upon a body for any purpose is therefore very dangerous and is not condoned by sensible men. Of course, the use of electrical shock was never intended to be therapeutic, but was intended only to bring about obedience by duress, and, as far as it can be discovered, to make the entirety of insanity a horror. Electrical shock deranges the electronic field in the vicinity of the body and is always succeeded by bad health or physical difficulties and never does otherwise than hasten the death of the person. It has been stated by people using electric shock that if they were denied euthanasia they would at least use partial euthanasia in the form of electric shock, brain surgery and drugs. These treatments in some large percentage of cases, however, effected euthanasia as they were expected to do."

Various other psychiatric methods are similarly the object of Hubbard's attack.

In *Dianetics: The Evolution of a Science* and in *Scientology: The Evolution of a Science*, he writes that narco-synthesis (i.e., narco-analysis) "produces slightly higher results than a magic healing crystal in the hands of an Australian medicine man."

In *Science of Survival* he writes, "Restraints and cold packs only succeed in shocking the patient into a deeper state of lethargy."

He further writes:

"Probably the most dangerous thing one can do to any aberrated mind is to place it under heavy sedation and try to treat it, or while it is under sedation place it in an atmosphere which is restimulative. Sedation of the insane is, shortly and abruptly, criminal, since it permits new perceptics to become entangled with an already confused mind under circumstances of perception which could not take place if the patient were not under sedation."

In *Dianetics: MSMH* Hubbard writes, "When one of these 'unconscious' periods was so probed - by the drug hypnosis called narco-synthesis - the patient usually became worse, not better."

Referring to the hospital treatment of psychotics, Hubbard writes in *Science of Survival*:

"Probably the worst thing that can happen to a psychotic is to be placed in the atmosphere normally provided for him by the state No better method of tailor-making psychotics could be devised than the usual institution, and it is probable that if the normal person were placed in such an institution, in such an atmosphere, he would become psychotic. Indeed, the incidence of psychosis overtaking attendants and psychiatrists in attendance in such institutions is alarmingly high. This is second only to psycho-surgery and shock treatment in the worsening of psychotics in a psychotic state. Rather than give psychotics such treatment it would be far kinder to kill them immediately and completely, and not partially as does psycho-surgery and electric shock."

Ironically, the Board heard evidence that the very institutions which Hubbard denounces in the last quotation have successfully treated unfortunate victims of his pernicious processes.

Hubbard has been equally intemperate in his bulletins and in magazine articles. A few examples must suffice. In HCO Bull. of the 24th July AD 10 (1960), Hubbard attacks the British Medical Association which, so he said, had maligned him. He writes:

"With what amazed surprise we viewed the recent attack upon us by the British Medical Association. With their hands caked with blood they sought to point a grisly finger at us and to bring down upon us the wrath of the government they claimed they controlled. Folly, thy name is medicine ... I have found that the British Medical Association in England has encouraged its doctors to spread vicious lies about us via their patients."

In Ability magazine, issue 31, Hubbard writes,

"Psychiatry, a dead duck in fact, is now pretty well known around as a dead duck by reputation. A gone goose. The reason: brain washing doesn't work."

In Certainty magazine Vol. 5, No. 1, AD 8 (1958), appears the following:

"Amongst the failed philosophies is communism and its offspring, the quack 'therapy' of Vienna, Moscow and Eastern Europe. This has posed as 'psychology' while at no time has it validated the psyche, or soul, thus being a contradiction in terms (just to start off with). In the U.S.A. at this date, 'psychology' as known for the last half century and more, is a discredited practice."

In HCO Bull. of the 15th June AD 10 (1960) Hubbard writes, "We are having Dickson investigated for Anti-social background, and if it ever comes to a court case, we'll ruin him." Dr. Dickson was and is the medical secretary of the Victorian Branch of the Australian (formerly British) Medical Association. A press statement concerning scientology attributed to Dr. Dickson had sprung, so Hubbard said, "From an effort by medical doctors to monopolize all healing to their own profit." Later evidence showed that the "investigation" of Dr. Dickson unearthed nothing.

In Com. Mag. Vol. 4, No. 10, October, 1962, Hubbard writes:

"The treatment of the insane today is far worse than it was two centuries ago and the brutality practised under the name of 'mental healing' cannot be regarded with equanimity by any civilized man.

We discover psycho-analysis to have been superseded by tyrannous sadism practised by unprincipled men themselves evidently in the last stages of dementia. This then is the end of the trail for psycho-analysis--a world of failure and brutality. Today men who call themselves analysts are merrily sawing out patient's brains, shocking them with murderous dlllg~, striking them with high voltages, burying them underneath mounds of ice, placing them in restraints, sterilizing them sexually, and generally conducting themselves much as their patients would were they given the chance."

In HCO Bull. of the 5th May, 1959, Hubbard writes that we have seen several practices which are

"not now classifiable as anything better than betrayal. Psychiatry and medicine are both good examples of this. The person who goes to a psychiatrist usually finds himself betrayed. He does not receive help, he receives brutality in the form of electric shocks, brain surgery and other degrading experiences. Even in the highest form of psychiatry it was common advice for the psychiatrist to tell the wife that the best cure for her troubles was to betray her husband, and vice versa."

The frequency and the intensity with which he vilifies psychiatrists and their work are the more serious because of the effect upon preclears, many of whom, when most needing psychiatric attention, are terrified at the thought of going to a doctor. This is one of the most wicked sides of scientology, for having made a massive onslaught on the person's mental integrity by its pernicious

practices, it then effectively prevents him from seeking assistance from a source likely to cure or ameliorate his condition. Many scientologists stated in evidence that they would prefer to go to a scientologist rather than to a doctor for the treatment of almost all conditions and they echoed Hubbard's hostility to the medical profession. The scientology viewpoint is that the medical profession is a great conspiracy, and that, especially in relation to mental health, "medical doctors and psychiatrists" are to be avoided.

A tragic and yet typical example of the dangerous way in which these fulminations against the medical profession take effect was dramatically illustrated in the case of one ardent but non-conforming scientologist who was excommunicated by Hubbard from scientology. Thereafter he suffered great mental tortures for a year and more, "with crocodiles chasing him around the streets of Brisbane", because, though he was desperately in need of psychiatric attention, he was yet unable to bring himself to seek it because of the deeply ingrained belief that psychiatry was, in Hubbard's words, "barbarity practised under the name of mental healing" and that the medical profession was concerned only to torture and debase.

The Anderson Report CHAPTER 23 DANGERS TO MENTAL HEALTH

Although there are several features of scientology which are to be condemned, its threat to health - particularly mental health - ranks paramount. The HASI's boast is that it is the world's largest mental health organization, an astonishing claim, the more so because it has flourished for some years in open competition with orthodox medicine and psychology. Thinly disguised as something of quality and value, scientology has insidiously infiltrated the community. In reality it is a dangerous medical cult, with special though not exclusive interest in mental health, and it is concerned primarily to discredit orthodox medicine and psychology and in lieu thereof offer its deluded victims dangerous hypnotic and other techniques which falsely masquerade as scientific.

The evil consequences of this deception have been dealt with at length in this Report: Chapter 18 deals with the dangerous hypnotic aspects of scientology processing; Chapter 19 deals with the spurious healing claims made by scientology; Chapter 21 deals with the general effects of processing. The hostility with which Hubbard so violently attacks the orthodox medical and psychological professions, dealt with in Chapter 22, produces a great fear and loathing of these professions in the minds of his adherents, who are thus conditioned to avoid at all costs medical practitioners, especially psychiatrists.

As already pointed out, this attitude may have tragic results, because not only may mental and physical conditions which require early attention for their successful treatment be neglected, but the unfortunate sufferer may embark upon a course of scientology processing which could produce a worsening of his condition, thereby greatly decreasing, and possibly destroying, the prospect of successful treatment by orthodox medicine. One psychiatrist illustrated the danger of scientology processing which he encountered in a patient who was a neurotic with paranoiac tendencies. The patient had been subjected to scientology processing and thereafter resisted all attempts by the psychiatrist to gain his confidence, a state necessary for successful psychiatric treatment.

This Chapter deals more particularly with the dangers resulting from the inability of scientology practitioners to observe mental or physical conditions, and the positive harm likely to result therefrom.

The Board heard expert medical evidence to the effect that there are numerous conditions, physical as well as mental, where skill and experience are required to detect them in their incipient stages when prompt treatment is tremendously important. A person exhibiting some degree of lethargy, a feeling of inadequacy, insecurity or anxiety may be suffering from a cerebral tumour, some form of glandular disorder, drug intoxication or some psychiatric condition, the detection of which is often difficult even by skilled practitioners. With many types of psychiatric illness it is important to treat the patient as early as possible and the prospect of ultimate successful treatment could be jeopardized by delays in receiving psychiatric attention. The expert evidence was to the effect that it was difficult to determine whether a person showing signs of depression was suffering from melancholia or from unhappiness or tension. Even the most experienced and skilled psychiatrist is aided by modern laboratory equipment, radiologists and well trained psychologists, and he proceeds with caution; the price of error may be the suicide of the patient, and inadequate medical training may well result in failure to locate, as the cause of the patient's depression, brain tumour, cerebral clot or other medical condition. The more trained and experienced the psychiatrist the

greater is his humility in his investigation, and the greater is the contrast between his investigation of the patient's condition and the cavalier way in which the HASI assumes that all conditions are amenable to indiscriminately applied scientology techniques. Scientologists almost invariably apply sledge-hammer methods, hitting savagely wherever they see a head. Such methods are often disastrous.

One of the very grave aspects of scientology is that its practitioners almost entirely lack any medical knowledge, and are unable to discern in a preclear symptom which would indicate to a medical practitioner the need for medical attention. Williams and other witnesses asserted that the HASI had a "filter system" wherein "the ill and insane are not encouraged to seek scientology as a cure", and that it was not the policy of the HASI to accept for processing or as a student a person having a history of mental illness or who appeared to have mental illness. The HASI uses printed forms of contract in which preclears and students acknowledge that the HASI is extending no guarantee of cure of any specific ailment the applicant may possess and that the course he is to undergo or treatment he is to receive is not in any way medical or psychiatric treatment but a series of exercises designed to increase ability. The HASI somewhat ingenuously thinks that the terms of these contracts ensure that their processes do not possess the quality of treatment of physical or mental conditions. It was said by the HASI witnesses that as a further safeguard, the applicant was given a "pre-processing security check" with the assistance of the E-meter. Williams said that people who were in fact undergoing psychiatric or other medical care were told that they were unacceptable for processing.

The whole routine, including the printed form, is all part of Hubbard's cunning in deluding his victims. In a perverse way it heightens the belief of the applicant that scientology will cure his complaint, for this unctuous negation by scientology of the claim to cure, while at the same time loudly proclaiming in its literature that it positively cures, suggests to the victim that scientology must be the great healer it claims to be because of the apparent modest disclaimer it so disarmingly makes.

The plain fact, however, is that the "filter" system does not filter; it was never intended to filter out those in need of psychiatric help or possessing mental troubles, for these are scientology's prime target. It is quite apparent from the evidence the Board heard and the files it examined that a substantial number of people suffering from various mental illnesses and in need of psychiatric care were accepted by the Melbourne HASI and by the Geelong and Hawthorn organizations for processing and training. Some of these persons had had earlier psychiatric treatment, and in many such cases that fact was known to the HASI and the other organizations before they were accepted for processing and training. Other persons who were evidently mentally ill were also accepted, or they developed obvious symptoms of mental troubles during their processing and training. After their sojourn in scientology, a significant number of these required and some received psychiatric treatment in State mental hospitals. How many in fact received psychiatric treatment in hospitals and from private practitioners could not be determined.

The Board heard evidence about one person who had been in and out of mental hospitals over a period of several years. He paid a total of £927 to the HASI. He was for a time a member of the HASI staff, at a time when the HASI knew he had been receiving psychiatric treatment. As a staff member he would be handling the affairs of preclears. Another person, after scientology processing, became a voluntary patient in a psychiatric hospital for four months. On his release the HAS[signed him up for 300 hours. After some scientology processing he had to return to the hospital. In several cases scientology processing reduced the mental health of the persons concerned to a state which required psychiatric treatment. Many who need psychiatric treatment as a result of scientology processing most probably have not had it, because of the fear and hate of

the medical profession which Hubbard has inculcated in them. Hubbard has admitted in HCO Pol. Lr. of the 10th October (probably 1960), that one of the consequences of scientology processing can be "nervous breakdown" involving "observation" in an asylum. He even regards the development of the "sad effect" as a good sign in processing.

Of the "sad effect", Hubbard writes,

"The Sad Effect. We could call this Tearculi Apathia Magnus and everyone would be in great awe of it. But I see no reason to follow the Latinated nonsense of yesterday's failured [sic] sciences. Call it something simple and the auditor will feel he can do something about it and even the preclear will cheer up a bit. So it is 'the Sad Effect'. This is a state of great sadness, apathy, and misery and desire for suicide."

It is such cases as this that scientologists enthusiastically process, and such states they are proud to produce.

The sequel to one of the demonstration sessions impressed upon the Board in a most dramatic and convincing way the great harm which flows from the ignorance of scientologists who are unaware of and incapable of understanding the dangerous nature of their techniques and the real condition of mental health of those on whom they are practising. The particular session demonstrated what was called "listen style auditing" (See Chapter 13). It was said that this was one of the simpler processes, quite a low grade process, and was designed to help people talk about their worries and problems and get them "off their chests", on the basis that people found relief when there was someone ready and willing to listen to them. It enabled the person with problems and worries to talk about them and the auditor, on this occasion being very literally one who listened, merely started the preclear talking and then sat silent, providing a receptive ear.

The demonstration session was of about thirty minutes duration. The auditor was Tampion and the preclear was a woman approaching middle age. She appeared to be nervous, excitable and enthusiastic. This woman had initially signed for 50 hours processing in June, 1962, but the director of processing, carrying out "the old routine" as directed by Hubbard, had certified that she required 300 hours before she could obtain a "stable gain", and she had then signed up for 300 hours. By February, 1964, she had had something in excess of 60 hours' auditing and had undergone some training. The session was conducted in a small room with only the auditor and the preclear present; the Board and others viewed proceedings in another room on closed television circuit. An E-meter was used.

The session began with the customary strict routine of "start of session", and the auditor then commanded the preclear to "Tell me about the right decisions you have made". The preclear thereupon commenced talking and almost without pause talked for the whole of the session about herself, her daughter, her husband and his woman friend, her neighbours, her work and her employment. It was a somewhat pitiful performance in which the preclear seemed to he talking herself into the belief that all the problems associated with the matters she mentioned were working themselves out and that she was understanding them better. At the end of the session, she said she had made gains. Nine days after the demonstration session this preclear was admitted as a patient to the care of the Mental Health Authority.

The Board is appalled at the realization that it witnessed this unfortunate woman being processed into insanity. At that early stage of the Inquiry the Board had not been informed of the potentially dangerous nature of this apparently simple and easy "listen style auditing". Subsequently a psychiatrist witness who read the transcript of this woman's demonstration session gave evidence that her behaviour in the session indicated clearly that she was in a state of mania rather than ecstasy which would have been readily apparent to a psychiatrist. Williams, Tampion and Mrs.

Tampion were present at this session and the woman was obviously regarded by them as a suitable subject for auditing. Williams is a "Doctor of Scientology", and Tampion, for his services in conducting the demonstration sessions, was subsequently awarded the "degree" of "Doctor of Scientology" by Hubbard. Neither Williams nor Tampion nor Mrs. Tampion had the slightest idea that there was anything mentally wrong with this preclear. Later, Williams unconvincingly sought to explain the need for hospitalization as being the stress of vaguely hinted at family troubles. Williams, when asked did he know what a schizophrenic was, was constrained to answer that he knew only in the broadest terms. Tampion claimed that because of his scientology training his ability to treat people with psychotic and neurotic conditions was superior to the ability of psychiatrists although he did not bother to give the time to treating such people.

The kind of treatment given to this unfortunate woman was the very kind which precipitated her breakdown. The Board heard expert psychiatric evidence to the effect that it was one of the well known traps in handling depressives to believe that by encouraging them to talk and "get things off their chests" one was doing some good. In dealing with a person showing signs of depression, psychiatrists have to exercise great care and judgement in determining whether it is advisable to allow the person to talk about himself or not. The traditional concept of a psychiatrist as one who encourages a person to sit down and talk about anything is not correct. In order to determine whether a patient should be allowed to talk about himself, the psychiatrist must be a highly trained physician, with insight of many branches of medicine, as well as of his own speciality, and the listener must be able to observe and appreciate the danger signs. In particular cases, positive harm may be done by encouraging a person in depression to talk about himself for the patient could be developing a pathological sense of guilt. and to allow him to elaborate on his guilt may develop in him such a sense of guilt that he may well attempt suicide.

There was further expert psychiatric evidence that such techniques as listen style auditing encourage a trust and dependency by the patient on the auditor, and tend to mobilize guilt and bring up emotions and anxiety in the preclear. Such anxiety tends to provoke more symptoms, more anxiety and perhaps more depression, leading to a worsening of the situation. In any event, being a good listener or interviewer is not sufficient. What is required is an understanding of the complexities which are involved, an understanding impossible of attainment by a person whose only claim is that he is a scientology-trained auditor.

The danger to mental health is further emphasized by the peculiar basis of scientology. Its practitioners use techniques based on impossible theories and directed towards the treatment of "conditions" said to be brought about by completely fictitious circumstances. Based on fantasy, scientology has built up a body of "knowledge" which is fanciful and simply not true: thus scientology treats its imaginary "diseases" or conditions with its own fantastic procedures. The alarming feature is that real diseases and conditions do exist and often the fantastic scientology procedures are positively harmful to the real condition. It is like applying an oxy-welding torch to a leaking rubber tube to weld on a patch of rubber over a hole that does not exist, the real trouble being a leaking valve.

It is unnecessary to dwell further at any length on the harm which scientology has done and can do to the mental health of the community. It is not to be doubted that scientology techniques worsen and prolong the mental troubles of the mentally ill and that they produce in even a normal person disturbances and anxieties which may precipitate mental trouble.

The Anderson Report CHAPTER 24 DANGERS OF COERCION

In scientology auditing very detailed notes are made by the auditor of all that a preclear divulges. A copy of the auditor's report in respect of each preclear is sent each week to Hubbard at Saint Hill, where it is kept. The office copy remains at the HASI. In the course of, say, 200 hours' auditing, a preclear's file may be handled by as many as ten persons at the HASI, all of whom have the opportunity of reading and familiarizing themselves with the contents of the file. There is thus no secrecy within the HASI, notwithstanding that the auditor is enjoined by the auditor's code not to impart the personal secrets of his preclears. The knowledge that these intimate files exist both at the HASI and in England is likely to be harmful to the mental health of the preclear concerned. A great part of the contents of these files is confessional material disclosed by the preclear at a time when the psychological mechanisms of repression are dulled or are not operating and normal inhibitions are suspended. A preclear frequently makes disclosures in an auditing session which he would not otherwise make except under great psychological stress or other form of compulsion. While a preclear is in the thraldom of scientology and his mind is dulled to reality, while he is a compliant victim willingly abasing himself, taking his courses and paying his fees and continuing to aspire with customary fervour to the elusive goals of scientology, he may not be troubled by the fact that the records of his intimate and frequently shameful secrets are preserved. But should he, whether or not he is still "in scientology", show any disposition to deviate or otherwise offend the HASI, he soon realizes the grave embarrassment of such records and the great influence which the HASI has over him. The potentialities of the situation are limitless, and the fears and anxieties of the non-conforming or sadly awakened preclear are greatly disturbing and very likely to enforce upon him subjection to the HASI under threat or fear of the threat that non-compliance might result in damaging disclosure of the contents of his file. Even if the preclear is still prepared to assume that the HASI will not divulge the secrets in his file, he is nevertheless in a condition of great subservience to the organization, because of the power which it is in a position to exercise as a result of the confessions he has made. The Board heard expert psychiatric opinion that the realization that confessional material disclosed in auditing and kept in records which could not be recovered by the preclear would be enough in some cases to induce potential and chronic anxiety and perhaps psychosomatic symptoms.

In fact, no evidence was presented to the Board to the effect that in Victoria any material obtained during auditing has been used for the purposes of what is conventionally described as "blackmail", that is, threatening to accuse any person of any infamous or other crime with intent to extort or gain from any person any valuable thing. (See Crimes Act 1958, section 124.)

But that is only one side of the picture. The existence of the files at the HASI and at Saint Hill creates a situation which tends to keep the preclear in permanent subjection to the HASI. It is a very brave man who would defy the HASI in such circumstances. The fact that it knows all his secrets, guilty or otherwise, confers on the HASI great power and authority over the preclear, and constrains the preclear to comply with the requirements of the organization and enables it to bring to heel even the most recalcitrant.

The evidence revealed a number of cases in which the fear of the HASI forced upon preclears a course of conduct in which they were not willing parties. One witness wrote to the secretary to the Board, offering to give evidence along lines which he admitted in the witness box would not have

been true. He had written the letter because the HASI had desired him to do so. Eventually he broke with the HASI, and bravely gave evidence hostile to scientology which exposed him to very rigorous cross-examination. The secretary received several letters from other scientologists written in consequence of a direction or request by the HASI that they be written. One young woman at the beginning of 1964 freed herself for a time from the shackles of the HASI. Early in the Inquiry she gave evidence which was hostile to scientology. At a later stage when she was recalled to give further evidence she contradicted much of what she had earlier said; in the meantime she renewed her association with a number of scientologists, with two of whom she was residing, and it became evident on the later occasion that she was not a free agent, that her will had been overborne by her scientology associates and that she moved in fear of the HASI and the consequences of being "outside" scientology.

Scientologists are kept in line by the HASI; any deviationist is summoned to the HASI for processing, and if he is the holder of certificates these may be cancelled or suspended, and only revalidated when the offender has sufficiently abased himself and has undergone a long period of processing at his own expense. The Board heard of several instances of scientologists having to humble themselves in order to be readmitted to the good graces of the HASI. All staff members, from Williams down, are regularly audited to ensure their continued conformity with scientology. Any scientologist whose loyalty is in any way doubted may be required to undergo a "security check."

A particularly effective means for keeping preclears in subjection to the HASI is the letter-writing technique. This technique involves the frequent writing of brief letters to any preclear who seems to be withdrawing from scientology. The letters are written by various members of the staff at intervals of two or three weeks and are of a friendly nature, but keep referring to the preclear's association with the HASI and to the preclear's problems and need of scientology. The effect on the preclear of this sustained barrage of letters is generally to bring him into the HASI once more. Examples of such letters are in Chapter 15. The continuous flow of letters serves to keep alive the preclear's fear of and subjection to the organization.

A very disturbing feature of the existence of the HASI records is that there is no guarantee that the HASI will continue to respect the confidence which preclears have reposed in the organization. It may be that the present HASI staff at Melbourne, and even Hubbard's subordinates at Saint Hill, will honour the trust which preclears have placed in them; but future personnel may not be as trustworthy. Indeed, even at this stage, Hubbard himself is not to be trusted to preserve confidences. If it suits his purpose, he is prepared to make known secret communications sent to him in the belief and with the intention that their contents were not to be disclosed. during the course of the Inquiry the Board heard evidence of a breach of such confidence, in the case of one preclear who went to extraordinary lengths to get back into the favour of Hubbard. It so happened that when Hubbard visited Victoria in 1959 this preclear, a most ardent scientologist at that time but a man who dared to think and to question, was out of favour with Hubbard. This preclear had been in scientology for about three years and had found in repeated processing much the same solace and escape as a drug addict finds in his drug. Because he had offended Hubbard, he was excommunicated and denied further processing. He felt he was desperately in need of processing, and, believing that if he abased himself sufficiently he would ingratiate himself with Hubbard and so be pardoned, he wrote to Hubbard pleading for permission to be audited. In his letter he made an abject confession of a long list of wrongful acts which he said he had committed, ranging from the stealing of five shillings from the mantelpiece when he was six years old to very disgusting and depraved behaviour in later life, some of which may well have been untrue. The confession was produced at the Inquiry and was used in an attempt to discredit this witness. It is apparent that it was written to Hubbard in the greatest confidence and for his eyes alone, but it was argued that the confession was not a disclosure made by a preclear during auditing, and therefore the "code" had

not been breached. From a legal point of view, it may well have been available as a means of testing the credit of the witness, but the publication of the contents of this document, manifestly intended to be of the most secret nature, illustrates Hubbard's irresponsibility and unreliability. One psychiatrist said in evidence that this preclear had done everything he could to regain favour with the organization, never expecting that his confession would be made public, and that psychiatrically its disclosure "was a dreadful and cruel thing to do".

The use which Hubbard sought to make of this confession is in line with his general attitude towards those who offend him. It is quite clear from correspondence between Hubbard and his Melbourne HCO that Hubbard was quite prepared to privately "blackmail" scientologists who offended him. In a letter dated the 24th May, 1960, Hubbard wrote to Mrs. Williams, the secretary of the Melbourne HCO, in these terms:

"Dear Eliz - re [name omitted]: Horner blew up in our faces and has had his certs. cancelled. We have criminal background on him. Rape of a girl pc in Dallas and countless others. This will do something to [name omitted]. Now, I firmly believe you will be able to find a criminal background this life on [names of two Melbourne scientologists deleted], as no such occurrence anywhere in the world has failed to find one. I'd grab him when he comes in and security check it into view. Run one on [two names omitted]. If they won't co-operate you have 'suspected criminal activities.' It's a thrupenny push now. Horner, Nina West, Sylvan Stein in U.S., [two names omitted], Kemp, possibly Wing and Smokey Angell, Nibs - all tie into a neat network. We're pulling it apart. Stein ran away. Tooley, by the way is in N.Z. with Stevens. Best Ron."

The person who was the subject of the letter was at this stage out of step with the HASI and he was informed by letter signed by Mrs. Williams, on behalf of Hubbard, that "all your certificates, rights, memberships. franchise in Scientology and in dianetics are suspended forthwith.... Should you make arrangements, and receive 500 hours of HGC processing at professional rates further attention may be given to your circumstances." Professional rates at that time were about 1 guinea and a half per hour.

It is now said that accusations against Horner, of "criminal background" and of "rape of a girl pc in Dallas and countless others", were unfounded. Subsequently Hubbard and Horner settled their differences and Horner returned to the fold. He is the author of Summary of Scientology and when last heard of was a leading overseas scientologist who probably would be extremely surprised to know of his "criminal background" so irresponsibly publicized by Hubbard in May, 1960. There was no justification for the accusations which Hubbard directed should be made against the two persons whose names were omitted above; Hubbard was merely irresponsibly asserting, as was his practice, that anyone who was out of line with scientology had a criminal or communist or homosexual background.

In the circumstances it is quite understandable that few former scientologists came forward volunteering evidence hostile to scientology. Those who had such thoughts would promptly have put them aside, quite evidently fearing that should they raise their voices in opposition to scientology they would be at the mercy of an organization which was in a position to wreak vengeance by revealing their secrets. There was, however, a small group of former scientologists who came forward as witnesses and were prepared to take the risk of giving evidence hostile to scientology. The Board is grateful to these witnesses who, though subjected to rigorous and embarrassing cross examination, gave evidence which, on a number of matters, was of material assistance to the Board.

Security Checks.

One means whereby the HASI sought to ensure that staff and students alike did not deviate from or have mental reservations concerning scientology theories was by the use of what were called "security checks". The "security" in contemplation in these checks was the security of the HASI. Williams said there were two kinds of security checks; one kind was "generalized", of which there were about a dozen different forms, and the other was "particularized", "for some particular mores, group or profession." It was said that from about the beginning of 1962 securitychecks had ceased for the most part for processing purposes and that thereafter they had been used only for security purposes.

These security checks probed deeply into the minds of preclears and, when used in conjunction with the E-meter, were very effective in establishing HASI domination over both students and staff members.

"The Only Valid Security Check", prescribed by HCO Pol. Lr. of the 22nd May, 1961, contained 150 questions, of which the following are examples:

Have you given your right name?

Are you here for a different purpose than you say?

Have you ever stolen anything; forged a signature, cheque or document; blackmailed anybody; been blackmailed; cheated; smuggled anything; entered a country illegally, been in prison; tried to act normal; indulged in drunkenness; done any reckless driving; hit and run with a car, burgled any place, embezzled money?

Are you guilty of anything?

Do you have a secret you are afraid I'll find out?

Have you ever assaulted anyone, practised cannibalism, been in gaol?

Have you ever been a drug addict, made anyone into a drug addict, peddled dope?

Have you ever raped anyone or been raped, been involved in an abortion, committed adultery, bigamy, practised homosexuality, had intercourse with a member of your family, been sexually unfaithful, practised sex with animals, practised sodomy, slept with a member of a race of another colour, committed culpable homicide, committed a justifiable crime, bombed anything, murdered anyone, hidden a body, attempted suicide, caused a suicide, kidnapped anyone, aided an informer, betrayed anyone for money, threatened anyone with a firearm?

Are my questions embarrassing?

Have you ever plotted to destroy a member of your family, had a member of your family in an insane asylum, ever been pronounced insane, looted any place, conspired with anyone, practised fraud, ever had anything to do with Communism or been a Communist, been a newspaper reporter?

Are you hiding anything?

Have you ever used hypnotism to procure sex or money, ill-treated children, practised sex with children or practised masturbation, taken money for giving anyone sexual intercourse, sexually coerced a servant?

Do you have any bastards?

Are you withholding anything?

Have you ever had any connexion with a brothel, coerced anyone into giving you sex, had anything to do with a baby farm, killed or crippled animals for pleasure, been a spy for the police?

Are you afraid of the police?

Have you ever committed a misdemeanour, a felony, a capital offence?

Have you ever done anything you are afraid the police might find out, falsified the books in any firm you worked for, criminally avoided taxes, counterfeited money?

Have you ever done anything your mother would be ashamed to find out?

How could you help mankind?

Have you ever controlled people?

What is Communism?

Do you feel Communism has some good points?

Have you ever injured Dianetics or Scientology, committed an overt on a scientology organization?

Do you have any overts on L. Ron. Hubbard, Mary Sue Hubbard?

Have you ever had any unkind thoughts about L. Ron Hubbard or scientology?

Do you know of any secret plans against scientology?

Do you plan to steal a scientology organization?

Do you deserve to be helped by scientology?

Are you upset about this security check?

What question in this check shouldn't I ask you again?

What unkind thoughts have you had while doing this check?

Subsequently, in HCO Pol. Lr. of the 29th June, 1961, a specific security check for scientology students was devised. It contained over 100 questions, of which the following are examples: -

Are you coming on this course in order to get away from someone, or something?

Are you here to get into something, to find out whether scientology works, to prove that scientology can't help you?

Are you coming on this course with the intention of killing off your body, with the intention of spinning, or going insane?

Are you presently taking tranquilizers, drugs, or medication of any sort?

Have you had sex with any other student, a staff member?

Are you trying to get another student (staff member) to have sex with you?

Have you coughed, or distracted others, during a lecture?

Have you criticized this course, or the organization, verbally or in writing to non-scientologists?

Have you been thinking unkind, or critical, thoughts about your preclear, L. Ron. Hubbard, your instructor, your auditor, other students, staff members?

Are you in disagreement with any of the stable data of scientology?

Have you secretly violated any course rule, or regulation; passed on restricted data of scientology to unauthorized persons, tried to give scientology a bad name?

Are you making any scientologist guilty of anything?

Have you done anything that would discredit Ron or Mary Sue Hubbard, or your instructors, by reason of their having trained you?

Is there anything that L. Ron Hubbard, or your instructors should mistrust you for that you haven't told them about?

Are you in communication with someone who understands more about scientology than L. Ron Hubbard?

Is anyone hostile to scientology assisting you financially on this course?

Is anyone here counting on you to keep a secret for him?

Have you any feeling of "injured innocence" at having been asked these questions?

Have you been critical of the data or quality of tapes?

Have you ever written and then destroyed critical messages addressed to L. Ron Hubbard?

How do you feel about these questions?

Staff Auditors and field auditors were subject to security checks as well, and HCO Pol. Lr. of the 7th July, 1961, sets out the "HGC Auditor's Sec. Check". This security check contained over 170 questions of which the following are examples: -

Have you ever permitted a preclear to take control of the session?

Have you ever startled a preclear when he was on a comm lag?

Have you ever permitted a preclear to have secrets from you?

Have you frequently run overtime?

Have you ever blamed the pc for running overtime?

Have you ever blamed the D. of P. or L. Ron Hubbard for your preclear's case not advancing?

Have you ever failed to flatten a process when it was still biting?

Have you ever jammed a preclear into a one-way flow and left him stuck there in it?

Have you ever audited badly?

Do you regard auditing as a punishment?

Do you feel that auditing is too good for psychotics, or cripples or criminals?

Have you ever been distressed because of a preclear's physical or mental pain under auditing?

Have you ever stolen from a preclear?

Have you ever stolen another auditor's preclear?

Is there any question about the way you audit which you would hate to have the D of P, or L. Ron Hubbard, ask you?

Is there any question the D. of P., or L. Ron Hubbard, should have asked you, and hasn't?

Do you hope you won't be found out?

Have you ever avoided receiving auditing yourself?

Have you ever mistrusted your E-meter?

Is there anything mysterious to you about an E-meter?

Have you ever let a preclear control you?

Do you think selling auditing is really a swindle?

Do you think there is anything wrong with invading a preclear's privacy?

Do you feel there is anything wrong with having your own privacy invaded?

Have you ever sought to prove that auditing would not work?

Have you ever done anything to slow down L. Ron Hubbard's research?

Have you ever wasted auditing time?

What do you wish you hadn't done?

Are scientologists' or Ron's goals really false?

Are you upset by this security check?

The Anderson Report CHAPTER 25 MORAL LAXITY

The evidence showed that sometimes lax, and even low, standards of morality existed at the HASI. One witness said that while scientology did not advocate free love or talk in terms of advocating free love, yet the normal barriers to free love which might inhibit some people were non-existent for a scientologist. Certainly, in session, the barriers to intimate sexual revelation are down, and teenage female auditors discuss the most intimate and disgusting sexual matters with their male preclears; and it would seem that there is little restraint on conversation between the staff on such matters. One female preclear, whose husband was a staff member, revealed in auditing either a real or imagined "affair" with another scientologist. In the course of office routine, the preclear's file came to her husband, who, reading of the "affair", merely endorsed the file, "Lacks morals."

Hubbard, especially in his earlier writings, wrote extensively about such topics as abortion and other sexual matters. Some examples of Hubbard's preoccupation with such subjects are set out in Chapter 6. Though he makes some attempt at denouncing such behaviour, he treats abortion in such a way as to present it as a common and almost usual event with any woman, and he engenders in the minds of his adherents, for whom his books are recommended and almost essential reading, such an attitude towards and familiarity with abortion and similar topics that they lose any quality of moral wrongdoing. The subject of abortions seemed to be almost a regular coffee break topic at the HASI, and the evidence shows that a number of female members of the HASI staff, married and unmarried, have had abortions. Furthermore, scientology processing involves the removal of all normal checks and restraints and readily conditions a person to regard as of little consequence departures from normal moral standards. One can well understand, then, how scientology-indoctrinated women, married and unmarried, set as their goals in auditing sessions the goal to have a miscarriage- which, with a little bit of outside assistance, they then claim scientology processing achieved for them.

Scientology theories are in part at least responsible for the distorted attitude of scientologists towards sexual matters, for it is argued amongst some scientologists that to seduce a girl of, say, fifteen years of age would be of no moment, for the thetan has had many sexual experiences and really such a girl would be 76 trillion and 15 years old. And it is said amongst scientologists that if a girl of 15 years were to be upset when introduced to sexual relations at that age it would be because she had some enormous overt on sex in a past life. On this basis, a particular scientologist may well have considered it of little account that he was living as man and wife with a young girl of 15 or 16 years, who later at the age of I7 years was a staff auditor at the HASI.

And yet it is into an atmosphere such as this that Mrs. Williams said she would recommend mothers to send their daughters to start upon the career of a scientology auditor.

In auditing sessions, where the preclear is not allowed to have inhibitions or to show reticence or reluctance to revealing and discussing the most intimate things, sexual matters are frequently discussed at length and in startling detail and sexual feelings are aroused and dwelt upon. It frequently happens that the auditor is a male and the preclear is a female, or vice versa - both participants often being in their early twenties and sometimes younger This unrestrained dwelling upon sexual topics is sought to be justified by scientologists on the basis that such discussions are "in session", as though that circumstance regularized the debased and erotic prying which

psychiatric evidence described as harmful and as bordering on voyeurism in the auditor.

In the evidence given by witnesses before the Board and in the files examined were many examples of quite shocking mental depravity. It is both unnecessary and undesirable to give full details of all these cases; a few typical examples will suffice to indicate the general nature of this particular form of perversion. Thus, in cases where the auditor was a young female and the preclear a male, such comments as the following appear in the files: "sexy as hell"; "pc gets often the urge to move down to his sex organs. If he does that he gets restimulated"; "pc has a bug about sending sexual beam at auditor"; "pc disturbed because he came to have auditing and now wants sexual intercourse." One female preclear, a widow, is recorded in four successive sessions as wanting to have sexual relations with her male auditor. In another case a female preclear, audited by a male, was run for 13 to 14 hours mainly on "sex withholds", the auditing command being, for part of the time, "What have you done that your mother would be ashamed to find out?" This preclear's file records that she "went into grief over being father's lover" in 1919, which, incidentally, was before the preclear was born.

It was quite common for preclears in auditing to tell in great detail about their sexual experiences in past lives. These hallucinations often related to the preclear's infidelity in an earlier existence, in which, not infrequently, the preclear was of the opposite sex to his or her sex in this lifetime. Acts of perversion by the preclear, both real and imagined, were readily discussed. One young man, oppressed by the thought of youthful acts of masturbation long since abandoned, was regularly processed at about six-monthly intervals on such earlier behaviour. One woman preclear recalled living on the island of Lesbos, and considered that she was the original Lesbian. She also believed she was Karl Marx in a previous life time; and another preclear, a male being audited at the same time, considered that he was her wife when she was Karl Marx. There was also the case of one scientologist who was worried because his wife was now living with another man who had been her husband in a previous life time.

One male witness, whose processing file contained a large number of references to disgusting sexual matters, when asked, "Did the sex of the auditor affect you in that regard?" answered with unexpected candour, "What do you think? A luscious doll sitting in front of you and you have to cough up these horrible sex withholds. Of course it did."

The following cases give some detail of the predominance of sexual matters in the course of a preclear's prolonged association with the HASI. In the case of one young woman, an earlier goal "to be a martyr" having been cleared up, it was considered by the HASI that another goal, "To create a beautiful universe", which was showing up, might be her basic goal. Early in January, 1963, she was accordingly run on process "R2-12 Listing" (see Chapter 13) for several sessions with the purpose of bringing up such a goal or until it died out or until an R.R.ing (rocket reading) item was found.

On at least eight different days, sessions of several hours were held during which the preclear was asked repeatedly the question, "Who or what would oppose scientology?" During the course of these sessions, the preclear gave a total of over 1,200 answers which ranged from the most absurd to the most disgusting of topics. The preclear frequently repeated a particular item, sometimes emphasizing it by adding a vulgar adjective or adjectives. Amongst the answers given, and sometimes repeated, were, "fairies", "wizard", "witch", "witch doctors", "bodies", "bad sex", "dirty sex", "sexual pleasure", "sex organs", "maggots", "Ron Hubbard", and details of various sexual intimacies between her husband and herself.

Thereafter, on a number of days, the preclear was run for several hours each day on the same

process, the questions put to her being, "Who or what would oppose a fairy?", "Who or what would a fairy oppose?", "Who or what is connected with scientology?", " Who or what would oppose witches?" The answers, running into several hundreds, were of a kind and degree similar to those given in the previously described sessions.

As a climax to this series of sessions, the preclear in the course of a 2 hour 35 minute session gave over 300 answers to the question, "Who or what would oppose fairies?", the answers being of a kind and degree comparable with those earlier mentioned, with, however, great emphasis on things that were "ugly", and including, incidentally, "an elephant who was not like an elephant should be" and "a triangle that did not meet at one of its corners."

From these sessions, and by the use of the E-meter on the preclear, it was decided that the "reliable item" was "ugliness" and " ugly things". After further processing, during which the preclear's goal was determined as being "to create a beautiful universe", there followed more "Who or what" processing, including at least two sessions in which the question was, "Who or what would the goal to create a beautiful universe oppose?" The answers given by the preclear totalled about 150, several of them being as disgusting as answers given in earlier sessions. In this session and in some of the earlier sessions, the auditor was a male.

Before she was married, this preclear in correspondence with Hubbard made confessions of her overts and withholds, including confessions of her misconduct with another member of the HASI staff, and other quite weird and sometimes disgusting matters, some of which were the sheerest fantasy about her thetan's existence in a "police" state millions of years ago when she was a member of a force of invaders who "used to swoop down on bodies, in and out in and out," giving the thetans of those bodies terrible frights and causing them to exteriorize.

This woman was formerly a school teacher who had qualified as HPA and for a time was employed as an auditor at the HASI.

One young male preclear who, to the knowledge of the HASI, had shortly before received psychiatric treatment and had revealed to the HASI his "compulsions" as "fears on girls", "feels he must keep away from small girls", was accepted by the HASI for auditing and was processed intermittently over about four years. During one course, when his auditor was a young woman, one of his goals was to "work towards not masturbating any more." The following year his goal was still, "to stop masturbating."

At one time he was required to write down his "goals". He compiled a list of over 600 goals, several of which were of a most depraved nature.

For a period, he was subjected to the "R2-12, Listing', process and, to the repeated question, "Who or what would oppose psychiatry?" he gave over 1,400 answers, many of them disgusting and containing obscene emphasis. Then he provided 750 answers to "Who or what would oppose social disease?" and 350 answers to "Who or what would social disease oppose?" From these answers a "reliable item " was found to be " a good smell accordingly, he was repeatedly asked in processing, "Who or what would a good smell oppose?" to which he gave over 500 answers. The nature of his answers to this series of questions was almost unbelievably filthy.

This unfortunate individual, mentally ill and with sexual problems, distressed because of an association with a young woman, and disappointed that auditing did not cure a skin irritation which he had, was required for years to wallow in the mire of his disturbed mind, shackled to scientology and goaded by its beastly methods.

One female preclear first became interested in scientology in 1954, when a psychologist who practised scientology audited her for about 300 hours for fees amounting to something over £200. The preclear considered that this auditing had done her harm. Thereafter she became acquainted with the HASI and between 1958 and 1964 she spent thousands of pounds on scientology. Though there was no economic necessity for her to take employment, she periodically did so to earn money to pay for her auditing. By February, 1958, she had had 271/2 hours auditing at the HASI and thereafter she embarked on a series of intensive auditing sessions. By October, 1958, when she had had over 122 hours auditing on the goal "to be clear", with a balance of about 150 hours to go, her auditor reported, "pc determined to be clear next Wed." In October, 1964, when she gave evidence before the Board, she was then currently undergoing a course of auditing, of which she had had over 100 hours, and she was being audited 3 half days a week; she was still not clear.

At some time subsequent to 1958, this preclear had Mrs. Williams as her auditor. After 44 hours of auditing in an intensive of 50 hours, the auditor's report read, "Pc had somatics in stomach, shoulder and head got very itchy and fidgetty solid silver ball was very real to begin with, 'the realest thing so far, very solid.' During process Pc had unreality, turn off and on. Brought ball out of head and put it back again." After the next session the auditor reported, "I found I had not got the basic rock, so scouted and found solid silver sphere that Pc had been hiding in her head. Ran resp [responsibility) on it for last part of session. pc tried to hide it again but caught her in time."

About the end of 1960 this preclear was required to answer a questionnaire; some of the questions and her answers to them were:

Question: "What was your original goal when you first became interested in Dianetics and Scientology?"

Answer: "To get out of the doldrums and near psychosis I was in." (This preclear had had psychoanalysis and therapy in 1952, 1953 and 1954, a circumstance known to the HASI).

Question: "Have you had any failure in being helped by Dianetics and Scientology?"

Answer: "Yes - 6 years ago I was processed by a scientologist and was worse at the end of it. Also I had processing with the goal to be clear and was disappointed that I didn't reach it (2 years ago)."

Question: "What are your future processing plans?"

Answer: "The processing I receive during training (HPA) then later when I can afford it I'll have what I can afford. I want to reach OT some day."

In 1961 when asked what goals she had for "this country" she named as her goals, "judges to be scientologists, also M.P.'s, in fact, anyone in prominent positions". By the time she gave evidence before the Board she said she no longer had such goals.

On the 6th February, 1961, Mrs. Gogerly, as director of processing, considered that she should have a further 100 hours processing for a start and would require 250 to 275 hours auditing to clear, though it might be less. Two days later, the preclear signed for 50 hours, to be taken about June, and felt at that stage she could not afford to sign for 100 hours, but said she would do so if she could.

On the 10th March, 1961, one of the staff auditors wrote to her the almost standard letter, part of which read, "Ron wants you to come in and get auditing immediately. He wants a clear in your

area."

Included in the course of auditing she had thereafter were a number of occasions when she was run on the "R2-12, Listing" process. To the question, "Who or what would oppose the organization?" she gave over 500 answers. To the question, "Who or what would you not communicate with?" she gave 750 answers, some of which were utterly revolting, disgusting and sexually perverted. To the question, "Who or what would oppose people who made you confess?" she gave 170 answers, and two days later, to the question, "Who or what would oppose the getting in first with your confession?" she gave 450 answers.

In some sessions highly personal and sexual matters were discussed with her various auditors, sometimes with a male auditor. Intimate secrets were extracted from her by such questions as, "What would you not be willing to tell an auditor?"

This preclear was an anxious type of person not quite sure what she was seeking; that she was on occasions mentally disturbed was evident. It was clear that she was in need of psychiatric assistance; she had not sought it in recent years. She found escape from reality in scientology, and considered that going into the HASI was like walking into sanity.

It should not be thought that the foregoing examples exhaust the cases in which matters of sex and perversion were dealt with in an obscene and uninhibited way. Nor should it be thought that they mark the limits of mental depravity reached. By way of further illustration, the transcript of part of the evidence of Mrs. Williams, given before the Board on the 18th September, 1964, is set out in Appendix 19. At that time she was HCO Secretary, and had recently returned to Melbourne after attending a special briefing course at Saint Hill, under the guidance of Hubbard. She was one of the highest ranking scientologists who gave evidence before the Board; as HCO Secretary, she was directly responsible to Hubbard.

Also included in Appendix 19 are details of some preclears' cases which further illustrate the degree of mental depravity reached in some scientology processes.

The Anderson Report CHAPTER 26 FAMILY DISCORD

It is clear from the evidence that scientology has been the cause of grave strain and tension in families, both in respect of relations within the family amongst its members and in respect of financial stringency resulting from excessive spending on scientology. The Board heard evidence from a number of non-scientologists whose spouses were in scientology. The number of such witnesses who came forward to give evidence of friction and dissension in existing marriages was understandably small, and it is obvious that many persons refrained from volunteering evidence adverse to scientology because they feared that the giving of evidence would cause further exacerbation of the already strained martial situation. Others, it is believed, refrained from coming forward to give evidence because of the fear that the other spouse might be victimized by the HASI on account of his or her inability to handle and control the non-scientologist.

Dalliance with scientology processing and training is not a cheap experience; oral evidence before the Board and the contents of files showed that sometimes grave financial embarrassment attended the squandering on scientology of large sums of money needed for the maintenance of the family. Quite apart from oral evidence, the contents of many files showed that serious family discord was frequently caused by a preclear's activities in scientology, which he or she pursued notwithstanding the dissension and unhappiness caused by such conduct.

Such harassed witnesses as gave evidence still had affection for their spouses and were genuinely distressed by the effects which scientology practices were having. They sought to adopt a tolerant and persuasive attitude towards their scientology partners, but such evidence as the Board heard rather suggested that in all but one of these cases such measures were attended with little success. The known exception was the case of a married couple, one of whom gave evidence, who confronted each other in the Board room and are believed to have become reconciled. They were the more fortunate ones. The Board heard evidence of broken marriages in which one partner was a scientologist and in which scientology was at least a contributing factor. This was not because scientology has as a goal the breaking up of marriages where one party is not a scientologist, but rather because, as one witness put it, if a person who was scientology-orientated had a spouse who was not scientology-orientated and was anti-scientology, and on that account there was no chance of reconciliation and no prospect of getting the non-conforming spouse into scientology, then the scientology spouse would be advised to consider giving up the marriage.

An article by Hubbard, distributed as a free pamphlet and reprinted in the Melbourne HASI *Communication* Magazine, Vol. 1, No. 19, is a vicious document, and is calculated to cause trouble, dissension, suspicion and distrust between husband and wife, where one is a scientologist. In part, this article, which is entitled "Why some fight scientology", reads -

"Unfortunately the person who does not want you to study scientology is your enemy as well as ours.

When he harangues against us to you as a 'cult' as a 'hoax' as a very bad thing done by very bad people he or she is only saying 'Please, please, please don't try to find me out'. Thousands of such protesting people carefully investigated by us have been found to have unsavory pasts and sordid motives they did not dare (they felt) permit to come to light. The wife or mother who rails against a family member who takes up scientology is, we regret to have to say, guided by very impure motives, generated in the morass of dread secrets long withheld. The father, husband or friend who frowns upon one knowing more about the mind is hiding something he feels would damage him.

'You had better leave scientology alone!' is an instinctive defence, prompted in all cases investigated by a guilty conscience.

Once they hear a few truths from scientology such people become afraid. They know we know. And if we know this much and if you are further informed, they feel you will find them out, The wife, protesting is hiding such things as infidelity or an unsavoury circumstance, one or many. The husband, protesting, is hiding a past with many blots upon it

".... Beware the person or group who fights Scientology, for that person fights Truth - not the truth of natural laws but the truth about himself."

The Board is disturbed by the effect which this article could have on a scientology-indoctrinated person who, accepting without question the truth of Hubbard's dangerous assertions, would read into any opposition to scientology an admission of guilt in respect of shameful conduct and sordid secrets.

The prospect of the break-up of a marriage is enhanced if one spouse is imbued with Hubbardian teaching that sexual relations are "a low tone activity" and that such relations between a person high on the tone scale and a person lower on the tone scale is an aberrative act on the part of the higher toned scientology-conditioned person. The Board heard evidence from the wife of a broken marriage who said that her husband declined to have sexual relations with her because he was on a higher level on the tone scale and she was "of too low tone activity", and "he had reached a higher tone where such physical activity was no longer necessary."

Where members of the family other than the parents have espoused scientology, unhappiness has frequently been caused. Some children have given up study and careers, dissipated their savings, left home and generally adopted towards their parents attitudes which have been variously aggressive, superior, provocative and condescending. In one instance, where violence occurred between parents and their scientologist son, the son went berserk, had to be restrained and was not pacified until Williams and another scientologist arrived.

The Anderson Report CHAPTER 27 SCIENTOLOGY AND RELIGION

From time to time during the Inquiry the question arose as to whether the Board was concerned to investigate scientology theories on the basis that scientology was a religion and the theories and practices of scientology were religious beliefs and practices. The Board repeatedly stated that it was not some form of ecclesiastical or theological commission charged to investigate religious beliefs as such, and that it was not concerned to determine whether the beliefs that were held by scientologists were religious or otherwise. The Board was concerned to ascertain the scientology theories and practices so far as they were held and engaged in in Victoria, and its task was the same in that respect whether or not the theories and practices could properly be described as religious.

The Board was further concerned to investigate the manner in which these theories and practices, whether religious or not, operated in Victoria and their impact on the community, and whether or not benefit was being obtained by or harm was being done to members of the community by such practices. The Board made it clear that scientologists were quite entitled to believe whatever they wished - freedom of thought was their right - and it was immaterial that the Board might disagree with a very large part of scientology and regard it as nonsense and contrary to reason. Scientologists are, of course, free to disagree with the Board's conclusion as to the validity of scientology theories, and there are already indications that they will disagree - that is their right.

However, the carrying into practice of such theories by pernicious techniques from which grave harm results is quite a different matter. In this community, there are different faiths and their members are free to hold whatever beliefs they wish, though their creeds differ and are even in positive disagreement on various tenets. The adherents of scientology, if it be a religion, are entitled to the same freedom of belief, and they have the same freedom, even though it is not a religion. This, however, as the Board repeatedly pointed out, was not the question.

Those who claim that their beliefs constitute a religion cannot, under the cloak of such "religion", pursue a course which is evil and a danger to the mental health of the community. A group of people, by claiming that its particular religion requires the killing of human beings by way of sacrifice, does not obtain a licence to kill according to its creed. Neither are the adherents of scientology entitled to practise on others their "skills" and techniques which have deleterious effects upon their victims. Nor are they entitled to proselytize by calculated deception.

The Board reiterates that the holding of beliefs is a private matter; it may even be that the holding of certain scientology beliefs is, with some scientologists, a matter of conscience, though that would seem to be inconsistent with the general theme of scientology. What the Board is concerned to do, pursuant to its terms of reference, is to draw attention to the harm which has been done in the past by the practice of scientology, and the harm which will inevitably result in the future should scientology be allowed to continue to exploit and mentally ruin the anxious victims of its deception.

The foregoing observations apply on the assumption that scientology is a religion; they are equally applicable should scientology as known, carried on, practised and applied in Victoria not be a religion. In fact, scientology is not a religion. Apart from an occasional reference to scientology as a religious brotherhood and a claim to have some affinity with Buddhism and other religions, no

claim was made at the Inquiry, except forlornly in the final stages, that scientology as known, carried on, practised and applied in Victoria was a religion.

When the Inquiry began, the stated attitude of the HASI was that scientology was a science and not a religion. However, towards the end of the Inquiry, when it became apparent to the HASI that the practice of scientology in Victoria had been revealed in a very unfavourable light, and that it had no evidence with which to controvert the impressive body of expert evidence to the effect that it was dangerous to the mental health of the community, an attempt was belatedly made to present it as a religion. No evidence was tendered to that effect, but the complaint began to be made that scientologists were being persecuted because of their religious beliefs, and the suggestion was that bigotry was rampant. This change of front was merely laying the foundation for the eventual withdrawal of the HASI from the Inquiry, and is dealt with elsewhere in this Report.

Though scientology has not been, and is not, a religion, some early attempts were made in Victoria to exploit the favourable attitude which the community usually adopts towards ministers of religion and these attempts were along lines suggested by Hubbard. In the early years of scientology in Victoria, about 1955-1958, some of its practitioners assumed ecclesiastical titles, one or more posing as a "minister" and adopting the title "reverend", and one the title "bishop".

There was no valid basis for the assumption of such titles; their adoption, and also the wearing of clerical garb which was practised for a short time, merely indicated compliance with directions from Hubbard to assume such poses to help gain entry into hospitals and other places where no one would "dare to hinder or oppose a man of God". In PAB 32, Hubbard wrote that "a society accords to men of the church an access not given to others. Prisons, hospitals and institutions, and those who manage them, cannot do otherwise than welcome men of the church." The practice of such infiltration was referred to as procurement or dissemination, its purpose being to inveigle the sick and the distressed into scientology. The "bishop", one Frank Turnbull, went to New Zealand in the late 1950's (For further details, see Chapter 15).

Scientology in Victoria has not developed the "founding churches of scientology" which have arisen in the United States and are located in Washington, Los Angeles, New York, Miami, Seattle, Detroit and Dallas. On such evidence as the Board heard concerning the founding churches, one would not classify their practices as those of a religion. What their beliefs are did not appear in evidence and a handbook of their ritual, which was produced in evidence, dealt with little more than how to conduct church services, weddings, christenings and funerals. The instructions as to sermons were that they must always be on some phase of scientology. The handbook did contain "The Church of Scientology Creed", which asserted that its adherents believed, in effect, in the right of man to be free, and included declarations "That all men have inalienable rights to their own religious practices and their performance", and "That the spirit alone may save or heal the body."

When, in 1962, the United States Food and Drug Administration began to investigate the E-meter as an instrument of healing, Hubbard promptly issued HCO Pol. Lr. of the 29th October, 1962, claiming that the E-meter was "a valid religious instrument, used in Confessionals", pointing out that all scientology corporations were religious in nature, and were incorporated as religious fellowships, and that, as to the London and Commonwealth offices, "they will soon be transferred to Church status when the founding Church of Washington DC is given full tax exemption." Williams said that nothing was done to implement the plan to transfer the Melbourne HASI to "Church status", and he repudiated the suggestion that tax exemption had anything to do with the proposal to give "Church status".

There is no justification for the claim that the E-meter is "a valid religious instrument, used in

Confessionals"; in the course of being audited with an E-meter, a preclear is forced to tell his most intimate and shameful secrets and he does not make such disclosures as part of a religious practice.

In the early days of scientology in Victoria scientology weddings were performed, the total of such ceremonies being probably under ten in number. In such cases as the Board heard about, the scientology "wedding" was supplementary to the conventionally recognized nuptials, except, perhaps, in one instance. In that case, a witness gave evidence that in about 1957 one of the "reverend" gentlemen mentioned above with whom the witness was then residing was summoned at 4 o'clock in the morning to a flat where a scientologist had a sudden urge to be married. The "reverend" and the witness proceeded to the flat where the expectant couple were and the scientology ceremony took place. The witness said that he felt some anxiety that the couple should embark upon the sea of matrimony with no more formal launching than a scientology ceremony his anxieties stemmed from the fact that adverse publicity might be given to scientology if such a secret leaked out - but, he said, his anxieties were allayed because eventually the couple were "churched", albeit somewhat tardily, a couple of weeks later.

Some scientologists claim to be "doctors of divinity" and assume the letters "D.D.", a distinction believed to have been bestowed by a Hubbardian institution in America. There are few such dignitaries in Australia. One curious instance, however, is that of "Doctor" Marcus Tooley, who practises in Sydney, with a branch office in Melbourne, under the name of "The American College", what he calls "dynamic psychology". Tooley was formerly a prominent scientologist and held the degree of "Doctor of Scientology" and a Hubbardian-bestowed "D.D.". The Hubbardian-bestowed "doctorates" are the only justification for Tooley's title of "doctor". Tooley now denies that he is a scientologist, and he claims that what he practises is a form of psychology which has developed either parallel with or independent of scientology. His case is dealt with in Chapter 29.

There is in Victoria one "minister" of the founding church, but this is the founding church of America, for as the witness explained, there is no founding church in Victoria. This "minister" is a woman who qualified for this title, and a certificate signed by Hubbard, by writing a 2,500 word thesis. She has confined the practice of her "ministry", so she said, to a very occasional scientology wedding service which she has conducted merely to oblige friends, and her services have been additional to the conventional ceremonies because she says that she was well aware that she had no authority to conduct a regular wedding service. She has found scientology consistent with her religion, so she said, but in her day she has had several religions and would probably have difficulty today in nominating her current religion.

A partiality for dalliance in numerous religions was a characteristic of a number of witnesses who were scientologists. They were anxious people seeking something to believe in, and the various beliefs they had savoured did not satisfy them, for such beliefs required the searcher to accept the hard realities of life. In their search for the truth, as they wanted it to be, they found in scientology a soporific which insulated their minds against reality and allowed them to treat as their creed anything which pleased or satisfied or quieted them. Scientology promised them immortality if only they could have a reality on the thetan; the scientology "hereafter", with no heaven or hell or purgatory but an unending progress beyond time as a clear or even an operating thetan, gave them just the sort of thing they wanted to believe in. They had not found such escape in the tenets of the major religions, Christian and otherwise. Instead, they found refuge from reality in scientology and were able to sustain their illusion of escape by continued scientology processing. One witness had subscribed at one time or another to as many as 23 different religions, some of them quite exotic, but he was somewhat exceptional.

When the Inquiry began, scientologists were very concerned to establish scientology as a science, a precise science, and the whole of the evidence which they then gave was directed towards this end. So far as religion was concerned, reference was made to the injunction of Hubbard - "change no man's politics, change no man's religion, deny sovereignty to no nation" - to demonstrate its non-religious quality. Witnesses said that a person could be a scientologist and could at the same time subscribe to any religion he wished and there would be no conflict. An undated pamphlet from the HASI states that "the HASI is non-religious - it does not demand any belief or faith and is not in conflict with faith. There are people of all faiths who use scientology in their every day living, better their relations with people and their environment." In HCO Bull, of the 19th August, 1959, which is entitled "To a Roman Catholic", Hubbard, in an endeavour to ingratiate scientology with members of that denomination, wrote that in Ireland scientology had operated without coming into conflict with the Church, and he added, "Scientology is not an heretic religion and demands no belief or faith and thus is not in conflict with faith." Some witnesses even said that scientology helped them understand and appreciate their own particular religion better, though they were considerably surprised and embarrassed when they were referred to some of Hubbard's writings, to be mentioned later, on matters religious which they could not reconcile with their particular religious beliefs.

The eighth dynamic in scientology is variously described as the urge towards existence, as infinity, the dynamic of the Supreme Being, the infinite or God dynamic, the Creator, the Infinite Nature; it is called the eighth dynamic, so Hubbard states, because the symbol for infinity stood upright makes the letter "8". Hubbard shows a curious reluctance to write about the eighth dynamic, and scientology writers somewhat ingenuously leave it to each other to discuss. Scientology, writes Hubbard, "does not intrude into the dynamic of the Supreme Being", and his claim is that scientology "embraces" only the first seven dynamics.

The Attitude of Scientology to Religion.

Except for the purpose of deceit, scientology has not been practised in Victoria on the basis that it even remotely resembles a religion. In advertisements and in the personal efficiency courses the HASI takes care not to disclose Hubbard's disparagement of religion. The directive, "Change no man's religion" is brought to the attention of the beginner, and scientology is paraded as quite consistent with and even as a help to all religion, of whatever denomination. It is only after the preclear has been conditioned by training and processing that he is likely to learn of Hubbard's cynical hostility to religion.

The attitude of Hubbard towards religion is one of bitter cynicism and ridicule, which gives the lie to his directive to "change no man's religion". In a warped and sneering fashion he snipes at all things sacred in much the same way as he attacks the medical profession, though there is generally less venom in his tone when he is dealing with religion.

Scientology is opposed to religion as such, irrespective of kind or denomination. The essence of Hubbard's axioms of scientology is that the universe was created not by God, but by a conglomeration of thetans who postulated the universe. Sometimes God is referred to as the Big Thetan. Many of the theories he propounds are almost the negation of Christian thought and morality.

Hubbard spent some of his youthful years in the Orient and probably absorbed some Eastern learning, being particularly interested in Buddhism. He considers that his travels and research have made him an authority on the customs and religions of the peoples of the world. In *Scientology:* 8-8008, after dealing with the Early Greeks, he writes that

"more modern man has fallen into the error of making God into the body of a homo sapiens and posting him somewhere on high with a craving for vengeance and a pettiness in punishment matched only by the degradation of homo sapiens himself. There are gods above all other gods, and gods beyond the gods of universes, but it were better, far better, to be a raving madman in his cell than to be a thing with the ego, cruelty and jealous lust that base religions have set up to make men grovel down."

These loosely expressed sentences are probably susceptible of several meanings, but their evident meaning is one disparaging of Christian belief, and this passage somewhat shocked certain scientology witnesses who had earlier said that they had not found anything in scientology which was opposed to their particular religion.

Whenever Hubbard mentions religion the note of disparagement is present. For instance, in PAB 130, Hubbard writes, "Purgatory and hell is a total myth, an invention just to make people very unhappy and is a vicious lie." In PAB 32, Hubbard writes, "Only a barbaric minister is a 'Man of God'. In all enlightened religions such men are called 'Men of Wisdom' ". In PAB 31 Hubbard writes.

"Religion does much to keep the assumption in restimulation, being basically a control mechanism used by those who have sent the preclear into a body. You will find the cross as a symbol all over the universe, and the Christ legend as an implant in preclears a million years ago." In PAB 37 Hubbard writes,

"For two thousand years Man has not had health, happiness, or immortality, yet they were promised to him two thousand years ago, and Scientology is delivering them today." . In HCO Bull. of the 21st January, AD 10 (1960), Hubbard writes,

"Some churches used a mechanism of confession. This was a limited effort to relieve a person of the pressure of his overt acts. Later the mechanism of confession was employed as a kind of blackmail by which increased contribution could be obtained from the person confessing." In PAB 31 it is written,

"A few operating thetans - scarcity - could lead to trouble. Witness the chaos resulting from the activities and other determinism technology of one operating thetan, 2,000 years ago. It is despicable and utterly beneath contempt to tell a man he must repent, that he is evil. Those who talk most about peace on earth and good-will among men themselves carry forward the seas of unrest, war and chaos."

In Certainty Magazine, Vol. 5, No. 10, it is written,

"Two and a half thousand years ago a handful of clears civilized half a billion people. What if we were all clear. Neither Lord Buddha nor Jesus Christ were OT's according to the evidence. They were just a shade above clear."

The "poet laureate" of scientology, Julian Cooper, wrote a poem, evidently with Hubbard's commendation, the text of which was distributed by the HCO as suitable for publication in magazines, part of which was:

"They that worship Jesus nailed upon a cross above an altar,

While an old priest chants

Do they know what they worship?

Do they know who they worship?

Can they recognize what they worship?

Bondage of unconfronted facsimiles implanted so as to control and invalidate, is what they

worship."

In HCO Bull. of the 18th July, 1959, appears this "Historical note":

"The whole christian movement is based on the victim. Compulsion of the overt-act motivator sequence. They won by appealing to victims. We can win by converting victims. Christianity succeeded by making people into victims. We can succeed by making victims into people."

In HCO Bull. of the 11th May, AD 13 (1963), popularly called the "Heaven" bulletin, Hubbard states that he has been to Heaven. He claims that the bulletin is

"based on over a thousand hours of research auditing, analyzing the facsimiles of the reactive mind, and with the help of a Mark V Electrometer. It is scientific research and is not in any way based upon the mere opinion of the researcher The contents of this HCO Bulletin discover the apparent underlying impulses of religious zealotism and the source of the religious mania and insanity which terrorized Earth over the ages and has given religion the appearance of insanity." In the bulletin, amongst many other things, he writes,

"The Goals-Problems-Mass implants, which are the apparent basic source of aberration and human travail, which began with the goal to Forget, were cynically done in Heaven'.

For a long while, some people have been cross with me for my lack of co-operation in believing in a Christian Heaven, God and Christ. I have never said I didn't disbelieve in a Big Thetan but there was certainly something very corny about Heaven et al. Now I have to apologize. There was a Heaven. Not too unlike, in cruel betrayal, the heaven of the Assassins in the 12th Century who, like everyone else, dramatized the whole track implants-if a bit more so

The symbol of the crucified Christ is very apt indeed. It's the symbol of a thetan betrayed "

He then tells of how he, apparently, visited Heaven on two occasions.

"The first time I arrived and the moment of the implant To Forget was dated at 43,891,832,611,177 years, 344 days, 10 hours, 20 minutes and 40 seconds from 10:02 P.M. Daylight Greenwich Time May 9, 1963. The second series was dated to the moment of the implant To Forget as 42,681,459,477,315 years, 132 days, 18 hours, 20 minutes and 15 seconds from 11:02 P.M. Daylight Greenwich Time May 9, 1963."

He tells further that the implants were electronic by nature and following the pattern of the GPM, that implanting was done on a non-visible thetan, which arrived in a ship in a doll body. On the first occasion there were 29 implanted goals, on the second, 21; on each occasion the first three goals were the same, namely, To Forget, To Remember, To Go Away.

On both occasions the Gates of Heaven were visible.

"The gates of the first series are well done, well built. An avenue of statues of saints leads up to them. The gate pillars are surmounted by marble angels. The entering grounds are very well kept, laid out like Bush Gardens in Pasadena, so often seen in the movies. Aside from the implant boxes which lie across from each. other on the walk there are other noises and sounds as though the saints are defending and berating

The second series, probably in the same place, shows what a trillion years of overt acts does (or is

an additional trickery to collapse one's time). The place is shabby. The vegetation is gone. The pillars are scruffy. The saints have vanished. So have the angels. A sign on one (the left as you enter) says 'This is Heaven'. The right has a sign 'Hell' with an arrow and inside the grounds one can see the excavations like archaeological diggings with raw terraces, that lead to 'Hell'. Plain wire fencing encloses the place. There is a sentry box beside and outside the right pillar. The roadway 'leading up' to the gates is deeply eroded. An effigy of Joseph, complete with desert clothing is seen approaching the gates (but not moving) leading a donkey which 'carries' the original Madonna and child from 'Bethlehem'. The implanting boxes lie on either side of this 'entering' path at path level."

There is a great deal more in similar vein, the research which Hubbard did upon himself and others producing a great amount of hallucinatory nonsense enthusiastically conjured up in response to suggestions made before and during hypnotic auditing sessions, yet accepted as reality and verified scientific fact by Hubbard and his followers.

The bulletin concludes,

"Further, we have our hands on an appalling bit of technology where the world is concerned. With rapidity and a Meter it can be shown that Heaven is a false dream and that the old religion was based on a very painful lie, a cynical betrayal.

What does this do to any religious nature of Scientology? It strengthens it. New religions always overthrow the false gods of the old, they do something to better man. We can improve man. We can show the old gods false. And we can open up the universe as a happier place in which the spirit may dwell. What more can you expect. This actually places us far beyond any other beings that are about. It puts us, through increased beingness and a restoration of life, in control of much destiny.

We have now only a few unsolved problems about life, huge though they may be, such as the construction of bodies and how does one establish the character of and communicate, if feasible, with beings who are making trees and insects. There are a few things like these. But I imagine when we finally manage to communicate with beetles under rocks and free them, we'll no doubt find the Creator of Heaven who 431/2 Trillion years ago designed and built the Pearly Gates and entrapped us all.

Good Lord, I'd hate to be guilty of that overt. But never mind - you aren't either. That guy is Gone (I hope!)".

Some scientology witnesses sought to explain that Hubbard really did not mean that it was Heaven that he was writing about in this bulletin, but either that it was "hell" or "a hell" over the entrance to which some "racketeer" had written "Heaven". The suggestion was, as Hubbard puts it in the bulletin, that "Before you went to Heaven you were not really very good or very bad, but you didn't think you had lived only once and you had a good memory and knew who you were and enjoyed life", and that when you went to this place which was got up to resemble Heaven, you were implanted with aberrative goals to forget and the like, which somehow conditioned you to believe in this present lifetime a variety of things about Heaven because of the deception practised by the implanters 43 trillion years ago. Actually no scientology witnesses could explain what Hubbard meant, except that Hubbard says what he means and that "an ordinary person" would not understand the bulletin.

Care is taken by Hubbard to ensure that while a student is on course he is kept away from influences which might be likely to assist him to resist the pernicious learning which he is

absorbing. In HCO Pol. Lr. of the 22nd November, 1961, appear the "Training Course Rules and Regulations", one of which reads -

"24. Do not engage in any rite, ceremony, practice, exercise, meditation, diet, food therapy, or any similar occult, mystical, religious, naturopathic, homeopathic or chiropractic treatment or any other healing or mental therapy while on Course without the express permission of the Director of Training."

Courses may be as short as two weeks or may extend over many months. Absence from salutary influences for such lengthy periods tends to increase the domination of the HASI over the preclear.

Williams, the grandson of a clergyman, said that scientology was particularly careful not to interfere with anyone's religion and that his own religious views had not been changed away from Christianity by reason of anything he had studied in scientology. One young woman, struggling to escape from scientology, said that one of the many side effects of the past lives teaching of scientology was that it "knocks your belief in the [Christian] church."

One scientology witness considered that it was consistent with his Christian teaching in relation to the human soul, that it might well have returned to Earth on a number of occasions after visiting implant stations in the intermediate period. Other witnesses agreed that scientology was inconsistent with their Christian beliefs. Some said that they saw no inconsistency. One considered scientology more christian than Christianity. Another said that Hubbard's disparagement of religion was merely the expression of scientific or philosophic hypotheses, and there was no inconsistency because religion was one thing and philosophy another, and the witness sought to justify the apparent difference between Christianity and scientology by saying in effect that one was not being inconsistent if, when writing in one frame of reference, one contradicted what was a fact in another frame of reference.

That scientology witnesses were partisan on the question of whether Hubbard was hostile to religion was, of course, apparent. An ordinary person, reading the writings of Hubbard, a man who "says what he means", can judge for himself.

In a community which is nominally Christian, Hubbard's disparagement of religion is blasphemous and a further evil feature of scientology.

Promise of Immortality.

Hubbard exploits in two ways what psychologists regard as "the great mystery and the great threat to us as human beings", namely, death. Hubbard promises immortality through scientology, and he warns that death awaits those who oppose scientology. In making such a promise, Hubbard ascribes immortality to the thetan and the suggestion is that a person cannot enjoy immortality unless he has a reality on the thetan and on scientology teachings.

In Certainty Magazine, Vol. 3, No 9, R. Kemp, who is described as "D.Scn., D.D.", writes,

"Of course all people are thetans (spirits) and all thetans are immortal and indestructible; they cannot under any circumstances actually and finally die, but they can get awful close to it "

The editorial note to *Scientology: 8-80* is sufficiently garbled to convey the impression that scientology, which claims to bestow the ability to heal the ill without physical contact and to cure the insane and the incapacitated, may also revive the dead or dying. In *Certainty* Magazine, Vol. 4, No.

12, Hubbard writes,

"Well, I've been working now for a lot of years to bring Dianetics and Scientology up to a point of super-magic. It was easy to get them up to magic "

and the article goes on to say that there was "enough data to know that dying wasn't fatal but *still* men died and dying often hurts."

The claim is made in PAB's Book 3, and repeated elsewhere that "Scientology can demonstrate that it can obtain the goals set for man by Christ, which are: Wisdom, Good Health and Immortality". In Infm. Lr. of the 24th November, 1963, Hubbard writes,

"No organization which can achieve individual immortality ever worries much about momentary hostility. And individual immortality is not only in our grasp but is now being achieved."

In HCO Infm. Lr. of the 1st September, AD 13 (1963), which tells of a young woman, who on her death had indicated that she would return to pick up the body of a girl and take up where she had left off, Hubbard writes, "We lose very few people by death, unlike the healing professions who lose their practitioners in hordes. But we do temporarily lose some by death. A dozen or so in the past few years." Though, when writing along these lines, Hubbard is writing about the thetan, so garbled is much of what he writes on this subject that his followers may be excused for believing - as some appear to have believed - that the immortality promised is immortality in respect of their earthly or human life. As recently as May, 1965, the Melbourne Communication Magazine, unequivocally states "An intensive at the Hubbard Guidance Centre puts you on the road to freedom from sickness misery and death."

The Auditor, the Saint Hill Journal of Scientology, mailed direct from East Grinstead to Victorians on the HASI mailing lists, in explaining in issue No. 8 (1965) what scientology really is asserts, "It is the only thing which can salvage you from sickness and eventual death".

It was said by one witness that, since scientology ensured good health by proofing against illness, scientology was, in effect, ensuring at least longevity, and very few if any people who had espoused scientology were known to have died.

Scientology is opposed to all forms of punishment, and Hubbard has told his followers that "purgatory and hell is a total myth, an invention just to make people very unhappy and is a vicious lie". Comforted by these and similar statements, deluded preclears are able to shut their eyes to the reality of normal human problems which they may have, and to find refuge in scientology theories about the immortality of the thetan which will keep on returning to earth again and again for eternity, without the obligation of accounting, according to Christian belief, for this life's conduct.

Hubbard repeatedly predicts death or disaster for those who oppose scientology. He writes from time to time of long lists of people who have died because they have opposed scientology and offers to produce the death list to the doubters. Specifically, in *Com.* Mag. Vol. 1, No. 19, Hubbard writes, "On the other hand, without any action taken against them, of twenty-one highly placed attackers, seventeen are now dead." Elsewhere he writes that each and every one of the "squirrels" of yesteryear have met with disaster, the biggest squirrel in Great Britain being recently found in the bankruptcy court. (A squirrel is one who has given up allegiance to Hubbard but still purports to practise scientology, picturesquely described in evidence by a former scientologist as a madman running around collecting nuts.) Hubbard lists other cases of individuals who, having acted in a way hostile to scientology, have suffered disaster, such as one tradesman who overcharged the

organization and contracted TB, and a bill collector who, when he realized how terrible it was to endeavour to collect a debt from Hubbard, committed suicide half an hour later.

In HCO Bull. of the 29th July, 1963, Hubbard reports with evident glee, "Government attacks have entered a more desultory stage. Meters will go to jury trial eventually and we will certainly win. The U.S. Government Attorney handling the case became terribly ill and had to resign it".

A number of witnesses gave evidence that they were aware of a general belief amongst scientologists that hostility to scientology rendered a person likely to die on that account, not by physical violence at the hands of scientologists but because he would be so overwhelmed by the enormity of his overt that death would ensue as the motivator. The effect of such a belief was observed in the case of one woman witness who believed that any person who turned against scientology could expect to die within twelve months. She tried to break away from scientology in the early stages of the Inquiry but she returned to the fold at a later stage partly, at least, because she feared she would suffer harm, even death, should she defect.

This fear of death is one of the elements which keep Hubbard's followers in subjection, and constrain them to continue in their acceptance of his teachings. They fear the human death which Hubbard has said will follow their defection, and they fear the loss of the immortality which acceptance of his teachings makes into a reality for them. Hubbard prescribes effective treatment as soon as doubts begin to arise. The waverer, while still a worried believer, is processed and his overts revealed and his continued subjection ensured by further indoctrination.

The Anderson Report CHAPTER 28 SCIENTOLOGY AND POLITICS

Hubbard writes, "Change no man's polities". The only qualification to scientology's claim that it is non-political is that it is wholeheartedly and enthusiastically opposed to Communism. Hubbard, however, has no love for governments in general. In *Ability* Magazine, Issue No. 5, Hubbard writes: "If we place the government on our chart of human evaluation we find a craven psychotic Governments are insane."

In Com. Mag. June, 1960, Vol. 2, No. 6, the Melbourne HASI reprinted an article by Hubbard entitled "Special Zone Plan". This article is capable of several interpretations: a commemoration of the tenth anniversary of some breakthrough of Hubbard, a review of progress during the previous ten years, an exposition on the eight dynamics, an enthusiastic appeal for more industry and activity on the part of scientologists, an appeal to exert more influence and control in circles or "zones" in which each scientologist moves, even a direction to take over governments. It had something for everybody, and scientologists read it and were impressed by whatever particular feature appealed to them.

For the housewife, Hubbard wrote:

"See this: a housewife already successfully employing scientology in her own home, trained to professional level, takes over a women's club as secretary or some key position. She straightens up the club affairs by applying comm exercises and making peace, and then, incidental to the club's main function, pushes scientology into a zone of special interest in the club - children, straightening up marriages, whatever comes to hand and even taking fees for it - meanwhile of course, going on being a successful and contributing wife."

For the junior executive, Hubbard wrote:

"Or this: a scientologist, a lesser executive or even a clerk in a company, trains as a professional auditor, and seeing where the company is heading, begins to pick up its loose ends by strengthening its comm lines or its personnel abilities Without 'selling' anybody scientology, just studies out the bogs [sic] and remedies them. If only as 'an able person' he would rapidly expand a zone of control, to say nothing of his personal standing in the company. This has been and is being done steadily across the world. Now that we have pre-sessioning, it's easy to straighten up other people. Our unreleased technology on handling third dynamic business situations is staggeringly large. You'd be surprised how easy it is to audit seniors. They and their families have so many troubles. Or how easy it is to spot the emergency-maker and audit him."

In respect of government, Hubbard wrote;

"And see this: a race is staggering along making difficulties for itself. Locate its leaders. Get a paid post as a secretary or officer of the staff of the leaders of that race. And by any means, audit them into ability and handle their affairs to bring co-operation, not trouble. Every race that is in turmoil in a nation has quasi-social groups around its leaders.

And this: a nation or a state runs on the ability of its department heads, its governors, or any other leaders. It is easy to get posts in such areas unless one has delusions of grandeur or fear of it. Don't bother to get elected. Get a job on the secretarial staff or the bodyguard, use any talent one has to get a place close in, go to work on the environment and make it function better. Occasionally one might lose, but in the large majority, doing a good job and making the environment function will result in promotion, better contacts, a widening zone.

The cue in all this is don't seek the co-operation of groups. Don't ask for permission. Just enter them and start functioning to make the group win through effectiveness and sanity."

Whatever may be Hubbard's aspirations in his special Zone Plan, it is evident that many scientologists believed that it was some sort of plan to dominate, by political means, countries and eventually the whole Earth. One preclear who had affiliations with the Australian Labour Party saw the Zone Plan as "a very able plan for infiltration and subversion of the key institutions of the country," the intention of the plan being "to create by those subversive means a scientology government"; and he was so enthusiastic about the possibilities which scientology offered for political domination that he concocted a plan to scientologize the Australian Labor Party. In a letter to Williams dated the 16th January, 1961, he wrote:

"My goals for the Zone Plan are to make my organization a Scientology Organization with all executives [being] HPA graduates, to use our publications to improve administration, management and communication in the Labor movement and interest the Australian Labor Party and Trade Union officials in taking scientology training. The Australian Labor Party as an organization using scientology principles would soon win a Government as soon as the next Federal election.

With Australia led by a government employing scientology principles we should soon have a civilization which can extend influence overseas."

He submitted his plan to Hubbard and obtained his approval. Mrs. Williams, who, according to *Com*. Mag. Vol. 2, No. 1 of January, 1960, "once vowed Australia will be the first scientology continent", approved of this preclear's ambition to scientologize the Australian Labor Party. His Zone Plan was not implemented. By a queer twist, this preclear sustained an "ARC break" with the Australian Labor Party, and while still in scientology he changed his political affiliations. Whether it was scientology which changed his politics does not appear.

Another executive had the ambition to take control of a large company and then to have as executives only those who were scientologists.

A police constable who resigned from the Victoria Police Force and occupied the post of "Director of Zoning" at the HASI, described the Zone Plan as a means of disseminating scientology in the various "zones" in which preclears moved, and he kept a card index to show the various zones - industrial, social, professional, &c, in which preclears moved.

In some of Hubbard's writings obscure references are made to the "1970 Plan". Williams said that the 1970 Plan was merely an expression used to describe the objective of the scientology organization to be reorganized within itself and "tidied up" by 1970.

The only overt political action of which evidence was given to the Board was in relation to the 1964 Victorian elections. The Melbourne HASI *Com.* Mag. Vol. 6, No. 5A for May, 1964, carried an article written by Williams, entitled "A Declaration of War", which stated: "It is the urgent duty of every scientologist in Victoria to get out and make certain that the Australian Labor Party is defeated completely and thoroughly and forever at the election." Williams said he wrote this article for humanitarian, not political, reasons, because the Australian Labor Party was hostile to the HASI which was entitled to defend itself.

The HASI did, however, explore the possibility of promoting scientology in various government departments, and a number of Victorian public servants have shown an interest in scientology. The number involved and the extent of their involvement is not known in detail, except that it would seem that the number is probably very small. The HASI considered the Education Department to be a good "area" and in the early days of scientology in Victoria some effort seems to have been made to infiltrate that Department, but there is no evidence that scientology made any headway in the Department as such, though a significant feature is that several highly placed staff members of the HASI are former school teachers. One scientologist, still a school teacher. formerly had her own

private school. She introduced scientology techniques into her school, apparently causing alarm amongst parents, and the school closed.

About nine years ago, a scientologist approached the Chief Probation Officer of the Children's Court, claiming that scientology could help to reform delinquent children. It is known that four children with Children's Court records went to the HASI. One, a retarded child, showed some signs of increased communication after doing some elementary exercises; a second child resumed his criminal activities. No information was available as to the effect, if any, which scientology had on the remaining two children. The Chief Probation Officer himself investigated scientology sufficiently to see that it offered no solution to the problems of child delinquency. The techniques used upon the child who became more communicative were of a very elementary kind; they cannot be claimed as exclusive to scientology and were not hypnotic procedures.

Hubbard has displayed great cunning in anticipating attacks upon himself and scientology by providing in advance a weapon with which to retaliate. Hostility to Communism finds a very prominent place in scientology. Scientology is almost rabidly opposed to Communism which Hubbard frequently denounces in flamboyant and intemperate terms. In doing so, he has been laying the foundation for his counter-attack on those who attack scientology, for it is guite standard technique for scientologists to stigmatize anyone opposed to them as communists, hoping thereby to blunt the attack on scientology and turn attention from scientology by the hostility they hope will be felt against its alleged communistic attackers. Scientology had its beginnings in the early 1950s in America in an era when McCarthyism was rampant, and Hubbard readily appreciated the value of discrediting any opposition and stifling criticism by denouncing any attack on scientology as communist-inspired. This technique has been repeatedly practised in Victoria. In one case at least the circumstances were almost amusing, except for the impudent threats which accompanied the attempts by the HASI to muzzle the critic. The occasion arose out of an answer given in 1960 by the Rev. Mr. L. Rumble, M.S.C, in a radio session called "Question Box" where, in answer to an inquirer who sought Dr. Rumble's advice on scientology, Dr. Rumble answered simply, "Have nothing to do with it", adding that "only credulous and gullible people will be impressed by the high-brow term, as a fruit-shop proprietor hopes simpletons will be by the description of himself as a 'Fruitologist' ". Dr. Rumble further said, "You flatter it by calling it a 'science'. Time magazine recently described 'scientology' as compounded of 'equal parts of science-fiction, dianetics and jabberwocky'!"

The reply was also published in a Catholic weekly newspaper, *The Tribune*, circulating in Victoria. Dr. Rumble's reply provoked two letters from the HASI, one to Dr. Rumble from Williams as Association Secretary for Australia of the HASI, and one to the editor of The Tribune, from Gogerly, who assumed for the occasion the pretentious title of "Director Government Relations". The Williams letter to Dr. Rumble read in part:

" by your utterance you have not only divorced yourself from the main international body of Catholic attitude to scientology but have allied yourself further with international subversive communism. By attacking scientology you are attacking one of the most ardent and vigorous groups in the field of combating communism and maintaining national security.

It is our policy to investigate subversives through our own channels as well as to co-operate with national security organizations.

Scientology has never been a threat to the honest and upright of any nationality or belief. Only those who tread dark paths or are on the payroll of a specialized interest seeking to profit by the sickness and troubles of man would fight a group of people trying to help man. May these sick and troubled offer their forgiveness."

Gogerly's letter to the Editor of *The Tribune* read in part:

"It is the policy of this Association throughout the world to investigate every attack on scientology. We have found

that the attackers have always had something to hide. Facts uncovered in these investigations have generally been given to government security services, police, etc. Dr. Rumble is now being investigated and has been apprised of this fact already. Any facts brought to light of interest to security services will be given to them. We generally in such investigations make public any facts which are of interest to the public, their safety, or elucidation."

The sequel to the threat to investigate and report Dr. Rumble is not known. There was no evidence that any such "investigation" took place and probably it did not, for few things would be more unlikely than a clandestine liaison between Dr. Rumble and Communism.

The HASI from time to time did in fact employ enquiry agents to investigate those it regarded as hostile to it. It appears that Dr. Dickson, the Medical Secretary of the Victorian Branch of the Australian Medical Association, was investigated by inquiry agents on behalf of the HASI to ascertain whether he had any communistic associations. He was found to be "clean" on this score. The Honorable J. W. Galbally, M.L.C. was also investigated by the inquiry agents on behalf of the HASI. What particular aspects were investigated did not appear from the evidence, but whatever it was, the result was also negative.

As part of his defence mechanism, Hubbard set up in America his own National Academy of American Psychology (NAAP), and in 1957 this organization impudently sent to all mental health practitioners in the U.S.A. - almost all of whom, if not all, were not members of or in any way associated with the "National Academy of American Psychology" and were unaware of its existence - a copy of the "Loyalty Oath", to which they were required to subscribe. This oath was objectionable in form and typically hubbardian in its inflammatory language. (See Appendix No. 18) The threat was made that those who did not subscribe were to be classified as "potentially subversive", those who "railed" against it were to be classified as "subversive" and their names submitted to government authorities for further action. This "oath" was to be submitted not only to all psychologists, psychiatrists and psychoanalysts, but also to "ministers of various denominations who engage in mental practice," and it was said that "a similar procedure may in the future be initiated in the British Commonwealth dependent upon the outcome of the American operation."

By this manoeuvre, Hubbard planned to anticipate attacks on himself and his organization by the American medical profession, and he was starting to arm himself for a counter attack with the retort that many or all American mental health practitioners were disloyal and possibly subversive since they had failed or refused to sign his "Loyalty Oath".

Such a scheme is in line with Hubbard's practice of "smearing" the founders of modern psychiatry and psychology by attributing to them communistic tendencies, and seeking thereby to discredit their sciences. The Melbourne HASI claims to be a member of N.A.A.P.

The Melbourne HASI claimed to be a valuable instrument for preserving national security, and that it co-operated with the Commonwealth Security Service on a number of occasions. This co-operation did not appear to extend beyond informing the Security Service of one or more pieces of information which a preclear or student had revealed concerning another person or persons. There is no reason to suppose that the HASI revealed anything to the Security Service about the preclear in question or that the air of mystery and importance surrounding the alleged disclosures to the Security Service was justified. The reticence of Williams on this matter rather suggested that it was merely another pretentious claim.

In his denunciation of Communism, Hubbard repeatedly refers to the Brainwashing Manual as an illustration of the manner in which Communism is striving for world domination. A curious aspect of scientology is that numerous passages in the Manual, with the substitutions indicated in Appendix 16 made, show a remarkable parallel between the brainwashing techniques attributed to

psychopolitics and Hubbard's techniques. Many scientologists, inspired by Hubbard, have the goal to scientologize and clear the whole world. Scientology techniques are directed towards domination of the mind. In this community, the normal person may regard as fatuous the thought that a sufficient number of persons could embrace scientology to make it a political menace to the community as a whole. The Board is not concerned to speculate, but, in a generation in which the conquest of the mind is the primary target, a new tyranny may well find the pernicious techniques of scientology amongst its armoury.

The Anderson Report CHAPTER 29: ON THE FRINGE

Such are the attractions of psychology and the mystery which surrounds it, that it readily lends itself to exploitation by persons who in fact have quite inadequate qualifications to justify their claim to understand and treat the mind. Hubbard and his organization have no monopoly in such exploitation, though they represent a massive proportion of self-styled authorities in this field. Many other persons in varying degrees have entered the field of mental health or skirmish on the fringe of it without adequate training, knowledge or experience.

The Board's terms of reference related specifically to scientology, but scientology is not the only potential danger. The Board heard evidence concerning other persons or organizations practising somewhat similar techniques under different names. In Chapter 30 are the names of some persons who have been scientologists, who now practise forms of psychology under other names, but who retain in their practices to a greater or lesser degree features of their scientology training. In particular, there are two organizations which are spreading from New South Wales and which have about their practices so many scientology features that they require individual mention in this Report. They are The American College and the Australian Centre of Applied Psychology, the proprietor of which is David Maxwell Tooley.

David Maxwell Tooley, also known as Marcus Tooley, carries on business in Sydney under the business names of The American College and Australian Centre of Applied Psychology. His addresses are 126 King Street and 187 Macquarie Street, Sydney. He first became interested in psychology while in a camp for conscientious objectors in New Zealand in about 1942. His acquaintance with dianetics and / or scientology began in 1950 in New Zealand where he attended a course run by one Turnbull, later "bishop" Turnbull. Tooley qualified as HPA in 1953, then went to America where in Philadelphia in 1954, he took his "Bachelor of Scientology". In the same year he became a "Doctor of Scientology" and obtained his "Doctor of Divinity" in Phoenix, Arizona, U.S.A., from the "Church of American Science", which is one of "the churches linked with Hubbardian scientology". On his return to New Zealand he conducted, separately from the HASI, some courses in scientology.

In 1955, he went to Sydney, New South Wales, where he established the "Scientology Centre", the name of which he changed to "The American College" after about twelve months. He denies that he was ever connected with the HASI or held a franchise; however, he was in good standing with the HASI and, amongst other things, taught scientology to students for examinations held by the HASI. He claims to have been excommunicated in about August, 1955, by Hubbard, but he attended the Fifth London Advanced Clinical Course, free of charge, at Hubbard's invitation in 1958. He was one of the auditors who contributed reports to a book produced by Hubbard and published in 1958, entitled *Have You Lived Before This Life?*, which contains some 43 case histories, being reports by auditors of preclears who have revealed past life experiences during auditing. Tooley's contribution related to a female preclear who had had experiences in Somaliland in 1630, when she had been a man, using anaesthetics while performing an operation, killing her (or his) wile and fighting a large lion with a black mane which killed the body which her thetan was then occupying.

In 1959, Tooley returned to Australia and resumed activities at The American College in Sydney. He admits that he has a couple of E-meters there, but says that he has never used an E-meter except

when on a Hubbard course. He states that the E-meters are kept merely for demonstration to students on advanced courses.

From about 1959, Tooley appears to have drifted from Hubbard and the state of suspended excommunication became effective. He claims to have developed his own form of psychology which he terms "dynamic psychology". He claims that it is based on the works of a variety of psychologists, that it is totally distinct from scientology and that he developed it quite independently of Hubbard and scientology. However, it contains axioms and dynamics which echo Hubbard's axioms and dynamics, though it has fewer axioms and more dynamics. Many of its terms are identical, it has a code similar to those of the auditor and the scientologist, it conducts IQ tests, it uses the Oxford Capacity Analysis, the copyright of which is in Hubbard; it has affinity, reality, communication, control, confronting, havingness, acknowledgments, infractions and a variety of other scientology features. Several of its techniques are similar to those of scientology; the coach "starts" a "session" much as a scientologist would; the "Dear Alice" training drills are parallel, specific extracts from *Alice in Wonderland* being prescribed by Tooley for such drills. There are drills which are almost identical with the CCH's and SCS. Tooley, though disclaiming any affinity with scientology, is frequently referred to as "Doctor" Tooley, the only basis for such title being the Hubbardian bestowed "Doctor of Scientology" and "Doctor of Divinity".

Even if Tooley's dynamic psychology was not largely borrowed from scientology, it contains many of the undesirable features of scientology; but the similarity between so much of scientology and dynamic psychology is explained rather by the fact that Tooley has pirated much of Hubbard's distorted appreciation of psychology than that he has, independently of Hubbard, arrived at much the same destination as Hubbard. In 1956, Tooley was the author of a publication entitled *New Hope for Man* which dealt specifically with dianetics and scientology and credited Hubbard with being the founder of scientology. In 1963, Tooley produced a "dynamic psychology" publication entitled *Living Successfully*. Very substantial parts of the texts of these two publications are identical. Tooley has merely applied another name to his earlier scientology writings. He even learned this technique from Hubbard who produced *Dianetics, The Evolution of a Science* in about 1950 (reprinted in 1958) and *Scientology, the Evolution of a Science* in 1950 [?]. Apart from the names, these two books are so identical in almost every respect that it requires word for word examination to ascertain the few instances where there are verbal differences.

The relevance of Tooley to scientology in Victoria is that in November, 1963, he established at 21 Bourke Street, Melbourne, two businesses of which he is the proprietor, namely, The American College and Australian Centre of Applied Psychology, being branches of his Sydney organizations. His Melbourne manager is Michael Andrade Birch.

The Board heard evidence from a scientologist and also from a senior policewoman, who in January and February, 1964 respectively attended courses of free lectures given by Birch at 21 Bourke Street, Melbourne. From this evidence and from the literature of "dynamic psychology" as sought to be propagated by Tooley, it is clear that it closely parallels scientology in many of its features.

Due to the attention which has been directed at scientology, The American College in Melbourne has not proceeded beyond the elementary stages of practice. A course of fourteen lectures in applied psychology and human relations is available at The American College for a fee of £24 terms, or £22 cash. A brochure relating to the course indicates that the course contains many of the features found in scientology. It is quite evident that The American College and Australian Centre of Applied Psychology are marking time until this Inquiry is over. Tooley has picked out of scientology many of its theories and techniques, abandoning, however, the ludicrous features but retaining

many of the techniques which are potentially harmful to mental health.

The incursion of Tooley into Victoria in the guise Of The American College illustrates the problem involved in seeking to take effective steps to control the dangers resulting from the unskilled practice of psychological techniques. for such practitioners may readily shed titles and descriptions which have become opprobrious and assume some other imposing or deceptive title, under which they are enabled to engage in the same discredited and dangerous techniques as those formerly practised .

It is organizations and individuals such as The American College and Tooley which constitute a potential danger to the mental health of the community, differing only in degree and intensity from the danger which scientology is.

The Anderson Report CHAPTER 30: CONCLUSIONS

If there should be detected in this Report a note of unrelieved denunciation of scientology, it is because the evidence has shown its theories to be fantastic and impossible, its principles perverted and ill-founded, and its techniques debased and harmful. Scientology is a delusional belief system, based on fiction and fallacies and propagated by falsehood and deception. While making an appeal to the public as a worthy system whereby ability, intelligence and personality may be improved, it employs techniques which further its real purpose of securing domination over and mental enslavement of its adherents. It involves the administration by persons without any training in medicine or psychology of quasi-psychological treatment, which is harmful medically, morally and socially.

Its founder, with the merest smattering of knowledge in various sciences, has built upon the scintilla of his learning a crazy and dangerous edifice. The HASI claims to be "the world's largest mental health organization". What it really is however, is the world's largest organization of unqualified persons engaged in the practice of dangerous techniques which masquerade as mental therapy.

No acceptable or recognized standards are prescribed for its practitioners, whose ignorance of orthodox medicine and psychology make them each a menace to the health of the community. They are the more dangerous because of their spurious air of competence and the tremendous amount of mis-directed energy which has gone into promoting the organization and devising techniques, the mentally crippling qualities of which are cleverly concealed.

That many scientologists sincerely believe in the virtues and the efficacy of scientology is apparent from the evidence. Some have become so dedicated to it and have served it so faithfully that their sacrifices cannot but excite compassion. These ardent devotees, though quite rational and intelligent on other subjects, are possessed of an invincible impediment to reason where scientology is concerned.

If one considers self-dedication a criterion, those persons deeply involved in scientology are broadly of two kinds. In the first place, there are professional scientologists who have forsaken many careers, interests and comforts in order to "live" scientology in every sense of the word. Of such a kind were many of the officials of the HASI and the proprietors of the Geelong and Hawthorn centres. ~ many instances they are husband and wife teams, occasionally with young families. Their dedication to scientology has involved great sacrifices, which, however, they were prepared to make to reach and maintain their role as the elite of a New Order. Though they were egregiously wrong in their reasoning and outlook, though they suffered and are suffering the ignominy of exposure and ridicule, many of them are still as fanatically adherent as they were when the Inquiry began, and they are still prepared to immolate themselves for their ideals. They spurn the pity one feels for them; the prospect of their rescue from their deluded state is problematical, since their continued intimate association with others of a like mind will probably result in their being permanently conditioned by auditing, co-auditing and other oppressive and compelling features of the HASI machine. Few, if any, of these dedicated scientologists whom the Board heard and saw seemed to be conscious that they were perpetrating a fraud masterminded by Hubbard. They seemed to be deluded, mistaken and almost innocent tools.

The other class of people who have been dedicated to scientology are those who have undertaken long courses of processing and training but who have not taken up scientology as a profession, though from time to time they may have been on the HASI staff and they may have done some auditing. While they have continued to be in communication with the HASI, their delusions and subjection to the HASI have been maintained. Their awakening will be distressing, for they once more must face the realities of life. As they emerge from the miasma of scientology there may be some residual effects, for Hubbard's ideas may be so entrenched in their minds that they will still feel compelled to shun the proper medical and other treatment which they may well require.

The overall picture of scientology is thus one of grave disquiet. In the foregoing Chapters of this Report, the Board has sought to deal as fully as the circumstances required with all relevant aspects of scientology in Victoria, and, with one exception, the matters raised in the twelve numbered paragraphs of the terms of reference have been already sufficiently discussed to make unnecessary anything more than a final comment and reference to those parts of the Report where the particular subject matter has been dealt with more extensively.

The one exception is in relation to paragraph 11 of the terms of reference which deals with scientology and children. This aspect is not dealt with in detail earlier in the Report because no extensive evidence was forthcoming in relation to it. Having regard to the effects which scientology processing has produced, not only on mentally disturbed adults but also on persons who were well within the range of normality, the potential danger of scientology processes to children has merely to be mentioned to be appreciated. It remains now to comment briefly on the subject matter of the twelve numbered paragraphs of the terms of reference.

- 1. The aims, objects and purposes thereof and whether it is capable of achieving to any and what extent such aims, objects and purposes.
- (a) Aims, Objects and Purposes.

The professed aims, objects and purposes have about them a laudable quality. In essence it may be said that the aim of scientology is to make the able more able.

(i) The stated purpose of the Hubbard Association of Scientologists International is "To disseminate scientology, to advance and profit its membership. To hold the lines of data of scientology clean and clear. To educate and process people towards the goal of a civilized age on earth second to none, and to survive on all dynamics."

In respect of Australia, the object is to make Australia the first scientology continent or nation, but the broad objective of scientology is to clear the whole world.

- (ii) The more immediate aim of scientology is said to be to make the able more able. This aim is variously described and presented to the public as a worthy method of improving personal efficiency; in reality this supposed aim is the means whereby unsuspecting persons are lured into the organization to be exploited by fraudulent practices and subjected to techniques dangerous to mental health.
- (iii) In a covert way, scientology and its companion science", dianetics, claim the ability to cure a great number of physical and mental ailments.
- (b) Whether scientology is capable of achieving such aims, objects and purposes.

- (i) The state of "clear" is a condition variously described and defined but may be simply expressed as free of all aberrations. It is a condition said to be capable of being produced by scientology auditing, but it is a fanciful and imagined state of ecstacy or exhilaration. During and for a short period after scientology auditing some few people have claimed to be clear, but the state is an illusory one. A concomitant of this illusory state is a large measure of detachment from reality. The state of "clear" is not attainable by scientology practices. What is claimed to be the state of clear is an euphoric state, where the mind of the "clear" is divorced from reality.
- (ii) As to the more immediate objective of making the able more able, it must be said that scientology achieves no real success in this respect. All but the most elementary of scientology processes are of a hypnotic nature and they are conducted in a manner calculated to have harmful effects on those subjected to them. The effect of such processes may be to give to the individual on whom they are practised a temporary and subjective feeling of elation, but no permanent benefit is obtained and no increased abilities result. Some claim is made that persons who have been processed obtain, in effect, a better understanding of themselves and are more able to cope with their problems, but the Board found such evidence unconvincing. Many of the individuals who claimed to have benefited were still under the influence of processing which they had recently undergone.

At the lowest level of scientology, on what was called the personal efficiency course, the HASI conducts free lectures from which a number of individuals claimed to have benefited. The elementary techniques practised at this stage are not exclusive to scientology, but are comparable to methods available to and used in schools, clubs and welfare and social organizations for the benefit of members and others who need and find help and comfort from companionship and friendship. Such benefits as resulted from the personal efficiency department were not attributable to scientology processes, but to the application of well known and well understood psychological principles.

- (iii) Neither scientology nor dianetics is capable of making good the claims they make for curing physical or mental ailments.
- (iv) Far from being the boon to mankind which it claims to be, scientology and its practices are harmful. The aims, objects and purposes are not capable of attainment. They are fictitious aims based on fantastic and illusory theories and on techniques which are potentially, and in many cases positively, harmful to the mental health of the community.
- 2. The persons and organizations by whom and the place or places at which the same is so carried on, practised, or applied.
- (a) The Hubbard Association of Scientologists International, a company incorporated in Phoenix, Arizona, U.S.A. and registered in Victoria as a foreign company, has carried on, practised and applied scientology at 157-159 Spring-street, Melbourne since about 1957, and is still currently so engaged. For a few years prior to 1964, until a date in the latter part of 1964, this company also carried on, practised and applied scientology at 2 Coates Lane, Melbourne.
- (b) Several business names have been registered which relate to scientology activities in Victoria. As the names of their proprietors and their addresses indicate, they are connected with the HASI. They are:

Melbourne Test Centre, proprietor, Peter Rogers Williams

Test Bureau of Australia, proprietor, Peter Rogers Williams

Academy of Personal Efficiency, proprietor, Lafayette Ronald Hubbard

Hubbard Guidance Centre, proprietor, Lafayette Ronald Hubbard

Academy of Scientology, proprietor, Lafayette Ronald Hubbard

Hubbard Communications Office, proprietor, Lafayette Ronald Hubbard. In each of the above cases, the address has been 157 and/or 159 Spring Street, Melbourne.

On the 15th February, 1965, the College of Applied Philosophy was registered, the proprietor being the Hubbard Association of Scientologists International, and the address 157-159 Spring Street, Melbourne.

- (c) Mrs Katherine Steele, formerly the proprietress of a private school at S Mount Street, Heidelberg, called "The Sylvester Co-Educational College", practised or applied scientology on pupils in the school from about 1955; to what extent could not be determined. The school subsequently closed about 1959. Mrs. Steele practised scientology at the above address in respect of a few children in 1960-61. Though still an ardent scientologist, she did not appear to be in practice as an auditor at the time she gave evidence in September, 1964.
- (d) Eric Bruce Lake and June Margaret Lake conducted at 68 Yarra Street, Geelong, The Australian College of Personnel Efficiency from September, 1959. They originally held a franchise from the HASI to practise scientology and later had the status of a district office. They were still in practice as scientologists when they gave evidence before the Board in October, 1964. Prior to September, 1959, Peter John Crundall and Phyllis Dawn Crundall had been in partnership with the Lakes in this venture.
- (e) Peter Francis Gillham and Yvonne Doreen Gillham have practised scientology in partnership since the 17th February, 1959. They formerly carried on business under the business name of The Melbourne College of Personnel Efficiency, but since the 1st June, 1961, the name has been The Melbourne College of Personal Efficiency. For several years until about the 12th December, 1964, they carried on business at 39 Riversdale Road, Hawthorn. Thereafter, they have carried on business at 235 Camberwell Road, Hawthorn. From the 1st June, 1961, Peter Francis Gillham has also traded as the Hawthorn Test Centre.
- (f) James Hamilton Momsen traded under the business names of The Momsen School for Marriage Education and Momsen School of Personal Efficiency at 2 Hartley Avenue, Caulfield, from the 12th April, 1960, and the 1st July, 1960, respectively to the 1st December, 1961. Thereafter he conducted and still conducts the Brahma School of Yoga at 407 Nepean Highway, Frankston. He describes himself as a hypno-therapist and psychologist. He studied scientology at the Hubbard organization in London in 1957, and obtained his HPA certificate. He does not hold a franchise from the HASI and denies that he currently practises scientology and claims to have evolved his own techniques. He has, however, practised scientology, which has influenced the techniques he now uses.
- (g) D'Arcy Swain Hunt, of 526 St. Kilda Road, Melbourne, who advertises as a psychologist and ability consultant, claims to have obtained in 1929 the degree of Bachelor of Arts or an equivalent degree at the Stamford University, U.S.A. His study in psychology was limited to his first year. He qualified for the HPA certificate and was intimately associated with dianetics and scientology for

several years. He was admittedly a dianeticist and/or scientologist from about 1951, when he arrived in Australia, until at least 1956. In his present business he does not now use an E-meter though he did use an E-meter up to 1956. His charges are £3 per hour. He denies that he is now a scientologist; however, some at least of his present procedures possess features which are found in dianetics and scientology. For the purpose of treating some mental conditions he still considers that it would be immaterial whether the treatment be given by a scientologist or a psychologist.

- (h) Roger David Meadmore and Helen McKinnell Meadmore conducted from the 1st February, 1960, to the 26th February, 1961, the business of Roger Meadmore and Associates, personal efficiency consultants, at 39 Riversdale Road, Hawthorn. Thereafter, they conducted the business successively at 505 St. Kilda Road, Melbourne, and 474 St. Kilda Road, Melbourne. They appear to have been inactive since about 1962. They held a franchise from the HCO and part at least of their activities was the practice of scientology, the fees for which ranged from £2 2s. to £10 10s. per hour.
- (*i*) Clifford Edward John Hardy, by occupation a boilermaker, conducted at 336 Victoria-parade, East Melbourne, from the 7th April, 1960, the business of applying and teaching scientology. For this purpose he traded under the names of College of Ability Development arid Problem Clinic. He appears to have had very few clients.
- (j) Between 1957 and 1958, a few people the precise number is uncertain endeavoured to set up in business as professional scientologists. They do not appear to have succeeded.
- (k) David Maxwell Tooley conducts at 21 Bourke Street, Melbourne, The American College and the Australian Centre of Applied Psychology. This man and his businesses are dealt with at length in Chapter 29.
- 3. The methods, processes, practices, techniques, principles and treatment involved or used or applied in the carrying on, practice, or application thereof.

These aspects are dealt with extensively in preceding Chapters of this Report. Particular reference is made to Chapter 12, dealing with the teaching of scientology; Chapter 13, dealing with scientology auditing processes; Chapter 14, dealing with the E-meter, and Chapter 18, dealing with the parallel between scientology and hypnotic techniques.

4. The circumstances in which and the conditions under or in accordance with which the same is carried on, practised or applied.

Scientology is carried on, practised and applied

- (a) in lectures;
- (b) in private session with only the student or preclear and the coach or auditor present;
- (c) in co-auditing sessions wherein one student audits another. A strict routine is followed in auditing sessions and frequently hypnotic techniques are employed. The matters mentioned under this heading are dealt with extensively in Chapters 12, 13, 14 and 18.
- 5. The circumstances surrounding, the principles involved in, the conditions relating to, and the methods applied in the teaching thereof.

A distinction is to be drawn between -

- (a) the giving or receiving of scientology processing and
- (b) the teaching of scientology principles and practices to enable students to attain scientology qualifications and so practise on others.

A person who is not and has not been a student may undergo elementary processing, but advanced processing is not available to him unless he is a student whose education in scientology is of a standard appropriate to the processing techniques involved. It is a rule that a person may only be processed at a level appropriate to his knowledge of scientology. Accordingly, eligibility for anything beyond elementary processing involves a requirement that the individual shall have been a student at the Academy of Scientology. Auditing is conducted both at the Academy and at the Hubbard Guidance Centre. An individual may make his introduction to scientology by being introduced by a friend, through free lectures conducted by the HASI, or through the Melbourne Test Centre, or some other associated concern which advertises IQ and personality testing and offers a variety of scientology advantages.

The teaching of scientology is dealt with at length in Chapter 12. 6. The nature and amount of fees charged or remuneration received by persons or organizations engaged in the carrying on, practice, application and teaching thereof.

Fees are charged both for administering processing and training students in scientology theory and techniques. At the HASI processing fees have varied from time to time. The ordinary rate was £4 14s. 6d. per hour for processing, but in various ways concessions were available because of various memberships, &c. and also where a "package deal" was done for a course of auditing for a specified period. For instance, an " intensive" of 25 hours was available for amounts ranging around £100; where long courses of processing were undertaken, e.g., for 250 or 300 hours, the rate was lower, in the region of about £2 per hour for the hours of processing actually given.

For processing, the Geelong organization has charged fees generally ranging from 3 to 41/2 guineas per hour.

For processing, the Hawthorn organization has charged up to 41/2 guineas. Other smaller charges were made for co-auditing and particular courses.

Individual auditors in "private practice" have charged various fees ranging around £2 2s. per hour. One such practitioner for a short period charged £10 10s. per hour.

For training, specific charges were made for individual courses. At the HASI, the following were the fees charged up to November, 1964

Hubbard Qualified Scientologist Course (HQS) £40 cash or £44 on terms Hubbard Professional Auditor Course (HPA) £140 cash or £160 on terms

From November, 1964, the scale was

Hubbard Qualified Scientologist £75
Hubbard Certificated Scientologist £75
Hubbard Professional Auditor £75
Hubbard Clearing Scientologist (a course in contemplation) £100

The actual amounts charged for processing and training were infinitely variable, the figures given being either the advertised, standard or normal figures.

A negligible amount of training was done at the Geelong and Hawthorn organizations.

For six years to the 30th June, 1963, the income of the HASI was £273,373.

Full details of the financial position of the HASI and of the Geelong and Hawthorn organizations appear in Chapter 5.

7. The instruments, machines, devices or other apparatus used in the carrying on, practice, or application thereof and the manner of use of and the effectiveness of the same.

The only instrument, device or apparatus used is an electrical instrument called an E-meter, for which fantastic claims are made that it records the reaction of the thetan to questions asked. In practice, the preclear holds two electrodes attached to the meter and is thus part of the circuit the auditor asks the preclear questions and the claim is made that the meter records the reaction of the preclear's reactive mind. A variety of deductions are then drawn from the reading of the meter. In reality, the E-meter is akin to a lie detector and is an efficient instrument for recording electrical resistance, and nothing more. The claims made for the E-meter are entirely unjustified.

This subject is dealt with at length in Chapter 14.

- 8. The illnesses or ailments, mental or physical, treated in the course of the carrying on, practice or application thereof, the effectiveness of such treatment, and whether such treatment is in any way harmful or beneficial or likely to be harmful or beneficial to the persons treated.
- (a) The claim is made for dianetics, which is part of scientology, and inferentially for the whole of scientology, that between them they can positively cure all psychosomatic ailments, which it is claimed represent 70 per cent. of man's illnesses.
- (b) These claims are entirely unjustified.
- (c) On the contrary, scientology techniques, beyond the elementary stages, are potentially and, in some instances, positively harmful to mental health.
- (d) Scientology is practised by "auditors" who have no medical training; they use dangerous techniques; they are unable to recognize symptoms and diagnose particular mental and physical conditions of ill health; they indiscriminately apply dangerous techniques irrespective of the circumstances; they not only administer the wrong treatment, but also poison their patients' minds against orthodox medicine and thus prevent them from obtaining proper medical treatment which they may require.

These aspects are dealt with in Chapters 19, 22 and 23.

9. The advertisements used in the carrying on, practice, or application thereof and whether the same have been in any and what manner false or misleading, and whether the same should be prohibited, regulated, or restricted in any and what way.

With few exceptions, in recent years the advertisements addressed to the public generally in the press, pamphlet and preliminary brochure have been confined to offering free lectures, with "no

obligation" and "no financial obligation", and promising to make the able more able, to increase confidence, improve IQ and personality, &c. These advertisements have not mentioned advanced scientology theory or techniques, but have been of the "personal efficiency" kind, appealing to the anxious, worried, lonely and ambitious.

Such advertisements are part of a large scale deception, and are designed to inveigle unsuspecting persons into the organization, where high pressure techniques are designed to coerce the gullible..

In advertisements with a more limited distribution addressed to scientologists and to persons on mailing lists, reference is made to scientology processing and courses, and extravagant claims for scientology and dianetics in relation to healing and to attaining the states of clear and OT are common. These advertisements appear in such documents as the monthly Melbourne HASI magazine *Communication*, which frequently contains articles by Hubbard. The advertisements are false in that they offer the impossible and unattainable goals of scientology, promising attainment through scientology processing.

Books by Hubbard and others on scientology and dianetics are available at the HASI and at some booksellers for purchase by the general public. These books are, of course, a form of advertising of Hubbard's "sciences". Several of these books deal with advanced features of scientology and dianetics.

Advertising by the HASI is dealt with in *Chapter 15*.

The question of prohibition, regulation or restriction of such advertisements is dealt with in Chapter 31, where the broad topic of advertising generally is considered.

10. Whether persons or organizations engaged in the carrying on, practice, or application thereof have indulged in any and what unlawful, improper, harmful or prejudicial practices or activities.

This term of reference was taken to refer to the practices of scientology and the activities of scientologists *qua* scientologists.

In scientology there is a very great scope for coercion. Very intimate matters are revealed by preclears in the course of auditing and records are kept by the organization of such revelations.

There was no evidence that blackmail in the legal sense had been practised; but the basis for such criminal activity exists in the files of the organization and preclears could readily be criminally exploited, though at this stage there is no evidence that it has happened.

However, that coercive use is made of the incriminating confessions contained in the HASI records is undoubted. These confessions, which are often of secret weaknesses and feelings, misconduct and prurient ambitions, are made in such circumstances that the organization develops domination over preclears.

The nature of the coercive use to which the HASI records could be put is dealt with in Chapter 24.

Though the practice of scientology has many undesirable features, such is the novelty of many of its activities that it is difficult to classify them precisely as being in breach of existing laws. That scientology practices and activities are improper and are harmful and prejudicial to mental health is evident. This is the most important and most serious feature of the practice of scientology. It is a

large subject, and is the burden of most of this Report; it is dealt with especially in Chapter 23.

11. Whether children under the age of sixteen years have been treated by any and what persons or organizations in the carrying on, practice, or application thereof the circumstances and conditions under which such children have been treated and whether such treatment has been or was likely to be in any and what manner harmful or beneficial.

The application of scientology processing to children does not appear to have developed in Victoria to any substantial extent.

What seemed to be half-hearted attempts were made to establish Saturday morning classes for children both at the HASI in Melbourne and at Geelong. These classes dealt with only the most elementary procedures, which were not novel or exclusive to scientology and were designed to do little more than teach children to mix and play games; they were psychological techniques and were of a kindergarten nature.

At the Melbourne HASI, Saturday morning classes were conducted from 1961 to late 1962 about 15 to 20 children at a time, aged 8 to 14 years, attended. Saturday morning shoppers rather used the HASI as a creche. These classes were not successful; the children exhausted the auditors and the classes were discontinued.

The evidence showed that in a few instances children with one or both parents in scientology received scientology processing. They received such processing with the approval of at least one parent. The number of children involved was small, the techniques do not appear to have been advanced and the Board did not hear evidence that in fact harm did result from such processing. It appeared from the evidence that one scientologist, Mrs. Katherine Steele, who conducted a private school, introduced scientology techniques into her establishment. Some of these techniques were potentially harmful, though the Board heard no evidence that lasting harm was done to any student at the school. Parents of students became aware of and apparently alarmed by the practices being engaged in, and the school closed. Mrs. Steele also had as private clients a few children to whom she applied scientology techniques. Mrs. Steele was herself "addicted" to past experiences involving extraordinary encounters with electrical phenomena; she claimed, amongst other things, to have been struck by lightning at least twice; her peculiar acquaintance with electricity was reflected in past experiences which her suggestible child preclears revealed when she processed them. The Board did not hear any evidence of permanent harm to these particular preclears.

The Board heard evidence from a school teacher from South Australia, a woman dedicated to her task of teaching brain-damaged children, who gave evidence of some success she had at such work by the use of what she claimed were scientology techniques. However, from the evidence of an expert psychologist of wide experience both in Australia and abroad, it is evident that the methods claimed to be scientological are well known and commonly practised in various parts of the world. They are not in any way the discovery of scientology, but are virtually identical with principles and methods which were known and developed before scientology existed. As already mentioned, the alleged scientology methods of handling young children are those of school and kindergarten psychology. The Board heard no evidence of alleged scientology treatment being applied to brain-damaged children in Victoria.

As to older children under sixteen years, the Board heard very little evidence. Though the public was invited to come forward with any helpful evidence, there was no public response as to the teaching of scientology to children, and the Board on this aspect had to rely, in the main, on evidence of scientologists and the contents of a very few files. However, there seems to be no reason to

suppose that the application of scientology to children in Victoria assumed any substantial proportions; and, except in cases where a parent is a scientologist, it is probable that children are not now being brought into contact with scientology. However, scientology is potentially the same danger to children as to adults, and the absence of positive evidence may well be due to the fact that the practice of scientology in relation to children has been desultory and negligible in Victoria. There is no reason to suppose that its application to children has been in any way beneficial.

A former Chief Probation Officer of the Children's Court, about eight years ago, investigated scientology sufficiently to satisfy himself that it was of no value in treating child delinquents. Of four delinquent children who had some treatment by scientologists, one appeared to obtain some benefit, one lapsed into criminal habits and no information was available as to the other two. The techniques applied were elementary and were not novel to scientology.

12. The terms and conditions under which and the circumstances in which persons are engaged for employment and employed by persons or organizations carrying on, practising or applying the same.

In Victoria, the Melbourne HASI at 157-159 Spring Street, Melbourne, was the only substantial employer of labour. The Geelong and Hawthorn organizations were each a husband-and-wife partnership whose staff establishments were very small; on the occasions when staff was employed, generally only nominal payments were made; some of the help was even voluntary. At the Melbourne HASI, the staff was remunerated on a unit basis, each member being allotted a particular number of units according to the position which he or she held. Staff was paid each week according to the number of units held.

The amount available for distribution amongst the unit holders was a fixed percentage of each week's receipts. it was a variable amount and the unit value varied accordingly. There were thus good and bad weeks. At the beginning of the Inquiry the staff of the HASI numbered approximately 70, some full-time and others part-time. The most senior and highest ranking in the organization were receiving an average of about £17 per week, and other full-time employees were receiving about £11 per week. Part-time employees were, of course, receiving less. A few appear to have been voluntary. Frequently, the weekly wage of an individual staff member fell far below the basic wage. By mid-November, 1964, the staff of the HASI was only about one third of what it had been at the beginning of the Inquiry, and the unit had fallen to 1s. 9d. At that time the chief executive officers of the HASI were averaging about £5 per week or less. Some may have been receiving slightly in excess of this amount by performing the duties of more than one position.

The remuneration of the HASI staff was not liberal. Long hours were frequently worked weekend work was common. Hubbard appears to have made it a practice to increase staff whenever the financial position of the HASI improved, thereby perpetuating the depressed rate of remuneration.

Further details of remuneration appear in Chapters 5 and 17.

The Anderson Report CHAPTER 31 RECOMMENDATIONS

The Inquiry has revealed the real nature of scientology and its serious threat to the mental health of the community, and it is evident that its continued practice should not be permitted.

There are indications that, temporarily at least, there has been a substantial curtailment of scientology activities in Victoria, due to unfavourable reaction to the publicity necessarily attendant upon the Inquiry. This has been reflected in the big decline in the income of the HASI, the drastic reduction in its staff and the discontinuance of the Academy of Scientology at 2 Coates Lane, Melbourne. There is reason to believe, also, that the Geelong and Hawthorn centres are now quite attenuated.

It appeared from the evidence that to be a scientologist has in some instances attracted ridicule and discrimination. To be, or to have been, a scientologist, should not, of itself, carry any opprobrium. For many people, dalliance in scientology has been an unhappy interlude-a somewhat unnerving experience unwittingly embarked upon and thankfully escaped from. Hubbard claims that once a person has been in scientology he is really never the same again; this may well be so because, when disillusionment dawns, there may still remain the seared memory of chastening experiences in scientology. Very often persons in scientology have been, and still are, normal people who, however, have lapsed into gullibility and have had disquieting experiences without which they would have been far better off. It is unfair to ostracize or otherwise punish these innocent victims of scientology's deception. "There, but for the grace of God, go 1," should be the fervent prayer of many who resisted the urge to sample scientology.

Unless some action is taken it is unlikely that the fortunes of scientology in Victoria will remain at tl1eir present low ebb for very long. Hubbard is very resilient; his organization is international, large, efficient and powerful, and can comfortably absorb the reverse which it is presently suffering in Victoria. Hubbard still has loyal henchman in Victoria. As well as the disillusioned persons mentioned above, there are those who are the hard core of scientology in Victoria. They are the professional scientologists; they may well be deluded and their dedication may be sincere, but they are the people who have been keeping the flag of scientology flying during the Inquiry, who sought to underplay certain features of scientology theory and practice - " to go very slowly and use watered down scientology" - and who proclaim in the Melbourne HASI *Communication* Magazine of January, 1965, that "We will be here teaching and listening when our opponents' names are merely misspelled references in a history book of tyranny".

Notwithstanding that their revenue has almost ceased and their numbers are decimated, they are waiting to re-establish themselves along much the same lines as formerly, though possibly under some disguise. The registration of a new business name by the HASI in February, 1965, namely, "The College of Applied Philosophy", almost suggests that plans are being laid for scientology to blossom forth as a new philosophy.

Scientology and Religion.

In considering the nature, extent and effect of recommendations which it deems proper to make, the Board has constantly kept in mind the distinction between the holding of beliefs on the one

hand and harmful practices based on such beliefs on the other. Such a distinction assumes added importance when the body of beliefs under consideration may reasonably be described as a religion. It is quite clear, however, that scientology in Victoria does not remotely resemble anything even vaguely religious, and no serious claim was made that it did.

Even if scientology could reasonably claim to be a religion, such are its practices that government action may be required to curtail or prohibit certain of its activities. In Adelaide Company of Jehovah's Witnesses Inc. v. The Commonwealth (1943) 67 C.L.R. 116, (1943) A.L.R. 193, the High Court of Australia unanimously ruled that, notwithstanding section 116 of the Australian Constitution which provides that " The Commonwealth shall not make laws for establishing any religion, and no religious test shall be required as a qualification for any office or public trust under the Commonwealth", the Parliament of the Commonwealth was not prevented from making laws prohibiting the advocacy of doctrines or principles which, though advocated in pursuance of religious convictions, were prejudicial to the prosecution of a war in which the Commonwealth was engaged. Thus, even where there are constitutional guarantees as to freedom of religious beliefs the advocacy of such beliefs may be curtailed or prohibited in the national interest. In the course of his judgment, the Chief Justice, Sir John Latham, referred with approval to a number of decisions of the Supreme Court of the United States dealing with restrictive action which the government was entitled to take to curtail or punish allegedly religious practices, even though constitutional safeguards for the freedom of religious belief existed. One such case, Reynolds v. The United States (1878) 98 U.S. 145, will serve as an illustration. In that case, Reynolds, a Mormon who had a religious belief in polygamy, and who had more than one wife, was convicted of bigamy. In announcing the unanimous decision of the Court, which refused to set aside Reynolds's conviction, Waite, C. J. said at p. 166, "Suppose one believed that human sacrifices were a necessary part of religious worship, would it be seriously contended that the civil government under which he lived could not interfere to prevent a sacrifice? Or if a wife religiously believed it was her duty to burn herself upon the funeral pile of her dead husband, would it be beyond the power of the civil government to prevent her carrying her belief into practice? So here, as a law of the organization of society under the exclusive domination of the United States, it is provided that plural marriages shall not be allowed. Can a man excuse his practices to the contrary because of his religious belief? To permit this would be to make the professed doctrines of religious belief superior to the law of the land, and in effect to permit every citizen to become a law unto himself. Government could exist only in name under such circumstances ".

Where there are no constitutional guarantees, the position of the State would be at least as strong.

Inadequacy of present laws.

There are many features of scientology which are to be condemned, but its disturbing features possess a certain novelty, and many of its aspects, criticized in this Report, are merely on the fringe of existing laws which have not been designed to deal with the particular problems which scientology presents.

The Board heard a submission by the Committee for Mental Health and National Security that much that the HASI had done was actionable civilly in fraud, negligence, for breach of contract and assault. The Board expresses no views on individual rights in respect of remediable grievances which persons may consider, or be advised, they have against the HASI or individual scientology practitioners.

The Board also heard submissions by the same Committee that the activities of the HASI and of scientology practitioners have revealed a variety of offences, including treason, sedition,

blasphemy, assault and conspiracy. Conduct of which one disapproves may readily be stigmatized by being vaguely denounced as criminal, without regard to the precise constituents of criminal offences and the insuperable difficulties of evidence and proof which often exist. The Board does not consider that the problems which scientology presents may appropriately be dealt with by invoking the criminal law in respect of the past conduct of the HAS1 and of scientology practitioners.

It is of this view for these reasons:

- (a) the real problem of scientology is not one of repeated and clearly defined breaches of existing laws, but rather of a special course of conduct not adequately controlled by law, namely, the invasion of the field of mental health and the propagating of harmful psychological practices. Such features as may be contrary to existing laws are merely incidental to the overall problem;
- (b) there are difficult questions of law as to whether such offences as have been enumerated and other similar offences have been committed and are capable of proof;
- (c) an unsuccessful prosecution would be claimed as a vindication of the past conduct of the HASI and of scientology generally;
- (d) even a successful prosecution for a breach of an existing law would be of little significance-like prosecuting a bank robber for driving his get-away car against the traffic lights.

The Board is of the view that scientology presents a problem which is not to be adequately dealt with by peripheral sniping. The Inquiry has revealed that many people, not necessarily scientologists, have taken advantage of the virtual absence of any control over practices or techniques which are concerned with the treatment of the mind and mental health generally. As the Board sees the matter, the problem which the Inquiry has revealed is not confined to practices and techniques which, for the time being, are labelled "scientology", but extends to a variety of pseudo-psychological practices which flourish under a variety of names.

Scientology, it is true, embodies a large number of fantastic theories and it so happens that the processes of scientology have been built around these theories. However, the processes of scientology could be administered with equal harm without one word of scientology theories being mentioned. People accept the fantasy of scientology theories because they have been processed and have become mentally crippled; it is not the fantasy that leads them into scientology though it may help to sustain their servitude.

The real dangers of scientology processing and of the harmful hypnotic and other techniques it uses are not in relation to the fantasy which its adherents believe; such beliefs are merely symptomatic. The harm is done by the processes used and it is the prohibition of such processes which is necessary.

Suppression of scientology by name insufficient.

An attempt to suppress or proscribe scientology and scientologists specifically by name would probably be ineffective, and, in any event, would be undesirable.

It would probably be ineffective because scientologists, who are not lacking in mental agility, would find ways of changing names of processes-even changing the name of scientology- and continuing to practise in much the same way under another guise. Hubbard changed the name of his "science from scientology to dianetics, and back again to scientology. Doctor Marcus Tooley, dealt with in Chapter 29, has his "dynamic psychology" and, as earlier mentioned, the HASI has

now registered the business name of "The College of Applied Philosophy".

Action against scientology specifically by name would be undesirable for at least two reasons: -

- (a) Though scientology is not, and does not claim to be, a religion, it is a system of beliefs. Notwithstanding its weird theories and the peculiar practices based on them, it is a system of beliefs which any person is at liberty to hold, just as whoever wishes may believe that the moon is made of green cheese. That the rest of the community believes that it knows better is not recognized as a reason for coercing the Green Cheese School to believe what the most recent lunar observations have revealed. The proscribing of persons for beliefs which they may hold has so many undesirable features as not to need elaboration.
- (b) An attempt specifically to proscribe scientology and scientologists would tend to bestow on them the quality of an oppressed minority suffering for its beliefs.

Hubbard is adept at turning to advantage any opposition to himself and his "science". He has often told his followers that they would meet opposition and persecution because of what they believe; when views are expressed or action taken hostile to scientology, such events are complacently accepted by his followers as further proof that "Ron is right". Misplaced compassion and sympathy for scientologists, who would parade as martyrs suffering for their faith, might well be engendered in sections of the community.

In the view of the Board, however, there are more effective and less controversial ways of dealing with the dangerous practices of scientology than by prohibiting it by name. In order to control scientology, it is necessary to strike at the heart of the problem. This involves an appreciation of the essential nature and quality of its processes. Hubbard claims that scientology is a form of psychology and the evidence shows it to be psychology practised in a perverted and dangerous way by persons who are not only lacking any qualifications which would fit them to practise psychology but who have been indoctrinated and trained in beliefs and practices which equip them to do no more than apply dangerous techniques harmfully and indiscriminately.

Control scientology by controlling Psychology.

There are no standards prescribed in Victoria for persons who wish to practise psychology. Very considerable use is made, of course, Of psychological theory and practice by the medical profession, especially in psychiatry and psycho-analysis. As used in medicine it is ancillary to the broader profession of medicine. It is, of course, a learned science in its own right and many universities and other recognized institutions of learning award degrees or diplomas in psychology, or award degrees and diplomas which comprehend or embrace the very extensive study of psychology

A person so qualified may properly describe himself as a psychologist and may practise psychology; yet, legally, so may the most ignorant and least qualified person. The absence of any legal restrictions has enabled many persons, including scientologists, to describe themselves in various ways and to practise psychology though quite unqualified to do so.

In Victoria there is no legal control of hypnosis which is generally a major technique in the routine of the pseudo-psychologist. In England, there are restrictions on stage hypnosis. In the *Hypnotism Act* 1952 (15 and 16 Geo. 6 and 1 Eliz. 2, c. 46), hypnotism is stated to include hypnotism, mesmerism and any other similar act or process which produces or is intended to produce in any person any form of induced sleep or trance in which the susceptibility of the mind of that person to suggestion or direction is increased or intended to be increased. Restrictions are imposed on

exhibitions, demonstrations or performances of hypnotism at or in connexion with an entertainment to which the public are admitted. They do not extend to the practice of hypnotism privately.

Despite the affinity between psychology and medicine no assistance in the control of the practice of psychology is to be obtained from Victorian legislation relating to the medical profession.

This year is the centenary of the *Medical Practitioners Statute* 1865 which enacted in substantially the same form as the law stands today sections which were the predecessors of section 24 of the *Medical Act* 1958. In Victoria, there is no statutory prohibition on the practice of medicine or surgery as such by other than legally qualified medical practitioners. Virtually the only control is that effected by section 24. Section 24 (1) provides that "No person shall be entitled to recover any charge in any court of law for any medical or surgical advice or attendance or for the performance of any operation or for any medicine which he has both prescribed and supplied unless he proves on the trial that he is registered in accordance with the provisions of Division two of this Part". Section 24 (3) makes it unlawful for any person unless registered "to pretend to be or to take or use the name or title of a physician doctor of medicine licentiate in medicine and surgery master in surgery bachelor of medicine doctor surgeon medical or general practitioner or apothecary or surgeon apothecary or accoucheur or licentiate or practitioner in midwifery or any other medical or surgical name or title".

After the passage of one hundred years and the tremendous advances in medicine and surgery, which cannot now be effectively practised except in association with a system of hospitals from which those regarded as unqualified are excluded, there are very few who would presume to practise medicine or surgery if not registered under the Medical Act. On the contrary the problem of the unqualified psychologist is substantial. Psychology is a relatively new science and similar safeguards in respect of its practice have not yet developed.

In the Board's view, rather more stringent provisions than those in the Medical Act are needed to curb the emergence of a class of unqualified pseudo-psychologists of whom scientologists are the prime examples. Prohibitions on the right to sue for fees and on the use of deceptive titles are, of course, necessary; but such minimal restrictions could readily be overcome by payment in advance or some form of psychological coercion afterwards, and by the dropping of deceptive titles which, though often used, are not a necessary part of the pseudo-psychologist's act.

Regulation of the Practice of Psychology.

In order adequately to control the practice of psychology, the Board considers that measures of a regulatory and economic nature are necessary. In essence, the Board envisages a system of registration of psychologists, and prohibition on the advertising of psychological services and on practising psychology for fee or reward, unless registered. The practice of scientology is a business, and its attraction lies in the opportunity it affords for making money. Without revenue to maintain its organization its activities could not continue, and even its most dedicated adherents would find it impossible to practise professionally unless financially endowed. The prescribing of minimum standards for the practice of psychology and a prohibition on receiving payment for substandard services appear, therefore, to be the most effective means of procuring the cessation of scientology practices, except in the most attenuated and honorary way.

Recommended Action.

Appropriate action to deal with scientology involves three steps:

1. Public Awareness.

As already mentioned the adverse publicity which scientology received during the Inquiry has seriously curtailed its activities. However, the HASI may well regard such a reverse as temporary and may be awaiting the time when interest in the Inquiry and this Report will have waned. Thereafter, either at 157-159 Spring Street, Melbourne, or elsewhere, it will, under its own name or some other name, seek to re-establish itself. The likelihood is that scientology is planning to extend, if it has not already done so, to new areas in Victoria. It is important, in the Board's view, that the public be repeatedly warned of the dangers to mental health of psychological techniques practised by unqualified persons.

2. Registration of Psychologists.

Psychologists Registration Board.

The system of registration of psychologists envisaged as a means of controlling the practice of psychology in Victoria is one which involves the establishment by statute of a Psychologists Registration Board and the prescribing of minimum standards for registration. In considering the structure of such a Board, it would be necessary to safeguard the interests of those who are not professional or academic psychologists, but for justifiable reasons should be permitted to practise psychology in particular circumstances and are entitled to be registered or, alternatively, to be exempted from registration. Some consideration is given later in this Chapter to the kind of persons who should be exempted.

It is not the function of this Board of Inquiry to determine minimum standards or qualifications. It is sufficient for the Board to say that, in its opinion, scientology qualifications should not entitle a person to registration as a psychologist.

Of prime importance in any system of registration would be the meaning to be given to the word "psychology". It would need to be defined in terms which ensured that scientology techniques came under control.

The full scope of such a definition involves a consideration of many factors as to which the Board heard little or no evidence. The techniques and practices of scientology include -

hypnotic and similar techniques,

IQ testing,

personality assessment,

the use of the E-meter.

Whatever else may be comprehended by "psychology ", the above particular activities should evidently be included.

Prohibited Practices.

The purpose of the proposed legislation should be to prohibit generally the practice of psychology for fee or reward by persons who are not registered or who are not exempted from registration.

It should also be an offence, unless in the case of permitted exceptions, for any unregistered person, for fee or reward,

(a) to practise hypnosis, defined in terms similar to, but not necessarily identical with, the definition

- of "hypnotism" in the English Act, and enlarged to describe more specifically scientology processes which involve a series of commands, repetitive or otherwise, designed or intended to be used for the purpose of inducing the absence or lessening of inhibitions and repressions;
- (b) to administer IQ tests or personality assessments or similar tests or assessments;
- (c) to use an E-meter or similar electrical device whether or not in conjunction with (a) or (b). Prohibition on Advertising.

It should also be an offence for any person, unless in the excepted classes or unless registered, to advertise -

- (a) that he is skilled in psychology or psychological or similar techniques, psychological or similar counselling, hypno-therapy, psycho-therapy or any other similar alleged skills; or
- (b) that he is available whether for fee or reward or not to render services in respect of any such skills, or
- (c) that he will render such services whether for fee or reward or not.

Prohibition of Deceptive Titles.

The trappings of scientology include the pretentious use of titles, referred to in detail in Chapter 15. The use by scientologists and others of the title "Doctor" in relation to their practising of psychological techniques is a calculated deception where they are not possessed of any recognized degree which justifies the use of the title. An unregistered person may not lawfully use the title of "Doctor" in the practice of medicine or surgery. A similar prohibition should exist in relation to those who practise any form of psychology. The Board is of opinion that the restriction on the use of deceptive titles and qualifications should go further. It considers that it should be an offence for a person practising any form of psychology to use any title, letters or abbreviations in association with his name, unless he possesses a degree, diploma, licence or other authorization to use it or them from a university, college or other institution recognized by the Psychologists Registration Board. The deceptive "degrees" of "doctor of scientology" and "bachelor of scientology", both in extenso and abbreviated to "D.Scn." and "B.Scn.", and the additions of such letters as "HCA" and "HPA" to the name of a person practising scientology or some other form of psychology should not be permitted.

Exemptions from Registration.

Exceptions to or exemptions from the requirement of registration would need to be carefully considered. Prima facie, persons proper to be excepted would seem to include -

legally qualified medical practitioners, dentists, ministers and other individuals engaged in recognized or usual activities of any bona fide religion, teachers in universities and registered or otherwise recognized schools and similar institutions, employees or other persons engaged by a person or organization for the purpose of determining the quality, capacity, ability or potentiality of persons employed or seeking employment, various social workers, officials of marriage guidance organizations, etc.

This list is by no means exhaustive; it is capable of being, and probably should be, substantially enlarged, for care has to be taken to guard against the establishment of too narrow a category of

persons who have the right to practise for fee or reward techniques of psychology which are in varying degrees of constant application in almost every occupation and walk of life.

The Board's recommendations, within its terms of reference, are of course limited to scientology, though the method recommended of dealing with scientology necessarily involves the surveillance of practices and conduct by persons other than scientologists.

3. The HASI Records.

In Chapter 24 and in other parts of this Report reference has been made to the nature and extent of records made and kept by the Melbourne HASI of the processing of preclears. These records, containing details of intimate and secret matters relating to preclears, are a potential threat to preclears who may be subjected to various forms of coercion. Preclears could be constrained to return to their earlier allegiance to scientology, and, in some hands, files could be used for the purposes of extortion and blackmail.

The Board has given consideration to the question of what action, if any, should be taken in respect of the HASI processing files. Though the files were not entitled to any form of privileged non-disclosure, the HASI created the false impression that the contents of its files were completely secret and that nothing in them would be disclosed, whatever the circumstances. Such a protection is considerably greater than the privilege which attaches to communications made by a patient to his doctor (see section 28 of the Evidence Act 1958), but, as a matter of law, no vestige of legal privilege attaches to the HASI files. Though the disclosure of the contents of the processing files is highly undesirable, to prohibit by legislation the disclosure of their contents would give to them an entirely unjustified quality of importance and security. The FIASI would probably boast that it was recognized by the State as of such importance that the contents of its files were by law made secret. However, while the files exist, they are a potential threat to the people to whom they relate.

All the HASI records, including the files, were tendered as exhibits before the Board, but their physical possession (other than of those which were specifically and individually tendered as separate exhibits) remained in Williams on his undertaking to produce them as and when required. The Board now sees no practicable way of procuring the files which are still in possession of the HASI, short of legislation requiring their surrender for the purpose of destruction. It should be pointed out that copies of preclear's processing reports are also kept by Hubbard at Saint Hill Manor in England, so that destruction of the files of the Melbourne HASI would not entirely remove the threat.

Interstate Aspects.

Though the Board was required to investigate scientology in Victoria and has made the foregoing recommendations for its curtailment in this State, it is proper to point out by way of final comment that while scientology is permitted to flourish in other States, any action taken in Victoria could not be fully effective. It seems certain that continental headquarters of the HASI would be transferred from Melbourne to another capital city from which residents of Victoria would be bombarded with a barrage of scientology literature. It is also probable that Victorians addicted to processing would readily travel interstate for courses in processing and training.

Commendations.

In concluding this Report, the Board wishes to thank the many persons whose co-operation very materially assisted the Board in its inquiry. Of these, certain persons merit particular mention.

The Board is grateful to counsel who appeared before it. Mr. Gordon Just, then of counsel (now his Honour Judge Just of the County Court), who appeared to assist the Board, brought to the Inquiry his customary courtesy, skill and thoroughness. Mr. J. R. O'Shea, who, until May, 1964, appeared as counsel for the scientology interests, and Mr. Ian G. Abraham, who thereafter and until November, 1964, appeared as counsel for the scientology interests, greatly helped the Board in the clear presentation of the views and the evidence of their clients; Mr. Abraham, in particular, bore the burden of his clients' cause for six difficult months during which he very competently and conscientious1y presented to the Board every relevant consideration in favour of his clients' case and tested evidence hostile to their cause. The Board regrets that his clients' instructions constrained Mr. Abraham to withdraw from the Inquiry on the 30th November, 1964. Mr. Warren C. Fagan, of counsel, who appeared for the Committee for Mental Health and National Security made an important contribution to the material placed before the Board by his cross-examination on a number of aspects of scientology which assumed increasing importance as the Inquiry proceeded. The Board is grateful to these counsel whose very considerable labours were of real value to the Board in its deliberations.

The evidence of the 151 witnesses was recorded by a team of shorthand writers, typists and their assistants under the direction of Mr. J. May, the Chief Government Shorthand Writer. These ladies and gentlemen performed the colossal task of recording nearly 4,000,000 words covering nearly 9,000 pages, with speed and accuracy. Their co-operation and efficiency resulted in a complete daily transcript being available shortly after the Board adjourned each day. The Board's task was greatly facilitated by the immediate availability of a complete and accurate transcript.

The Board also wishes to thank Mr. John Herbert Joseph Campbell, who conducted a thorough investigation of the financial affairs of the scientology organizations in Victoria and whose very valuable services to the Board are not to be measured by the relatively short time which he took to give his evidence, which was almost unique at the Inquiry for it was accepted by all concerned without challenge, and he was virtually not cross-examined. Mr. Campbell's evidence threw light on certain features of scientology in Victoria, which might not have been fully appreciated without his thorough investigation.

The Board also wishes to thank the Director of the Royal Botanic Gardens, Mr. R. T. M. Pescott, for the use of the theatrette at the National Herbarium, where almost all the proceedings of the Board were held. Though the presence for many months of the Board and its entourage involved Mr. Pescott in some inconvenience he readily made available for the Inquiry adequate facilities in idyllic surroundings.

The Board reserves for final mention the secretary to the Board, Mr. Ian Edward Crook. The Board cannot too highly commend Mr. Crook for the consistently efficient and thorough manner in which he cheerfully and unobtrusively carried out the very burdensome duties which fell to him. He performed the tasks normally performed by at least two persons, for he worked in collaboration with counsel assisting the Board in sifting material and preparing it for presentation to the Board, and he carried out the very substantial secretarial duties required of him both while the Board was sitting and while it was in recess. His services to the Board in the preparation of this Report were of incalculable assistance and were beyond the call of duty. The Board is aware that his Honour Judge Just shares the Board's views regarding Mr. Crook. The Board considers it fitting that its Report should end with its grateful acknowledgement to the secretary whose constant attention, intelligence, efficiency and skills sustained the Board and made more bearable the burdens involved in the conduct of the Inquiry and the writing of this Report.

Given under my hand and seal this twenty-eighth day of September, 1965.

KEVIN ANDERSON

Owen Dixon Chamber, 205 William Street, Melbourne, Victoria.